BILL T. ARNOLD JOHN H. CHOI

# A GUIDE TO Biblical Hebrew Syntax

CAMBRIDGE www.cambridge.org/9780521826099

CAMBRIDGE

hand of

This page intentionally left blank

## A Guide to Biblical Hebrew Syntax

A Guide to Biblical Hebrew Syntax introduces and abridges the syntactical features of the original language of the Hebrew Bible or Old Testament. Scholars have made significant progress in recent decades in understanding Biblical Hebrew syntax. Yet intermediate readers seldom have access to this progress because of the technical jargon and sometimes-obscure locations of the scholarly publications. This guide is an intermediate-level reference grammar for Biblical Hebrew. As such, it assumes an understanding of elementary phonology and morphology, and it defines and illustrates the fundamental syntactical features of Biblical Hebrew that most intermediate-level readers struggle to master. The volume divides Biblical Hebrew syntax, and to a lesser extent morphology, into four parts. The first three cover the individual words (nouns, verbs, and particles) with the goal of helping the reader move from morphological and syntactical observations to meaning and significance. The fourth section moves beyond phrase-level phenomena and considers the larger relationships of clauses and sentences.

Bill T. Arnold is Director of Hebrew Studies and Professor of Old Testament and Semitic Languages at Asbury Theological Seminary. He is the coauthor of *Encountering the Old Testament: A Christian Survey* and author of the forthcoming volume on Genesis in the New Cambridge Bible Commentary Series.

John H. Choi is a Hebrew Teaching Fellow at Asbury Theological Seminary.

# A Guide to Biblical Hebrew Syntax

BILL T. ARNOLD Asbury Theological Seminary

JOHN H. CHOI Asbury Theological Seminary



CAMBRIDGE UNIVERSITY PRESS Cambridge, New York, Melbourne, Madrid, Cape Town, Singapore, São Paulo

Cambridge University Press The Edinburgh Building, Cambridge CB2 2RU, UK

Published in the United States of America by Cambridge University Press, New York

www.cambridge.org Information on this title: www.cambridge.org/9780521826099

© Cambridge University Press 2003

This publication is in copyright. Subject to statutory exception and to the provision of relevant collective licensing agreements, no reproduction of any part may take place without the written permission of Cambridge University Press.

First published in print format 2004

 ISBN-13
 978-0-511-18650-9
 eBook (EBL)

 ISBN-10
 0-511-18650-9
 eBook (EBL)

 ISBN-13
 978-0-521-82609-9
 hardback

 ISBN-10
 0-521-82609-8
 hardback

 ISBN-13
 978-0-521-53348-5
 paperback

 ISBN-13
 978-0-521-53348-5
 paperback

Cambridge University Press has no responsibility for the persistence or accuracy of URLS for external or third-party internet websites referred to in this publication, and does not guarantee that any content on such websites is, or will remain, accurate or appropriate.

# **Contents**

Preface

have	vi
page	лі

1	INTROI	DUCTION	1
2	NOUNS		4
	2.1.	Nominative	6
		2.1.1. Subject	6
		2.1.2. Predicate Nominative	6
		2.1.3. Vocative	7
		2.1.4. Nominative Absolute	
	2.2.	Genitive	7 8
		2.2.1. Possessive	8
		2.2.2. Relationship	9
		2.2.3. Subjective	9
		2.2.4. Objective	9
		2.2.5. Attributive	10
		2.2.6. Specification	10
		2.2.7. Cause	11
		2.2.8. Purpose	11
		2.2.9. Means	11
		2.2.10. Material	12
		2.2.11. Measure	12
		2.2.12. Explicative	12
		2.2.13. Superlative	13
	2.3.	Accusative	13
		2.3.1. Object	14
		2.3.2. Adverbial	18
	2.4.	Apposition	21
	1	2.4.1. Species	22

#### Contents

		2.4.2.	Attributive	22
		2.4.3.	Material	23
		2.4.4.	Measure	23
		2.4.5.	Explicative	24
	2.5.	Adjectiv	ves	24
		2.5.1.	Attributive	25
		2.5.2.	Predicate	26
		2.5.3.	Substantive	26
		2.5.4.	Comparative and Superlative	27
	2.6.	Determ	ination	28
			Referential	29
		2.6.2.	Vocative	30
		0	Naming	30
		2.6.4.		31
		2.6.5.	Generic	31
			Demonstrative	32
		2.6.7.	Possessive	32
	2.7.			32
			Cardinal Numbers	33
		2.7.2.	Ordinal Numbers	35
3	VERBS			36
	3.1.	Stem		37
		3.1.1.	Qal	37
		3.1.2.	Niphal	38
		3.1.3.		41
		3.1.4.	Pual	46
		3.1.5.	Hithpael	47
		3.1.6.	1	48
		3.1.7.	Hophal	52
	3.2.	Aspect		53
		0	Perfect	54
			Imperfect	56
	3.3.	Modals		60
		3.3.1.		61
			Imperative	63
			Cohortative	65
	$3 \cdot 4 \cdot$	Nonfini		66
			Infinitive Construct	67
			Infinitive Absolute Participle	73 77

4

3.5.	Verbal	Sequences	83
	3.5.1.	Imperfect plus waw Consecutive	84
		Perfect plus waw Consecutive	87
	3.5.3.	Commands in Verbal Sequences	91
	$3 \cdot 5 \cdot 4 \cdot$	Interruptions in Verbal Sequences	92
PARTIC	LES		95
4.1.	Prepos		95
	4.1.1.	אַחֲרִי ֻ ⁄ אַחַר	96
	4.1.2.	אָל־ ∕ אַל	97
	4.1.3.	אַצָל	101
	4.1.4.	אֶת־ ⁄ אֵת	101
	4.1.5.	בָּ בֵין	102
	4.1.6.	בִּין	106
	$4 \cdot 1.7 \cdot$	בָּעַד־ / בַּעַד	107
	4.1.8.	יַעַן	108
	4.1.9.		109
	4.1.10.	7	110
	4.1.11.	לְמָעַן	115
	4.1.12.		115
	4.1.13.		116
	4.1.14.		119
	4.1.15.	עַר	120
	4.1.16.		120
	$4 \cdot 1 \cdot 17 \cdot$		124
	4.1.18.		125
$4 \cdot 2 \cdot$	Advert		127
		אַזַי / אָז	127
	4.2.2.	,ĭ Į	129
	4.2.3.	אַל	130
	$4 \cdot 2 \cdot 4 \cdot$		130
	4.2.5.		132
	4.2.6.		134
	$4 \cdot 2 \cdot 7 \cdot$		134
	4.2.8.	כה	135
	4.2.9.	Ć.	135
	4.2.10.	ي چا	136
	4.2.11.	<b>5</b>	137
	4.2.12.	מאד	138
	4.2.13.	עוד	139
	4.2.14.	עַתָּה	139

## Contents

	.2.15. רַק	140
	.2.16. 回遊	140
	בָּמִיד. .2.17. הַמָּיִר	
	Conjunctions	142 143
	.3.1. X	143 143
	.3.2. DX	143
	-3-3- <b>-</b> 3	143
	-3-3- -3-4- 'P	140
	··3·4· 7 ··3·5· ]	149
	Particles of Existence/Nonexistence	156
	.4.1. []X	156
	.4.2. U	150
	רביים איניים יוביאיים. הנה and הנה Particles	157
	.5.1. הְנָה	158
	.5.2. <u>ה</u> בי	159
		-99
5 CLAUSES	AND SENTENCES	162
	Jominal and Verbal Clauses	164
	.1.1. Nominal Clause	165
	.1.2. Verbal Clause	167
	ubordinate Clauses	171
	.2.1. Substantival Clause	171
	.2.2. Conditional Clause	173
	.2.3. Final Clause	174
5	.2.4. Temporal Clause	176
	.2.5. Causal Clause	178
	.2.6. Comparative Clause	179
	.2.7. Exceptive Clause	180
	.2.8. Restrictive Clause	181
	.2.9. Intensive Clause	181
	.2.10. Adversative Clause	181
	.2.11. Circumstantial Clause	182
	.2.12. Concessive Clause	183
	.2.13. Relative Clause	184
	.2.14. Disjunctive Clause	186
	Additional Sentence Types	186
	.3.1. Interrogative Sentences	187
	.3.2. Oath Sentences	188
5	.3.3. Wish Sentences	189
5	.3.4. Existential Sentences	191

# Contents

5.3	.5. Negative Sentences	191
5.3	.6. Elliptical Clauses and Sentences	192
Appendix I: Stem	n Chart	193
Appendix II: Exp	oanded Stem Chart	194
Glossary		195
Sources Consulted	d	207
Subject Index		213
Scripture Index		216

# Preface

This book is intended to introduce basic and critical issues of Hebrew syntax to beginning and intermediate students. It grows out of eighteen collective years of experience in teaching Biblical Hebrew to seminarians. Each year, we teach or supervise the instruction for approximately 180 students in preparation for ordained ministry or other religious professions. Our experiences led us to conclude that a significant gap exists between, on the one hand, the current scholarly understanding of Hebrew syntax, based on significant progress in the discipline in recent decades, and on the other hand, the understanding of Hebrew syntax among our students. The problem seemed compounded by the lack of an intermediate-level grammar, holding a position between beginning grammars and advanced reference grammars. In addition, the ever-growing demands on theological education today have resulted in less time to master Biblical Hebrew. Often the first thing omitted in a beginning Hebrew course is an overview of syntactical features. Our purpose, then, has been simply to bridge the gap, as best we can, between our students and the best of current research on Biblical Hebrew syntax.

This book, then, is not intended to replace the standard reference grammars, which we have consulted constantly in the process, but to present to beginning and intermediate students a means of entry into the latest scholarship on Biblical Hebrew. To this end, we have included extensive references in the footnotes where appropriate. In particular, we have been most influenced by the unity brought to bear on the Hebrew verbal system by Bruce K. Waltke and Michael O'Connor. We have also consulted frequently the grammars by Joüon-Muraoka and Gesenius-Kautzsch-Cowley, and to some degree those by van der Merwe-Naudé-Kroeze and Meyer. Through interaction with these and other sources of scholarship, we feel that we have, at several points, introduced innovations in our explanations of Biblical Hebrew syntax in an attempt to refine the way we read and interpret the Bible today.

We express appreciation to our colleagues Joseph R. Dongell, David L. Thompson, Lawson G. Stone, and Brent A. Strawn for helpful suggestions on several points, and especially Dr. Stone for permission to use his chart in Appendix 2. We also benefited greatly from the comments and suggestions of the anonymous external reviewers hired by Cambridge University Press. The editors of the Press have been exemplary in every way, and we note especially Andrew Beck, who has been a source of encouragement from the beginning. In addition, Phyllis Berk and Janis Bolster made many improvements during the production process.

# **1** Introduction

t the heart of biblical interpretation is the need to read L the Bible's syntax, that is, to study the way words, phrases, clauses, and sentences relate to one another in order to create meaning. Biblical Hebrew is a language far removed from us in time and culture. Mastering it is a noble but daunting task. Students often learn to discern the elementary phonology and morphology in order to "read" the biblical text. But we believe exegesis (or the extraction of a text's meaning) requires more than phonology and verb parsing. Achieving a deep-level reading requires a grasp of a text's syntactical relationships, a topic that most beginning grammars do not present in detail. Thus, our task has been to help the reader grasp the building blocks of Biblical Hebrew, that is, the syntactical specifics that constitute meaning. These are the linguistic details through which the most profound of all statements can be made, and have been made - those of Israel's monotheism and the nation's covenant relationship with YHWH.

We have defined and illustrated the fundamental morphosyntactical features of Biblical Hebrew. The volume divides Hebrew syntax, and to a lesser extent morphology ("the way words are patterned or inflected"), into four parts. The first three cover individual words (nouns, verbs, and particles) with the goal of helping the reader move from morphological and syntactical observations to meaning and significance. The fourth section moves beyond phrase-level

phenomena and considers the larger relationships of clauses and sentences. Each syntactical category begins with at least one paragraph, giving definition to that grammatical category. This is followed by a list of the most common exegetical possibilities for that particular grammatical phenomenon. We have provided at least one example (and in most cases more than one) for each syntactical function. Each example is followed by a translation, in which the syntactical feature in question is italicized and underlined where possible. The translations are often related to the NRSV, although we have frequently taken the liberty of altering the translations at points in order to illustrate better the particular syntactical feature under discussion. This is followed by the biblical reference. All examples are taken directly from the Hebrew Bible; on occasion, certain prefixed or conjoined particles, which have no bearing on the syntactical principle being illustrated, have been omitted for the sake of clarity in the English translation.

The categories for classification presented here are by no means exhaustive, which would have required a book many times this size. We have made frequent reference to the leading reference grammars for additional information. We have also omitted discussions of elementary phonology and morphology, including difficult forms or spellings that may be unique or exceptional in some way, all of which are covered sufficiently by numerous beginning grammars. In our footnotes we have frequently included references to the elementary grammars so as to encourage the reader to consult a familiar source in order to review an elementary detail of phonology or morphology, which may have been forgotten. For example, our discussion of "determination" (section 2.6) reminds the reader that one of the ways a noun may be marked as definite is with the prefixed definite article. Since all beginning grammars explain the morphological details of the definite article, with examples of the various forms it takes depending on the noun it marks, we have not repeated that information here. Instead, we direct the reader to review the

beginning grammars where needed.<sup>1</sup> We have also omitted entirely, or in some cases briefly summarized, certain theoretical and complex grammatical issues that regularly make the standard reference grammars unintelligible to the intermediate student. We have, however, included many discussion footnotes dealing with these issues in order to provide additional background information that we believe will be of particular interest to advanced students and scholars. In this way, we have attempted to create a user-friendly volume of modest size.

For the most part, the features defined and illustrated here pertain to the language used in the extended narratives of the Pentateuch and the Historical Books, along with prose sections of the Prophets and Writings. This language is sometimes known as Classical Biblical Hebrew, although we refer to it simply as Biblical Hebrew (BH).<sup>2</sup> At times, we make further observations on Late Biblical Hebrew (LBH), by which we mean the language of most of the biblical books written after the exile (1–2 Chronicles, Ezra-Nehemiah, Esther, Daniel, selected Psalms, Song of Songs, Ecclesiastes, and portions of others).<sup>3</sup> Although LBH has features that are often unique, it also shares many features with BH. Thus, in some cases, we have used examples from both BH and LBH to illustrate the continuity of certain grammatical features of the Hebrew language.

- <sup>1</sup> For more on morphology, students may now consult the convenient "How Hebrew Words Are Formed" in Landes 2001, 7–39.
- <sup>2</sup> "BH" will be used throughout for "Biblical Hebrew." All other abbreviations may be found in Patrick H. Alexander et al., eds., *The SBL Handbook* of Style for Ancient Near Eastern, Biblical, and Early Christian Studies (Peabody, Mass.: Hendrickson, 1999), 121–52.
- <sup>3</sup> This list is only partial, since it depends to a large degree on interpretive issues about which scholars are not agreed. For more on the distinction between BH and LBH, see Polzin 1976, 1–2; Rooker 1990; and Sáenz-Badillos 1993, 50–75 and 112–29.

# 2 Nouns

**B** y comparing evidence from early Semitic languages, scholars have concluded that pre-biblical Hebrew, and most likely all the Semitic languages of the second millennium B.C.E., had a declension system for the nouns (i.e., inflections), using cases parallel to those of Indo-European languages.<sup>1</sup> Thus, endings were used to mark a subject case (the nominative, ending in singular *-u*, plural *-ū*, and dual  $-\bar{a}$ ), an adjectival case, which was used also with all the prepositions (genitive in *-i*, *-i*, and *-ay*), and an object case that also had many adverbial uses (accusative in *-a*, *-i*, and *-ay*). However, the case endings were almost completely lost in all first-millennium Northwest Semitic languages, and they were certainly lost throughout all attested Hebrew.<sup>2</sup>

- <sup>1</sup> Akkadian retains the cases in most dialects, as does Classical Arabic. Among the Northwest Semitic languages, Amorite, Ugaritic, and the Canaanite glosses in the Tell Amarna texts – all from the second millennium B.C.E. – retain the case endings. On the preservation of cases in Amarna letters written by Canaanite scribes, see the important discussion of Rainey (1996, 1:161–70), although note his preference for "dependent" case over "genitive."
- <sup>2</sup> Garr 1985, 61–63; Sáenz-Badillos 1993, 23; Moscati 1980, 94–96; Bergsträsser 1983, 16–17; Harris 1939, 59–60; Joüon and Muraoka 1993, 277–78; Bauer and Leander 1991, 522–23. Earlier grammarians believed the unaccented Hebrew ending ד, used on certain nouns denoting direction, was a vestige of the old accusative case ending (so ד, ארצר), the

Nouns

Biblical Hebrew compensates for the lack of case endings through a variety of means, primarily word order (as in modern English) and syntactical relationships, as well as through the use of prepositions. So the nominative case is most frequently discerned by word order and the lack of other markers. The genitive is marked by the construct relationship (section 2.2), and the accusative primarily by the definite direct object marker אֶת־/אֶת and other syntactical relationships (section 2.3). Although we are able to trace the history of the three case functions in ancient Hebrew by comparing other Semitic languages, some authorities believe we should abandon the traditional case terms (especially "nominative") when describing BH syntax because the language does not mark the cases morphologically.<sup>3</sup> While we admit BH does not mark the cases with specific noun inflections, our objective is to identify and describe the functions of the noun.<sup>4</sup> Since the nouns in BH function syntactically in the same distinct "cases" as its parent language, it is still helpful to distinguish three case functions in BH using the traditional terminology: nominative, genitive, and accusative.5

so-called directive  $\exists$ ,  $\neg$ , or *he locale*). However, Ugaritic has a separate adverbial suffix *-h* in addition to an accusative case ending *-a*, proving beyond doubt that the *he locale* in Hebrew is not a remnant of the accusative (Waltke and O'Connor 1990, 185; Seow 1995, 152–53; and for the older – now outdated view – cf. Kautzsch 1910, 249). The closest BH comes to having cases is in the declension of the personal pronoun (cf. van der Merwe, Naudé, and Kroeze 1999, 191).

<sup>3</sup> Kroeze 2001, 33–50.

- <sup>4</sup> Regarding the nominative, Kroeze accepts "subject" as a designation for category 2.1.1, but proposes the following alternative designations for the others: "copula-complement" for *predicate nominative*, "addressee" for *vocative*, and "dislocative" for *nominative absolute* (2001, 47). If the reader remembers that we are describing the *syntactical* functions of these nouns rather than their grammatical morphemes, we believe the traditional terminology is more helpful.
- <sup>5</sup> It should be remembered that pronouns may serve in all these functions as well.

#### 2.1 Nominative

Since a noun's case function is not marked morphologically, the nominative can be detected only by the noun's or pronoun's word order, by its agreement in gender and number with a verb (although with many exceptions), or by the sense of the context. Generally, the nominative may be categorized as follows.<sup>6</sup>

#### 2.1.1 Subject

The noun or pronoun serves as the subject of an action: וַיֹּאמֶר אֲלֹהִים, "<u>God</u> created" (Gen 1:1), בָּרָא אֲלֹהִים, "And <u>God</u> said" (Gen 1:3). In the same way, when used with stative verbs the noun or pronoun may serve as the subject of a state: מָלְאָה הָאָרֶץ הָמָלָאָה הָאָרֶץ. "<u>the earth</u> is filled with violence" (Gen 6:13).

#### 2.1.2 Predicate Nominative

The noun or pronoun is equated with the subject by a "to be" verb (stated or implied): הְרָוָה מֶלֶךֹ, "Yhwh <u>is king</u>" (Ps 10:16). In this example, the *subject* noun (section 2.1.1) is Yhwh, and the *predicate nominative* is "king." The predicate nominative is often a clause of *identification*, in which case the word order is likely subject-predicate: אָרָי הָרָהָ (I <u>am Yhwh</u>" (Exod 6:2), אָרָה הָאָיָשׁ, "You <u>are the man</u>" (2 Sam 12:7).<sup>7</sup> However, the word order is flexible, as this clause of *description* illustrates, also with subject-predicate order: הַמָשׁפְּשִׁרֹיְהָנָה אָמָת (Ps 19:9 [Eng 19:10]). The predicate nominative is one

 <sup>&</sup>lt;sup>6</sup> Van der Merwe, Naudé, and Kroeze 1999, 247–49; Kautzsch 1910, 451–55;
 Waltke and O'Connor 1990, 128–30; Lambdin 1971a, 55; Chisholm 1998, 61; Williams 1976, 10.

<sup>7</sup> Andersen 1970, 31-34.

of several ways nominal clauses are constructed (see section 5.1.1,a).

#### 2.1.3 Vocative

The noun designates a specific addressee and normally has the definite article (see section 2.6.2): הְנֵה חֲנִית הָמֶּלֶן, "here is the spear, <u>O king</u>" (1 Sam 26:22 Qere). The vocative noun stands separate from the clause's syntax and is often juxtaposed to a second-person pronoun (or pronominal suffix) reflecting the direct speech: הְמָלֶן הַמֶּלֶן, "as your soul lives, <u>O King</u>" (1 Sam 17:55), הְיָבֶר־סַתֶר לִי אָלֶיך הַמֶּלֶן הַמֶּלֶן, "I have a secret message for you, <u>O King</u>" (Judg 3:19). The second person may be expressed by the imperative: הוֹשִׁיעֵנוּ אֲלֹהֵי יִשְׁעֵנוּ (1 Chr 16:35), הוֹשִׁיעֵנוּ הָהָן הַנָּגָרָ וֹשָׁעַנוּ [Eng 12:1]).

#### **2.1.4** Nominative Absolute

The noun is isolated from the following sentence (sometimes by an intervening subordinate clause or series of appositional terms) and then resumed by a pronoun serving as the subject of the sentence: א הָאָשֶׁר הָוֹא הָאָרָ (<u>Yhwh, he</u> is God" (1 Kgs 18:39), "הַאָשֶׁה אֲשֶׁר נָתַזְה עָמָרִי הָוֹא נָתְנָה לִי מִן־הָעֵין וָאֹכַל , "The <u>woman</u>, whom you gave to be with me, <u>she</u> gave me fruit from the tree, and I ate" (Gen 3:12).

The nominative absolute is also known as *casus pendens* (Latin "hanging case") or as *focus marker*.<sup>8</sup> These are general designations for a grammatical element isolated outside a clause, usually at the start of the clause.

<sup>&</sup>lt;sup>8</sup> Other designations include *dislocated construction* and *pendens construction*. The sentence constituent taking up the noun or pronoun again may be called the *resumptive*. Joüon and Muraoka 1993, 586–88; Waltke and O'Connor 1990, 692; and for a distinction from so-called *fronting*, see van der Merwe, Naudé, and Kroeze 1999, 336–39.

#### 2.2 Genitive

Most relationships that exist between two nouns are expressed in BH by means of the genitive construction. English usually expresses this genitive relationship between two nouns with the word "of." For example, in the phrase "the daughter of the king," the noun "king" acts as a genitive modifier of "daughter." Thus, the genitive relationship often denotes a possessive sense: "the king's daughter." However, the genitive relationship is used in BH to denote a wide variety of other uses besides possessive. For example, "the word of truth" does not mean "truth's word," but rather "the true word."<sup>9</sup>

The genitive relationship is marked grammatically in Hebrew by means of the construct state, in which two (or more) nouns are bound together to form a construct phrase or chain. In the structure *construct plus genitive*, the genitive modifies the construct in some way, frequently as some sort of attributive adjective.<sup>10</sup> This list includes the most common ways in which a genitive modifies the preceding noun(s) or adjective(s).<sup>11</sup>

#### 2.2.1 Possessive

The genitive has ownership of the construct: בֵּית־יְהוָה, "the temple <u>of YHWH</u>" or "<u>YHWH</u>'s temple" (1 Kgs 6:37), צְׁבְאוֹת יְהוָה, "the hosts <u>of YHWH</u>" or "the hosts <u>possessed</u>"

- <sup>9</sup> English also uses noun-noun compounds (as in "notebook" and "airport") and apostrophe-*s* (Jeremiah's confession) in ways similar to the Hebrew construct phrase (Waltke and O'Connor 1990, 141).
- <sup>10</sup> Joüon and Muraoka 1993, 463–77; Waltke and O'Connor 1990, 136– 60; Kautzsch 1910, 410–23, esp. 416–19; Williams 1976, 10–12; Chisholm 1998, 62–64; van der Merwe, Naudé, and Kroeze 1999, 197–200. The genitive may also be called the *postconstructus* (van der Merwe, Naudé, and Kroeze 1999, 192). On the genitive in epigraphic Hebrew, see Gogel 1998, 240–45.
- <sup>11</sup> The genitive function may also be expressed by the object of a preposition or a pronominal suffix on a noun (Waltke and O'Connor 1990, 138). The genitive categories listed here also apply to these.

<u>by YHWH</u>" (Exod 12:41). That which is possessed may be body parts: <sup>12</sup> אָשָׁקָתִי־מֶלֶן "lips of <u>a king</u>" or "a <u>king</u>'s lips" (Prov 16:10), or a characteristic (דְרַת־מֶלֶן ", "the glory of <u>a king</u>" or "a <u>king</u>'s glory" (Prov 14:28).

The direction of ownership may be reversed resulting in a *possessed* genitive, in which case the genitive is owned by the construct: בַּעַל־הַבַּיָת, "the owner <u>of the house</u>" or "<u>the house's</u> owner" (Exod 22:7 [Eng 22:8]), יְהָוֶה צְּבָאוֹת, "YHWH <u>of hosts</u>" or "YHWH <u>who possesses hosts</u>" (1 Sam 1:11).

#### 2.2.2 Relationship

The genitive marks someone standing in association with the construct: בְּיֵי הַכָּיֶכֶן ", "the <u>king</u>'s sons" (2 Sam 13:23), "הָרָיִי," (2 Sam 13:23), "הָרָיִי," (2 Sam 13:23), "הָרָיָיָי," (1 Chr 29:22). This genitive is common with pronominal suffixes: "הָאָרָיָיָרָיָיָ brother" (Gen 20:13). The genitive of human relationship normally refers to kinship, although other social structures are possible.<sup>13</sup>

#### 2.2.3 Subjective

The genitive is the subject of a verbal notion expressed by the construct noun: ישׁע אָלהִים, "the salvation of <u>God</u>" (Ps 50:23), הְכָרִיהוָה "<u>Yhwh</u>'s word," (Gen 15:1), הְכָמַת שָׁלמֹה, "<u>Solomon</u>'s wisdom" (1 Kgs 5:10).

#### 2.2.4 Objective

The genitive is the object of a verbal notion expressed by the construct noun: יְרָאָת יְהוָה, "the fear of <u>Yhwh</u>" (Ps 19:8 [Eng 19:9]), שׁר שְׁנִיִים, "the violence of the <u>oppressed</u>" or "the violence done <u>to the oppressed</u>" (Ps 12:6). Sometimes the genitive identifies the recipient of the action implied in the construct: מְשֵׁר־אָרָצָהָ, "rain <u>for your land</u>" (Deut 28:12).

<sup>&</sup>lt;sup>12</sup> Waltke and O'Connor's "genitive of inalienable possession" (145).

<sup>&</sup>lt;sup>13</sup> Waltke and O'Connor 1990, 145.

The direction is reversed in the *action* genitive, in which the genitive expresses a verbal notion denoting action directed toward the constuct: עָם עֶבְרָתִי, "the people of <u>my wrath</u>" or "the people who are the <u>objects of my wrath</u>" (Isa 10:6).

# 2.2.5 Attributive

The genitive denotes a quality or attribute of the construct. In translation, the genitive often becomes an adjective: גְּבּוֹר חֵיל, "a man of <u>worth</u>" or "a <u>valorous</u> man" (Judg 11:1), אָשָׁת חֵיל, "a woman of <u>worth</u>" or "a <u>valorous</u> woman" (Ruth 3:11), הָר־קָרְשׁוֹ, "the mountain of <u>his holiness</u>" or "<u>his holy</u> mountain" (Ps 48:2 [Eng 48:1]), הָכָבוֹד הַכָּבוֹד, "the king <u>of glory</u>" or "<u>the glorious king</u>" (Ps 24:7), הָכָן שָׁרָשׁ, "joy <u>of perpetuity</u>" or "<u>everlasting</u> joy" (Isa 61:7).

The nouns בעל , איש, בעל , איש, and בעל , איש, often occur with genitives to express the possessor of a quality, which is one way BH compensates for the lack of genuine adjectives:<sup>14</sup> way BH compensates for the lack of genuine adjectives:<sup>14</sup> Sam 16:7), איש הקרים, "a man <u>of blood</u>" or "a <u>bloodthirsty</u> man" (2 Sam 16:7), איש הקרים, "a man <u>of words</u>" or "an <u>eloquent</u> man" (Exod 4:10), איש הקרים, "the son <u>of death</u>" or "one <u>who deserves death</u>" (1 Sam 20:31), העיה כוון, "a son <u>of five hundred years</u>" or "<u>five hundred years old</u>" (Gen 5:32), קעל כָּוָך , "master <u>of wings</u>" or "<u>winged creature</u>" (Prov 1:17).

# **2.2.6** Specification

The reverse of the *attributive* genitive, the *specification* genitive is characterized by a quality or attribute of an adjectival construct: קְשָׁה־עֶּרֶף, "stiff <u>of neck</u>" or "stiff-<u>necked</u>" (Exod 32:9), יְפָה־תֹּאֶר (stiff <u>of neck</u>" or "<u>good-looking</u>" (Gen 39:6), הַאָרָף אָפָרָים, "unclean <u>of lips</u>" or "unclean <u>with regard to the lips</u>" (Isa 6:5), הַאָרֶך אָפִים, "long <u>of nostrils</u> [as source of anger]" or "patient <u>with regard to anger</u>"

<sup>&</sup>lt;sup>14</sup> Joüon and Muraoka 1993, 468–69.

(Exod 34:6). The specification genitive may also be called the *epexegetical genitive*.<sup>15</sup>

#### 2.2.7 Cause

The genitive may be caused by the construct: רְּהַחְרְמָהַ "the spirit of <u>wisdom</u>" or "the spirit <u>that causes wisdom</u>" (Exod 28:3). The causal relationship may move in the opposite direction, so that the genitive is perceived as causing the construct: הוֹלֵת אָהַרָה, "sick <u>of love</u>" or "sick <u>because of love</u>" (Song 2:5), הוֹלֵת אָהַרָה, "those exhausted <u>of hunger</u>" or "exhausted people <u>because of hunger</u>" (Deut 32:24).<sup>16</sup>

#### 2.2.8 Purpose

The genitive denotes the intended use of the referent of the construct: צֹאן מָבְחָה, "sheep <u>of slaughter</u>," or "sheep <u>intended for slaughter</u>" (Ps 44:23 [Eng 44:22]), אָבְנִי־קָלָע "slingstones" or "stones intended for the sling" (Zech 9:15).

Closely related to the idea of *purpose* is the *geni*tive of result, which denotes the result of an action implied in the construct: מוסר שלומנו, "the chastisement <u>of our welfare</u>" or "the chastisement <u>that resulted in our peace</u>" (Isa 53:5), כוֹס הַתַּרְשֵׁלָה, "the cup <u>of staggering</u>" or "the cup which results in staggering" (Isa 51:17).

#### 2.2.9 Means

The genitive is the instrument by means of which the action implied in the construct is performed: חַלְלִי־חָרָב "wounded <u>of the sword</u>" or "those wounded <u>by the sword</u>" (Isa 22:2), שֶׁרְפוֹת אָשׁ, "burned <u>of fire</u>" or "burned <u>with fire</u>"

<sup>&</sup>lt;sup>15</sup> Waltke and O'Connor 1990, 151.

<sup>&</sup>lt;sup>16</sup> In which case the *causational genitive* is very close in meaning to 2.2.9, the *genitive of means* (Joüon and Muraoka 1993, 468; Waltke and O'Connor 1990, 146, and cf. note 22).

(Isa 1:7), שָׁמָא־עָבָּשׁ, "impure <u>of a corpse</u>" or "impure <u>by</u> reason of contact with a corpse" (Lev 22:4).

When the genitive is personal, this becomes a *genitive of agency*, which is similar to the *subjective genitive*: מָכָה אָלהִים, "stricken <u>by God</u>" (Isa 53:4).<sup>17</sup>

#### **2.2.10** *Material*

The genitive denotes the material from which the construct is made: אָרוֹן שֵׁץ, "an ark <u>of wood</u>" or "a <u>wooden</u> ark" (Deut 10:1), כְּלֵי־כֶּסֶך, "vessels <u>of silver</u>" or "<u>silver</u> vessels" (Gen 24:53), שֵׁבֶם בַרְזֶל, "a rod <u>of iron</u>" or "an <u>iron</u> rod" (Ps 2:9).

#### **2.2.11** Measure

The genitive marks the thing measured by a numeral in the construct: שָׁנִי בְנִים, "two <u>sons</u>" (Gen 10:25), שָׁנִי בְנִים, "seven <u>days</u>" (Gen 8:10). The construct may be a nonnumeric quantifier, such as רב/רוֹב, "multitude" or , "multitude" or , "all"; הָבָל-לָבוֹן, "genty of <u>grain</u>" (Gen 27:28), all your <u>heart</u>" (1 Sam 7:3).<sup>18</sup>

## 2.2.12 Explicative

The genitive is a specific member of a general category or class denoted by the construct, and typically specifies the proper noun for the construct: נְהַר־פְּרָת, "the river <u>Euphrates</u>" (Gen 15:18), אֶרִיץ מִצְרִים, "the land <u>of Egypt</u>" (Gen 41:19), בְּתוּלַת יִשְׂרָאֵל, "the garden <u>Eden</u>" (Gen 2:15), נָן־עָרָן "Virgin <u>Israel</u>" (Amos 5:2), "Daughter <u>Zion</u>"

<sup>&</sup>lt;sup>17</sup> On the *genitive of agency* as a variation of subjective genitives, cf. Waltke and O'Connor 1990, 143.

<sup>&</sup>lt;sup>18</sup> On the special connotations of כָּלֹ/כֹּל/כֹּל/ כֹּל/כֹּל/כֹּל depending on the determination or indetermination of its genitive, see Kautzsch 1910, 411, and Joüon and Muraoka 1993, 518–19.

(2 Kgs 19:21). Note that "of" is often unnecessary in English translations of the *explicative genitive*.

### 2.2.13 Superlative

The genitive can be divided into parts, of which the construct is the best, most, or the greatest form or expression of the whole: מְרָחֵר קְרָרֵינוּ, "the choice <u>of our graves</u>" or "the choicest <u>of our graves</u>" (Gen 23:6). Frequently a plural genitive will occur with the singular construct of the same noun to express the superlative: קְרָשֶׁים, "holy <u>of holies</u>" or "<u>most</u> holy" (Exod 29:37), מָלֶרִים, "king <u>of kings</u>" or "<u>the greatest</u> king of all" (Ezek 26:7), מָלֶרִים, "the song <u>of songs</u>" or "<u>the choicest</u> song" (Song 1:1).<sup>19</sup>

## 2.3 Accusative

As we have said, the accusative function of the noun is not indicated morphologically as in Greek or Latin, but similar functions are expressed through other grammatical features. The use of the definite direct object (DDO) marker  $\[mu]{\label{eq:sigma}}$  is one such grammatical tag, although it hardly exhausts the forms taken by the accusative. In later biblical texts, the preposition taken by the accusative. In later biblical texts, the preposition  $\[label{eq:sigma}$  is used as the DDO, probably due to Aramaic influence.<sup>20</sup> So, for example, in the Hebrew examples that follow, some accusatives are marked as pronominal suffixes on the verb. In others, the accusative will appear as the object of a prepositional phrase, since some Hebrew verbs mark objects with specific prepositions (e.g.,  $\[mu]{\label{eq:sigma}}$  plus  $\[mu]{\label{eq:sigma}}$ , "to

<sup>19</sup> A similar meaning is possible (although not universally accepted) when a divine name occurs as the genitive: אָכַוְ־ְהָוָה, "like a <u>splendid</u> garden (literally: a garden <u>of YHWH</u>)" (Isa 51:3), אַרְהַרְהָוָה, "a <u>very deep</u> sleep (literally: a deep sleep <u>of YHWH</u>)" (1 Sam 26:12). For more on the use of divine references to express a superlative, see Thomas 1954, 209–24.

<sup>&</sup>lt;sup>20</sup> Joüon and Muraoka 1993, 447–48; Kautzsch 1910, 366. Cf. Lam 4:5; Ps 145:14; Ezra 8:24; 2 Chr 25:10.

cling to, " בְּקַע plus בְּקָע, "to prevail," בְקַע plus רְּיָלְחַם, "to fight").<sup>21</sup> The English translations of the examples may look simplisitc, while the Hebrew illustrates the variety of ways for marking the accusative function.

As we have seen, nouns in the genitive modify other nouns by means of the bound construction with those nouns (or by means of prepositions). Nouns in the accusative modify verbs, denoting either the direct object of a verb or serving as adverbial modifiers. Although BH lacks explicit indicators of the accusative beyond the DDO, its functions can be identified by comparison with other Semitic languages where the accusative case is marked by an ending (sg. -*a*, pl. -*i*, du. -*ay*).<sup>22</sup> This list includes the most common ways in which the accusative is used.<sup>23</sup> We have divided the uses into two groups, the first listing examples of verbal objects in which the accusative is the recipient of a verb's action. The second group lists adverbial accusatives, in which the accusative modifies the circumstances surrounding the action of the verb.

#### 2.3.1 Object

The simplest form of the accusative is the direct object of a transitive verb: בָרָא אֶלֹהִים אָת הַשְׁמִים וְאָת הָאָרָץ, "God created <u>the heavens</u> and <u>the earth</u>" (Gen 1:1), צַּנְה יְהוָה אֶת־מֹשֶׁה "Yhwh commanded <u>Moses</u>" (Exod 40:32).<sup>24</sup> However, the

- <sup>21</sup> See Joüon and Muraoka 1993, 440–43, and Waltke and O'Connor 1990, 166.
- <sup>22</sup> Since Hebrew no longer marks accusatives morphologically, some grammarians prefer to avoid such designations as "direct object" and "adverbial accusative," referring instead to a verb's "nominal complements" and "nominal adjuncts" (van der Merwe, Naudé, and Kroeze 1999, 241–45, and see the discussion in Waltke and O'Connor 1990, 162–63).
- <sup>23</sup> Joüon and Muraoka 1993, 440–63; Meyer 1992, 412–20; Waltke and O'Connor 1990, 161–86; Kautzsch 1910, 362–76; Williams 1976, 12–14; Chisholm 1998, 64–66; van der Merwe, Naudé, and Kroeze 1999, 241–47.
- <sup>24</sup> The DDO is primarily a marker of this *accusative of object*, although it occasionally also appears with certain adverbial accusatives (Joüon and Muraoka 1993, 444).

simplicity of the direct *object accusative* changes depending on the nature of the verb itself. The following variations are possible.

(a) Affected – The object, which exists before and apart from the action of the verb, is reached by the action: ויַרָא אָלהִים אָת־הָאוֹר, "and God saw <u>the light</u>" (Gen 1:4), ויַיָעוֹר־לוֹ, (and he helped <u>him</u>" (2 Sam 21:17), "בָּרֶכְךָ יְהָוָה וִיִשְׁמְרֶךָ, "May YHWH bless <u>you</u> and keep <u>you</u>" (Num 6:24).

(b) Effected – The object did not exist prior to the verbal action, but is brought about by the action and is therefore the product or result of the action: וְלְבְנִים, "let us make <u>bricks</u>" (Gen 11:3), יָרָע זְרַע זָרַע זָרָע (Gen 11:29).<sup>25</sup> The effected accusative is rare except with a denominative verb of the same root. It is also concrete and external to the verbal action, and therefore distinct from the *internal accusative* (see the following).

(c) Internal – The object is an abstract noun of action, and implies an action in most cases identical with that of the verb:<sup>26</sup> הַחָרוֹ פַּחַרוֹ [literally: they feared a <u>fear</u>]" (Ps 14:5). The object itself may precede the verb: הַמָאָה יְרוּשָׁלָם, "Jerusalem sinned <u>grievously</u> [literally: sinned a <u>sin</u>]" (Lam 1:8). As these two examples illustrate, the object is a noun formed from the same root as the verb (so technically also *cognate accusative*), is generally indefinite, and as the following examples show, is frequently modified by

- <sup>25</sup> This same idiom elsewhere (קוֹרְעָ אָה־וֹרְעָהָ), Deut 11:10, and compare 22:9) is not an *effected accusative* because as the context makes clear and as all the grammars recognize, the seed existed prior to the sowing and is not brought about by the action. Thus, not every *cognate accusative* is an effected accusative (Joüon and Muraoka 1993, 449–50; Waltke and O'Connor 1990, 166).
- <sup>26</sup> For the following, see Joüon and Muraoka 1993, 450–51; Kautzsch 1910, 366–67; and Waltke and O'Connor 1990, 167.

an attributive adjective: אָרָרָה גְּרֹלָה, "and Isaac trembled <u>violently</u> [literally: trembled a great <u>trembling</u>]" (Gen 27:33), וַיֵּךְ יְהוָה בָּעָם מַכָּה רַהָּה מָאֹר, "and Yhwh struck the people with a very great plague [literally: struck . . . with a very great <u>striking</u>]" (Num 11:33).

The concrete noun קוֹל, "voice," takes special uses of the *internal accusative*. Although it has no corresponding verb, קוֹל appears to be used in a way similar to the internal accusative with verbs expressing an emission of voice (Joüon and Muraoka 1993, 451; Kautzsch 1910, 367– 68). Thus, we find it as an internal accusative noun of action with איקר, בָּכָה , בָּכָה , מָרָא מוט usually modified as well: אָמָר אָל־פָּל־אָישׁ יִשְׂרָאָל קוֹל רָם "Then the Levites shall declare <u>in a loud voice</u> to all the Israelites" (Deut 27:14), וִיְבָרָך אָת כָּל־קְהַל יִשְׂרָאָל קוֹל נָרוֹל, "and he [Solomon] blessed all the assembly of Israel <u>with a loud voice</u>" (1 Kgs 8:55).

(*d*) *Complement* – The object is a noun used with certain intransitive verbs, which take on new transitive meanings through a modification of their original meanings (Kautzsch 1910, 368–70; Waltke and O'Connor 1990, 168). Verbs often taking the *complement accusative* include רִיבֹּר אָלְמָנָה, רְיבּוּ אָלְמָנָה, מָשָּׁרָ אָיָבִי יְכָל רִיבּוּ אָלְמָנָה אַיְבִי יְכָלְתִין, and שָׁכַב So, for example: רְיבּוּ אַלְמָנָה, "lest my enemy say, 'I have prevailed <u>against him</u>'" (Ps 13:5), my enemy say, 'I he king is delighted <u>with you</u>" (1 Sam 18:22).

Most of these occur in the Qal stem, but occasionally even the reflexive stems (Niphal and Hithpael) may take such an accusative:

# וַיִּתְפָּרְקוּ כָּל־הָעָם אֶת־נִזְמֵי הַזָּהָב אֲשֶׁר בְּאָזְנֵיהֶם

"and all the people took off <u>the rings of gold</u> from their ears" (Exod 32:3). Other verbs sometimes taking the *complement accusative* are verbs of robing and disrobing (עָרָה), and פָשָׁט, עָרָר, אָרָרָע, בָּרָשָ), verbs of fullness and emptiness (אָרָה), אָבָרָש

נְזַל ,פָּרַץ, גְּבַר , אוֹ , and others), as well as verbs of dwelling (יְשָׁב), יְשָׁב).<sup>27</sup>

(e) **Double** – The causative derived stems (Piel and Hiphil) and certain other verbs sometimes have more than one object.<sup>28</sup> Both of the accusatives are often direct objects, meaning we may speak of a *double object accusative*, although occasionally the second of two objects is a *complement accusative* or an *adverbial accusative*. In the case of the latter, we may speak rather of a *complex accusative*.

(e.1) In causative sentences, the Piel or Hiphil of verbs that are Qal transitives frequently require two accusatives. Typically, the subject of the causation (often a person) is used as a second accusative: לְמַרְהֵי אָתְכֶם חָקִים וּמִשְׁפַּטִים, "I have taught you statutes and ordinances [literally: I have caused you (subject of causation/accusative #2) to learn *statutes and ordinances* (accusative #1)]" (Deut 4:5), ויאַכָלְנִי אָת הַמָּגְלָה הַוֹּאַת, "and he [Yhwh] fed me this scroll [literally: and he caused me (subject of causation/accusative #2) to eat *this scroll* (accusative #1)]" (Ezek 3:2). This *double accusative* is common with verbs of fullness and emptiness, robing and disrobing, and other causative constructions. In some, however, the first object is not a direct object accusative, but rather an adverbial accusative (see section 2.3.2): מְלָא אֹתָם הַכְמָת־לָב, "He [Yhwh] filled them with skill [literally: he caused them (subject of causation/accusative #2) to be full of skill (accusative #1)]" (Exod 35:35).

(e.2) The *double accusative* is also required by the nature of certain verbs that are not necessarily verbs of causation. So verbs of making, forming, naming, or counting often require two accusatives, the second

<sup>&</sup>lt;sup>27</sup> See Kautzsch 1910 (369–70) for details.

<sup>&</sup>lt;sup>28</sup> For details, see Kautzsch 1910, 370–72; Joüon and Muraoka 1993, 451– 54; Waltke and O'Connor 1990, 173–77; and van der Merwe, Naudé, and Kroeze 1999, 243–44.

יַרְכָנִה אֶת־הָאַרָנִים מִזְבָּח מִזְבָּח (object #1] <u>into an altar</u> [object #2]" (1 Kgs 18:32), יַרָּרָה אֶלְהִים אֶת־הָאָדָם עָפָר (1 Kgs 18:32), יַרָּיָר אָלָהִים אָת־הָאָדָם עָפָר (1 Kgs 18:32), יַרָיָר אָלָהִים אָלַהִים אָלַהִים אָרַהָאָדָם עָפָר (1 Kgs 18:32), "Then YHWH God formed <u>man</u> [object #1] <u>of dust</u> [object #2]" (Gen 2:7), יָרָאָרָז יָרָאָרָז (Gen 30:6). Likewise, verbs of speaking and giving often require two accusatives: יָרָקוֹן לָלַוּ אָת־הָאָרָק'ן (biject #1] <u>this land</u> [object #2]" (Deut 26:9), יָרָאָרָז הַאָרָק אָתַכָּק בָּעַת הָהוֹא אָת כָּל־הַדְּבָרִים אָשֶׁר תַעֲשׁוֹן, "And I commanded <u>you</u> at that time <u>all the things</u> that you should do" (Deut 1:18).

#### 2.3.2 Adverbial

In addition to the simple direct object of a verb, some accusative nouns are indirectly subordinated to the verb. Rather than specifying the person or thing directly affected by the verb, such accusatives modify more precisely the circumstances under which an action or situation occurs. In other words, they serve as *adverbial* accusatives.<sup>30</sup>

(a) Place – The accusative signifies location with verbs of motion or dwelling: צא הַשָּׁרָה, "go out (into) <u>the field</u>" (Gen 27:3), אַצָּא אַשׁוּר, "he went (into) <u>Assyria</u>" (Gen 10:11), רַיְרָדוּ כָּלִישָׂרָאֵל הַפָּלְשְׁתִים, "and all Israel went down (to) <u>the Philistines</u>" (1 Sam 13:20), האָשָׁה בֵּית פָּרְעֹה, "and the woman [Sarai] was taken

<sup>30</sup> Often such syntactical relationships are designated by prepositional phrases, especially taking the place of accusatives of time or place (Kautzsch 1910, 377–84). Instead of the *accusative of place*, the *he locale* (Lambdin 1971a, 51–52; Seow 1995, 152–53) is often used. On the *adverbial accusatives* in general, see Joüon and Muraoka 1993, 455–59; Kautzsch 1910, 372–76; Waltke and O'Connor 1990, 169–73; Williams 1976, 12–14; van der Merwe, Naudé, and Kroeze 1999, 244–45; and Chisholm 1998, 64–65.

<sup>&</sup>lt;sup>29</sup> See the accusatives of *material* and *product* in 2.3.2,f, and 2.3.2,g.

(into) <u>Pharaoh's house</u>" (Gen 12:15), וְיַעֲמֹד פָּתַח הָאָהֶל (Gen 12:15), "and he [YHWH] stood (at) <u>the doorway of the tent</u>" (Num 12:5). The location may specify the spacial extent of the verbal action: וְיָרְדֹך אֶחֲרִיו דֶרֶך שֶׁרְעַת יְמִים, "and he pursued after him <u>a seven-day journey</u>" (Gen 31:23), "and he pursued after him <u>a seven-day journey</u>" (Gen 31:23), "and we went (through) all that great and terrible wilderness" (Deut 1:19). Sometimes the accusative of place is used with verbs that have no specific locational quality: ""and you yourself will hear (in) <u>heaven</u>" (1 Kgs 8:32).

(b) Time – The accusative locates an action in time or states its duration: וָאָרֹ הָעָיָן, "and I came <u>today</u> to the spring" (Gen 24:42), דְּלֶך אָעֶרְדְּלֹך (Gen 24:42), "הָוָה בֹּקָר אָשֶׁרָי בֹּקָר אָשֶׁרָ בֹּקָר אָשֶׁרָ דְּלֹך (Gen 24:42), "in <u>the morning</u> you hear my voice; <u>in the morning</u> I plead my case to you" (Ps 5:4), אָבָרי מַקטיר עַל־הָאָרֵץ אַרְבָּעִים יום וְאַרְבָּעִים לְיָלָה, "I will send rain on the earth <u>forty days and forty nights</u>" (Gen 7:4), "I will send rain on the earth <u>forty days and forty nights</u>" (Gen 7:4), "(Ps 61:5).

(c) Manner – The accusative is an indefinite noun describing the way in which an action or situation is performed: וְיָשֶׁכֹן יִשֶׂרָאֵל בָּטַח. "and Israel will swell <u>in safety</u>" (Deut 33:28), דְּבְרֵי הַנְּבְיָאִים פָּה־אֶחָד מוֹב אָל־הַמֶּלֶן. "and Israel will swell <u>in safety</u>" (Deut 33:28), דְּבְרֵי הַנְּבִיאִים פָּה־אֶחָד מוֹב אָל־הַמֶּלֶן. "מוֹם אָל־הַמָּלָן. words of the prophets <u>with one accord</u> are favorable to the king" (1 Kgs 22:13), דְּבָרֵי מִוֹל אֶחָד מוֹם אָלָד הַמָּלָן. "and all the people answered <u>with one voice</u>" (Exod 24:3). Adjectives and participles may also serve as accusatives of manner: נוֹעָל הָרָי הַנָּבָר הַוֹלֵך עַרִירי of Shechem" (Gen 33:18),<sup>31</sup> וְיָבֹא יְעָכְם לָוֹל דְּרָי הַנָּהַן, "and Jacob came <u>safely</u> into the city of Shechem" (Gen 15:2), בִיְבָרִי הַנָּהַרְיָם, הַנָּהַרָהָ בָּא מְדַלֵּג עַל־הָהָרִים, "and I go childless" (Gen 15:2), כָּרָהָרָים, הַנָּא מְדֵלֵג עַל־הָהָרִים, "the voice of my beloved; behold he comes <u>leaping</u> upon the mountains" (Song 2:8).

<sup>&</sup>lt;sup>31</sup> Notice that this example also has a second accusative, which is an *accusative of place:* "into the city of Shechem."

(*d*) State – The accusative is an indefinite noun describing a feature, state, or quality of the *subject* at the time of the verbal action:<sup>32</sup> אַשָּׁשָׁה רָאָשָׁים מָלַשָּׁה רָאָשָׁים "And raiders came out of the camp of the Philistines <u>in three companies</u>" (1 Sam 13:17), ביאָרָרָים, "and the Arameans went forth <u>as maurading bands</u>" (2 Kgs 5:2), "עָרָי אָרָיָך", "remain <u>as a widow</u> in your father's house" (Gen 38:11).<sup>33</sup>

The accusative may also describe a feature, state, or quality of the *object* at the time of the verbal action: אָרָיָהַמָבוּל מֵיָם, "and I myself [YHWH] am bringing a flood <u>of waters</u>" (Gen 6:17), "march on, my soul, <u>in strength</u>" (Judg 5:21), הְרָאָרָי יְהָוֶה אֹתְךָ מֶלֶךְ עַל־אָרָם, "YHWH has shown me that you will be <u>king</u> over Aram" (2 Kgs 8:13).

Like the accusative of manner, the accusative of state may be expressed by adjectives or participles: אָרֶד אָל־בְּנִי אָבֶל שָׁאֹלָה, "I will go down to the grave to my son <u>mourning</u>" (Gen 37:35), אַנִי מְלָאָה הָלָכְתִי, "I went away <u>full</u>" (Ruth 1:21), אָנִי הָעָם בֹּכָה, "and Moses heard the people <u>weeping</u>" (Num 11:10).

(e) Specification – The accusative clarifies or explains further the verbal action, which would otherwise be generally or ambiguously stated:<sup>34</sup> הָרָרְלָלוּ, "he was diseased <u>in his feel</u>" (1 Kgs 15:23), לא נַכָּנּוּ נָכָּפּוּ נָכָּפּוּ not strike him <u>to death</u> [literally: <u>with respect to life</u>]" (Gen 37:21), רְק הַכָּפָא אֶגְדַל מְמֶדְ), "it is only (with respect to) <u>the throne</u> that I shall be greater than you" (Gen 41:40), <u>the throne</u> that I shall be greater than you" (Gen 41:40), (with respect to) <u>the flesh of your foreskins</u>" (Gen 17:11).

<sup>&</sup>lt;sup>3<sup>2</sup></sup> This use of the accusative is similar to the *accusative of manner*, and some grammars combine the treatment of the two (cf. Kautzsch 1910, 374–75).

<sup>&</sup>lt;sup>33</sup> Note that this example contains two accusatives, the first an *accusative of state* (widow) and the second an *accusative of place* (father's house).

<sup>&</sup>lt;sup>34</sup> May also be called the *epexegetical accusative* or the *accusative of limitation*.

(f) Material – The accusative denotes the matter used in the verbal action.<sup>35</sup> This use of the accusative is common with verbs for making, forming, or building, which typically take two accusatives; the first an *object accusative*, the second an *accusative of material*: רְיָרְאָל אֶכן, "and all Israel stoned him <u>with stones</u>" (Josh 7:25), "and all Israel stoned him <u>with stones</u>" (Josh 7:25), "and all Israel stoned him <u>with stones</u>" (Josh 7:25), "and <u>from dust</u>" (Gen 2:7), בְיָבֶר יְהָוָה אֶלֹהִים אֶת־הָאָדָם עָפָר you shall make two cherubim <u>of gold</u>" (Exod 25:18), you shall make two cherubim <u>of gold</u>" (Exod 25:18), (1 Kgs 7:27).

At times, it is impossible to determine by morphology alone whether the relationship between two nouns is genitival, accusatival, or appositional (compare sections 2.2.10, 2.3.2,f and 2.4.3).<sup>36</sup>

(g) Product – The accusative denotes the result of the verbal action. This use of the accusative is common with verbs for making, preparing, or shaping, which typically take two accusatives; the first an *object accusative*, the second an *accusative of product*: אָת־בָּנִיו שׁבְּטִים לְיִשָּׁרָאָל, "and he [Samuel] made his sons *judges* over Israel" (1 Sam 8:1), "and he built the stones *into an altar*" (1 Kgs 18:32), "in and he built the stones *into an altar*" (1 Kgs 18:32), הַלָּכְחַתָּ סֹלֶת וְאָבִית אֹתָה שָׁתִים עֶשָּׁרָה חַלּוֹת, "and you shall take choice flour, and bake it *into twelve cakes*" (Lev 24:5).

# 2.4 Apposition

In addition to the three main case functions (nominative, genitive, and accusative), BH also uses nouns in apposition, that is, in simple juxtaposition in order for the second noun (the *apposition*) to modify or elucidate the first

<sup>&</sup>lt;sup>35</sup> For more on this and the next (*accusative of product*), see Kautzsch 1910, 371–72 and Waltke and O'Connor 1990, 174–75.

<sup>&</sup>lt;sup>36</sup> Jouon and Muraoka 1993, 460–61; Waltke and O'Connor 1990, 173.

(the *leadword*).<sup>37</sup> Such nouns generally agree in gender, number, and definiteness, have the same function in the syntax, and refer to the same person, place, or thing in the external world. The apposition may often be translated by an adjective or prepositional phrase.<sup>38</sup> This list includes the most common ways in which apposition is used.<sup>39</sup>

#### 2.4.1 Species

The apposition denotes the species or subclass to which the leadword belongs: זְרָחִים שֶׁלְמִים, "peace-offerings [literally: sacrifices, <u>peace-offerings</u>]" (Exod 24:5). This usage is common when the leadword is a common noun for a person: אָישׁ כָּהַן, "a priest [literally: a man, <u>a priest</u>]" (Lev 21:9), "an Egyptian [literally: a man, <u>an Egyptian</u>]" (Exod 2:11), אָיָשָׁר הָנָּבִיא, "the young prophet [literally: the young man, <u>the prophet</u>]" (2 Kgs 9:4), "brothers [literally: men, <u>brothers</u>]" (Gen 13:8).

## 2.4.2 Attributive

The apposition denotes a quality or attribute of the leadword. In translation, the apposition often becomes an adjective: אָמָרִים אָמָרִים אָמָרִים (Prov 22:21), אָלְהִים אָמָרִים גָחָרִים (Jer 10:10), (Prov 22:21), אָלָהִים אָמָרִים גָחָרִים (Jer 10:10), הָרָרִים נָחָרִים (Tech 1:13), הָלָשׁוֹן רְמִיָּה (from a <u>deceitful</u> tongue" (Ps 120:2).

- <sup>37</sup> Pronouns may also occur as the leadword (Waltke and O'Connor 1990, 232–33; Williams 1976, 15–16; and van der Merwe, Naudé, and Kroeze 1999, 230).
- <sup>38</sup> Kautzsch 1910, 423–27; Waltke and O'Connor 1990, 226–34; Joüon and Muraoka 1993, 477–81; van der Merwe, Naudé, and Kroeze 1999, 228–30; Williams 1976, 15–16.
- <sup>39</sup> On the emphatic and distributive uses of *repetitive apposition*, see Waltke and O'Connor 1990, 233–34, and van der Merwe, Naudé, and Kroeze 1999, 230. On apposition in epigraphic Hebrew, see Gogel 1998, 237–40.

#### 2.4.3 Material

The apposition denotes the material from which the leadword is made. Similar to the *attributive apposition*, this often becomes an adjective in translation: הְשָׁבְתָּת הַזָּהָם "gold cords [literally: the cords, <u>the gold</u>]" (Exod 39:17), "<u>gold</u> cords [literally: the cords, <u>the gold</u>]" (Exod 39:17), ", <u>הָאָב</u>ן הַבְּדִיל, "the <u>bronze</u> oxen" (2 Kgs 16:17), ", <u>הָאָב</u>ן הַבְּדִיל, "the <u>bronze</u> oxen" (2 Kgs 16:17), "<u>the alloy</u> stone (i.e., plummet)" (Zech 4:10), הַבָּקָר הַנָּחַשָּת "<u>brass</u> cymbals" (1 Chr 15:19). Occasionally the apposition may omit the definite article: הַמַבּוּל מֵיִם, "the flood (of) <u>waters</u>" (Gen 6:17), אַשֵׁרָה כָּל־עֵיָן, "Asherah made of <u>every</u> <u>kind of wood</u>" (Deut 16:21).

## 2.4.4 Measure

The apposition specifies the thing measured or weighed, in which case the leadword is a measuring unit: סָאָה־סֹלֵת, "a seah [unit of dry measurement] of <u>flour</u>" (2 Kgs 7:1), "two <u>silver</u> talents" (1 Kgs 16:24).

Although the *apposition of measure* is not particularly common, a variation of it using numerals is. The apposition thus specifies the thing numbered, while the leadword is a numeral: שָׁבְעָה בָנִים וְשָׁלוֹשׁ בָּנוֹת, "seven <u>sons</u> and three <u>daughters</u>" (Job 1:2), הַמְשָׁה אָנָשִׁים, "five men" (2 Kgs 25:19), שָׁנִים הֶעַשָּׁר אָישׁ, "seventy sons" (Judg 8:30), שָׁבִעִים בָּנִים, "the twelve men" (Josh 4:4).40 Sometimes, however, the order is reversed, in which case the apposition is the numeral and the leadword specifies the thing numbered. This is often the case in inventories or enumerated lists: עִזִים מַאתַיָם וּתִיַשִׁים עֵשָׂרִים רְחֵלִים מַאתַים וָאֶילִים עֲשָׂרִים, "two hundred female goats and twenty male goats, two hundred ewes and twenty rams" (Gen 32:15), בַקר שְׁנַיִם אֵילִם חֵמִשָּׁה עַתּוּדִים חֵמִשָּׁה כִּבַשִּׁים בִּנִי־שָׁנָה חֵמִשָּׁה, "two oxen, five rams, five male goats, and five male lambs a year old" (Num 7:17, and compare Num 28:19).

<sup>&</sup>lt;sup>40</sup> Compare the *genitive of measure* in 2.2.11.

## 2.4.5 Explicative 41

The apposition is a particular member of a general category denoted by the leadword, and typically gives the proper noun for the leadword: הָאָרֶ שְׁלְמֹה, "King <u>Solomon</u>" (1 Kgs 1:34), הַאָרָן פָּרָת, "the River <u>Euphrates</u>" (1 Chr 5:9), הָאָרֶץ בְּנָעָן, "Canaan [literally: the land, i.e., <u>Canaan</u>]" (Num 34:2). Occasionally the order is reversed, in which case the apposition specifies the category and the proper noun is the leadword: שָׁלְמֹה הַמֶּלֶך, "<u>King</u> Solomon" (1 Chr 29:24).<sup>42</sup>

This becomes an *apposition of relationship* when the leadword is a noun of kinship: אֶת־אָבִיל, "his brother <u>Abel</u>" (Gen 4:2). The order here may likewise be reversed: הֶבְל אָחִיו, "Abel <u>his brother</u>" (Gen 4:8).<sup>43</sup>

## 2.5 Adjectives

As might be expected, nouns may also be modified by adjectives. BH makes no morphological distinction between nouns and adjectives (although the adjective has no dual form).<sup>44</sup> The distinction between a noun and an adjective is left to the inherent lexical meanings of the terms and to their syntactical functions. Since most beginning grammars detail the uses of the adjective quite clearly, their syntactical features will only be summarized here.<sup>45</sup>

- <sup>41</sup> The *explicative genitive* is more common than this *explicative apposition* (see 2.2.12 and Joüon and Muraoka 1993, 479).
- <sup>42</sup> When the apposition is a proper noun requiring a preposition or DDO (אָר־/אָר־/אָר), the particle is generally repeated on both words. If the lead-word has the proper noun, the particle is not repeated. See Waltke and O'Connor 1990 (232), Joüon and Muraoka 1993 (479–80), Kautzsch 1910 (425), and Williams 1976 (15).
- <sup>43</sup> Compare the genitive of relationship (2.2.3).
- <sup>44</sup> Plural adjectives are thus used when modifying dual nouns (van der Merwe, Naudé, and Kroeze 1999, 231). Occasionally, adjectives can carry a vowel pattern distinct from the nouns.
- <sup>45</sup> Lambdin 1971a, 13–15; Seow 1995, 72; Pratico and Van Pelt 2001, 61–64;
   Kittel, Hoffer, and Wright 1989, 251; Kelly 1992, 45–47; Weingreen 1959, 32–33.

Adjectives modify nouns by describing their state or condition. This list includes the most common ways in which the adjective is used. For more of what follows, the reference grammars may be consulted.<sup>46</sup>

## 2.5.1 Attributive

The adjective ascribes a quality to a noun. The *attributive adjective* forms a phrase with the noun it modifies, having a single function in the sentence. The adjective stands in apposition to the noun it modifies, normally following it and agreeing in gender, number, and definiteness: אָישׁ בָּרוֹל, "a <u>great</u> man" (2 Kgs 5:1), אָשָׁה חֵכָםָה, "a <u>wise</u> woman" (2 Sam 14:2), הַעִיר הַבּּרֹלָה, "the <u>great</u> city" (Gen 10:12).

If the noun occurs in construct state with another noun, the attributive adjective follows the entire genitive construction: אָישׁ אָלהִים קָרוֹשׁ, "a <u>holy</u> man of God" (2 Kgs 4:9). Wherever both nouns and the adjective are of the same gender, ambiguity may occur, and it can only be made clear by broader context. A series of attributive adjectives may occur after the noun modified, usually with a demonstrative adjective occurring in the final position: הָהָר הַטוֹב הָאָה (<u>this great</u> nation" (Deut 4:6).<sup>47</sup>

Rarely, the adjective precedes the noun it modifies, most often with numerals and רְבּוֹת/רְבִים, probably because "many" was thought to be similar to numerals in function: הבים צַיָּרִים (Jer 16:16).

The definite article of the attributive adjective is sometimes omitted, perhaps because certain adjectives were thought inherently definite (such as רְבָים and , as

<sup>&</sup>lt;sup>46</sup> Kautzsch 1910, 427–32; Waltke and O'Connor 1990, 255–71; Joüon and Muraoka 1993, 521–25; Meyer 1992, 371–76; Horsnell 1999, 193–201; Chisholm 1998, 66–67; van der Merwe, Naudé, and Kroeze 1999, 230– 37; Williams 1976, 16–18. On the adjectives in epigraphic Hebrew, see Gogel 1998 (202).

<sup>&</sup>lt;sup>47</sup> For more details of the syntactical peculiarities of two or three adjectives following a noun, see Kautzsch 1910 (428).

well as the numeral אָחִיכֶם אָחִירָם אָחִיכָם אָחִיכָם אָחִיכָם אָחִיכָם אָחִירָם, "one"): אָדָחִיכָם אָחִי brother" (Gen 43:14), הָרָאשׁ אָחָד, "<u>one</u> company" (1 Sam 13:17), הַגּוֹיִם רַבִּים, "<u>many</u> nations" (Ezek 39:27). At other times, the noun being modified is inherently definite and therefore anarthrous, even while the adjective retains the article: יוֹם הַשָּשׁׁי, "the <u>sixth</u> day" (Gen 1:31).

## 2.5.2 Predicate

The adjective expresses an assertion about a noun requiring a form of the verb "to be" in translation. The *predicate adjective* often precedes the noun it modifies and agrees in gender and number, but is always indefinite:<sup>48</sup> (אָרוֹל שָׁנוֹי, "my punishment <u>is great</u>" (Gen 4:13), מוֹב הְעֵרֹי, "the tree <u>was good</u>" (Gen 3:6), אָרֵייָהָן, "righteous <u>is YHWH</u>" (Ps 145:17), הָשָׁר הְבָר־יְהָעָה, "the word of YHWH <u>is upright</u>" (Ps 33:4). The predicate adjective is often used in nominal clauses of description (see section 5.1.1,b).

## 2.5.3 Substantive

The adjective functions as a noun, most often occurring with the definite article: הַחְכָמִים, "the <u>skillful men</u> [literally: <u>the wise men</u>]" (Exod 36:4), though also without the definite article, as in הַכְמִים הָשָׁמָשׁ ", "<u>wise men</u>" (Job 5:13).

The substantive adjective may serve in any of the normal functions of a regular noun: as nominative (הַקְּמֹן), "the <u>young one</u>" [Gen 42:13]), as either noun of the genitive relationship (בִית בָּרוֹל, "<u>great</u> house [literally: house of a <u>great one</u>]" [2 Kgs 25:9]),<sup>49</sup> as accusative (literally: house of a <u>great one</u>]" [2 Kgs 25:9]),<sup>49</sup> as accusative (<u>present</u>), "<u>the righteous</u>" [Eccl 3:17]), as apposition (בִרֹקָל בְרָכָּב), "the <u>righteous one</u>, my servant" [Isa 53:11]), or as object of a preposition (בִּרֹקַל בְרָכָּב), "we will ride upon <u>swift horses</u> [literally: upon a <u>quick one</u>]" [Isa 30:16]).

<sup>&</sup>lt;sup>48</sup> The tense of the translation must be determined from the Hebrew context.

<sup>&</sup>lt;sup>49</sup> And see the *genitive of specification* in 2.2.13.

## 2.5.4 Comparative and Superlative

Hebrew has no inflected forms of the adjectives to express degrees of the adjectival quality, such as in English: bigbigger-biggest, great-greater-greatest, and so on. Instead, BH uses the adjectives in a number of syntactical variations to express the comparative and superlative degrees.

(a) Comparative – A higher degree of the adjectival quality is expressed when the preposition מן is used with the noun being surpassed (see section 4.1.13,h). The preposition identifies the standard against which the noun is being compared, and typically the adjective has no definite article: אָרָה מְרָנִיאָל, "you are <u>wiser than</u> Daniel" (Ezek 28:3 *Qere*), קָכָם אָרָה מְרָנִיאָל, "<u>taller than</u> all the people" (1 Sam 9:2), "<u>stronger than</u> a lion" (Judg 14:18), נְרָוֹל יְהָוֹה מְרָנִיאָל הִים, "Yhwh is <u>greater than</u> all gods" (Exod 18:11).<sup>50</sup>

Sometimes the *comparative* denotes a condition that is too little or too much in force for attainment:<sup>51</sup> קּרְכָבֵד מְמָךְ הַדְּכָר (Exod 18:18), הַמְעַם מִכָם, "*too slight a thing* for you" (Isa 7:13). In such uses, the preposition מון may be attached to an infinitive construct rather than a noun: גַּדְוֹל עֲוֹנִי מִגְּשָׁא , "my punishment is *too great* to bear" (Gen 4:13), גָּדוֹל עֲוֹנִי מְשָׁם רְב מִשֶׁבָת יַחְדָוֹ , "Their property had become *too great* for them to live together" (Gen 36:7), קיֹה הָלַבְנִי יְהָוָה קָטֹן מֵהְכִיל אֶת־הָעֹלָה , "For the bronze altar which was before YHWH was *too small* to hold the burnt offering" (1 Kgs 8:64).

- <sup>50</sup> At times the comparative is required even without the preposition גָלן, and at other times the adjective is replaced by stative verbs. In some cases, the context must determine when the comparative is intended (see Kautzsch 1910, 430–31).
- <sup>51</sup> See Waltke and O'Connor's *comparison of capability* (1990, 266) and Joüon's *elliptical comparison* (1991, 523–24); also see Kautzsch (1910, 430).

(b) Superlative – The highest degree of the adjectival quality is expressed when the adjective is made definite, whether by a definite article, a genitive construction,<sup>52</sup> or a pronominal suffix: דָקָשָׁן, "the <u>youngest</u> [of eight sons]" (1 Sam 16:11), יקָשׁן בָּנָין, "the <u>youngest</u> of his sons" (2 Chr 21:17), "from <u>the greatest</u> of them to <u>the least</u> of them" (Jonah 3:5).

ד ה superlative adjective may rarely take the preposition אין with a noun: הַיָּפָה בַּנְשִׁים, "the <u>most beautiful</u> among women" (Song 1:8), האָיש הָרָך בָּך (<u>The most refined person</u> among you" (Deut 28:54).<sup>53</sup>

#### 2.6 Determination

Nouns in BH are indeterminate (i.e., indefinite) unless marked otherwise. They may be marked as determinate/ definite in three ways: a) by a prefixed definite article, b) by a pronominal suffix, or c) by occurring in the construct state bound to a definite noun.<sup>54</sup> Proper nouns and other appellatives are also definite by nature and do not require other determination, since they name a specific person or thing.

In general, indefinite nouns tend to focus on the class or quality of the person, place, or thing, while definite nouns focus on a unique or particular identity. Although BH has no *indefinite* article, the numeral "one" (אָחָר/אָאָקד) is used to

<sup>53</sup> The superlative may be expressed in a number of other ways, including the superlative genitive (2.2.13), repetition, the use of אָרָר מָאָר סי מָאָר סי מָאָר מָר מָרָל מָרָל מָרָל after an adjective, or מְכָל מָרָל מָכָל מָכָל 1990, 198-201; Kautzsch 1910, 431-32; Waltke and O'Connor 1990, 267-71; van der Merwe, Naudé, and Kroeze 1999, 236-37; and Williams 1976, 17-18. On the distinction between the absolute superlative (excels all others in its class) and the comparative superlative (excels in some particular quality or condition), see Ben Zvi, Hancock, and Beinert 1993, 192, and Waltke and O'Connor 1990, 267-71.

<sup>54</sup> In all three cases, the mark of determination causes morphological changes, for which, see the beginning grammars.

<sup>&</sup>lt;sup>52</sup> That is, a *substantive* adjective serving as a construct in bound relationship with a noun, as in גָּרֹלִי הָעִיר , "*the greatest men* of the city" (2 Kgs 10:6).

denote indetermination, especially in the books of Judges, Samuel, and Kings (see section 2.7.1,b).<sup>55</sup> This often requires the translation "certain": וְיָהִי אָישׁ אֶחָד מִן־הָרְמָתִים, "There was a *certain* man from Ramathaim" (1 Sam 1:1).

The use of the definite article in BH is similar to the use of the article in English. However, there are also several uses of the article in BH that are dissimilar to use in English. In these cases, definiteness in the English translation will often not correspond to the definiteness of Hebrew. It is important, therefore, not to expect a literal, word-forword translation of the definiteness of nouns, which may lead to awkward, or even incorrect English phrasings. Additionally, since the definite article developed late in the history of Classical Hebrew,<sup>56</sup> its sparse use in poetry is a key distinction between prose and poetry in general. Compare, for instance: זַכָּר לְאֵלֶף דּוֹר, "He has remembered his covenant to eternity, the word which he commanded to a thousand generations" (Ps 105:8), יַקָם עַד־עוֹלָם, "<u>The word</u> אָשֶׁר דִּבַּרְהָ עַל־עַבְדְךָ וְעַל־בֵּיתוֹ הָהֵם עַד־עוֹלָם, "<u>The word</u> which you spoke concerning your servant and his house, confirm it forever" (2 Sam 7:25).

The following categories are useful when considering the uses of the definite article in BH.<sup>57</sup>

#### 2.6.1 Referential

A noun with the definite article can refer to a person or thing already introduced into the context: ווֹיאמָר אֵלהִים יָהִי אוֹר נִיְהָי־אוֹר נִיָּרָא אֵלהִים אֶת־הָאוֹר כִּי־מוֹב.

- <sup>55</sup> Joüon and Muraoka 1993, 513; see also Waltke and O'Connor 1990, 251– 52; Pratico and Van Pelt 2001, 49; and van der Merwe, Naudé, and Kroeze 1999, 187.
- <sup>56</sup> Garr 1985, 89; Joüon and Muraoka 1993, 507; Seow 1995, 157; Andersen and Forbes 1983, 165–69.
- <sup>57</sup> Meyer 1992, 367–70; Joüon and Muraoka 1993, 505–21; Waltke and O'Connor 1990, 235–52; Kautzsch 1910, 401–13; Chisholm 1998, 72–75; van der Merwe, Naudé, and Kroeze 1999, 187–91; and for comparison with epigraphic Hebrew, see Gogel 1998, 235–37.

"God said, 'Let there be light,' and there was light. And God saw that <u>the light</u> was good" (Gen 1:3-4); אָרָקָר בָּקָר, בָּקָר, בָּקָר, בָּקָר, הַבָּקָר, "And he (Abraham) took a calf ... then he took curds and milk, and <u>the calf</u>" (Gen 18:7-8). If the person or thing is particularly well known, it need not have been mentioned previously: וְיָשְׁמֹן אֹתָם יֵעֵקֹב בַּחַת הָאֵלָה אֲשֶׁר עִם־שָׁכָם "מָמָל אַתָּם יַעַקֹב בַּחַת הָאֵלָה אֲשֶׁר עִם־שָׁכָם "מוּל", "and Jacob hid them [the foreign gods] under <u>the oak</u> that was near Shechem" (Gen 35:4).

#### **2.6.2** Vocative

A noun with the definite article can denote a specific addressee (see section 2.1.3): בָּן־מָי אָבָה הַנְּעַר, "Whose son are you, *young man*?" (1 Sam 17:58), הוֹשָׁעָה הַמֶּלֶך, "Help, <u>O King</u>!" (2 Sam 14:4).<sup>58</sup>

## 2.6.3 Naming

The definite article can mark a common noun as a proper noun. Since proper nouns themselves denote particular persons, places, or things, they do not normally take the definite article: הַשָּׁה, "Moses" (Exod 2:10), דְּוָה, "David" (Ruth 4:22), הָדָרָשָה, "YHWH" (Gen 2:4). However, some nouns (or primitive appellatives) are still in the process of becoming proper names:<sup>59</sup> הַבְּרָשָׁה, "Gibeah [literally: <u>the</u> hill]" (Judg 19:14), הַיָּרָשָׁה, "<u>the</u> Nile [literally: <u>the</u> stream]" (Gen 41:1), הַיָּרְבָשָׁה, "<u>the</u> Jordan [literally: <u>the</u> river]" (Gen 13:10), הַיָּרְבָשָׁה, "Lebanon [literally: "<u>the</u> white (mountains?)]" (Josh 9:1).

Related to this category is the *solitary* use of the definite article (see next section), in which appellatives referring

<sup>59</sup> Joüon and Muraoka 1993, 505; Meyer 1992, 370.

<sup>&</sup>lt;sup>58</sup> However, the vocative frequently omits the definite article. See Meyer 1992, 27; Kautzsch 1910, 405; Joüon and Muraoka 1993, 508.

to unique persons, places, or things are on their way to becoming a name: דְאָל הִים, "God [literally: <u>the</u> God]" (Gen 5:22).

Also related to this category is the *gentilic* use of the definite article, often used collectively (i.e., in the singular): הַכְּנַעֵּנִי, "<u>the</u> Canaanites" (Gen 10:18), but occasionally in the plural: הָעָבְרִים, "<u>the</u> Hebrews" (Gen 40:15).

### 2.6.4 Solitary

The definite article marks a unique person, place, or thing: הַשָּׁמְשׁ, <u>ה</u>להן הַגָּרוֹל, "<u>the</u> high priest" (Josh 20:6), הַשֶּׁמְשׁ, "<u>the</u> sun" (Gen 15:12), הַיָּרָחַ, "<u>the</u> moon" (Deut 4:19), <u>א</u>ת הַשְׁמִים וְאָר הָאָרָץ, "<u>the</u> heavens and <u>the</u> earth" (Gen 1:1).

Certain titles may take the definite article because they are understood by everyone to designate someone unique: הַמָּלֶך דָּוֹך , "King David [literally: <u>the</u> King David]" (2 Sam 5:3), הַמֶּלֶך שֶׁלֹמֹה, "Long live King Solomon [literally: <u>the</u> King Solomon]" (1 Kgs 1:34), הַשְּׁר, "<u>the</u> chief" (1 Chr 15:5).<sup>60</sup>

#### 2.6.5 Generic

A noun with the definite article denotes a class of persons or things: הַנְּמָל , "<u>the</u> camel" (Lev 11:4), בָּאֲשׁ, "with <u>the</u> fire" (Josh 11:9), הָרָעֵב, "a hungry person [literally: <u>the</u> hungry]" (Isa 29:8), בָּאַרי, "like <u>the</u> lion" (Num 24:9).

When used with a plural noun, the *generic* use may indicate that all the individuals of the class are included: הָרָשָׁעִים, "<u>the</u> wicked" (Ps 1:4), הַכּוֹכְבִים, "<u>the</u> stars" (Gen 1:16), הַלּוֹיָם, "the nations" (Gen 10:32).

<sup>&</sup>lt;sup>60</sup> The noun הַשָּׁר could be used in various constructions for titles, military and royal, although in LBH, it appears to have become a title standing alone with the article (*HALOT* 3:1351-52).

Use of the definite article to mark *abstract* nouns is related to this category: הַמְלוֹכָה, "<u>the</u>kingship" (1 Sam 11:14),

> וַיּמְלֵא אֶת־הַחָּכְמָה וְאֶת־הַתְּבוּנָה וְאֶת־הַדֵּעַת לַעֲשׂוֹת כַּל־מַלַאכָה בַּנָּחֹשֵׁת

"he [Hiram] was full of <u>skill, intelligence</u>, and <u>knowledge</u> in working bronze" (1 Kgs 7:14).

#### 2.6.6 Demonstrative

The definite article has a "pointing out" (i.e., deictic) force when used with nouns referring to present time.<sup>61</sup> Such *demonstrative* uses of the article typically take on adverbial force: פּה עִפְעוּ עִמִר הַיּוֹם לְפָנֵי יְהוָה אֱלֹהֵינוּ, "standing here with us *today* before YHWH our God" (Deut 29:14 [Eng 29:15]), לִינוּ פֿה הַכַּיְכָה, "this year you will die" (Jer 28:16), 22:8), הַשָּׁבָה אָקָה מֵת, "*this time* I have sinned" (Exod 9:27).

## 2.6.7 Possessive

The definite article occasionally denotes ownership of the noun, and requires a possessive pronoun in translation: וְהַחֲנִית בְּיֵר־שָׁאוּל (ז Saul had <u>his spear</u> in his hand" (1 Sam 18:10), וְהָשְׁרָאשׁ הַמְשָׁה (Saul had <u>his spear</u>), "Israel bowed himself on the head of <u>his bed</u>" (Gen 47:31), איי קור בעל הַרָאשׁ [King Eglon] rose from <u>his seat</u>" (Judg 3:20).

#### 2.7 Numerals

Numbers in BH are written as words, rather than represented graphically as with the Arabic digits used in English

<sup>&</sup>lt;sup>61</sup> It is often assumed that Hebrew's definite article originated from a prefixed pronominal with demonstrative force, explaining its occasional demonstrative use in BH as a vestige of its history. See Blau 1976, 43; Kautzsch 1910, 404; Joüon and Muraoka 1993, 506; Bergsträsser 1983, 23–24; Garr 1985, 87–89; but see the challenging discussion in Lambdin 1971b, 315–33.

(so "thirty-two" rather than "32").<sup>62</sup> Most are derived from substantives, but unfortunately their syntax is varied. Thus, the numeral "one" most frequently functions as an adjective, following the noun it modifies and agreeing in gender. But the numerals from "two" to "ten" function as nouns that may stand either before or after the noun they modify when appositional (for which, see section 2.4.4), or may stand before the noun in the construct state (see section 2.2.11). While the numeral "two" exhibits gender agreement with the noun it modifies, those from "three" to "ten" take feminine forms with masculine nouns, but masculine forms when modifying feminine nouns.<sup>63</sup> All beginning grammars of BH explain and illustrate these syntactical peculiarities of the numerals, and as a result we include here only a few additional categories useful when considering the uses of the numerals.<sup>64</sup>

Numerals are morphologically divided into the cardinal and ordinal variations.

#### 2.7.1 Cardinal Numbers

(a) Measure – Indicating the amount or quantity of something denoted by the noun being modified: שָׁלִשָה, (<u>"three</u> sons" (Gen 6:10), שָׁלֹשֶׁה אָנָשִׁים, "<u>three</u> men" (Gen 18:2), שָׁלִשָּׁה אָנָשִׁים, (<u>three</u> men" (Gen 18:2), (<u>three</u> weiter), (Josh 15:60).

(b) Indetermination – The numeral "one" (אַחַר or אָחָד) may follow an indefinite noun to mark an unnamed individual: וִיְהִי אָישׁ אָחָד מִן־הָרָמָתִים, "There was <u>a certain</u>

- <sup>62</sup> Most languages of biblical times used number signs, although Ugaritic preferred to spell numbers out completely as in BH (Segert 1984, 52–54).
- <sup>63</sup> This inexplicable feature of Hebrew numerals is limited to the cardinals. With double-figure numerals (11 to 19), the lowest number includes this reverse gender, while the tens take the gender of the noun being modified.
- <sup>64</sup> Kautzsch 1910, 286–92 and 432–37; Meyer 1992, 204–11; Joüon and Muraoka 1993, 525–30; Waltke and O'Connor 1990, 272–89; van der Merwe, Naudé, and Kroeze 1999, 263–70.

man of Ramathaim" (1 Sam 1:1), אַשָּה אָחַת, "<u>a certain</u> woman" (Judg 9:53), גָבִיא אָחָד, "<u>a certain</u> prophet" (1 Kgs 20:13). Occasionally this use takes on an emphatic nuance, often reflected in English by "single" or "same": מוכ, often reflected in English by "single" or "same": מוכ, often reflected in English by "single" or "same": אָבָרֶיך אָחִים אָבָחָנוּ בְּנֵי אָישׁ־אָחָד אישר, "We, your servants, are twelve brothers, the sons of <u>the same</u> man" (Gen 42:13), קבִיַת אָחָד, "on <u>the same</u> day" (1 Sam 2:34), "in <u>the same</u> house" (1 Kgs 3:17).

(c) Multiplication – The cardinals may express multiples of the noun being modified. Multiplicatives of "four" and "seven" are attested in the feminine duals of the cardinals:<sup>65</sup> הארבעקים ארבעקים, "and he shall restore the lamb <u>four times over</u>" (2 Sam 12:6), הארעקים, "<u>sevenfold</u>" (Gen 4:15).<sup>66</sup> Occasionally the feminine dual of "two" may be used for "twice": אַרָת וְלָא אוֹסִין וְלֹא אָאָעָה וּשָׁחַים וְלֹא אַרָּבָרְתִי וְלֹא אָאָעָה וּשָׁחַים וְלֹא "I have spoken once, and I will not answer; <u>twice</u>, but I will not again" (Job 40:5), בַּעַם וּשָׁתִים (Neh 13:20).

The singular also may be used for *multiplication*, often with פָּעָמִים הפָּעָם for "occurrence, times":<sup>67</sup> הָּעָמִים בַּשָּׁנָה (<u>three times</u> a year" (Exod 23:17), שָׁלֹש פְּעָמִים שָׁבָע, "<u>three times</u> (Gen 33:3),

וְיָסַפְּתִּי לְיַסְרָה אֶתְכֶם שֶׁבַע עַל־חַמּאתֵיכֶם

"and I will continue to punish you <u>sevenfold</u> for your sins" (Lev 26:18), מָאָה פָּעָמִים, "<u>one hundredfold</u>" (2 Sam 24:3).

(*d*) *Distribution* – Repetition of a cardinal number will often express a distributive sense: שַׁנֵים שָׁנֵים בָאוּ אֶל־נֹחַ אֶל־נֹחַ,

- <sup>65</sup> The ending may in fact be an adverbial ending rather than a dual proper (Joüon and Muraoka 1993, 327).
- <sup>66</sup> The forms are feminine possibly because of the omission of פַּעָם or רַשָּׁלָם, "times" (Kautzsch 1910, 436).
- <sup>67</sup> HALOT 3:952; see also גָּגָל (HALOT 3:1185) and גָּ (HALOT 2:388, DCH 4:82). A variation occurs when פַּעָם occurs in the dual (פַּעָמָיִם) without the numeral (Num 20:11).

"<u>Two by two</u>, they came to Noah into the ark" (Gen 7:9). The repeated numerals may be joined with the conjunction waw : אָאָבְּעֹת רְגָלִיו שָׁשׁוָשָׁשׁ, "<u>Six</u> fingers on his hands, and <u>six</u> fingers on his feet (2 Sam 21:20). The distributive concept may also be expressed by the repetition of the numeral with its object: אָשָּׁרְפִים עִׁמְרִים מִמַעַל לוֹ שֵׁשׁ כְּנָפִים שֵׁשׁ כְּנָפִים לָאָחָר stood above him, each one having <u>six wings</u>" (Isa 6:2). A numeral prefixed with the preposition *lamed* may also mark the distributive: וְסַרְנֵי פְּלְשָׁתִים עִרְרִים לְמֵאוֹת וְלָאֲלָפִים, "And the lords of the Philistines were marching <u>by the hundreds</u> and <u>by the thousands</u>" (1 Sam 29:2).

(e) Succession – The cardinal "one" (אָחַת or אָחָד) may function as an ordinal when denoting the first of a small list of items: וְיָהִ־בְּקֶר וֹם אָחָד, "And there was evening, and there was morning; the <u>first</u> day" (Gen 1:5), הַמוּר הָאָחָד, "the <u>first</u> row" (Exod 39:10).

## 2.7.2 Ordinal Numbers<sup>68</sup>

Succession indicates consecution or sequence, especially in formulae expressing dates:<sup>69</sup> בַּיּוֹם הָרָאשׁוֹן, "on the <u>first</u> day" (Exod 12:15), בַּיּוֹם הָרָאשׁוֹן, "in the <u>ninth</u> year of Hoshea" (2 Kgs 17:6), וּבְשָׁבָה שָׁנָה לְיוֹרָם, "In the <u>eleventh</u> year of Joram" (2 Kgs 9:29), בן שׁנִי, "a <u>second</u> son" (Gen 30:7).

- <sup>68</sup> The ordinals function as attributive adjectives, following the noun and agreeing in gender and definiteness (see 2.5.1). There are no separate forms for ordinals above the number 10 (review the beginning grammars for more on the forms).
- <sup>69</sup> However, the cardinals are frequently used instead of the ordinals in the expressions of date (Kautzsch 1910, 435; Waltke and O'Connor 1990, 284–86).

# 3 Verbs

**B** iblical Hebrew has no *tenses* in the strict sense of the term. By this statement, we mean that Hebrew does not locate an action or state in time by means of specific morphology. Of course, this does not mean that Hebrew fails to express time relations entirely, only that it does so through a variety of syntactical and contextual features, rather than through verbal inflections or grammatically realized tenses.<sup>1</sup>

Rather than a morphologically precise tense system, Hebrew uses an *aspect* system to indicate first, the contour of a situation in time, and second, the type of action. The

<sup>1</sup> Caution is in order in such statements as "BH has no tenses." We have expressed the majority opinion in this paragraph on the general nature of the Hebrew verbal system, but scholars are not unanimous. Older grammarians believed that the *perfect* denoted past time and the *imperfect* the present-future, and that therefore Hebrew did, in fact, employ a tense system. But during the past century, this view gave way to the current understanding of *aspect* rather than *tense*. See Kautzsch 1910, 117; Bauer and Leander 1991, 268-78; Bergsträsser 1962, 2.9-10; Endo 1996; Sáenz-Badillos 1993, 73; Waltke and O'Connor 1990, 346-47; Chisholm 1998, 85-86; Emerton 2000b, 191-93; and van der Merwe, Naudé, and Kroeze 1999, 141-43. Some prefer to say that Hebrew does, in fact, express tense, though in a less complete way (Bauer 1910; Joüon and Muraoka 1993, 354-58; and more emphatically, Rainey 1986, 7, and 1988; and Zevit 1988 and 1998, 39-48). It is likely that such arguments have overstated the case against aspect and in favor of tense in the prefixed form (Huehnergard 1988, 20-21).

first grammatical category (*Aspekt*) identifies the action as either undefined or progressive ("perfect" or "imperfect"). The second category (*Aktionsart*) refers to the type of action with regard to voice, fientivity, transitivity, causation, and various reflexive actions.<sup>2</sup> The latter (*Aktionsart*) is marked by verbal stems, the former (*Aspekt*) by verbal conjugations.

## 3.1 Stem

In general, the Qal and Niphal have no element of causation in their predication, in distinction to the other stems.<sup>3</sup> The Piel, Pual, and Hithpael have "causation with a patiency nuance," and the Hiphil and Hophal represent "causation notion with an agency nuance."<sup>4</sup>

## **3.1.1** *Qal (the G stem)*

The Qal<sup>5</sup> is the "simple active" stem. The voice of the primary subject is active (see vertical axis of the appendixes). Since there is no element of causation, the Qal has no secondary subject (first column of the horizontal axis of Appendix II). Hence, the stem is "simple."

- <sup>2</sup> This aspectual explanation (i.e., speaking of *Aspekt*, "contour of action," and *Aktionsart*, "type of action") follows modern linguistic theory (e.g., Comrie 1976), and has become the consensus of scholars working on BH (Waltke and O'Connor 1990, 346–50; and for a different view, Zevit 1998, 41–48).
- <sup>3</sup> Variously known as *derived stems, patterns, themes,* or better, the Hebrew designation, *binyanim,* "structures."
- <sup>4</sup> Waltke and O'Connor 1990, 355.
- <sup>5</sup> The Qal is also known as the "G stem" (for German *Grundstamm*, "basic stem"). The verbal stems in other Semitic languages have different names, making it difficult to study them together. Scholars use a standardized list of sigla for the verbal stems, thus facilitating comparisons across the various languages. We will include the siglum for each stem in parenthesis after its name. The advanced student would do well to learn the designations in addition to the traditional stem name.

(a) Fientive – describes action, motion, or change of state.<sup>6</sup> Fientive verbs may be transitive (לְקַח אֹתוֹ אֶלֹהִים, "God <u>took</u> him" [Gen 5:24], הָנָה אֶלֹהִים אֶת־הָאָדָם, "YhwH God <u>formed</u> humanity" [Gen 2:7]), or intransitive (אָלָהִים אָלִהִים, "but also many of the people <u>fell</u> and died" [2 Sam 1:4], "אָרָד כָּל־בֵּית אַחָאָר [2 Kgs 9:8]). Several fientive verbs may be both, as these examples of אום מלא מרי הַנָּהָן הָעָנן מְלָא יָמָיך (גַרָהָרָבָה נָבָל מָן־הָעָם וָיָמָת), "the cloud <u>filled</u> the house of YhwH" (1 Kgs 8:10), דָּרָשָׁר יָמִיָר (צַר מָן־הַעָבן יָמִרָר), "when your days are <u>complete</u>" (2 Sam 7:12).<sup>7</sup>

(b) Stative – describes a state or quality of the subject. English translations often make no distinction between stative verbs and predicate adjectives: וְיָכְבָר לֵב פָּרְעָה "and Pharaoh's heart <u>was heavy</u>" (Exod 9:7), "And yet, this <u>was insignificant</u> in your eyes" (2 Sam 7:19), אָרָי זְכְרָתָן עוֹד זֹאת בַעִינִיך 23:2).

### 3.1.2 Niphal (the N stem)

The "simple middle-passive" and "simple reflexive" stem. The voice of the primary subject is middle-passive or reflexive (see vertical axis of the appendixes). Since there is no element of causation, the Niphal has no secondary subject (first column of the horizontal axis of Appendix II). Hence, the stem is "simple."

Niphal was originally the simple reflexive, that is, the reflexive of the Qal. It gradually acquired the passive meaning of the Qal as well when the Qal passive fell from

<sup>&</sup>lt;sup>6</sup> Since the term "active" is regularly used to denote voice, the term "fientive" is used to describe this type of action (Waltke and O'Connor 1990, 363).

<sup>&</sup>lt;sup>7</sup> The question of transitivity or intransitivity is a less important distinction for the verb than that of fientive versus stative, because stative verbs may be transitive as well as the expected intransitive. Thus, stative verbs such as "to dress," "to don" (לבש) and "be full," "to fill" (מלא) may be transitive (Joüon and Muraoka 1993, 357).

use.8 Grammarians sometimes use the expression "mediopassive" for the Niphal, although Waltke and O'Connor refer to a "medio-reflexive notion," from which they derive four specific meanings (1990, 380): a) middle, b) passive, c) adjectival (simple adjectival, ingressive, gerundive), and d) double-status (reflexive, benefactive, reciprocal, tolerative, causative-reflexive). They state further, "In all the specific uses of the Niphal, we find the common notion(s) that the action or state expressed by the verb affects the subject (as in the middle voice) or its interests (as in the reflexive). Species 1, 2, and 3 can plausibly be associated with the middle notion, and the others with the reflexive. Even in the double-status uses, where the subject is both actor and patient of the action, the primary option is that the subject is affected by the action."9 The following semantic categories are helpful in classifying verbs in the Niphal.

(a) Passive – frequently the passive of the Qal. This is its simplest meaning, though not necessarily its primary one. So קבר ("bury") and אכל ("eat") may become "be buried" and "be eaten" in the Niphal: קבית אָקב, "there I will <u>be buried</u>" (Ruth 1:17), and בּביֵת אָקר יֵאָכִל, "in one house <u>it shall be eaten</u>" (Exod 12:46).

Hebrew's use of the Niphal is an "incomplete passive" (Lambdin 1971a, 176). The passive in English is a construction, not a category of verbal meaning. Thus, "the students read the lesson" has a passive construction, "the lesson was read by the students." By the incomplete passive we mean this same transformation minus the agent: "the lesson was read." The speaker is not concerned with specifying the agent of the action. All Hebrew passives belong to this category; constructions with a specified agent are virtually nonexistent.<sup>10</sup>

<sup>&</sup>lt;sup>8</sup> Joüon and Muraoka 1993, 149.

<sup>9</sup> Waltke and O'Connor 1990, 380.

<sup>&</sup>lt;sup>10</sup> Examples of the infrequent complete passive include Deut 33:29; Gen 9:6, 11; Exod 12:16; 1 Sam 25:7.

(b) Middle – Verbs that are normally used in passive constructions may be used in a quasi-active sense, but with the object as the subject of the verb. Unlike the incomplete passives, middle verbs are active in form, but the meaning (i.e., voice) is, in a sense, reversed. The object of the active verb has become the subject of the middle verb: נְפִרָק הָשָׁרֵץ בְּקוֹלָם, "the heavens <u>opened</u>" (Ezek 1:1), נותבָקע הָשָׁרֵץ בְקוֹלָם, "the earth <u>split</u> with their sound" (1 Kgs 1:40).

(c) **Reflexive** – action for, or concerning, oneself. The subject is at the same time object (recipient) of the action, so that this use of the Niphal may be called the double-status use.<sup>11</sup> In many languages a reflexive pronoun is added to clarify the object, but in the Hebrew Niphal verb, no reflexive pronoun is needed.<sup>12</sup>

In most reflexive uses, the subject and object refer to the same person or thing: אַבְּקְהָה מֵאוֹיְבָי, "and <u>I will avenge myself</u> on my enemies" (Isa 1:24), "and <u>I will avenge myself</u> on my enemies" (Isa 1:24), "[If] they <u>sell themselves</u> to you, you shall not make them serve as slaves" (Lev 25:39). The reflexive may also occur in conjunction with a prepositional phrase, as in the frequent Niphal imperative (Exod 34:12), "guard yourself for yourself," or "<u>take great care</u>" (Exod 34:12), בי נְשָׁבְּעָהִי נָאָם־יְהָוָה, "I have sworn by myself," declares YHWH" (Gen 22:16).

Additionally, the reflexive may denote permission (*tolerative*): אָדְרָשׁ, "I will <u>allow myself to be enquired</u>" (Ezek 36:37), אָדְרָשָׁ, "He will respond to them (literally: he will <u>allow himself to be entreated</u>) and heal them" (Isa 19:22), גִדְרַשָּׁהִי לְלוֹא שָׁאָלוּ נִמְצֵאָהִי לְלוֹא בְקַשׁׁנִי *I permitted myself to be sought* by those who did not inqure,

<sup>&</sup>lt;sup>11</sup> The subject is both agent and patient (or undergoer) of the action in Waltke and O'Connor's terminology (1990, 387–91).

<sup>&</sup>lt;sup>12</sup> Cf. the "tolerative," Williams 1976, 138; Waltke and O'Connor 1990, 389–90.

<u>I permitted myself to be found</u> by those who did not seek" (Isa 65:1). The *reciprocal* construction may be seen as a plural variation of the reflexive: יבָל־בְּנֵי לֵין, "and all the children of Levi <u>gathered (themselves) around</u> him" (Exod 32:26). In one root particularly (דע), the Niphal denotes a *causative-reflexive* meaning, similar to the Hithpael: notes a *causative-reflexive* meaning, similar to the Hithpael. notes a *causative-reflexive* meaning, similar to the Hithpael.

(d) Stative – describes the state of the subject produced by the verbal action:<sup>13</sup> וְשָׁרָת 'יָפְתָח "on the sabbath day [the gate] <u>shall be opened</u>" (Ezek 46:1), אָאָל'תַעָצֶבוּ, "do not <u>be grieved</u>" (Neh 8:11). The stative Niphal is often equivalent to a simple adjectival state, and is also often the same as the passive in translation. So for example, the verb מָשָׁבָרוּ לְפָנִי־יְהָוָה," or "to be <u>in pieces</u>": כִי וְשָׁבְרוּ לְפָנִי־יְהוָה, "for they <u>were broken</u> before YHWH" (2 Chr 14:12[Eng 2 Chr 14:13]).<sup>14</sup>

### 3.1.3 Piel (the D stem)

The Piel is the "causative (with patiency nuance) active" stem and is the most elusive of the Hebrew stems.<sup>15</sup> One leading scholar recently signaled a warning about the futility of searching for a unifying central understanding of the stem's use: "[I]t is pointless to try to find a single . . . explanation to account for all of the

<sup>&</sup>lt;sup>13</sup> Lambdin's resultative (1971, 177).

<sup>&</sup>lt;sup>14</sup> Some verbs in the Niphal have little or no Qal counterpart (ex., נְלְחֵם and (נְסָתֹר). Others may have an "ingressive-stative" connotation, describing "coming" into a certain state, rather than "being" in such a state (ex., certain occurrences of נְבָא see Ben Zvi, Hancock, and Beinert 1993, 120).

<sup>&</sup>lt;sup>15</sup> Joüon and Muraoka 1993, 151.

transformative power of the D stem. It is simply a form." $^{16}$ 

Traditionally the Piel had been considered intensive in meaning. Older grammars defined the idea as "to busy oneself eagerly with the action indicated," and even associated the doubling of the second radical of verbs in Piel as an outward expression of this intensification.<sup>17</sup> However, in light of today's deeper understanding of the Semitic languages generally, we can no longer refer to the Piel as bascially and primarily intensive.<sup>18</sup> Furthermore, it has long been assumed that as the *derived* stems are built morphologically upon the base form of the Qal, so the semantic use and function of those stems may be defined primarily by means of association with the Qal (as intensive of the Qal, causative of the Oal, passive of the Oal, etc.). We now recognize that although such semantic associations may be present in many verbs in the Qal and Piel, this is not always the case, and each stem should be regarded as independent.<sup>19</sup> Many Piel verbs simply cannot be placed in one of the categories that follow. In some cases, the verb may occur in both Qal and Piel with no discernible difference in meaning.

The Piel frequently expresses the bringing about of a state. Thus, the Piel focuses on causation and the outcome

<sup>16</sup> Kaufman 1996, 282. See also Joosten's conclusion that the various functions of the Piel cannot be reduced to one underlying basic function (Joosten 1998, 227).

<sup>17</sup> Kautzsch 1910, 141; and see Blau 1976, 52; Bauer and Leander 1991, 323–29; Martin-Davidson 1993, 136–37. The view that the doubling of the middle consonant is unassociated with intensification may need to be reconsidered in light of recent linguistic work on iconicity, that is, the iconic nature of language (cf. Kouwenberg 1997).

<sup>18</sup> The Piel in recent decades has been recognized as the key to the Hebrew verbal system. Albrecht Goetze opened the discussion to new approaches with his famous survey of the Akkadian D-stem (1942, 1–8), and subsequently the significance of his work for the West Semitic languages was investigated by Ernst Jenni (1968). For useful surveys of these developments, see Waltke and O'Connor 1990, 354–59; Fassberg 2001, 243–44. For caveats on Goetze's arguments, see Kaufman 1996, 281–82.

<sup>&</sup>lt;sup>19</sup> Richter 1978, 73.

of the action, though with a patiency nuance rather than an agency nuance (as in Hiphil). The foregrounded interest is not the event that happens to the subject, but rather the condition attained by it. It is for all practical purposes an adjectival causation predicate. Jenni's important study proposed a basic distinction between the Piel and Hiphil as the difference between the imposition of a state (adjectival) and the imposition of process (verbal).<sup>20</sup> So using as an example the verb הַיָה ("live" in the Qal), the Piel is "to cause to be alive," whereas the Hiphil is "to cause to live." The distinction is between causing to be something or to do something. Specifically with Qal intransitive verbs, we use the term *factitive* to denote a cause producing a state (as distinct from a cause producing an event, or *causative*).<sup>21</sup> This factitive use of the Piel designates, without regard to the process, the bringing about of the state depicted by an adjective. The object experiences this action as an "accident" (signifying that a quality or situation is not essential to the person or thing in question). With some Qal transitive verbs the Piel is *resultative*, meaning that it denotes the bringing about of the outcome of the action designated by the base root, which action can be expressed in terms of an adjective, and without regard to the actual process of the event.<sup>22</sup> In our analysis, we follow Waltke and O'Connor's practice of reserving causative for the Hiphil, which causes an action rather than a state. In the Piel, the object of causation is in a state of suffering the effects of an action and is inherently passive in part (see the second column of the horizontal axis of Appendix II).

The following semantic categories are helpful in classifying verbs in the Piel.

<sup>&</sup>lt;sup>20</sup> Jenni (1968), and see also Lambdin 1969, 388–89. For dissenting voices, see Joosten 1998 and Fassberg 2001, 243–44.

<sup>&</sup>lt;sup>21</sup> "Factitive," from Latin *factitare*, "to do often," "to practice," "to declare (someone) to be."

<sup>&</sup>lt;sup>22</sup> Waltke and O'Connor 1990, 400.

(a) Factitive – makes transitive many verbs that are intransitive in the Qal (mostly stative, although a few fientive verbs are intransitive). By transitivizing verbs that are Qal intransitives, the Piel factitive denotes a cause producing a state (rather than a cause producing an action; see Hiphil causative in section 3.1.6,a). So typically, Qal intransitives become factitive Piels:<sup>23</sup> (and it is in your hand to make great" (1 Chr 29:12), הְבָיָדְ לְנֵהֵל (and I will <u>make</u> your name <u>great</u>" (Gen 12:2), הַקִרְשֶׁכֶם (Gen 12:2), "I YHWH sanctify you" (Exod 31:13), הִבָּיָהָה מָקַרָשֶׁכָם יָהָוָה מַקִירָשֶׁכָם יְהָוָה (1 Sam 2:6), וויאַבָּדַם יְהָוָה them" (Deut 11:4).

Such a use is frequently *resultative*, meaning that the Piel can bring about a state (or "result") corresponding to the verbal notion of a Qal transitive:<sup>24</sup> קרבר אֶת־הַמַצְרֹת, "<u>he broke down</u> the pillars" (2 Kgs 18:4), למִרְתִי אֶתְכֶם חֻקִים וּמִשֶׁפְּשִׁים ordinances" (Deut 4:5).

(b) Denominative – indicates a derived verbal idea related to a noun or substantive. The nominal form is primary and the verb secondarily derived from it. Piel is the stem most commonly used to form denominatives (though see also the Hithpael and Hiphil denominatives): (thou

<sup>&</sup>lt;sup>23</sup> Waltke and O'Connor 1990, 400–04; Joüon and Muraoka 1993, 154– 56; Seow 1995, 173–75; Lambdin 1972, 193–95. These roots typically become causative Hiphils, a relationship that may be illustrated as follows: Qal intransitive::Piel factitive::Hiphil causative (cf. Waltke and O'Connor 1990, 400).

<sup>&</sup>lt;sup>24</sup> Waltke and O'Connor 1990, 404–10; Lambdin 1972, 193; van der Merwe, Naudé, and Kroeze 1999, 80–81. Thus, the profile would be as follows: Qal transitive::Piel resultative::Hiphil causative (cf. Waltke and O'Connor 1990, 404). For similar uses of the Piel in the Canaanite dialect of the Amarna letters, Rainey 1996, 2:138–68.

Moses" (1 Kgs 8:56); זַמְרָה וְקוֹל זְמְרָה בְּכָנוֹר בְּכָנוֹר הַכָּנוֹר וְקוֹל זְמְרָה (<u>Sing praises</u> to YHWH with the lyre, with the lyre and the sound of <u>melody</u> [literally: <u>praise song</u>]" (Ps 98:5).

(c) Frequentative – pluralizes the action denoted by the Qal. Many verbs in the Piel reflect multiple, repeated, or busy action. This use of the Piel may be *iterative* over time, or *pluralic* through space, and often takes on the nuance of an *intensive*, although the intensive force is difficult to identify in most occurrences:<sup>25</sup> (all day long <u>I go around</u> mourning" (Ps 38:7), קלה יוֹם לְרַר הַכָּלָהִי (the righteous individual) <u>follows</u> my statutes [literally: <u>he walks about</u> in my statutes]" (Ezek 18:9), הָרָס הָהָרְסָם וְשָׁבֵר הַשָּׁבֵר הַשָּׁבֵר הַשָּׁבַר הַשָּׁבַר הַשָּׁבַר (Ps 109:10), הָרָס הָהָרְסָם וְשָׁבֵר הַשָּׁבֵר הַשָּׁבַר הַשָּׁבַר הַשָּׁבַר הַשָּׁבַר הַשָּׁבַר הַשָּׁבַר (Exed 23:24).

(d) Declarative – involves some kind of proclamation, delocution, or estimative assessment, although the precise nature of these verbs and their relationship to the factitive is debated: דְּפַצְהָי צָרְכָן, "I desire <u>to justify</u> you [literally: <u>to declare</u> you <u>righteous</u>]" (Job 33:32), "<u>clear me</u> [literally: <u>declare</u> me <u>innocent</u>] of hidden faults" (Ps 19:13), "(Lev 13:44).

As noted, there are simply some uses of the Piel that cannot be categorized and are, in fact, identical in meaning to the Qal stem, or have an equivalent simpleactive meaning: בְקַשׁ יְהוָה לוֹ אִישׁ כַּלְבְבוֹ for himself a man after his own heart" (1 Sam 13:14), for himself a man after his own heart" (1 Sam 13:14), וויַעַשׁ נֹחַ כְּכֹל אֲשֶׁר צָוָה אֹתוֹ אֱלֹהִים כֵּן עָשָׂה , "Noah did according to all that God had <u>commanded</u> him, so he did" (Gen 6:22).

<sup>&</sup>lt;sup>25</sup> Waltke and O'Connor 1990, 414–16; Seow 1995, 174; Lambdin 1971a, 194.

## 3.1.4 Pual (the Dp stem)

The Pual is the "causative (with patiency nuance) passive" stem and is the passive counterpart to the Piel. As such, it contains the same four functions of the Piel, though in passive voice (factitive/resultative, denominative, frequentative, and declarative).

## (a) Factitive/resultative

כְּתֵפֹת עָשׁוּ־לוֹ חֹבְרֹת עַל־שְׁנֵי קִצְווֹתוֹ חֻבָּר

"They made attaching shoulder pieces for it [i.e., the ephod]; <u>it was attached</u> at its two ends" (Exod 39:4), (Exod 39:4), "The flax and the barley were <u>ruined</u>" (Exod 9:31).

(b) Denominative – בָּהֶם אֲשֶׁר כָּפָּר בְּהֶם "and they shall eat them [i.e., the food] by which <u>attonement is made</u>" (Exod 29:33).

(c) Frequentative – וְהַנֵּה שֶׂרָך מֹשֶׁה וְהַנֵּה שָׁרָך "Then Moses diligently inquired about the goat of the sin offering, and <u>it was burned up</u>" (Lev 10:16).

# (d) Declarative

אֵלֶה בִּקְשׁוּ כְתָבָם הַמִּתְיַחְשִׁים וְלֹא נִמְצָאוּ וַיְגֹאֲלוּ מִן־הַכְּהָנָה

"These searched their genealogical records, but they could not be found; so they <u>were considered unclean</u> for the priesthood" (Ezra 2:62)

One striking feature of the Pual is the high percentage of times it occurs in the participle.<sup>26</sup> The participle comprises no less than 40% of all Pual forms, most of which denote a thing or person for or in which a new condition has been attained: וְהַגָּה יְדוֹ מְצֹרַעַת כַּשֶׁלֶג , "and behold, his hand <u>was leprous</u>, as white as snow" (Exod 4:6), רְשָׁע מֵכְתִּיר אֶת־הַצַּדִּיק עַל־כַּן יֵצָא מִשְׁפָּט מְעָקָל "Wickedness surrounds the righteous; therefore, judgment comes forth *perverted*" (Hab 1:4),

וּמֵהֶם מְמָנִים עַל־הַכֵּלִים וְעַל כָּל־כְּלֵי הַכּּןֶדֶשׁ

"And some of them *were appointed* over the vessels and all the vessels of the sanctuary" (1 Chr 9:29).

#### 3.1.5 Hithpael (the HtD stem)

The Hithpael is the "causative (with patiency nuance) reflexive" and "causative (with active secondary subject) reflexive" stem.

The rudimentary meaning of the Hithpael is the doublestatus (*reflexive-reciprocal*) counterpart of the Piel and secondarily as a passive form.<sup>27</sup> However, Hithpael verbs are intransitive and often have a reflexive or reciprocal meaning in relation to their active counterparts of the Qal, Piel, or Hiphil type from the same root.<sup>28</sup> The following semantic categories are helpful in classifying verbs in the Hithpael.

(a) Reflexive – indicates that the subject of a verb is also its implied object. Stated another way, the direct object, though unspecified, is also the subject of the verb: הְתְחַבָּא הָאָרָם וְאָשֶׁתוֹ מִפְּנֵי יְהָוָה , "they <u>hid themselves</u>" (1 Sam 14:11), <u>hid themselves</u> from the presence of YHWH" (Gen 3:8), <u>hid themselves</u> from the priests had not <u>sanctified them-</u> <u>selves</u>" (2 Chr 30:3).

In some verbal roots, the subject of the *reflexive* Hithpael is not the same as the direct object (which may be expressed), but is rather the same as the indirect object:<sup>29</sup>

<sup>&</sup>lt;sup>27</sup> Joüon and Muraoka 1993, 159–60; Waltke and O'Connor 1990, 424–32; Kautzsch 1910, 149–51; Chisholm 1998, 82; van der Merwe, Naudé, and Kroeze 1999, 82–84; Lambdin 1971a, 248–50.

<sup>&</sup>lt;sup>28</sup> Lambdin 1971a, 249.

<sup>&</sup>lt;sup>29</sup> Waltke and O'Connor 1990, 430; Seow 1995, 298.

יָהוֹנְתָן אֶת־הַמְעִיל, "Jonathan <u>stripped himself</u> of the robe" (1 Sam 18:4).

(b) Reciprocal – indicates that two or more subjects act in relationship to each other: לְכָה נְתְרָאָה פָּנִים, "Come, let us <u>look one another</u> in the face" (2 Kgs 14:8), "Come, let us <u>look one another</u> in the face" (2 Kgs 14:8), "his servants <u>conspired with one another</u> against him" (2 Chr 24:25), גָּאָשָׁמַע מִדְּבֶר אָלִי (Ezek 43:6).

(c) Iterative – indicates repeated action:

לַהַט הַחֶרֶב הַמִּתְהַפֶּכֶת

"the flame of a <u>revolving</u> sword [literally: <u>turning repeatedly</u>]" (Gen 3:24), וַיְהָאָבֵל עַל־בְּנוֹ יָמִים רַבִּים, "and he [Jacob] <u>mourned</u> for his sons many days" (Gen 37:34), <u>אָת־הָאָל</u>הִים הַתְהַלֶּךְ־נֹחַ (Gen 6:9).<sup>30</sup>

(*d*) *Denominative* – indicates a derived verbal idea related to a noun or substantive:

ּוּבִנְבִיאֵי שׁׁמְרוֹן רָאֵיתִי תִפְּלָה הַנַּבְּאוּ בַבַּעַל וַיַּתְעוּ אֶת־עַמִּי אֶת־יִשְׂרָאֵל

"Among the *prophets* of Samaria I saw a disgusting thing: <u>they prophesied</u> by Baal and led my people astray" (Jer 23:13), יויָהאָרָך יְהוָה מְאָר בְיָשְׂרָאֵל, "Yhwh was very <u>angry</u> with Israel" (2 Kgs 17:18).

**3.1.6** Hiphil (the H stem)

The Hiphil is the "causative (with agency nuance) active" stem. Unlike the Piel causative, where the focus is on the

<sup>&</sup>lt;sup>30</sup> On the special nuances of הל'ם in the Hithpael and the possibility that it is an "Akkadianism" borrowed into Hebrew, see Waltke and O'Connor 1990, 427–29.

bringing about of a state or condition, the Hiphil causative expresses the cause of an action. The Piel causative tends to focus on the outcome of the action, though with a patiency nuance (i.e, the object is transposed passively into a new state or condition). The subject of a Piel causative changes the object into the state or condition (see Piel *factitive* and *resultative* in section 3.1.3,a). By contrast, the Hiphil causative generally has to do with the causing of an event, this time with an agency nuance. In the Hiphil causation, the object participates in the event denoted by the verbal root.<sup>31</sup>

Generally, the fundamental causative nuance of a Hiphil will be determined by the specific type of verbal root used, and often by the meaning of that root in the Qal and whether it is transitive or intransitive in the Qal. The following semantic categories are helpful in classifying verbs in the Hiphil.

(a) Causative – indicates the bringing about of the sense of a verb occurring in the Qal or Niphal. But this causation takes a variety of distinctive connotations, depending on the nature of the root involved.

(a.1) Qal transitives usually become doubly transitive in the Hiphil; that is, they take two objects, a "causing" object and an object of the verbal idea expressed in the root.<sup>32</sup> In this example from Deuteronomy 3:28, the object of causation is represented in the translation by "them," and the object of the verbal idea is "the land": "them," and the object of the verbal idea is "the land": "for he [Joshua] <u>will cause</u> them <u>to inherit</u> the land."

However, sometimes the second object is omitted, leaving the predicate singly transitive. This is frequently

<sup>&</sup>lt;sup>31</sup> Jenni 1968, 20–52; Waltke and O'Connor 1990, 433–36; Joüon and Muraoka 1993, 162–64.

<sup>&</sup>lt;sup>32</sup> The terminology "doubly transitive" is that of Lambdin (1972, 211). Waltke and O'Connor refer to "three-place predicates," by which they mean one subject and two objects (Waltke and O'Connor 1990, 441).

the case when the objects of causation are personal, or where they are abstract nouns or speech acts:<sup>33</sup> or shout or <u>cause</u>, "Do not shout or <u>cause</u> your voice <u>to be heard</u>" (Josh 6:10). Thus, verbs that are transitive in the Qal, but which may appear without an object, may be either singly or doubly transitive in the Hiphil: may be either singly or doubly transitive in the Hiphil: "And the Lord <u>had caused</u> the Aramean army <u>to hear</u> the sound of chariots, and of horses, the sound of a great army" (2 Kgs 7:6). In this case, שמע takes two objects: the Aramean army and the sound. But the same root may have only one object in the Hiphil causative: "הַשְׁמַעָּתָ דָרָן "From the heavens <u>you uttered</u> judgment" (Ps 76:9).

(a.2) Qal and Niphal intransitives become singly transitive in the Hiphil; that is, they take one object.<sup>34</sup> Most common in this category is a large group of Qal intransitive fientive verbs designating motion: אַרָּיָדָהָרָ בַאָרָר בַאָרָ בַאָרָ (I am Yhwh who <u>brought you</u> from Ur of the Chaldeans" (Gen 15:7), <u>brought you</u> from Ur of the Chaldeans" (Gen 15:7), יָרָאָשֶׁב יְהָוָה עֵּבְהֶם אָרֹבֵי הָיָם waters of the sea upon them" (Exod 15:19).

Verbal roots that are specifically stative in Qal tend also to become singly transitive in the Hiphil. Here the term "factitive" may be used to indicate the transitivizing nature of the Hiphil, as in the Piel usage: אל הַרְכָב בְּרָכֶם, "Do not <u>harden</u> your hearts" (Ps 95:8), אלי הַרְכָב בַּרְעָה, Will <u>harden</u> Pharaoh's heart" (Exod 7:3), "I will <u>harden</u> Pharaoh's heart" (Exod 7:3), Sea of Reeds" (Josh 2:10), הַרְבָרָ הַרָּבָר הַשָּרָ הַרָּבָרָ *great [or magnified]* your loyal kindness" (Gen 19:19).

<sup>&</sup>lt;sup>33</sup> Waltke and O'Connor 1990, 442-43.

<sup>&</sup>lt;sup>34</sup> Waltke and O'Connor 1990, 436–41; Lambdin 1971a, 212.

<sup>&</sup>lt;sup>35</sup> Seow 1995, 182. On the difficulty of maintaining a distinction between the Piel factitive and the Hiphil transitivizing causative with Qal stative roots, see Waltke and O'Connor 1990, 437–38. In general, they conclude that verbs in the Piel factitive "direct attention to the results of the situation apart from the event," while the Hiphil "refers to the process."

(b) Stative – designates an intransitive causation of verbal roots that are also stative in the Qal, in which the action remains with the subject itself:<sup>36</sup> הַקְרִיב , "and Pharaoh <u>drew near</u>" (Exod 14:10), וְפָּרְשָׂה , "and you <u>became</u> <u>presumptuous</u>" (Deut 1:43).

At times this Hiphil stative is more specifically *in-gressive*; that is, it designates the entry into a state or condition and the continuation in the state or condition: חַנֹךְ לַצַּעָר עַל־פִּי רַרְכּוֹ גַם כִּי־וַיְקִין לֹא־יָסוּר מִכָּגָּם, "Train a youth in the right way to live, and when <u>he grows old</u>, he will not depart from it" (Prov 22:6).

At other times, an object simply ellides, leaving an intransitive sense of certain roots in Hiphil:<sup>37</sup> הַכְּשָׁיב בְּקוֹל תְּכָלֶתִי, "He [God] <u>has given heed</u> [literally: <u>inclined</u> (the ear)] to the sound of my prayer" (Ps 66:19). Finally, the absence of an object may also explain certain adverbial Hiphils:<sup>38</sup> הַשְׁרִבֹּתָ, "you <u>did well</u>" (1 Kgs 8:18), הַשְׁרִבֹּתָ, "they <u>are corrupt</u>" (Ps 14:1).

(c) **Declarative** – involves proclamation or delocution, or an *estimative* assessment:<sup>39</sup>

וְתִצְדִיקוּ אֶת־הַצַּדִיק וְהִרְשִׁיעוּ אֶת־הָרָשָׁע

"and they [judges] will <u>pardon</u> [or <u>declare righteous</u>] the righteous and <u>condemn</u> [or <u>declare wicked</u>] the wicked" (Deut 25:1).

(d) **Denominative** – indicates a derived verbal idea related to a noun or substantive. The nominal form is primary

<sup>37</sup> Joüon and Muraoka 1993, 164.

<sup>&</sup>lt;sup>36</sup> Waltke and O'Connor 1990, 439; Joüon and Muraoka 1993, 163; Seow 1995, 183; Lambdin 1971a, 212. Waltke and O'Connor refer to "one-place or inwardly transitive" Hiphils (Waltke and O'Connor 1990, 439).

<sup>&</sup>lt;sup>38</sup> Joüon and Muraoka 1993, 164.

<sup>&</sup>lt;sup>39</sup> Waltke and O'Connor 1990, 438–39; Joüon and Muraoka 1993, 163; Seow 1995, 163.

and the verb secondarily derived from it:<sup>40</sup> אַלִיכֶם "and he (YHWH) did not <u>give ear</u> to you" (Deut 1:45), "and the Philistine (Goliath) came forward <u>morning</u> and <u>evening</u>" (1 Sam 17:16).

(e) Permissive – Used rarely, the Hiphil designates an action that is agreeable to the object and allowed by the subject (Lambdin 1972, 212; Waltke and O'Connor 1990, 445): הָרְאָר אָרִים גַם אָר־וֹרְעָר "God allowed me to see your children also" (Gen 48:11). Waltke and O'Connor observe a number of other uses of the Hiphil similar to the permissive, in which the relationship between the principal subject and the secondary subject requires a variety of modals in translation: compulsion, solicitude, toleration, bestowal (1990, 445–46).

In addition to the difficulty of the overlap between the semantic meaning of certain Piel and Hiphil forms, there are many Hiphil verbs that simply defy classification in any of these categories. Some roots are both transitive and intransitive in different contexts, and we would do well to analyze each derived stem as an independent form without insisting on a discernable connection to the other derived stems.

## 3.1.7 Hophal (the Hp stem)

The Hophal is the "causative (with agency nuance) passive" stem and is the passive counterpart to the Hiphil. As such, it contains the same five functions of the Hiphil, though in passive voice (causative, stative, declarative, denominative, permissive). Most frequently in the Hophal, the subject is caused to be in the event signified by the verbal root, whereas the subject of the Pual, by contrast, is made into a state or condition.

<sup>&</sup>lt;sup>40</sup> Waltke and O'Connor 1990, 443–45; Joüon and Muraoka 1993, 163–64; Lambdin 1971a, 213; Seow 1995, 182.

(a) Causative – הַלְּגֵע בָּאִישׁ הַזֶּה וּבְאָשְׁתּוֹ מוֹת יוּמָת "anyone, who molests this man or his wife shall certainly <u>be put to death</u>" (Gen 26:11), קח־נָא אֶת־בִּרְכָתִי אֲשֶׁר הֻבָאת לָן "please take my gift which has <u>been brought</u> to you (Gen 33:11).

(b) Stative – הַפּׁךְ יְדְרָ וְהוֹצִיאֵנִי מִן־הַמַחֵנָה כִּי הָחֲלֵיתִי, "turn around (literally: turn your hand), and take me out of the battle, for <u>I am wounded</u>" (1 Kgs 22:34).

(c) Denominative – וְהָמְלֵחַל א הָמְלֵחַ, "you were not <u>rubbed</u> <u>with salt</u>" (Ezek 16:4), חַיֵּת הַשְׁלְמָה־לְךָ, "the beasts of the field <u>will be at peace</u> with you" (Job 5:23).

The change of voice from Hiphil to Hophal means further that in the case of Qal transtives, the primary subject of the Hiphil is no longer designated, and so the double transitivity is dropped. Such roots will designate one object or none at all: אָרֶבְקָה אָת־בְּבְרִי עֵשָׁו, "and the words of Esau were told to Rebekah" (Gen 27:42). Similarly with Qal and Niphal intransitives, the Hophal is most often also intranstive so that no object is represented: (Mic 1:4).<sup>41</sup>

## 3.2 Aspect

As we have seen, the grammatical category *Aspekt* identifies the action as either undefined or progressive ("perfect" or "imperfect").<sup>42</sup> Conjugations indicate aspect in that they denote the relation of an action to the passage of time, especially in reference to its completion, duration, or repetition.<sup>43</sup>

- <sup>41</sup> For more on the subtle changes from Hiphil to Hophal, see Waltke and O'Connor 1990, 449–51, and Lambdin 1971a, 243–44.
- <sup>42</sup> See Waltke and O'Connor 1990, 479–95; Joüon and Muraoka 1993, 359– 65; Chisholm 1998, 86.
- <sup>43</sup> As regards terminology, we have retained the traditional grammatical designations "perfect" and "imperfect." We recognize that these

## 3.2.1 Perfect

The perfect views a situation from the outside, looking upon it as a complete whole.<sup>44</sup> It may refer to an action or state in the past, present, or future, although it tends to view it as a complete situation or action that is temporally undefined (making it similar to the Greek "aorist" tense).<sup>45</sup> Although not required by the morphology, the perfect is frequently used for actions or states reported in the past, often requiring past tense translations. It may also refer to the perfect state, that is, to an event and a state resulting from that event (making the perfect similar to the Greek perfect). The following semantic categories are helpful in further classifying verbs in the perfect.<sup>46</sup>

terms are wholly inadequate, especially for the perfect, as we hope will be clear from our discussion. But a survey of the most widely used beginning grammars reveals a tendency to retain the traditional nomenclature (Hostetter 2000, Seow 1995, Kelley 1992, Lambdin 1971a, Pratico and Van Pelt 2001, etc.). Most of the reference grammars use a combination of terms: hence Joüon's "perfect" and "future," Meyer's interchanging "afformative conjugation"/"perfect" and "preformative conjugation"/"imperfect," and Waltke and O'Connor's "suffix (perfective)" and "prefix (non-perfective)." The reader needs to be aware of the inadequacy of the traditional terms and of the variety of designations for them in the reference grammars (see Rainey 1996, 2:365–66).

- <sup>44</sup> Note the careful distinction between viewing a situation "as a complete whole" and viewing a completed action or situation. The perfect does not denote completed action, only that the whole action is in view, which may be placed in the present or future as well as the past. This distinction is clearer when one considers the origins of the BH perfect among secondmillennium West Semitic languages as a development from a form related to the Akkadian "stative" or permansive plus personal suffixes, so that it was probably perceived originally as a timeless nominal clause (Moscati 1980, 133; Bergsträsser 1983, 20–23; Rainey 1996, 2:347–66). On the need to abandon "stative" for the Akkadian permansive, see Huehnergard 1987, 229–32.
- <sup>45</sup> However, in LBH, the perfect is used only for the past (Sáenz-Badillos 1993, 129).
- <sup>46</sup> Kautzsch 1910, 309–13; Waltke and O'Connor 1990, 479–95; Joüon and Muraoka 1993, 359–65; Bergsträsser 1962, 2:25–29; Chisholm 1998,

(a) Complete – The action or state is viewed as a complete whole, with the beginning and end in view. Usually required is either simple past, present perfect, or pluperfect in translation: וְלָחֹשֶׁךְ קָרָא לְיָלָה, "and the darkness he <u>called</u> night" (Gen 1:5), שֶׁכָח אָל, "God <u>has forgotten</u>" (Ps 10:11), וְלָא־יָדִע יִצְלְב כִּי דְחֵל גְּנֶבְתַם, "Jacob did not know that Rachel <u>had stolen</u> them" (Gen 31:32).

(b) Stative – a state of affairs or a condition usually expressed through a stative verb. Since a stative verb does not denote a "once-and-for-all" completed action, it regularly requires duration, usually present tense in translation: אין ריקט די גענער אין איט (Isa 1:15), "געין איט (Isa 1:15), "דָרָר־הַתּעֵרָה הַזּאַת אָשֶׁר שָׁנָאַת" (Jer 44:4).

(c) Experience – with a fientive verb denoting a state of mind, usually requiring the present in translation: 'לא יָדַעָּת', "I do not <u>know</u>" (Gen 4:9). This is common with verbs of perception (דרע) or attitude (שנאתי מָאַסִתּי חַגֵּיכֶם :(אָהב), "I <u>hate</u>, I <u>despise</u> your festivals" (Amos 5:21), שָׁנָאתִי מְאַסִתּי הַגֵּיכֶם :(אָהב), "I <u>hate</u>, I <u>despise</u> your festivals" (Amos 5:21), אָגי יְדַעָּתִי אֶת וְדְנָךְ וְאֵת רֹעַ לְבָבֶךָ (אַתרי (I Sam 17:28), "I the ones <u>I love</u> have turned against me" (Job 19:19).

(d) **Rhetorical future** – expresses a vivid future action or situation, which is not yet a reality but considered a certainty from the speaker's rhetorical point of view.<sup>47</sup> As

<sup>86–89;</sup> van der Merwe, Naudé, and Kroeze 1999, 144–46; Williams 1976, 29–30.

<sup>&</sup>lt;sup>47</sup> Often called the *prophetic perfect*, although this designation may be misleading if we make too much of the distinction between the *rhetorical perfect* and the *future imperfect*. For summaries of the problem, see Waltke and O'Connor 1990, 465 (but see their 490 also), and Joüon and Muraoka 1993, 363; Chisholm 1998, 88; van der Merwe, Naudé, and Kroeze 1999, 146, and especially 364; Williams 1976, 30.

a rhetorical device, the perfect presents future events as if they have already occurred, which often requires the present or future in translation: אָכָרַ הַבְּלִי־דְעַמִי הַבְּלִי־דְעַמִי "therefore my people <u>will go into exile</u> for want of knowledge" (Isa 5:13), דְּרַךְ כּוֹכָב מֵיַשֵׁקֹב (Isa 5:13), דְרַךְ כּוֹכָב מֵיַשֵּׁקֹב , "A star <u>will come out</u> of Jacob" (Num 24:17), דְּרַק כּוֹכָב מֵיַשֵּׁקֹה But you shall give it now, and if not, <u>I will take</u> it by force" (1 Sam 2:16), שָׁשִׁיתִי אֶת־הַדְּבָר הַזֶּה (2 Sam 14:21).

(e) **Proverbial** – denotes actions, events, or facts that are not time conditioned, and considered to be general truths. Also known as the *gnomic perfect*, this usually requires the present tense: וְתֹר שָׁמְרוּ אֶׁת־עֵת בֹּאָנָה), "and the turtledove, swallow, and crane <u>observe</u> the time of their coming" (Jer 8:7), יְבַשׁ חָצִיר נְבֵל צִיץ, "The grass <u>withers</u> and the flower <u>fades</u>" (Isa 40:7).

(f) Performative – describes an action that occurs by means of speaking, usually requiring the present: נְחַרִּ עָל כָּל־אֶרֶץ מִצְרָיִם, "<u>I appoint</u> you over all the land of Egypt" (Gen 41:41),

```
בַּיּוֹם הַהוּא כָּרַת יְהוָה אֶת־אַבְרָם בְּרִית לֵאמֹר לְזַרְעֵך נָתַתִּי
אֶת־הַאָּרֵץ הַזֹאת
```

"on that day, Үнwн made a covenant with Abraham saying, 'To your descedants, *I give* this land'" (Gen 15:18).

## 3.2.2 Imperfect

The imperfect has an even less precise time value than the perfect.<sup>48</sup> In general, the imperfect views actions, events, or states from the inside, meaning that the situation is underway or in process. The speaker or writer views the situation

<sup>&</sup>lt;sup>48</sup> Joüon and Muraoka 1993, 372. Although the term "imperfect" is less than completely satisfying for this aspect of the BH verb, we retain it for the sake of convenience (see Rainey 1986, 7, and 1996, 2:227–28).

as still continuing, in the process of accomplishment, just taking place, or imminent. The imperfect may assume an indicative mood so that the speaker or writer makes an objective statement, which, however, makes no reference to the beginning or end of the situation. As with the perfect aspect, context must determine whether the situation is past, present, or future.<sup>49</sup> At other times, the imperfect may assume a subjunctive or hypothetical mood, describing a dependent situation that is only contingent or possible. Thus, the speaker or writer uses the imperfect when viewing an action as nonspecific, habitual, or probable.

This great variety of uses for the imperfect, together with its close association to the modals (see section 3.3), can be explained partially by developments in the history of the Hebrew language near the end of the second millennium B.C.E. On the basis of comparative Semitic evidence, it appears that the prefixed forms in BH were preceded by four distinct forms: an indicative imperfect (presentfuture) *yaqtulu*, a subjunctive *yaqtula*, a jussive *yaqtul*, and a preterite *yaqtul.50* When Hebrew words lost their final vowels (approximately 1100 B.C.E.), these forms became largely indistinguishable. This may explain why the Bible's uses of the imperfect conjugation include a preterite use and several "contingent" uses (vestiges of the subjunctive in Proto-Hebrew). In addition, the volitive modals (see section 3.3) suggest that *yaqtulu* and the jussive *yaqtul* had merged completely by the time BH was written.51

- <sup>49</sup> However, in LBH, the imperfect is used primarily for the future (Sáenz-Badillos 1993, 129).
- <sup>50</sup> The evidence further suggests the presence of an energic form (see Rainey 1986, 10–12, and 1996, 2:221–64; Joüon and Muraoka 1993, 388; Waltke and O'Connor 1990, 496–97). Some scholars also speak of an accent distinguishing Proto-Hebrew jussive and preterite, although this is a moot point (see especially Zevit 1998, 49–65, and compare Joüon and Muraoka 1993, 388).
- <sup>51</sup> For more, see Meyer 1992, 18–20, 26–27, 381–84; Waltke and O'Connor 1990, 455–78, 496–502 (esp. 497), 543–47, 566–68; Joüon and Muraoka

The following semantic categories are helpful in further classifying verbs in the imperfect.  $^{52}$ 

(a) Future – describes an action anticipated or announced: בִּיֹמֶלֶךְ יְמְלֹדְ שָׁלֵינוּ, "for a king <u>will reign</u> over us" (1 Sam 12:12), "for a king <u>will reign</u> over us" (1 Sam 12:12), "you <u>shall go</u> to your ancestors in peace" (Gen 15:15). The future imperfect may be used to denote action that is in fact past as far as the speaker or writer is concerned, but prior to some other action: אָלָרְאָבֹרְאָבֹרְאָבֹרָ בְּשָׁלוֹם, "קרֹאוֹת מָה־יִקְרָאוֹת מָה", "and he [YHWH] brought [them] to the man to see what <u>he would call</u> them" (Gen 2:19). Likewise, it may denote action in the future that is prior to another anticipated action: וְהָבֵּלִיתָ עוֹלָה בַּעֲצֵי הָאָשֶׁרָה אָשֶׁר הַכָּרָא.

(b) Customary – denotes an action occurring regularly or customarily. The action may be in the past, in which case the customary imperfect is *iterative*, emphasizing the repeated nature of the action: בָּלִים, בָּלִים, יָשָׁשָׁה אַיּוֹב בָלִים, "This is what Job <u>would always do</u>" (Job 1:5). Customary action may also designate the present, in which case it frequently takes on a *proverbial* connotation: בֵּוֹ תָּכָם יְשָׁמָח־אָב

(c) Progressive – indicates action that is underway or continuing as the writer or speaker describes it. Such

1993, 387-89; Sáenz-Badillos 1993, 48-49; Rainey 1986, 4-19; Smith 1991, 12.

- <sup>52</sup> Kautzsch 1910, 313–19; Waltke and O'Connor 1990, 496–518; Joüon and Muraoka 1993, 365–73; Lambdin 1971a, 100; Bergsträsser 1962, 2:29–36; Chisholm 1998, 89–94; van der Merwe, Naudé, and Kroeze 1999, 146–49; Williams 1976, 30–32.
- <sup>53</sup> For the iterative and proverbial, Waltke and O'Connor speak of "past customary non-perfective," which implies that the situation no longer holds true at the time of the utterance, and "habitual non-perfective" (1990, 502–03 and 506).

actions or events of the present are usually expressed with participles, but the progressive imperfect emphasizes the present continuous nature of the action, particularly in questions: אָנָה הֵלֵכִי, "where <u>are you going</u>?" (Gen 16:8), לְמָה הַעֲמֹד בָּחוּץ, "why <u>are you standing</u> outside?" (Gen 24:31).

(d) Contingent – expresses action that is dependent upon other factors in the context. A wide variety of such uses is possible, often requiring the use of modal helping verbs: may, can, shall, might, could, should, and so on.

(d.1) *Conditional* – expresses the action of the protasis or introductory clause of a conditional sentence: גָּיָאָלֶךְ בְּגִיא צַלְמְוָת (Ps 23:4), בִי־אָלֶך בְּגִיא בְּסְרֹם חֲמִשִׁים צַּדִּיקִם, "If <u>I find</u> in Sodom fifty righteous people" (Gen 18:26).

(d.2) *Permission* – indicates that the subject is permitted to take action, usually requiring "may":

מִכּּל עֵץ־הַנְּן אָכֹל תֹאכֵל

"You <u>may freely eat</u> of every tree in the garden" (Gen 2:16), יְהְוָה מִי־יָגוּר בְּאָהֵלֶך מִי־יִשְׁכֹן בְהַר קִרְשֶׁך, "O Yhwh, who <u>may abide</u> in your tent? Who <u>may dwell</u> on your holy hill?" (Ps 15:1).

(d.3) *Obligation* – denotes that some action should (or should not) be done: מַעֲשִׁיהַ אָשֶׁר לֹא יֵעֲשׂר אָ יֵעָשׁר עָאָדָ "You have done things to me that <u>ought</u> not <u>to be done</u>" (Gen 20:9), לֹא יֵעָשֶׁה כֵן בְּיָשָׂרָאֵל, "such a thing <u>is not done</u> in Israel" (2 Sam 13:12).

(d.4) Command – expresses a strong injunction or prohibition: קנים הַשְׁשֶׂה אָת־הַהָּבָה, "<u>you shall make</u> rooms for the ark" (Gen 6:14),

כּל אֲשֶׁר־יָלֹק בִּלְשׁוֹנוֹ מִן־הַמַּיִם כַּאֲשֶׁר יָלֹק הַכֶּלֶב תַּצִיג אוֹתוֹ לְבָד

"all who lap the water with their tongue, just as a dog laps water, <u>set</u> them <u>apart</u>" (Judg 7:5).

The use of the imperfect (or the *waw*-consecutive perfect) may connote a command perceived by the speaker as continually valid, or one that need not be performed immediately. Such use of the imperfect may also occur when a superior speaker presents laws or instructions.<sup>54</sup> With the negative particle, such imperfects become prohibitions, which is common in legal literature: לא הַעָּכִי אָ הַרָּבִי הָבָרִי הָבָרִי הָבָרִי הָבָרִי הָנָבִיא הַחוּא . "<u>you shall not steal</u>" (Exod 20:15), לא הַעָּכִי הַבָּרִי הַבָּרִי הַבָּרִי הַבָּרִי הַנָּבִיא הַחּוּא . "<u>you shall not steal</u>" to the words of that prophet" (Deut 13:4 [Eng. 13:3]).

(e) **Preterite** – designates specifically past time situations when occurring after אָר האָת לֵיהוָה, and בִּבֶי יְשָׁרָאַל אֶת־הַשִּׁידָה הוֹאָת לֵיהוָה, "Then Moses and the Israelites <u>sang</u> this song to YHWH" (Exod 15:1), for a constraint of the field <u>was</u> yet in the earth" (Gen 2:5).

Technically, the preterite was a third conjugation that was dying out in BH. Comparison with other Semitic languages reveals that the shortened form of several weak roots – and consistently the Hiphil prefixed forms – could serve in prose as a genuine preterite with אָרֶם, and בְּמֶרֶם and in archaic poetry without them.

# 3.3 Modals

In addition to the conjugations expressing statement of fact (perfect and imperfect), Hebrew also has three modals expressing volition: jussive, cohortative, and imperative. These forms evolved independently in the early history of Hebrew, and they therefore constitute a class by function rather than by morphology; that is, they are treated together only because of similar uses, not because they constitute a distinct conjugation.  $^{55}$  Together they comprise the variety of ways BH expresses will or volition.  $^{56}$ 

#### 3.3.1 Jussive

The reader should recall that most roots show no distinction between the jussive and the corresponding imperfect forms. The only exceptions are Hiphil forms, Qal forms of middle weak roots, and all stems of final- $h\bar{e}$  roots.<sup>57</sup>

The jussive expresses the speaker's desire, wish, or command where a second or third person is the subject of the action, although the third person is more common.<sup>58</sup> The following semantic categories are helpful in classifying the jussive.

(a) Command – A superior uses the jussive with an inferior as subject: יְהֵי אוֹר, "<u>let there be</u> light" (Gen 1:3), הְּעוֹף יְרֶב בָּאָרִץ, "and <u>let</u> birds <u>multiply</u> on the earth" (Gen 1:22), שֶׁבְעַת יְמִים הוֹחֵל עַר־בּוֹאַי אָלֵיך, "seven days <u>you shall wait</u> until I come to you" (1 Sam 10:8).

At times, this connotes permission:

וְיַעַל לִירוּשְׁלַם אֲשֶׁר בִּיהוּדָה וְיָבֶן אֶת־בֵּית יְהוָה אֶלֹהֵי יִשְׂרָאֵל

- <sup>55</sup> For the hypothesis that the jussive had an independent origin from the imperfect and that they subsequently merged, see discussion under 3.2.2.
- <sup>56</sup> Kautzsch 1910, 319–26; Waltke and O'Connor 1990, 564–79; Joüon and Muraoka 1993, 373–79; Bergsträsser 1962, 2:45–53; Chisholm 1998, 103–07; van der Merwe, Naudé, and Kroeze 1999, 150–53; Williams 1976, 34–35.
- <sup>57</sup> Kautzsch 1910, 130–31; Hostetter 2000, 70, 97, and 140; Seow 1995, 209, 235–36, and 279–81; Ross 2001, 150, 211, 261, and 277.
- <sup>58</sup> In the early stages of Hebrew, the shortened prefix conjugation for expressions of volition could be used in all three persons. But in BH, the jussive occurs only in the third and second persons, although the third is by far the most common (Waltke and O'Connor 1990, 567; Joüon and Muraoka 1993, 138–39 and 376; and Chisholm 1998, 103). Most of the examples included here are third person.

"<u>let him go up</u> to Jerusalem which is in Judah, and <u>let him</u> <u>rebuild</u> the house of YHWH, the God of Israel" (Ezra 1:3), העהי אָשָׁה לְבֶן־אֲדֹנִיךְ כַּאֲשֶׁר דְבָר יְהוָה, "and <u>let her become</u> the wife of your master's son, as YHWH has spoken" (Gen 24:51). At other times, the concept of *invitation* is more appropriate: קַהִי נָא אָלָה בֵינוֹתֵינוּ, "<u>let there be</u> an oath between us" (Gen 26:28).

(b) Wish – An inferior uses the jussive with a superior as subject: יְחֵי הַמָּלֶך, "<u>long live</u> the king" (1 Sam 10:24), and now <u>let</u> Pharaoh <u>select</u> [literally: <u>see</u>] a man who is discerning and wise" (Gen 41:33).

The speaker may use the *jussive of request* by referring to himself or herself indirectly when addressing a superior: 'קרחי' (קרח' (קרח' (קרח' (קרח' (קרח' (קר') ), "and may my life <u>be saved</u>" (Gen 19:20). The use of "servant" is a common honorific substitute for a first-person pronoun in such requests: אַעְרְדָן יָשָׁר־נָא עַרְדָן "and now please <u>let</u> your servant <u>remain</u>" (Gen 44:33). Such a request is often followed by the particle 'יָשָׁר־נָא עַרְדָן הַיָּשָׁר "<u>let</u> your servant <u>return</u>" (2 Sam 19:38 [Eng 19:37]). At times, this connotes <u>advice</u>: מַשְׁרַן הַמֶּלֶך פַּקִירִים "(<u>let</u> he king <u>appoint</u> overseers" (Esth 2:3).

Where the jussive expresses a wish for God to take action, the jussive denotes *prayer*: יְסֵך יְהוָה לִי בֵּן אַחֵר , "<u>may</u> Yhwh <u>add</u> to me another son" (Gen 30:24), יְכֵם יְהוָה אֶת־דְּכָרוֹ (<u>may</u> Yhwh <u>establish</u> his word" (1 Sam 1:23).

(c) **Benediction** – A superior speaks of God as the subject of the jussive in order to pronounce a blessing for a third party: יְבָרֶכְהָ אָלֹהִים עָמָּד, "and <u>may</u> God <u>be</u> with you" (Exod 18:19), יְבָרֶכְהָ יְהוֶה וְיָשֶׁמְרֶךָ יְהוֶה וְיָשֶׁמְרֶדָ, "Yhwh <u>bless</u> you and <u>keep</u> you" (Num 6:24),<sup>59</sup> הַשָּׁר וָאָבֶ

<sup>&</sup>lt;sup>59</sup> For more on the six jussive verbs (only two in form) in the Aaronic blessing, see Waltke and O'Connor 1990, 566.

now <u>may</u> YHWH <u>show</u> steadfast love and faithfulness to you" (2 Sam 2:6).

(d) Prohibition – with אָל אָל הָשָׁטָלוֹ בְּאַל־הַעַט לוֹ בְאַל־הַעַט לוֹ בְאוֹבָה (do not do anything to him" (Gen 22:12), אַל־בָאַרָה בִּינִי וּבִינִיך הַבִינִי וּבַינִיך (<u>do not do</u> anything to him" (Gen 22:12), אַל־נָא הָהִי בִינִי וּבַינִיך הַיַנָי (<u>let there be no</u> strife between you and me" (Gen 13:8), "<u>let there be no</u> strife between you and me" (Gen 13:8), אַל־נָא יִחַר לַארֹנָי (Gen 18:30).

#### 3.3.2 Imperative

The imperative expresses the speaker's command or instruction directly in the second person. It is interesting to note that the imperative is used only in positive expressions of volition, never in negative commands. Rather, negative commands are expressed with the imperfect preceded by  $\aleph \circ \aleph$ ,  $\forall \aleph \circ \aleph$  with the imperfect for *immediate prohibition* (section 4.2.3) and  $\aleph \lor \vee$  with the imperfect for *permanent prohibition* (section 4.2.11).

The following semantic categories are helpful in classifying the imperative.

(a) Command – The speaker urges immediate action: (a) Command – The speaker urges immediate action: שָׁרָבְּרָךְ מֵאַרְצְךָ י (Gen 12:1),<sup>61</sup> לָדְ מַאַרְצָךָ אַבוֹתֶיךָ, "<u>go forth</u> from your country" (Gen 12:1),<sup>61</sup> שׁוּב אָל־אָרֶץ אַבוֹתֶיךָ, "<u>return</u> to the land of your ancestors" (Gen 31:3), באֹרַבִּיתָדְ אָל־הַתַּבְה וָכָל־בִּיתָד אָל־הַתַּבָה, "<u>go into</u> the ark, you and all your household" (Gen 7:1), the ark, you and all your household" (Gen 7:1), <u>בארַבַר אָל־פַרְעָה מֶעָרְיִם</u>, "<u>go</u>, <u>speak to</u> Pharaoh king of Egypt" (Exod 6:11). Less immediate action may

- <sup>60</sup> It is difficult to assign any particular meaning to the few uses of おっか with the jussive. See Kautzsch 1910, 322.
- <sup>61</sup> Note that the use of reflexive vith imperatives is common (see 4.1.10). Some have called this the centripetal use of the preposition (Muraoka 1978, 495–98; Waltke and O'Connor 1990, 207–08).

also be commanded by the imperative, although the imperfect is more common: אָרָאָה עָשָׂה עָם־עֲבְדֵיךָ, "and <u>deal (at that time)</u> with your servants according to what you observe" (Dan 1:13). To this category belong also those imperatives connoting *request* or *wish*: belong also those imperatives connoting *request* or *wish*: (Isa 5:3). In contradistinction to commands expressed by indicative forms (such as imperfects or converted perfects, see sections 3.2.2,d and 3.5.3, respectively), such imperatives are usually perceived by the speaker as urgent and usually valid for one occasion.<sup>62</sup>

(b) Permission – The speaker grants permission for the action, which the recipient of the imperative desires to take. This use of the imperative is frequently preceded in the context by an actual request (often a jussive or cohortative of wish): אַעֵּלֶה־נָּא וְאָקְבָּרָה אֶת־אָבִי ... עֵׁלֵה וּקְבֹר אֶת־אָבִיךָ (Gen 50:5–6), אַלְכָה נָא...לַךְ לְשָׁלוֹם (Gen 50:5–6), שׁלָכָה נָא... לֵךְ לְשָׁלוֹם (Externation description of the terms of the terms).

(c) Promise – The speaker assures that the recipient of the imperative will take the action in the future, although the action itself is normally outside the power of the person receiving the order:<sup>63</sup> אַלָּר וְהַצְּלָח (1 Kgs 22:12), "go up to Ramoth-Gilead and <u>triumph</u>" (1 Kgs 22:12), "go up to Ramoth-Gilead and <u>triumph</u>" (1 Kgs 22:12), "and in the third year <u>sow</u>, <u>reap</u>, <u>plant</u> vineyards, and <u>eat</u> their fruit" (Isa 37:30).

A long form is possible in all derived stems consisting of a suffix  $\square_{\tau}$ , which results in certain other vowel changes

<sup>&</sup>lt;sup>62</sup> For other subtle distinctions between imperatives and commands expressed by indicative forms, see Shulman 2001.

<sup>&</sup>lt;sup>63</sup> See Waltke and O'Connor's discussion of "heterosis" in which one grammatical form is exchanged for another. So a promise is made more emphatic by the use of an imperative where a present-future imperfect would be expected (1990, 572).

(קְּטָלָה or קְּטָלָה in the Qal strong verb). This has sometimes been taken as stylistic, emphatic, or polite, but appears to have no discernible difference in nuance from the regular form.<sup>64</sup> Recent studies have suggested it may have been used when the verbal action was directed toward the speaker.<sup>65</sup>

Likewise the particle  $\checkmark$ , which is frequently added to all three volitive modals (imperatives, jussives, and cohortatives), evinces no discernible difference in meaning. Some authorities continue to suspect that it originally added a nuance of "energy" (so translated "please" or "I pray"),<sup>66</sup> while others have noted its "logical" connotations, implying that the wish of the volitive is a logical consequence of a preceding statement or situation.<sup>67</sup> In any case, the particle could easily be left untranslated.

### 3.3.3 Cohortative

The cohortative expresses the speaker's desire, wish, or command, with the first person used as the subject of the action. In general, the cohortative emphasizes the determination behind the action, or one's personal interest in it. The following semantic categories are helpful in classifying the cohortative.

(a) **Resolve** – The speaker expresses self-determination to take an action that is within the speaker's ability: אָרֶדְה־בָּא וְאָרָאָה , אֶרְדָה־בָּא וְאָרָאָה , אָרְדָה־בָּא וְאָרָאָה in your stead-fast love" (Ps 31:8), אָרִידִי, (<u>I will exult</u> and <u>rejoice</u> in your steadfast love" (Ps 31:8), אָרָדְרָה בָּא לִירִידִי , אָרַבְרָה בְּמַר נַבָּשׁ ווו the speaker's ability:

 $<sup>^{64}</sup>$  Joüon suggests it is purely euphonic (1910, 143 and 378).

<sup>&</sup>lt;sup>65</sup> Fassberg 1999, 7-13.

<sup>&</sup>lt;sup>66</sup> Joüon and Muraoka 1993, 350–51.

<sup>&</sup>lt;sup>67</sup> Lambdin 1971a, 170–71.

bitterness of my soul" (Job 10:1), אָשׁיָרָה וַאַוּמָרָה לִיהוָה, "*I will sing* and *I will make melody* to YHWH" (Ps 27:6).<sup>68</sup>

(b) Wish – The speaker expresses the desire to take an action that requires the consent of another: אָשְׁבְרָה בְאַרְצֶךָ אָשְׁבְרָה בָאַרְצָרָ אָשְׁבְרָה בָּאַרָה אָל־אָחַי אָשְׁבְרָה בָּא וָאָשׁוּבָה אָל־אַחַי (Et me go back, tet me go back, tet me go back, tet me return to my kindred in Egypt" (Exod 4:18), <u>let me return</u> to my kindred in Egypt" (Exod 4:18), head" (2 Sam 16:9), אָלְכָה בָּשְׁבֵּלִים וַאַלַקְטָה בַשְׁבֵּלִים, (2 Sam 16:9), ביֹשְׁבֵּלִים, among the ears of grain" (Ruth 2:2).

At times, the consent seems doubtful, or the speaker incapable: אָבָה יְהוָה אַל־נָא נֹארְדָה הְּנָפָשׁ הָאִישׁ הַזֶּה, "please, O YHWH, <u>do not let us perish</u> on account of this man's life" (Jonah 1:14), על־מִי אֲדַבְּרָה וְאָשִ'רָה, "to whom <u>shall I speak</u> and <u>give warning</u>" (Jer 6:10).

(c) Exhortation – Plural speakers encourage one another to take action, or to help in the action: הְבָה וָלְבְנָה לְבַנִים, "come, <u>let us make bricks</u>" (Gen 11:3), (Gen 11:3), וְנַהְשָׁבָה עַל־יִרְמָיָהוּ מַחֲשָׁבוֹת (Ps 2:3), יוָנַהְשָׁבָה עַל־יִרְמָיָהוּ מַחֲשָׁבוֹת, "and <u>let us devise</u> plans against Jeremiah" (Jer 18:18).

# 3.4 Nonfinites

The Hebrew nonfinites are not inflected for *Aspekt*; that is, they make no reference to the action as undefined or progressive (as do the perfect and imperfect conjugations, see section 3.2). Furthermore, the two Hebrew infinitives are not

<sup>&</sup>lt;sup>68</sup> Here belong third-*hê* verbs that make no distinction between imperfect and cohortative in form: אָרָם בְצַלְמֵנוּ, "*let us make* humankind in our image" (Gen 1:26, and compare Gen 2:18 and others). See Kautzsch 1910, 210.

marked by person, gender, or number.<sup>69</sup> The participle is not marked by person, but is so marked for gender and number. The lack of specific inflection leaves these "nonfinites" less restricted in their uses than the conjugations or modals.

The nonfinites are based on verbal roots, but hold a middle position between verbs and nouns. The infinitives are "verbal nouns," displaying some uses similar to those of nouns and others similar to those of verbs. The participle is a "verbal adjective," being formed as a noun with gender and number, and often having functions like an adjective.

#### 3.4.1 Infinitive Construct

The infinitive construct comes closest to the uses of infinitives in other languages.<sup>70</sup> It is both "atemporal" and "apersonal," meaning that only the context determines the time/aspect features of the action as well as the subject of the action itself.<sup>71</sup> It has a wide and flexible range of uses, most of which occur in relation to another verb. It often functions as a noun in that it may be governed by prepositions and may take pronominal suffixes. Rather than being negated by  $\nabla \mathbf{x} \circ \mathbf{x}' \mathbf{x}$ , the infinitive construct takes  $\nabla \mathbf{y}$ .

Technically, the infinitive construct has no semantic function in itself: "The functions of an infinitive refer either to the syntactic function that it fulfills in a clause, or to the semantic relationship between itself and the finite verb."<sup>72</sup> The following semantic categories are helpful in classifying the infinitive construct.<sup>73</sup>

- <sup>72</sup> Van der Merwe, Naudé, and Kroeze 1999, 154.
- <sup>73</sup> Kautzsch 1910, 347–52; Waltke and O'Connor 1990, 598–611; Joüon and Muraoka 1993, 432–39; Meyer 1992; 399–403; Lambdin 1971a, 128– 29; Bergsträsser 1962, 2:53–60; Chisholm 1998, 77–78; van der Merwe, Naudé, and Kroeze 1999, 153–57; Williams 1976, 35–37. On the infinitive construct in epigraphic Hebrew, see Gogel 1998, 270–71.

<sup>&</sup>lt;sup>69</sup> Hebrew is not alone among the Northwest Semitic languages in having two infinitives (Harris 1939, 41; Garr 1985, 183–84).

<sup>&</sup>lt;sup>70</sup> Blau 1976, 47; Bergsträsser 1983, 56; Joüon and Muraoka 1993, 432.

<sup>&</sup>lt;sup>71</sup> Joüon and Muraoka 1993, 439.

(*a*) *Nominal* – when used as a noun or in place of a noun. Thus, the infinitive construct may function in nominative, genitive, or accusative functions.<sup>74</sup>

(a.1) Examples of the infinitive construct as *nominatives*:

עֵנוֹשׁ לַצַּדִיק לא־טוֹב

"<u>to impose a fine</u> on the innocent is not good" (Prov 17:26), שוֹב שֶׁבֶת בְּאֶרְץ־מִדְבָּר מֵאֲשֶׁת מְדְיָנִים וְכָעַס, "It is better <u>to live</u> in a desert land than with a contentious and fretful wife" (Prov 21:19), שוֹב תַּתִי אֹתָה לֶךָ (Prov 21:19), שוֹב תַּתִי אֹתָה לֶךָ.

(a.2) Examples of the infinitive construct as genitives: עָת סְפוֹד וְעֵת רְקוֹד, "a time <u>of mourning</u> and a time <u>of dancing</u>" (Eccl 3:4), עָת סְפוֹד וְעֵת רְקוֹד, "in the day <u>when you eat</u> (literally: <u>of your eating</u>) from it, you shall surely die" (Gen 2:17), לארעת הַאָּסַרְ הַמִּקְנָה, "it is not time for the cattle <u>to be gathered together</u> [literally: time <u>of being gathered together</u>]" (Gen 29:7). This category also includes the numerous times the infinitive construct occurs after a preposition (see sections 3.4.1,b–d, f, g).

(a.3) Examples of the infinitive construct as *accusatives*: לא אָרַע צָאָת וָבֹא, "I do not know how <u>to go out or to come in</u>" (1 Kgs 3:7), יְכֹר שָׁמְדִי לְפָנִיךּ, "remember how <u>I stood</u> before you [literally: <u>my standing</u> before you]" (Jer 18:20). Just as nouns may be *adverbial accusatives* (see section 2.3), so the infinitive construct may take on *adverbial* force when following a finite verb (often a Hiphil or Piel denominative). In such cases, the infinitive construct takes on the function of the main verb in the English translation: תוּהָרַעַ לְעַטוֹת, "why <u>have done</u> evil" (1 Kgs 14:9), (Exod 2:18).

To this category of the infinitive construct as accusatives belong many verbs that require an infinitive to complete their meanings. Thus, the infinitive construct may serve

<sup>&</sup>lt;sup>74</sup> The *nominal* infinitive construct may occur in any of these functions with the preposition *\*.

as a *verbal complement*: לא יוּכַל לְרָאוֹת, "he is not able <u>to see</u>" (Gen 48:10), לא־יֹאֶכֶה יְהוָה סְלוח לו, "Yhwh will not be willing <u>to pardon</u> him" (Deut 29:19). The most common verbs requiring such a complement include ידע (and other verbs of observation or cognition), ידע (Hiphil, to begin), יסך (Hiphil, to continue), בקש (Piel, to seek), יכל (to stop), יכל (to be capable of), מאן (to refuse), זכל (to allow), and אבה (to be willing).<sup>75</sup>

(b) *Temporal* – locates the action of the finite verb by relating it to the action of the infinitive construct. The uses of various prepositions on the *temporal* infinitive construct denote the specific moment in time.<sup>76</sup>

(b.1) The preposition  $\supseteq$  plus the infinitive (section 4.1.5,b) connotes action that is *simultaneous* with that of the finite verb, and may be translated "as," "when," or "while":

"<u>when</u> David <u>returned</u> from killing the Philistine, the women came out of all the towns of Israel" (1 Sam 18:6), ביוֹם הַהוּא בְּשֶׁבֶת עַבִּזְי יְשָׁרָאֵל לְבָטַח, "on that day <u>when</u> my people Israel <u>are dwelling</u> securely" (Ezek 38:14).

(b.2) The preposition  $\supseteq$  plus the infinitive (section 4.1.9,c) connotes action that occurs *immediately preceding* that of the finite verb, and may be translated "the moment when," or "as soon as":

וַיְהִי כְּבוֹא אַבְרָם מִצְרָיְמָה וַיִּרְאוּ הַמִּצְרִים אֶת־הָאִשֶׁה כִּי־יָפָה הַוּא מָאֹד

- <sup>75</sup> Van der Merwe, Naudé, and Kroeze 1999, 154, and Waltke and O'Connor 1990, 602.
- <sup>76</sup> Such constructions using the infinitive construct are especially frequent after יְיָהֵי (Kautzsch 1910, 347–48). For several other prepositions in addition to the ones discussed here, see Waltke and O'Connor 1990, 604–05.

"<u>As soon as</u> Abram <u>entered</u> Egypt, the Egyptians saw that the woman was very beautiful" (Gen 12:14), כְּשָׁמֹעַ עַשָּׁו אֶת־דְּבְרִי אָבִיו וַיְצְעַק צְעָקָה גְּדֹלָה וְמָרָה עַד־מָאֹד "<u>When Esau heard</u> his father's words, he cried out with an exceedingly great and bitter cry" (Gen 27:34). Similar to above, ק may also point to simultaneous action:

וַיְהִי כִרְאוֹת הַמֶּלֶךְ אֶת־אֶסְתֵּר הַמַּלְכָּה עֹמֶדֶת בָּחָצֵר נָשְׂאָה הֵן בְּצֵינָיו

"<u>The moment</u> the king <u>saw</u> Queen Esther standing in the court, she obtained favor in his sight" (Esth 5:2), (Esth 5:2), הָיָא כְעֹרְחַת עָלְתָה וָאָר וְהָיא כְעֹרְחַת עָלְתָה וָאָר (<u>As it was budding</u>, its blossoms came out" (Gen 40:10).

(b.3) The preposition עד plus the infinitive (section 4.1.15,b) connotes that the action of the finite verb occurs *during the period extending to* the action of the infinitive, and may be translated "until": ניא אוּכַל לְעַשׂוֹת דְּבָר עַר־בּאַך שָׁמָה, "I can do nothing <u>until</u> you <u>arrive</u> there" (Gen 19:22),

בְּזֵעַת אַפִּיך תֹאכַל לֶחֶם עַד שׁוּבְדָ אֶל־הָאֶדָמָה

"By the sweat of your face you shall eat bread <u>until</u> you <u>return</u> to the ground" (Gen 3:19), וַיָּאָבַק אַישׁ עָמוֹ עַד עַלוֹת הַשָּׁהַר, and a man wrestled with him <u>until</u> daybreak [literally: <u>until the coming up of</u> the dawn]" (Gen 32:25).

(b.4) The preposition אָחֵרֵי plus the infinitive (section 4.1.1,b) connotes that the action of the finite verb occurs *subsequent* to the action of the infinitive, and may be translated "after":

וּיִמָּלֵא שִׁבְעַת יָמִים אַחֲרֵי הַכּוֹת־יְהוֶה אֶת־הַיְאֹר

"seven days passed <u>after</u> YHWH <u>had struck</u> the Nile" (Exod 7:25), ויְהִי אֲחֵרִי הֵסַבּוּ אֹתוֹ וַתְּהִי יִרְ־יְהָוָה בָּעִיר, "<u>After</u> they <u>brought</u> it <u>around</u> [that is, the ark of the covenant to Gath], the hand of YHWH was against the city" (1 Sam 5:9). (c) Purpose – expresses the reason the action of the finite verb took place (most commonly used with ל, see section 4.1.10,d): רְיָבֵא אֶל־הָאָרָם לְרָאוֹת מַה־יָקְרָא־לוֹ, "and he [YHWH] brought [them] to the man <u>in order to see</u> what he would call them" (Gen 2:19), אַנִי יְהוָה אֲשֶׁר הוֹצֵאתִיך מֵאוּר כַּשִׁדִּים לָתֶת לְךָ אֶת־הָאָרֶץ הַוֹאת. "I am YHWH who brought you out from Ur of the Chaldeans <u>in order to give</u> to you this land" (Gen 15:7), "He is coming out <u>in order to meet</u> you" (Exod 4:14). Besides לְמַען (section 4.1.11,a) and בַּעֲבוּרָר

(d) Result - expresses the outcome or consequence of the action of the finite verb. This form of the infinitive normally occurs with  $\frac{1}{2}$ , and may be translated "and so," "so as," or "so that": וַיִּתְמַכְּרוּ לַשֲשׂוֹת הָרֵע בְּעֵינֵי יְהוֶה לְהָכָעִיסוֹ, "and they sold themselves to do evil in the sight of YHWH, so provoking him to anger" (2 Kgs 17:17), מָדוּשָ מָצָאתִי חֵן בִּשֵינִיך לְהַכִּירִנִי, "why have I found favor in your sight <u>so that you took notice</u> of me" (Ruth 2:10), וִיְגָרָשׁ שְׁלֹמה אֶת־אָבְיָתָר מִהִיוֹת כֹהן לַיהוָה לְמַלֵּא אֶת־דְבַר יְהוָה, "and Solomon banished Abiathar from being priest to Үнwн, so fulfilling the word of Үнwн" (1 Kgs 2:27), בְּדָ בָּחַר יְהוָה אֶלהֶיך לְהִיוֹת לוֹ לְעַם סְגָלָה, "Yhwh your God chose you and so you became his special people" (Deut 7:6). The distinction between *purpose* and *result* is not always clear, but result generally emphasizes the idea of sequence or progression (Joüon's "consecution" [1991, 436]). Besides , the infinitive construct of result also occurs with למען.77

(e) **Obligation** – expresses the burden of responsibility or necessity when used in verbless clauses: $^{78}$ 

- 77 Williams 1976, 36.
- <sup>78</sup> Other "modal" uses of the infinitive construct are also possible in verbless clauses. See Waltke and O'Connor 1990, 609–10.

<u>to give</u> you ten shekels of silver and a belt" (2 Sam 18:11). This use of the infinitive construct often takes a negative particle, and at times the negative obligation becomes *pro-hibition*: הַלוֹא לְכֶם לְרַעַת אֶת־הַמִּשֶׁפָּט, "is it not for you <u>to know</u> justice" (Mic 3:1), הַלוֹא לְכֶם לְרַעַת אֶת־הַמִּשֶׁפָט, "no one <u>was permitted to enter</u> the king's gate clothed with sack-cloth" (Esth 4:2).

(f) Imminence – expresses action expected soon or about to take place in constructions with ל plus infinitive construct in verbless clauses: יְהָוָה לְהוֹדִישָׁם, "Yhwh <u>will soon save me</u>" (Isa 38:20), וּבְרִיתוֹ לְהוֹדִישָׁם, "and he [YHwH] is <u>about to make known</u> his covenant to them" (Ps 25:14), הָרָה לְלַת הָרָה לְלַת, and his [Eli's] daughter-in-law, the wife of Phinehas, was pregnant, <u>about to give birth</u>" (1 Sam 4:19). For imminent action in past-tense narration, the construction adds a form of הַהָיָה הַרָּה לְבוֹא מִצְרִימָה Egypt" (Gen 12:11), "when he was <u>about to enter</u> Egypt" (Gen 12:11), "imminent action in set the sun was <u>at the point of setting</u>" (Gen 15:12), "and as the sun was and the gate was <u>about to be shut</u> at dark" (Josh 2:5).

(g) Specification – used with ל after a verb to clarify or explain further the preceding verbal action, which would otherwise be generally or ambiguously stated. In translation, the *specification* infinitive construct often requires "by ...-ing"):<sup>79</sup> ואָהָם מוֹסִיפִים הָרוֹן עַל־יִשְׁרָאֵל לְחַלֵּל אָת־הַשָּׁבָת (mand you are adding to the wrath on Israel <u>by profaning</u> the sabbath" (Neh 13:18),

הָלַך אַחֲרֵי בְּכָל־לְבָבוֹ לַעֲשׂות רֵק הַיָּשֶׁר בְּעֵינָי

<sup>&</sup>lt;sup>79</sup> Also known as *epexegetical, gerundive,* or *explanatory* infinitive construct (Waltke and O'Connor 1990, 608–09; Joüon and Muraoka 1993, 437– 48; Kautzsch 1910, 351; Williams 1976, 36).

"he followed me with all his heart, <u>by doing</u> only that which was right in my sight" (1 Kgs 14:8). The specification infinitive construct often has לְלֶכֶת or לְעֵשׂוֹת after יַשׁׁ:

"and the Israelites shall keep the sabbath, <u>by observing</u> the sabbath throughout their generations, as a perpetual covenant" (Exod 31:16),

"and you must keep the charge of YHWH your God, *by walking* in his ways" (1 Kgs 2:3).

To this category belongs the nearly ubiquitous לאמר (Qal infinitive construct of אמר), which is used to frame reported speech, frequently following verbs of speaking or thinking:<sup>80</sup> ווִדְבָר אֶלהִים אָל־נֹח לְאמר, "and God spoke to Noah, <u>by saying</u>" (Gen 8:15), ווִדְבָר אָלהִים לָאמר, "and God blessed them, <u>by saying</u>" (Gen 1:22). Such usage has taken on a stereotypical formula to mark reported speech.

### 3.4.2 Infinitive Absolute

The infinitive absolute is also both "atemporal" and "apersonal," meaning that only the context determines the time/aspect features of the action, as well as the subject of the action itself. This results from the fact that the infinitive absolute, like the infinitive construct, is also uninflected. On the other hand, unlike the infinitive construct, the infinitive absolute does not take prepositions or pronominal suffixes. Even though the two Hebrew infinitives appear similar in form, they apparently had completely different origins and histories.<sup>81</sup>

<sup>81</sup> Bergsträsser 1962, 2:61; Waltke and O'Connor 1990, 581–82.

<sup>&</sup>lt;sup>80</sup> A wide variety of verbs are used with אמר. For this term as a "complementizer introducing a highly marked form of direct speech," see Miller 1996, 163–212.

The following semantic categories are helpful in classifying the infinitive absolute.<sup>82</sup>

(a) Nominal – when used as a noun or in place of a noun. Thus, the infinitive absolute may serve in nominative, genitive, and accusative functions.

(a.1) Examples of the infinitive absolute functioning as *nominatives*: וְקָרוֹב לְשָׁמֹעַ מְהֵת הַכְּסִילִים וְבַח (and <u>to</u> <u>draw near</u> to listen is better than the sacrifice offered by fools," (Eccl 4:17 [Eng. 5:1]), ווָם־הֵיטֵיב אֵין אוֹתָם *well-doing* is not in them" (Jer 10:5), הַכָּר־פְּנִים לֹא־טוֹב, "<u>to show partiality</u> is not good" (Prov 28:21).

(a.2) Examples of the infinitive absolute functioning as *genitives*:<sup>83</sup> וּבְרוּחַ בְּעֵר, "and by the spirit <u>of burning</u>" (Isa 4:4), מְדֶרֶך הַשְׁכֵל, "from the way <u>of understanding</u>" (Prov 21:16). On rare occasions, the infinitive absolute may occur after a preposition.<sup>84</sup>

(a.3) Examples of the infinitive absolute functioning as accusatives: הֶבְשׁוֹ בְּבָשׁוֹם בְּרָע וּבְחוֹר בַּשּׁוֹם, "he shall eat curds and honey by the time he knows <u>to refuse</u> the evil and <u>to choose</u> the good" (Isa 7:15), לְמְדוּ הֵימֵב "learn <u>to do good</u>" (Isa 1:17), לא יֵאֲהַב־לֵץ הוֹכָח לוֹ (literally: <u>to reprove himself</u>]" (Prov 15:12).

(b) *Emphatic* – affirms and intensifies the authenticity or conviction of the verbal action when used with a finite verb of the same verbal root.<sup>85</sup> In one of the most common uses

- <sup>84</sup> Williams 1976, 38; Kautzsch 1910, 340.
- <sup>85</sup> By using the same root, the *emphatic* infinitive absolute becomes paronomastic; that is, it uses word play to create the emphasis. Earlier grammarians thought of this use as an *internal accusative* or an *accusative*

 <sup>&</sup>lt;sup>82</sup> Kautzsch 1910, 339–47; Waltke and O'Connor 1990, 580–97; Joüon and Muraoka 1993, 420–31; Meyer 1992, 403–07; Lambdin 1971a, 158–59; Bergsträsser 1962, 2:61–67; Chisholm 1998, 76–77; van der Merwe, Naudé, and Kroeze 1999, 157–62; Williams 1976, 37–39. On the infinitive absolute in epigraphic Hebrew, see Gogel 1998, 271.

<sup>&</sup>lt;sup>83</sup> Examples of genitives are rare, and may have different explanations. See Meyer 1992, 404, and Williams 1976, 38.

of the infinitive absolute, the infinitive normally precedes a finite verb of the same root, and is best translated by adverbs such as "certainly," "surely," "indeed," "definitely," and so on:

וְאַבְרָהָם הָיוֹ יִהָיֶה לְגוֹי נְּדוֹל

"
and Abraham <u>will surely become</u> a great nation" (Gen 18:18), הַעָר הַעָר הַעָר הָעָר מָצָר הָעָר מָרָר, "Gilgal <u>will certainly</u> go into exile" (Amos 5:5), הַלְאָלָל בָּלֹה יְגָלָה יְגָלָה יָבָלָה יָבָעָר destroyed" (Deut 4:26), הַשָּׁמִר הָשָׁמִרוּן, "Yuwu will be <u>utterly</u> destroyed" (Deut 4:26), הַעָּמָר הָשָׁמִרוּן, "Yuwu will <u>by no means</u> leave the guilty unpunished" (Nah 1:3). Sometimes the particle אַר טָרָך טַרָרָ מַרָרָ טַרָרָ טַרָרָ with has <u>surely</u> been torn to pieces" (Gen 44:28).<sup>86</sup>

At times the *emphatic* infinitive absolute follows the finite verb of the same root: "אַרָא יָצוֹא יָצוֹא יָצוֹא יָצוֹא יָצוֹא יָצוֹא יָצוֹא יָצוֹא יָצוֹא אַרָרּשָ (2 Kgs 5:11). Sometimes this postpositive position seems to connote continuous action or repetition: יָלְשָׁוֹא צָרַרְּ צָרוֹך in vain the refining *goes on*" (Jer 6:29), אָרְרִים אָמוֹר, "they *keep on* saying" (Jer 23:17), "Ided שָׁפּוֹט (Jer 23:17), "and he *constantly* acted as a judge" (Gen 19:9). However, while some grammarians continue to accept this durative connotation of the postpositive infinitive absolute, others doubt its validity.<sup>87</sup>

of internal object (Joüon and Muraoka 1993, 421–22). But comparative evidence from Ugaritic has shown that this use of the infinitive absolute is an extension of its use in the nominative case, and as such is an "appositional" nominative (Meyer 1992, 405) or an absolute complement resulting in what may be called an "intensifying infinitive" (Waltke and O'Connor 1990, 584–88).

- <sup>86</sup> The *emphatic* infinitive absolute is also used to strengthen contrasts, indignant questions expecting a negative answer, and conditional clauses. On these, see Kautzsch 1910, 343; Joüon and Muraoka 1993, 422–23; Waltke and O'Connor 1990, 587; or van der Merwe, Naudé, and Kroeze 1999, 159).
- <sup>87</sup> Of those who take the first position, see Meyer 1992, 63; Williams 1976, 38; Horsnell 1999, 73; Chisholm 1998, 77; of those who doubt this, see Waltke and O'Connor 1990, 585; Joüon and Muraoka 1993, 425; Seow 1995, 250–51.

Lambdin implies that the postpositive position is more common with intransitive roots.  $^{\$8}$ 

When the emphatic infinitive absolute follows a finite verb of the same root, it may be coordinated with a second infinitive absolute for simultaneous or durative action: ווֹיָצָא יָצוֹא (and it [the raven] <u>kept flying back and forth</u>" (Gen 8:7),

# וְהַמְאַפּף הֹלֵך אַחֲרֵי אָרוֹן יְהוֶה הָלוֹדְ וְתָקוֹעַ בַּשׁוֹפָרוֹת

"the rear guard came after the ark of YHWH, while they were <u>blowing the trumpets continually</u>" (Josh 6:13 *Qere*).<sup>89</sup> The use of the infinitive absolute for הלך in this function appears to have resulted in a formulaic use of the word, which is quite common: אַקר אָרָכָה אַרָרָ וּבָכָה אַרָרָ, "and her husband went with her, <u>weeping as he walked</u> behind her" (2 Sam 3:16).

(c) Manner – describes the way in which an action or situation is performed or the attendant circumstances of the action.<sup>90</sup> This is an infinitive absolute of a different root than the finite verb (and hence not the *emphatic* infinitive absolute). It typically follows the finite verb and modifies it adverbially: רְרָפוּ מָהֵר הָרְחַפּ מָוֹר מָחַרּץ לַמְחַנָה הַרְחַפּ מָן־הַמָּחַנָּה, "pursue <u>quickly</u>" (Josh 2:5), דְּרָפּר מָן־הַמָּחַנָה הַרְחַפּ מָן־הַמָּחַנָּה, "and he [Moses] pitched it [the tent of meeting] outside the camp, <u>a good distance</u> [literally: <u>far off</u>] from the camp" (Exod 33:7), בּחַרָּק מְקָהַ מְמָנָה הָדַרָּ (Exod 30:36), ויַעַּמְדוּ הַכֹּהְנִים נְשָׁצִי הָאָרוֹן בִּרִיִדִיהָוָה בָּחָנָהָ הַנָרָה בָּחַנָרָ ווּשׁצי ווּשַׁצי הָאָרוֹן בָּרִית־יִהּנָה הַחָרָ אַרָרָה בָּחַנָרָ וּשַרָּאַרוֹן מָרַרָּק מָוָביּרָה בַּחָנָרָ ווּשַרָּאַר הַמָּרָן הַבּחַנָּה ווּשַרָּאַרוּ מָמַנָּה הָבָרָה בָּחַנָרָה בָּחַנָרָ הַבָּחַנָּה מַוּצי ווּשַׁצי הָאָרוֹן בַּרִית־יִהּנָה בָּחָנָרָה בָּחַנָרָה בָּחַנָרָה ווּשַׁצי הָאָרוֹן בָּרִית־יִהּנָה בָּחַנָרָה בַּחַנָרָה הַכָּחַנָּה (Exod 30:36), יוּשַׁמָרוּ הַכּחַנִים נְשָׁצִי הָאָרוֹן בַּרִית־יִהּנָה בּחַנָרָם וּשַׁצי הַאָרוֹן בּרִית־יִהּנָה בּחַנָרָם הַיָּחָביו ווּשַׁצי הַאָרוֹן בַרִית־יִהּנָה בּחַנָרָם ווּשַׁצי זּחָר מוּשַרָּה מוּזי מּשׁרָי מוּר בּחַנָּיָה הַיָּרָה ווּשַרָּים ווּשַׁצי זּהַרָּרָה בָּחוֹן בַרִית־יִהּנָה בּחַנָים נַשָּצי הָאָרוֹן בָּרָה בָּחַנָים וּמָאַרוּ הַכּחַנָים נַשָּאַי הָאָרוֹן בָּרִיחַיּהוּ מּשָּרָה מוּשַרָּים מוּשַאַי הָאַרוֹן בָּרַיחַ מּשָׁצי הַרָּה מוּשָּרָה מוּשָׁצי הַרָּחָן מּשַרָּרָה מּשָּרָה מוּשָּרָיה מּשָּרָה מָבּחָנָים נַשָּאַר הַיַרָּה מַיָּרָה מַרָּהָה מּשָּרָה מּשָּשָּאי הַאָּרוֹן בּרִית־יִהּנָה בּחָנָרָה מוּשָּרָה מוּשָּאי הַאַרוּ מּשַרָים וּשָאַי הַירָיה מּשָּרָה מַרָּהָין הַיַרָּה מּשָּאַר הַיּאַרָין הַיַרָר מַשָּאָי הָיָרָ מּשָּרָי הַיַרָּין מּשָּרָה מּשָּרָה מוּשָּיָ הַיַין הַיַרָּין מּשָּרָין מּשָּרָה מוּשָּיָן הַיַין מּשָּין מּשָּרָים מוּשָּין הַיין ג

<sup>&</sup>lt;sup>88</sup> Lambdin 1971a, 158.

<sup>&</sup>lt;sup>89</sup> Note that the infinitive absolute has been restored in this verse in the *Qere* (קלוך), while the *Ketib* has an active participle (קלוך).

<sup>&</sup>lt;sup>90</sup> May be called an *adverbial complement* or an *adverbial infinitive* (Waltke and O'Connor 1990, 588–89).

Certain Hiphil and Piel infinitive absolutes were used for *manner* so naturally and frequently that they came to be treated as adverbs:

# אַחְאָב עָבַד אֶת־הַבַּעַל מְעָט יֵהוּא יַעַבְדֶנוּ הַרְבֵּה

"Ahab served Baal a little; Jehu will serve him <u>much</u>" (2 Kgs 10:18), סָרוּ מַהֵר מִן־הַדֶּרֶךְ אֲשֶׁר צִוּיְהָם, "they have <u>quickly</u> turned aside from the way I commanded them" (Exod 32:8). The infinitives most often used in this way are הַשְׁכֵם (much, many), הֵימֵב (well, thoroughly), הַשְׁכֵם (early), הַרְהֵק (far), and מָהָר (quickly).

(d) Verbal substitute – serves the function of a main verb, whether finite verb, modal, or even an infinitive construct:<sup>91</sup> רְגוֹם אֹתוֹ הָאָרָנִים כָּל־הָעֵרָה, "all the congregation <u>shall stone</u> him" (Num 15:35), רְגוֹם אֹתוֹ לָכֶם כָּל־זָכָר, (Gen 17:10), "every male among you <u>shall be circumcised</u>" (Gen 17:10), "every male among you <u>shall be circumcised</u>" (Gen 17:10), "the among you <u>shall be circumcised</u>" (Gen 17:10), "circ אֶת־יוֹם הַשֶּׁבָּת לְקַדְּשׁוֹ (Gen 41:43), זָכוֹר אֶת־יוֹם הַשֶּׁבָּת לְקַדְּשׁוֹ the sabbath day, to keep it holy" (Exod 20:8).

## 3.4.3 Participle

The participle is formed like a noun (taking gender and number), but has functions like the adjectives (section 2.5), and is thus a "verbal adjective." Like the other nonfinites, the participle is "atemporal," meaning that only the context determines the time/aspect features of the action.<sup>92</sup> Typically, the active participles (Qal active, Piel, and Hiphil)

<sup>91</sup> It has been suggested that this use of the infinitive absolute only occurs where one would expect a *waw*-consecutive form, and that this usage is related to the disappearance of the *waw* consecutive in LBH. See Rubenstein 1952, 262–67, and compare the use of the participle as a main verb in pre-exilic narrative, as discussed in Smith 1991, 27–33. For evidence of the same function surviving also in first-millennium Phoenician texts, see Garr 1985, 183–84.

<sup>92</sup> Joüon and Muraoka 1993, 409.

imply continuing or progressing action, whether past, present, or future. This durative idea is in fact more strongly expressed by the participle than by the imperfect (Joüon and Muraoka 1993, 412).<sup>93</sup> The passive participles connote completed action, often resulting in translation in a relative clause with a perfect or preterite verb (Lambdin 1971a, 18–19 and 158). In certain uses as an adjectival predicate, the participle has secondarily become a temporal form; that is, it may become a substitute for the imperfect.<sup>94</sup> When negated, the participle takes "X."

The following semantic categories are helpful in classifying the participle.  $^{95}$ 

(a) Attributive – ascribes a quality to a noun. Like the *attributive adjective*, the *attributive participle* forms a phrase with the noun it modifies, having a single function in the sentence. It stands in apposition to the noun, normally coming after it and agreeing in gender, number, and definiteness: אוֹבוֹרוֹעַ וְטוֹיָה, "a <u>listening</u> heart" (1 Kgs 3:9), הֹבוֹרוֹעַ וְטוֹיָ, "a <u>hearing</u> ear" (Deut 4:34), אוֹן שׁׁמִעָּת, "a <u>hearing</u> ear" (Prov 15:31).

Frequently, the participle requires a relative clause in translation (introduced by "who," "which," or "that"), in which case the attributive participle is usually definite:

מִבְּכוֹר פַּרְעֹה הַיֹּשֵׁב עַל־כָּסְאוֹ

- <sup>93</sup> The durative or progressive action of the active participles may be represented by a continuous line (i.e., \_\_\_\_\_) while the durative quality of the imperfect expresses repetition or multiplicity, and may be represented by a row of dots (i.e., .....). See Horsnell 1999, 78.
- <sup>94</sup> Specifically in LBH, the participle alone functions as a present tense, and in periphrastic clauses it indicates frequentative action (Sáenz-Badillos 1993, 129).
- <sup>95</sup> Kautzsch 1910, 355–62; Waltke and O'Connor 1990, 612–31; Joüon and Muraoka 1993, 409–18; Meyer 1992, 407–12; Lambdin 1971a, 18–19 and 157–58; Bergsträsser 1962, 2:68–74; Horsnell 1999, 78–84; Chisholm 1998, 67–70; van der Merwe, Naudé, and Kroeze 1999, 162–63; Williams 1976, 39–40.On the participles in epigraphic Hebrew, see Gogel 1998, 272–73.

"from the firstborn of Pharaoh <u>who sits</u> on his throne" (Exod 11:5), הַיוֹם הַבָּא, "the day <u>that is coming</u>" (Jer 47:4), הַכָּכֶּך הַמוּשָׁב, "the money <u>that was returned</u>" (Gen 43:12), "tablets of stone, <u>which were</u> <u>written</u> by the finger of God" (Exod 31:18).<sup>96</sup>

The attributive participle marks neither time nor aspect, both of which must be determined from the context. Thus, הַבָּא הַבָּא may mean "the army that came," "the army that is coming," or "the army that will come." In the context of 2 Chronicles 28:9, it can only mean "the army that came," since it is in a narratival context recounting past events.

(b) Predicate – expresses an assertion about a noun or pronoun in a nominal clause (i.e., in a clause without a finite verb). Like the *predicate adjective*, the *predicate participle* agrees in gender and number with the noun it modifies, and is always indefinite. The participle may stand before or after the noun or pronoun it modifies: אָרָרָם לָאָל שֶׁלִיוֹן (The heavens <u>are telling</u> the glory of God" (Ps 19:2 [Eng Ps 19:1]), דָרוּך אַרָרָם לָאָל שֶׁלִיוֹן, "<u>blessed is</u> Abram by God Most High" (Gen 14:19).

Because of its verbal character, the predicate participle may have a subject or a direct object, or it may be modified by an adverb or prepositional phrase.<sup>97</sup> As always, the participle does not express tense, but rather the durative or progressive action or condition that may occur in past, present, or future time, depending on the context.

(b.1) *Present* – expresses duration in present time, which is the participle's most natural verbal function due to its durative nature. The *predicate participle* indicating *present* action becomes a substitute for

<sup>&</sup>lt;sup>96</sup> See Waltke and O'Connor's *attributive relative* as a subcategory under *relative participles* (1990, 621–22).

<sup>&</sup>lt;sup>97</sup> The uses of the *predicate participle* are so similar to finite verbs that some grammars have a separate category of the participle as *verbal substitute*, or "participles as verbs" (see Kelley 1992, 200–01; van der Merwe, Naudé, and Kroeze 1999, 162).

the imperfect: אָלָכִי בֹּרְחַת, "I <u>am fleeing</u>" (Gen 16:8), לארכן אָנַחְנוּ עֹשִׁים, "we <u>are not doing</u> what is right" (2 Kgs 7:9).<sup>98</sup> This may denote general truths just revealed, or truths valuable for all time: רָרָע אָלְהִים, "for God <u>knows</u>" (Gen 3:5), רָרָע אָלָהִים עָמָרֶז לְעוֹלָם עַמְרָז, "a generation <u>goes</u> and a generation <u>comes</u>, but the earth <u>remains</u> forever" (Eccl 1:4). Occasionally the present predicate participle occurs after הַנָּה פָלְשֶׁתִים נַלְחָמִים בַּקְעִילָה <sup>6</sup>(Eccl 1:4). The express immediate action or conditions requiring observation (hence, the English, "behold"):<sup>99</sup> הַנָּה פַּלְשָׁתִים נַלְחָמִים בַּקַעִילָה הַנָּה פַלִשְׁתִים נָלְחָמִים בַּקַעִילָה 150%.

(b.2) Past – expresses duration in past time: וְנָהָר יֹצֵא מֵעֵדָן, "and a river <u>was flowing out</u> of Eden" (Gen 2:10), וְנָהָר יְבָא עֹשָׁה יְהוָה מַצְלִיחַ, "whatever he [Joseph] <u>was doing</u>, YHWH <u>was prospering</u> [or whatever he <u>did</u>, YHWH <u>prospered</u>]" (Gen 39:23), אַשָּׁר־הוּא עֹשָׁה יְהוָה בָּן־חַנִית מְהַנָשָׁא (Gen 39:23), "Now Adonijah son of Haggith <u>was exalting himself</u> [or, <u>exalted himself</u>]" (1 Kgs 1:5), בעַרַרסְרָם, "Lot <u>was sitting</u> in the gateway of Sodom" (Gen 19:1), "Sihon king of the Amorites, <u>who was living</u> at Heshbon" (Deut 3:2).

The *past predicate participle* frequently occurs after הַנָּה to express immediate action which occurred in the past (see b.1): הְנָּה אֹרְחַת יַשֶׁמְעֵאלִים בָּאָרִים "behold, a caravan of Ishmaelites <u>was coming</u>" (Gen 37:25), הְהַנָּה שְׁלְשָׁה אָנָשִׁים נִאָּרִים עָלִיו (behold, three men <u>were standing</u> near him" (Gen 18:2), הַנָּה אָנְחָנוּ מָאַלְמִים אָלָמִים בְּתוֹך הַשָּׁרָה (behold, we <u>were</u>

- <sup>98</sup> Though specific time reference is still determined by context only. In Mishnaic Hebrew, the participle actually becomes a finite verb for the present tense, while it continues to be used for frequentative or iterative action in both the past and future (Segal 1927, 155–57).
- <sup>99</sup> The use of הַשָּׁה plus the participle has been identified in recent rhetorical studies as one of the ways Hebrew narrators change points of view. This appears to be especially common in circumstantial clauses in which a narrator shifts the point of view to that of a character in order to narrate what the character sees as the action unfolds (Berlin 1983, 63). On הַשָּׁה as a focus particle drawing attention to the content of the clause that follows, see van der Merwe, Naudé, and Kroeze 1999, 328–30.

*binding* sheaves in the field" (Gen 37:7),

וִיְהִי הֵם מְרִיקִים שַׂקֵּיהֶם

"and they were emptying their sacks" (Gen 42:35).

The normally atemporal nature of the verbal participle is clarified with the addition of the verb היה to express past time clearly: וּמֹשֶׁה הָיָה רֹעֶה אֶת־צֹאן יִתְרוֹ חֹתְנוֹ, "and Moses <u>was pasturing</u> the flock of Jethro his father-in-law" (Exod 3:1), הְבָקר הִיוּ חֹרְשׁוֹת, "the oxen <u>were plowing</u>" (Job 1:14), הַבְּקָר הָיָוֹ חֹרְשׁוֹת, "you <u>have been rebellious</u> against YHWH" (Deut 9:24),

וָאָהִי צָם וּמִתפּלֵל לִפְנֵי אֵלהֵי הַשָּׁמִיִם

"and I *was fasting* and *praying* before the God of heaven" (Neh 1:4).

(b.3) Future – expresses duration in future time: אָנֹכִי מַמְטִיר עַל־הָאָרָץ אַרְבָּעִים יוֹם וְאַרְבָּעִים לְיָלָה min upon the earth forty days and forty nights" (Gen 7:4), למוֹעֵר הַזֶּה אַתִּ חֹבֶקֶת בַן, "At this season, in due time, you shall embrace a son" (2 Kgs 4:16 Qere).

Most frequently, the *future predicate participle* is used to announce approaching action, or action that is already in progress:<sup>100</sup> אָשֶׁר הָאָלהִים עֹשֶׁה הֶרְאָה אֶת־פַּרְעֹה (that which God <u>is about to do</u>, he has shown to Pharaoh" (Gen 41:28), Gen 41:28), "that which give are about to destroy this place" (Gen 19:13). As with the present and past predicate participles, the future predicate participle may indicate especially impending action when introduced with cate especially impending action when introduced with <u>am bringing</u> a flood of waters upon the earth" (Gen 6:17), <u>am bringing</u> a flood of waters upon the earth" (Gen 6:17), (Deut 31:16), הִנְיָ מֹתְנִי בְּנָרָ מְכָיַא (look, I <u>will deliver it</u> into your hand today" (1 Kgs 20:13). As with the past predicate

<sup>&</sup>lt;sup>100</sup> This announcement of imminent action, which has already in fact begun to take place, is a powerful literary tool in the hands of the Old Testament prophets. Announcements of judgment frequently use this future *predicate participle* for rhetorical effect.

participle, the future may occur with היה to express future time more clearly: וְכִםָּא דְוִר יִהְיֶה נָכוֹן לִפְנֵי יְהוָה עֵר־עוֹלָם, "and the throne of David <u>shall be established</u> before YHWH forever" (1 Kgs 2:45).

(c) Substantive – functions as a noun, most often occurring with the definite article: בְּיָמִי שָׁפֿט הָשׁׁפָטִים, "In the days when <u>the judges</u> ruled" (Ruth 1:1), קְנָה שְׁמִיִם וָאָרֶץ, "<u>the maker</u> of heaven and earth (Gen 14:19), קְנָה שְׁמִיִם וָאָרֶץ, "woe to <u>the one who builds</u> his house without righteousness" (Jer 22:13). The definite article may be omitted in poetry: וְהָלְכוּ בְּאוֹלִים, "but <u>redeemed people</u> shall walk [there]" (Isa 35:9). As a noun, the substantive participle functions in all the ways nouns are used. Thus, it may take pronominal suffixes: יְהָנָה שָׁמְרֶי הָשָׁמְרֵי (Ps 121:5), may stand in the construct state: אָשׁמְרֵי הָשׁמְרֵי (Ps 121:5), may stand in the construct state: שׁׁמְרֶרִי הָשׁמְרֵי (Ps 103:18). They may also serve in any of the nominative, genitive, or accusative functions.

The substantive participle serving in the genitive construction may fulfill any of the possible genitive constructions (section 2.2). However, one unique use of the participle occurs when certain verbs (especially בוֹא) and בוֹא) serve as constructs bound to another noun, rather than having an accusative or a prepositional phrase as one might expect:<sup>101</sup> בָּאֵי הַתַּרָה, "<u>those entering</u> at the gate of his city" (Gen 23:10), בָּאֵי הַתַּרָה, "<u>those coming out</u> of the ark" (Gen 9:10), יוֹרְדִי־בוֹר, "<u>those going down</u> to the pit" (Isa 38:18), אֹרְלֵי שֶׁלְהַן אִיוָבֶל, "<u>those eating</u> at Jezebel's table" (1 Kgs 18:19).

Certain verbal roots were used so widely as substantive participles that they in essence became nouns, most denoting vocations or other identifying roles: שוֹפָּט, "judge (one who judges)," סוֹפָר, "scribe (one who recounts),"

<sup>&</sup>lt;sup>101</sup> Joüon and Muraoka 1993, 416.

יוֹצֶר, "potter (one who forms)," אוֹיָב, "enemy (one showing enmity)," and so on.

### 3.5 Verbal Sequences

A specialized form of the conjunction waw may be prefixed to finite verbal forms in Hebrew narrative for functions not normally associated with the simple waw conjunctive.<sup>102</sup> Grammarians have used a variety of names for this feature of grammar: waw consecutive, waw inversive, relative waw, or waw conversive.<sup>103</sup> Each of these labels identifies the same phenomenon with a slightly different emphasis on the function of the waw in forming relationships between clauses. Whereas grammarians previously held that the waw simply converted the meaning of one verbal aspect to another (e.g., an imperfect form converted to a perfect meaning), more recent work demonstrates that the function of the waw with a finite verbal form affects not only the aspect of the verb but also the verbal ideas themselves. While the two main forms presented here (imperfects and perfects with waw consecutive) have different origins and connotations, in general we may consider succession in time or progression as the main idea.<sup>104</sup>

The following patterns are important to remember when noting verbal sequences in Hebrew narrative.<sup>105</sup>

- <sup>102</sup> The function of *waw* described here is rare in poetry. For the distinctions between narrative (discursive speech) and dialogue (direct speech), see van der Merwe, Naudé, and Kroeze 1999, 164–65; Miller 1996, 1–4 and 14–38. In general, discursive speech (narration) has a relatively limited variety of such functions, whereas dialogue (direct speech) has much more syntactical diversity.
- <sup>103</sup> See the beginning grammars for morphological features of the *waw* consecutive. On the terminology "*waw* consecutive," see Rainey (1986, 6).
- <sup>104</sup> For our best current understanding of the history and function of the waw consecutive, see Smith 1991, esp. 12–15.
- <sup>105</sup> Joüon and Muraoka 1993, 379–408; Meyer 1992, 386–89 and 393–94; Waltke and O'Connor 1990, 519–63; Kautzsch 1910, 326–39; Williams 1976, 33–34; Chisholm 1998, 94–112 and 119–33; van der Merwe, Naudé, and Kroeze 1999, 163–72; and for epigraphic Hebrew, see Gogel 1998, 260–68.

### 3.5.1 Imperfect plus waw Consecutive

In general, the imperfect with *waw* consecutive will connote the same aspect of the perfect, which will often precede it in the previous clause.<sup>106</sup> However, the imperfect plus *waw* consecutive pattern may also succeed a clause with an imperfect or a participle, or even a nominal clause.<sup>107</sup> In any case, it generally serves as a substitute for the perfect aspect, adding only the idea of succession. The following nuances will be helpful in interpreting the *waw* consecutive plus imperfect.

(a) Sequential – expresses temporal sequence, describing an action or situation subsequent to a previous action or situation: וְהָאָדְם יְדַע אֶת־חַוָּה אָשָׁחוֹ וְהַהָר וַהַּלֶד אֶת־קַיָן, "Now the man knew his wife Eve, <u>and</u> she <u>conceived</u> and <u>and then</u> <u>bore</u> Cain (Gen 4:1),

וַיַּעֲבֹר דָּוִד הָאֵבֶר וַיַּעֲמֹד עַל־רֹאשׁ־הָהָר

- <sup>106</sup> So it seems likely that this is really the original preterite tense of prebiblical Hebrew (see 3.2.2,e).
- <sup>107</sup> Several beginning grammars refer to a "governing verb" that influences the aspect and meaning of the prefixed waw consecutive in fixed relationships, such as perfect plus imperfect, or nonfinite plus imperfect (see Kelly 1992, 210–16; Seow 1995, 226–27). While these definitions are not inaccurate, and we recognize that a verb functions in sequence, we have chosen, for several reasons, not to define verbal sequences solely by the identification of the governing verb. Firstly, the identification of a governing verb in even simple narrative sequences can be quite complex and confusing. Secondly, as our examples show, the imperfect waw consecutive is often employed without a governing verb. Thirdly, the function of these verbal sequences is not as clear-cut as the beginning grammars tend to portray them. For instance, even though many stress the importance of the perfect plus imperfect sequence in narrative, the reader needs to be aware that a narrative rarely begins with a perfect verb but, in fact, more often begins with a waw consecutive plus imperfect (Niccacci 1990, 37; for an important exception see his 47). Furthermore, the function of these verbal sequences seems to differ on the basis of the narratival or discursive nature of the text at hand (Niccacci 1990, 35-45), a difference that is not given enough coverage by the beginning grammars, and often not appreciated by beginning and intermediate students.

"Then David crossed over to the other side, <u>and then</u> <u>he stood</u> on the top of the mountain" (1 Sam 26:13), וַיָּבֹא מֹשֶׁה וְאַהְרֹן אֶל־פָּרְעֹה וַיַּעַשׂוּ כֵן כַּאֲשֶׁר צִוָּה יְהוָה, "So Moses and Aaron came to Pharaoh, <u>and then they did</u>just as YHWH commanded" (Exod 7:10).

Within narrative, an action or situation that is not sequential to the main verb, but prior to it (equivalent to the English pluperfect or past perfect), is often presented by avoiding the ייקטי pattern and replacing it with the conjunctive waw on a finite form (see section 3.5.4,a). But, occasionally, waw consecutive plus imperfect may be used to mark such anterior constructions, which must be discerned from the context (Joüon and Muraoka 1993, 390– 91; Zevit 1998):

# וּיִדַּבֵּר אֲלָהֶם אֲבִיהֶם אִי־זֶה הַדֶּרֶךְ הָלָךְ וַיִּרְאוּ בָנָיו אֶת־הַדֶּרֶךְ אֲשֶׁר הָלַךְ אִישׁ הָאֱלֹהִים אֲשֶׁר־בָּא מִיהוּדָה

"Their father spoke to them, 'Which way did he go?' <u>Now</u> <u>his sons had seen</u> the way that the man of God who came from Judah <u>had gone</u>" (1 Kgs 13:12).

(b) Consequential – expresses logical result, describing an action or situation resulting from a previous action or situation: אָת־יוֹסֵך וַיְהִי אָישׁ מַצְּלִיחַ, "Yhwh was with Joseph, <u>and so</u> he became a successful man" (Gen 39:2), Joseph, <u>and so</u> he became a successful man" (Gen 39:2), "<u>Thus</u> the heavens and the earth were completed" (Gen 2:1), <u>"Thus</u> the heavens and the earth were completed" (Gen 2:1), (יָקַלַע וּבָאָרֶץ, "<u>So</u> David prevailed over the Philistine with a sling and a stone" (1 Sam 17:50). This consequential use of the ליק pattern may be used to conclude a narrative:

"Then he would return to Ramah because his house was there; and there he judged Israel. And <u>he built</u> there an altar to YHWH" (1 Sam 7:17),

וַיְהִי יְהוֶה אֶת־יְהוֹשֵׁעַ וַיְהִי שָׁמְעוֹ בְּכָל־הָאֶָרֶץ

"<u>So YHWH was</u> with Joshua, and his fame <u>was</u> in all the land" (Josh 6:27).

(c) Narratival – Due to the frequency of its use, the waw consecutive plus imperfect came to be used on its own, independent of a preceding verbal clause, in order to begin a narrative sequence or a new section of narrative: אָלָיו יְהָוָה בְּאֵלְיֵי מְמָרָא אָלִיו יָהוָה הַאָלִיו יְהָוָה בָאַלְיֵי מָמָרָא to him [Abraham] by the oaks of Mamre" (Gen 18:1). This narratival use of the ויִקְמָשׁ pattern may even begin a book: וִיִקְרָא אָלִימָשָׁה וְיָדְבָר יְהוָה אֵלִיו מָאֹהֶל מוֹעֵד יָוִיקָרָא אָל־מֹשָׁה וְיָדְבַר יְהוָה אֵלִיו מָאֹהֶל מוֹעַד (YHWH <u>called</u> to Moses and spoke to him from the Tent of Meeting" (Lev 1:1), יוָדְבַר יְהוָה אֶל־מֹשֶׁה בְמִדְבַר סְיַנֵי (<u>Then</u> YHWH <u>spoke</u> to Moses in the wilderness of Sinai" (Num 1:1).

The narratival use is especially common with the imperfect of the verb דָּיָה to introduce a new narrative or section of a narrative: וְיָהִי אֵישׁ אֶחָד מִן־הָרָמָתִיִם צוֹפִים, "<u>There was</u> a certain man from Ramathaim Zophim" (1 Sam 1:1), a certain man from Ramathaim Zophim" (1 Sam 1:1), (Jonah 1:1).

(d) Epexegetical – clarifies, expands, or paraphrases the clause that precedes it (the simple conjunctive waw may be used on nouns to perform a similar function; compare section 4.3.3,d): אָבָל אָשֶׁה־אַלְמָנָה אָנִי וַיְמָת אִישׁי, "Truly, I am a widow; <u>my husband has died</u>" (2 Sam 14:5), "Truly, I am a widow; <u>my husband has died</u>" (2 Sam 14:5), "Truly, I am a widow; <u>my husband has died</u>" (2 Sam 14:5), "Truly, I am a widow; <u>my husband has died</u>" (2 Sam 14:5), "Truly, I am a widow; <u>my husband has died</u>" (2 Sam 14:5), "Truly, I am a widow; <u>my husband has died</u>" (2 Sam 14:5), "Truly, I am a widow; <u>my husband has died</u>" (2 Sam 14:5), "Truly, I am a widow; <u>my husband has died</u>" (2 Sam 14:5), "Change Moses, <u>saying</u>, "Because I drew him out of the water"" (Exod 2:10), באָרָבָרָי, "Gen 31:26).

On occasion, the clarification of the verbal action can repeat the main verbal idea:

"The chief cup-bearer did not remember Joseph; *<u>he forgot him</u>*" (Gen 40:23),

הַלוֹא־הוּא אָבִיד קָנָד הוּא עַשְׂד וַיְכֹנְנָד

"Is he not your father, who created you, who made you and <u>established you</u>?" (Deut 32:6), גוְדָוִד נְס וִיָּמְלֵט בַּלֵיְלָה הוּא "David fled; <u>he escaped</u> that night" (1 Sam 19:10).

Nearly identical to this use is *specification*, in which *waw* consecutive plus imperfect makes a previous verbal idea more specific:

וַיִּשְׁכְּחוּ אֶת־יְהוֶה אֱלֹהֵיהֶם וַיַּעַבְדוּ אֶת־הַבְּעָלִים וְאֶת־הָאֲשֵׁרוֹת

"They forgot YHWH their God, <u>they served</u> the Baals and the Ashtoreth" (Judg 3:7), אָנָא לְאָל הָיו וַיְכָפּר עַל־בְּנֵי יִשְׂרָאָל "He was zealous for his God; <u>thus he made atonement</u> for the people of Israel" (Num 25:13).

(e) Dependent – Following temporal clauses, the וְיָקְטָּל pattern will present an action or situation that is dependent on the temporal clause: בְּשָׁרָאָרְאָרָאָר אָרָאָרָעָרָ עַזְיָהוּ וָאָרְאָה אָת־אַרָנִי "In the year of King Uzziah's death, <u>I saw</u> the lord" (Isa 6:1), "In the year of King Uzziah's death, <u>I saw</u> the lord" (Isa 6:1), Eagling the lord (Isa 6:1), בְּשָׁרָה וַיָּקָם קַיִן אָל־הֶבָל אָחִיו field, Cain <u>rose up</u> against Abel, his brother" (Gen 4:8), field, Cain <u>rose up</u> against Abel, his brother" (Gen 4:8), heard the law, <u>they separated</u> all the aliens from Israel" (Neh 13:3).

### 3.5.2 Perfect plus waw Consecutive

The perfect with *waw* consecutive will connote the same aspect of the imperfect, which will often precede it in the previous clause. However, *waw* consecutive plus perfect may also succeed a clause with an imperative, a perfect, a participle, an infinitive functioning as a finite verb, or a nominal clause. In any case, it generally serves as a substitute for the imperfect aspect, adding only the idea of succession. The following nuances will be helpful in interpreting the *waw* consecutive plus perfect. (a) Sequential – expresses temporal sequence, describing an action or situation subsequent to a previous action or situation:

קַרַע יְהוֶה אֶת־מַמְלְכוּת יִשְׂרָאֵל מֵעָלֶידָ הַיוֹם וּנְתָנָה לְרֵעֲדָ

"Yhwh has torn the kingdom of Israel from you today, <u>and he has given</u> it to your neighbor" (1 Sam 15:28), אַצַא וְהָיִתִי רוּחַ שֶׁקֶר בְּפִי כָּלֹ-וְבִיאָיו, (1 Kgs 22:22), אַצַא וְהָיִתִי רוּחַ שֶׁקֶר בְּפִי כָּלֹ-וְבִיאָיו, (1 Kgs 22:22), וּבְיוֹם פָּקָדִי וּפָּקַדְתִי עֲלֵיהֶם חַטָּאהָם, (1 Kgs 22:22), וּבְיוֹם פָּקָדִי וּפָּקַדְתִי שָׁלִיהֶם חַטָּאהָם, (1 Kgs 22:22), וּבְיוֹם פָּקָדִי וּפָּקַדְתִי שָׁלִיהֶם חַטָּאהָם, (1 Kgs 22:22), וּבְיוֹם פָּקָדִי וּפָּקַדְתִי עֲלֵיהֶם חַטָּאהָם, (1 Kgs 22:22), וּבְיוֹם פָּקָדִי וּפָּקַדְתִי שָׁלִיהָם חַטָּאהָם, (1 Kgs 22:22), וּבְיוֹם פָּקָדִי וּפָקַדְתִי שָׁלִיהָם חַטָּאָהָם, (1 Kgs 22:22), וּבְיוֹם פָּקָדִי וּפָקַדְתִי שָׁלִיהָם חַטָּאָהָם, (1 Kgs 22:22), וּבְיוֹם מָשָׁאָהָם הָרָה וְיֹלֶדֶת בֵּן וְקָרָאָת שָׁמוֹ עַמָּנוּ אֵל (1 kgs 1 kgs 22:22), וּבְיוֹם מָשָׁאָהָם הָרָה וְיֹלֶדֶת בֵּן וְקָרָאָת שָׁמוֹ עַמָּנוּ אֵל הַעַּלְמָה הָרָה וְיֹלֶדֶר בַּוֹן הַיָּרָם הַיָּרָאָד מָשָׁמוֹ עַמָּנוּ אַל

(b) Consequential – expresses logical result, describing an action or situation resulting from a previous action or situation: אָרָי הָוָה וְהוֹצֵאָרִי אֶרְכָם, "I am Yhwh, and <u>I will bring you out</u>" (Exod 6:6),

> וּבְנֵי יִשְׂרָאֵל פָּרוּ וַיִּשְׁרְצוּ וַיִּרְבּוּ וַיַּעַצְמוּ בִּמְאֹד מְאֹד וַתִּמְלֵא הָאֶרֶץ אֹתָם

"The people of Israel were fruitful, and increased greatly, and multiplied and became exceedingly mighty, <u>so that the land was filled</u> with them" (Exod 1:7), אָרָהָרָיָן יְהָוָה אָרְכֶם בְּעַמִים וְנִשְׁאַרְתָם מְתֵי מְסָפָּר , "Yhwh will scatter you among the peoples, and <u>you will be left</u> few in number" (Deut 4:27).

(c) Volitional – expresses a command or wish:

וַאָָהַבְהֶם אֶת־הַגָּר כִּי־גֵרִים הֲיִיתֶם בְּאֶרֶץ מִצְרָיִם

"<u>You shall love</u> the stranger, for you were strangers in the land of Egypt" (Deut 10:19), וּקְשָׁרְתָּם לְאוֹת עַל־יָרֶך "<u>You shall bind</u> them as a sign upon your hand" (Deut 6:8). This volitional use of the perfect plus waw consecutive is often preceded by a clause with a jussive, imperative or cohortative, or an infinitive absolute functioning as an imperative (see section 3.4.2,d), and functions to express *consecution*:<sup>108</sup> יִבַקְשׁוּ לַארֹנִי הַמֶּלֶךְ נַעֲרָה בְתוּלָה וְשָמְדָה לִפְנֵי הַמֶּלֶךְ (יַבְקְשׁוּ לַארֹנִי הַמֶּלֶך נַעֲרָה בְתוּלָה וְשָמְדָה לִפְנֵי הַמָּלֶך "Let them seek a young virgin for my lord the king, <u>and</u> <u>let her attend</u> the king" (1 Kgs 1:2),

לא־יָמוּשׁ סֵפֶר הַתּוֹרָה הַזֶּה מִפִּידְ וְהָגִיתָ בּוֹ יוֹמָם וְלַיְלָה

"This book of the law shall not depart from your mouth, but <u>you shall meditate</u> on it day and night" (Josh 1:8). The waw consecutive plus perfect after a modal, or an equivalent infinitive absolute, may emphasize succession, so that the execution of its action or situation is dependent upon the preceding one: אָשֶׁה כָּאֲשֶׁר דְּבֶר וּפְגַע־בּוֹ וּקְבַרְתּוֹ he has said and strike him down <u>and then bury</u> him" (1 Kgs 2:31), אָשָׁרְהָ בְּאַחַד חַמָּלְמוֹת וְלֵנוֹ (1 Kgs 2:31), בָּגַה־לְךָ בִיָת בְּירוּשָׁלַם וְיָשָׁבְתָּ שָׁם one of these places, and [then] <u>spend the night</u>" (Judg 19:13), בְּגַה־לְךָ בִיָת בְּירוּשָׁלַם וְיָשָׁבְתָּ שָׁם house in Jerusalem, <u>and stay</u> there" (1 Kgs 2:36).

Frequently, the idea of succession can be nuanced to express purpose or result:<sup>109</sup> אָקָה וְיָרַשְׁנוּ אֹתָה, "Let us go up at once and *[so] let us possess it*" (Num 13:30), נעמים בַּיָרְדָן, "Go, <u>and wash</u> seven times in the Jordan" (2 Kgs 5:10), הָלוֹדְ וְרָחַצָּהָ שֶׁבַע־פְּעָמִים בַּיָרְדָן (1 Sam 15:30), וְשׁוּב עָמֵוּ לָהֶם וְחָיוּ וְלֹא יָמָתוּ (1 Sam 15:30), (1 Sam 15:30), ווֹאָת עֲשׁוּ לָהֶם וְחָיוּ וְלֹא יָמָתוּ, "Do this to them, <u>so that they may live</u> and not die" (Num 4:19).

(*d*) *Apodictic* – in a conditional sentence, expresses logical result, describing an action or situation resulting from (or dependent on) a previous action or situation: אָם־חָפּץ בְּנוּ יְהוָה וְהֵרִיא אֹתְנוּ אֶל־הָאָרֵץ הַוֹּאָר is pleased with us, <u>then he will lead</u> us into this land"

<sup>&</sup>lt;sup>108</sup> Lambdin 1971a, 119; Seow 1995, 243-44.

<sup>&</sup>lt;sup>109</sup> The perfect plus *waw*-consecutive construction most often follows an imperative, and is frequently identical to the person, gender, and number of the imperative. As the last two examples in this paragraph illustrate, however, there are some first- and third-person examples of this construction that carry a successive idea from an imperative (Joüon and Muraoka 1993, 399–400).

(Num 14:8), אִם־יִהְיָה אֲלֹהִים עָמָּדִי... וְהָיָה יְהוָה לִי לֵאלֹהִים , "If God will be with me... <u>then YHWH will be</u> my God" (Gen 28:20–21), אָטָר אָרָנוּ וְיָצָאנוּ אֵלֶיך , "If there is no one to deliver us, <u>then we will surrender</u> to you" (1 Sam 11:3).

Quite often, *waw* consecutive plus perfect functions not to present the apodosis, but to provide an expansion of the idea in either the protasis, or the apodsis: אָם־שָׁמוֹעַ הָּשָׁמְעוּ בְּלְלִי וּשְׁמְרָשָּם אָת־בְּרִיתִי אָם־שָׁמוֹעַ הָּשָׁמְעוּ בְּלְלִי וּשְׁמַרְשָּם אָת־בְּרִיתִי obey my voice <u>and keep</u> my covenant" (Exod 19:5), cy הַעַּזְבוּ אֶת־יְהְוָה וַעֲבַרְשָּם אֶלהי נֵכָר coverate for eign gods..." (Josh 24:20),

> ּכִּי־יִהְיֶה לְהֶם דֶּבָר בָּא אַלֵי וְשָׁפַּמְתִי בֵּין אִישׁ וּבֵין רֵעֵהוּ וְהוֹדַעְתִי אֶת־חֻפֵּי הָאֶלהִים וְאֶת־תּוֹרֹתָיו

"If they have a dispute, it comes to me, <u>then I judge</u> between a person and that person's neighbor, <u>and I make known</u> the statutes of God and his instructions" (Exod 18:16).

Verbal sequence relationships are not limited to *waw* consecutive forms, but can also be expressed by verbs that are prefixed with the simple *waw* conjunctive. Thus, a *waw* conjunctive on a perfect<sup>110</sup> or imperfect<sup>111</sup> form can also express succession or consecution. Consider: עָשִׁיהִי לִי בָּנֹת וְםָרְדָסִים וְוָטֵעָהִי בְהָם עֵין כָּל־בָּרִי.

<sup>110</sup> Particularly in LBH, the *waw* conjunctive plus perfect is viewed as a substitute for the *waw* consecutive plus imperfect. This use is based on the theorized influence of Aramaic, which lacked *waw*-consecutive forms (Smith 1991, 27–33). However, evidence from the Tel Dan inscription indicates that there very well could have been a *waw* consecutive plus imperfect construction employed in Aramaic (see Emerton 1994, 255–58; and 2000, 35). This is in addition to a ninth-century Aramaic inscription in which this construction is found (*KAI* 202, line 11).

<sup>111</sup> It should be remembered that certain imperfect forms take a shortened form before the *waw* consecutive and a normal form (so-called longer form) with *waw* conjunctive. Thus, the imperfect of these verbs makes a morphological distinction between imperfect with *waw* consecutive and imperfect with *waw* conjunctive. See beginning grammars for more on the morphology of the *waw* consecutive. for myself gardens and parks, *and [then] I planted* in them all kinds of fruit trees" (Eccl 2:5),

וַיִּשְׁתּחוּ לְכָל־צְבָא הַשְׁמַיִם וַיַּעֲבֹד אֹתָם וּבָנָה מִזְבְּחוֹת בְּבֵית יִהוָה

"He worshiped all the hosts of heaven, and he served them, <u>and he built</u> altars [for them] in the house of Үнwн" (2 Chr 33:3-4), כִי מִי עָמֵד הְסוֹד יְהוָה וְיֵרֶא וְיִשֶׁמֵע אֶת־הְכָרוֹ "But who has stood in the council of Үнwн, <u>that they</u> <u>should see and hear</u> his word" (Jer 23:18),

הַאֵין פּה נָבִיא לַיהוָה עוֹד וְנִדְרְשָׁה מֵאוֹתוֹ

"Is there not yet a prophet of YHWH here, <u>so that we may</u> <u>inquire</u> of him" (1 Kgs 22:7).

Additionally, the *waw* conjunctive on a finite verbal form expresses all the various functions of the *waw* conjunction (see section 4.3.3), in particular, verbal hendiadys (section 4.3.3,g), coordination (section 4.3.3,b), or continuation.<sup>112</sup>

## 3.5.3 Commands in Verbal Sequences

As noted, the perfect plus *waw* consecutive may express volition (see section 3.5.2,c). BH may also use the *waw* conjunctive with imperfects and volitives (and at times perfects) to express sequential commands, which may connote succession or purpose.<sup>113</sup>

(a) Succession – shows an action that temporally follows the main verb: אַל־נָא יָחַר לָארֹנָי וַאֲרַבָּרָה, "May my lord not be angry, <u>and I shall speak</u>" (Gen 18:30), שאַל אֶת־נְשָׁרָיךּ וְיַגִּידוּ לְךָ <u>will tell</u> you" (1 Sam 25:8),

וְנִזְעַק אֵלֶיך מִצְּרָחֵנוּ וְחִשְׁמַע וְתוֹשִׁיעַ

<sup>&</sup>lt;sup>112</sup> Niccacci 1990, 40, and Waltke and O'Connor 1990, 540–41.

<sup>&</sup>lt;sup>113</sup> Lambdin 1971a, 119; Seow 1995, 243; Niccacci 1990, 187.

"We will cry out to you in our distress, and <u>you will hear us</u>, and <u>save us</u>" (2 Chr 20:9), לֵךְ אֶרֶץ יְהוּדָה, "Go, <u>and [then] enter</u> into the land of Judah" (1 Sam 22:5).

(b) Purpose – expresses the purpose or motivation behind the main verb: וְהָרִיאָה לִי וְאֹבֵלָה, "Bring it to me, so that I may eat" (Gen 27:4),

אָסַרָה־נָּא וְאֶרְאֶה אֶת־הַמַּרְאֶה הַגָּדֹל הַזֶּה

"Let me turn <u>in order to see</u> this great sight" (Exod 3:3), וְהַרְגְנוּם וְהַשְׁבַּתְנוּ אֶת־הַמְּלָאָכָה, "Kill them, <u>and</u> <u>[so], put a stop</u> to the work" (Neh 4:5 [Eng. 4:11]), <u>[so], put a stop</u> to the work" (Neh 4:5 [Eng. 4:11]), "I will let the people go, <u>that</u> <u>they may sacrifice</u> to YHWH" (Exod 8:4).

## 3.5.4 Interruptions in Verbal Sequences

The *waw* conjunctive employed with a verbal idea is also used in narrative texts to present an idea that is strongly disjunctive from the *waw* consecutives in context. This disjunction occurs with both perfect and imperfect verbs, but typically, the *waw* conjunctive is not prefixed onto the verbal form, but an associated noun (that is, either the subject or the object of the noun).<sup>114</sup> The nonsuccessive idea presented by the *waw* conjunctive can express several nuances.

(a) **Distinct subject** – the *waw* conjunctive points to a seemingly successive action performed by a different subject, thereby placing emphasis on an agent who is distinct from the agent of the main verb, and, in effect, presenting an opposition of subjects:

וַיֵּט מֹשֶׁה אֶת־מַמֵּהוּ עַל־הַשְּׁמַיִם וַיהוָה נָתַן קֹלֹת וּבָרָד

<sup>&</sup>lt;sup>114</sup> This "*waw*+subject+finite verb" sequence is variously diagrammed as "w...qatal/w...yiqtol" (Joüon and Muraoka 1993, 390 and 397) or "waw x qatal" (Niccacci 1990, 49–72).

"Moses stretched his staff to the heavens, *and YHWH sent* thunder and hail" (Exod 9:23),

"He sent the men away and they left. <u>But he himself</u> went and stood before his master" (2 Kgs 5:24-5).

(b) Simultaneous action – The waw conjunctive expresses an action that is not successive, but simultaneous with the main verb: ויִקְרָא אֶלהִים לָאוֹר יוֹם וְלַחֹשֶׁךְ קָרָא לְיָלָה "God called the light day, <u>and he called</u> the darkness light" (Gen 1:5), ניֵלֶך שְׁמוּאֵל הָרָמָתָה וְשָׁאוּל עָלָה אָל־בֵּיתוֹ וּבְעַת שָׁאוּל (Samuel went to Ramah, <u>and Saul went up</u> to his house at Gibeah of Saul" (1 Sam 15:34), יוֹר יוֹם (Gen 12:12).

(c) Anterior action – The *waw* conjunctive signals an action or situation that is out of chronological order in the narratival flow, and is equivalent to the English pluperfect or present perfect in narratives of past time. On occasion, parenthetical remarks within a narrative will be presented in this manner:<sup>115</sup>

<sup>115</sup> Some assert that this pluperfect use is expressed not only by the waw conjunctive but also by the perfect plus waw conjunctive (Waltke and O'Connor 1990, 541–42, and Johnson 1979, 41; and cf. רְיָרָא עָשָׁו בִיֹרָה אָרִם"), "Esau saw that Isaac blessed Jacob, and <u>had sent him</u> to Padan-Aram" [Gen 28:6]). Niccacci further states that the perfect plus waw conjunctive expresses background or parenthetical comments in narrative (1990, 35). Recently, Zevit has argued that authors of BH narrative could mark pluperfect or present-perfect time unambiguously by following a narratival past-time clause with the following sequence: waw conjunctive plus subject (noun or pronoun) plus perfect finite verb (Zevit 1998, 15–37).

"Laban went into Jacob's tent, and into Leah's tent, and into the tent of the two maid-servants, but he did not find [them]. He came out from Leah's tent, and went into Rachel's tent, but <u>Rachel had taken</u> the household idols..."(Gen 31:33-34),

```
וּשְׁמוּאֵל מֵת וַיִּסְפְּדוּ־לוֹ כָּל־יִשְׂרָאֵל וַיִּקְבְּרֶהוּ בָרָמָה וּבְעִירוֹ
וְשֵׁאוּל הֵסִיר הָאֹבוֹת וְאֶת־הַיִּדְּעֹנִים מֵהָאֶרֶץ
```

"Samuel was dead, and all Israel mourned for him, and they buried him in Ramah, his city. And Saul <u>had expelled</u> the mediums and the spiritists from the land" (1 Sam 28:3),

"The king of Israel disguised himself and went into battle. Now, the king of Aram *had commanded* the thirty-two officers of his chariots" (1 Kgs 22:30–31).

# **4** Particles

n addition to nouns and verbs, Hebrew syntax employs another category of words, often referred to as *particles*. This is an inclusive term often used for a wide-ranging number of parts of speech. Here we will use it to refer to prepositions, adverbs, conjunctions, the particles of existence and nonexistence, and the particles and הנה מה מה מה יום.<sup>1</sup>

### 4.1 Prepositions

In general, prepositions in BH are difficult to categorize because their function in the syntax may be viewed from a variety of perspectives. As relational terms standing before nouns (or noun equivalents), prepositions may be viewed primarily as "nominal" in orientation and function.<sup>2</sup> On the other hand, most of the Hebrew prepositions have no obvious morphological connection to nouns, and their nature as "particles" is frequently evident in their function as adverbs or conjunctions. However, a third perspective, the "semantic" perspective of prepositions, has become increasingly more prominent in modern grammars, and is important to consider here. In

<sup>&</sup>lt;sup>1</sup> Because Hebrew grammars traditionally include so many parts of speech under the rubric, some have chosen to use "other word classes," or something of the like (van der Merwe, Naudé, and Kroeze 1999, 271).

<sup>&</sup>lt;sup>2</sup> On these three perspectives of Hebrew prepositions, see Waltke and O'Connor 1990, 187–90.

essence, this perspective means that some prepositions in particular uses have connotations determined by the verb with which they are used. In other words, the meaning of certain prepositions is not so much determined by morphological origins or by use with specific nouns as it is by the particular pattern of *verb plus preposition plus object*. Because of the use of these patterns, it is insufficient to study only the use of prepositions in isolation, as we will present them here. It will also be necessary to consider the way individual prepositions are used with particular verbs, which is ultimately a lexical matter. Since the meaning of some prepositions is determined by the verbs that govern them, it is necessary for the exegete to learn particular prepositions used with certain verbs, or to use the lexica for specific verbs to determine their meaning when used with a given preposition.

The following are the most common prepositions (in alphabetical order) and their uses.<sup>3</sup>

#### אַחֲרִי / אַחַר 4.1.1

(a) Spatial – indicates localization, especially "behind" a place: אָחָרִי הַמִּשְׁכָן, "<u>behind</u> the tabernacle" (Num 3:23), עוֹמֵר בָּחָלֵנוּ, "he is standing <u>behind</u> our wall" (Song 2:9).<sup>4</sup> Related to this spatial use is the *directional* use of אָחָר אָחָר הוא אָחָר הוא אָחָר הוא אָחָר הוא אָחָר הוא אָחָר הוא אָחָר בּחָלֵנוּ, In BH, the four directional compass points are often expressed from an eastward-facing position, so that left and right represent north and south respectively, while front and back represent east and west. Hence, "behind" (אָחָר הְנָה אָחַר הָבָה יָשָׁרִים, "it is <u>west</u> of Kiriath-jearim" (Judg 18:12).

- <sup>3</sup> Waltke and O'Connor 1990, 190–225; van der Merwe, Naudé, and Kroeze 1999, 277–94; Williams 1976, 44–63; Kautzsch 1910, 377–84; Bauer and Leander 1991, 634–47; Joüon and Muraoka 1993, 482–92. In addition to these, the lexica are also frequently important for the study of these prepositions, especially *DCH* and *HALOT*.
- <sup>4</sup> The particle in its plural construct (אָחֶר) also serves as a substantive, referring to "back parts, rear" (DCH 1:199–200; BDB 30).

(b) Temporal – points to an event that comes chronologically "after" another event. The preposition normally precedes the noun or verb denoting the first event: דָּאָלֶה הָדְּבָרִים הָאָלָה (1 Kgs 21:1), וְיָהִי אָחָר הַדְּבָרִים הָאָלָה, "after these things" (1 Kgs 21:1), אָחָרִי בּאוֹ מֵחֶבְרוֹן, "It happened <u>after</u> these things" (2 Sam 5:13), אָחַרִי הוֹלִידוֹ אָת־ײַמָת he became the father of Seth" (Gen 5:4).

At times, אָחָר אָדָר serves as an adverb of time expressing *manner*, and is best translated by the English adverb "afterward": אָחַר הַלָּך, "<u>afterward</u> she may go" (Gen 24:55), "and <u>afterward</u> she bore a daughter" (Gen 30:21), האָחָרוּ אֶל־פַּרְשָׁה וָאָהַרוּ וָאָדָר וַ גָּאָ הָוָאַחַר בָּאוּ מֹשָׁה וְאָהַרוֹ וַיֹּאַמְרוּ אָל־פַּרְשָׁה, "and <u>afterward</u> Moses and Aaron went and said to Pharaoh" (Exod 5:1).<sup>5</sup>

(c) Metaphorical – denoting a behavior patterned after or according to that of another, or in support of another: היקר קלק לבי, "and he <u>followed</u> [literally: <u>went after</u>] the sins of Jeroboam" (2 Kgs 13:2), יביקר לבי, "my heart has <u>followed</u> [literally: <u>gone after</u>] my eyes" (Job 31:7), heart has <u>followed</u> [literally: <u>gone after</u>] my eyes" (Job 31:7), "and both you and the king who reigns over you will <u>follow</u> [literally: <u>be after</u>] YHWH your God" (1 Sam 12:14).

# אָל־ / אָל 4.1.2

(a) Terminative – marks movement "to" or "into" something, especially when the goal of the movement is reached: כִּי-חָבֹא אֶל־הָאָרֶץ, "when you come <u>into</u> the land" (Deut 17:14), כָּאַרֶרָד אוּרְיָה אָל־בֵּיתוֹ, "Uriah did not go down <u>to</u> his house" (2 Sam 11:10), ילא־יָרָד אָל־הָיָה, "and he [Moses] went out <u>to</u> his brothers" (Exod 2:11), he [Moses] went out <u>to</u> his brothers" (Exod 2:11), גוי לא־יִדְעוּך אָלִיך יָרוּצוּ shall run <u>to</u> you" (Isa 55:5). This may become "on(to)"

<sup>&</sup>lt;sup>5</sup> When occurring in the plural construct form, the adverbial takes the compound אָחֵרי־כֵן (see *HALOT* 1:36; *DCH* 1:197; BDB 30).

or "upon" when the motion is vertical:<sup>6</sup> וְיָפּל יְהוֹשֻׁעַ אֶל־פְּנָיו אֵלֶיך, "and Joshua fell <u>on</u> his face" (Josh 5:14), אָל־הָאָליך אָלֶיך, "may YHWH make his face shine <u>on</u> you" (Num 6:25), "and the hand of YHWH was heavy <u>upon</u> the people of Ashdod" (1 Sam 5:6).

The motion or movement takes on a *directional* connotation when the goal is not reached, sometimes resulting in a translation "toward(s)" or "in the direction of": אָשׁתַחָוָה אָל־הֵיכָל־כָּרְשָׁך, "I will bow down *toward* your holy temple" (Ps 5:8), יוֹפְנוּ אָל־הַמָּרְבָּר your holy temple" (Ps 5:8), ויִפְנוּ אָל־הַמָּרְבָּר Israelites] turned *toward* the wilderness" (Exod 16:10), גאָרָאָה וְהַנֵּה־יָר שְׁלֹתְה אָלִי , "and I looked and a hand was extended *to* me" (Ezek 2:9).

Often in the *terminative* use of אָל־/אָל, as with some of the other uses that follow (see especially the *declarative* and *perceptual*), the preposition also serves to mark a simple *indirect object*:7 וְנָתַן אָלֵיךָ אוֹת, "and he gives [<u>to</u>] you a sign" (Deut 13:2), באָלִי וְאָרָרְכָם לְפָנִיו אָל־עֵשָׁו "bring them <u>to</u> me that I may bless them" (Gen 48:9), gers before him <u>to</u> Esau" (Gen 32:4).

(b) Estimative – expresses interest/advantage or indifference/disadvantage in something, and often requires "for" or "against" in translation: כָּי הָנָנִי אֲלִיכָם, "see now, I am <u>for</u> you" (Ezek 36:9), אָקַבְּצָה אָל־נִי הָמָלֶך אָת־כָּל־יִשְׁרָאָר, "and let me gather all Israel <u>for</u> my lord the king" (2 Sam 3:21), "ILE Sam Jier (Isa 7:6), הַנָּרָקַשְׁנָה אָל־יִשְׁרָאָר (Isa 7:6), הַנָּרָקַשְׁנָה אָל־יִשְׁרָאָר (Isa 7:6), הַנָּרָקַשָׁנָה אָל־יִשְׁרָאָר הָיָר הַמָּלַד (Isa 7:6), הַנָּרָקַשְׁנָה אָל־יִשְׁרָאָר הַפָּל־יִשְׁרָאָר (Isa 7:6), הַיָּרָקָשָׁרָאָר (Isa 7:6), הַרָּקַשְׁנָה אָל־יִשְׁרָאָר (Isa 7:6), הַרָּקַשְׁנָה אָל־יִשְׁרָאָר (Isa 7:6), הַרָּקַשְׁנָה אָל־יִשְׁרָאָר (Isa 7:6), הַרָּקַשְׁנָה אָל־יִשְׁרָאָר (Isa 7:6), הַרָּקַשָּנָה אָל־יִשְׁרָאָר (Isa 7:6), הַרָּקַשָּרָאָר (Isa 7:6), הַרָּקַשְׁנָה אָל־יִשְׁרָאָר (Isa 7:6), הַרָּקַשְׁנָה אָל־יִשְׁרָאָר (Isa 7:6), הַרָּקַשָּנָה אָל־יִשְׁרָאָר (Isa 7:6), הַרָּקָשָׁרָאָר (Isa 7:6), הַרָּקַשָּרָאָר (Isa 7:6), הַרוֹן אַרְ־יִהָוָה אָל־יִשְׁרָאָר (Isa 7:6), הַרָּקָשָׁרָאָר (Isa 7:6), הַרוֹן אַרִייָרָהָשָׁרָאָר (Isa 7:6), הַרָרָשָׁרָאָר (Isa 7:6), הַרוֹן אַרִייָהָנָה אָל־יִשְׁרָאָרָל (Isa 7:6), הַרוֹן אַרַייָהָנָה אָל־יִשָּרָאָר (Isa 7:6), הַרוֹן אַרָּרַיָּשָׁרָאָר (Isa 7:6), הַרוֹן אַרַייָרָהָאָר (Isa 7:6), הַרוֹן אַרַייָרָהָאָר הַרוֹן אַרַייָהָנָה אָל־יִשָּרָאָר (Isa 7:6), הַרוֹן אָרַייָרָאָר (Isa 7:6), הַרוֹן אַרִייָנָה אָל־ישָׁרָאָר הַיָּרָה הַיָּרָאָר הַיָּרָאָר הַיָּרָאָר הַיָּרָאָר הַיָּרָאָר הַיָּרָאָר הַיָּרָה אָל־ייָרָיָרָאָר הַיָּרָאָר הַיָּרָאָר הַיָּרָאָר הַיָּרָה אָרייָרָרָה הַיָּרָי הָיָר הַיָּר הַיָרָין הַיָּר הַיָרָר הַיָרָה הַיָר הַיָר הַיָר הַיָר הַיָר הַיָּר הַיָר הַיָר הַיָר הַיָר הַיָר הַיָר הָיָין הַיָרָי הָרָיָר הַיָּרָה הַיָּר הַיָר הַיָר הַיָין הַרָיָר הָיָרָין הָיָר הַיָר הָיָר הַיָּר הַיָּרָי הָיָר הַיָר הַיַר הַיָר הַיָר הַיָר הַין הַיָר הַיָר הַיָר הַיָר הַיָר הַיָּר הַיָר הַיַין הַיָין הַיָיר הַיָר הַיָר הַיָר הָיָין הַיָר הַיָר הַיָר הָיָר הַיָר הַיָר הַיָין הַיָר הַיָּר הַיָּרָייָר הַיָּרָין הַיָין הַיָי הַיָר הַיָר הָיָר הָיָר

 $<sup>^{6}</sup>$  See *DCH* 1:260–64 for the distinction between horizontal and vertical motion.

<sup>&</sup>lt;sup>7</sup> Or the so-called datival use (see Waltke and O'Connor 1990, 193).

[Jer 1:19]), but occurring with many other verbs as well: וַיָּפְשָׁטוּ אֶל־הַגְרְעָה (Judg 20:37), "and they marauded <u>against</u> Gibeah" המה־לִי וָלָך כִּי־בָאתָ אֵלֵי לְהַלָּחֵם בָּאַרְצִי, "what is there between us that you have come <u>against</u> me to fight against my land" (Judg 11:12).<sup>8</sup>

(c) Declarative – involves proclamation or locution (marking the object/recipient of a verb of speech), and at times with an *estimative* connotation as well. An extension of the *terminative* use of אָל־/אָל , the *declarative* marks the recipient of verbs of speech (אָל־/אָל־, אָל־, אָל־, אָל־, אָל־, אָל־, אָל־, אָל־, פּרוּ עָל־לֵב יְרוּשָׁלִם וְקָרָאוּ אָלֵיה, אָל־הָאָשֶׁה , and he [the serpent] said to the woman" (Gen 3:1), וּאָלִיך אָליָה אָלִיך, אַל־ב יְרוּשָׁלִם וְקָרָא אָלֵיה, אָל־הָאָשָׁה, and proclaim to Jerusalem, and cry to her" (Isa 40:2), "speak tenderly to Jerusalem, and cry to her" (Isa 40:2), "speak tenderly to Jerusalem, and cry to her" (Jonah 3:2), it [Nineveh] the message that I tell [to you" (Jonah 3:2), "and he related it to his father and to his brothers" (Gen 37:10).

(d) Perceptual – marks one's disposition or predilection toward another person or place. As another extension of the *terminative*, the *perceptual* marks the recipient of verbs of perception (דא, האה, etc.), aptitude (דא, עבר, רדע, בין, שמר), or emotion (דע, ידע, בין, שמר), etc.): עגב, רנן), or emotion (דע, שנע, יהנה אָל־עָנְיֵך (Gen 16:11), שַמַע יְהוֶה אָל־עָנְיֵך Abram" (Gen 12:7), לָמָה לָא שָמַרְהָ אָל־אַרְרָם, "Why have you not kept watch <u>over</u> your lord the king" (1 Sam 26:15), לא יְבִינוּ אָל־פָּעָלָת יְהוֶה, "they do not regard [literally: understand <u>about</u>] the works of YHWH" (Ps 28:5), (Ezek 20:9), יו [YHWH] made myself known <u>to</u> them" (Ezek 20:9), לְבָיוּ הְבָשָׁרִי וְהָנָה, "I [YHWH] were the iniquity of his fathers be remembered <u>before</u> YHWH" (Ps 109:14), לָבִי וְבָשָׁרִי וְרָנָוו אָל־דָבָי, "my heart and my flesh sing for

<sup>&</sup>lt;sup>8</sup> For many other examples, see *DCH* 1:267–68, section 6.

joy <u>to</u> the living God" (Ps 84:3), וַהַּעְנֵב עַל־מְאָהֲבֶיהָ אֶל־אֲשׁוּר), "and she [Oholah] lusted after her lovers, <u>after</u> the Assyrians" (Ezek 23:5).

With verbs of writing, giving, or commanding, the *perceptual* use of אֶל'-/אָל', may connote "addressed to" or "intended for": אָל'-וֹאָל דִוּך סַפָּר אָל'־וֹאָל, "and David wrote a letter <u>intended for</u> Joab" (2 Sam 11:14), קלי אָסָר, "and [let] a letter [be] <u>addressed to</u> Asaph" (Neh 2:8).

(e) Addition – denotes circumstances over and above certain others, requiring such translations as "in addition to," or "as well as": אָשָׁה אָל־אֲחֹתָה לֹא חָקָה, "You shall not marry a woman <u>in addition to</u> her sister" (Lev 18:18), הוֹסַפְּתָ חָכְמָה וָמוֹב אָל־הַשְׁמוּעָה אֲשֶׁר שָׁמְעָתִי הוֹסַפּתָ חָכְמָה וָמוֹב אָל־הַשְׁמוּעָה אֲשֶׁר שָׁמְעָתִי her report which I heard" (1 Kgs 10:7).

(*f*) *Spatial* – indicates localization, especially "at," "in," "by," "in the vicinity of," or "alongside":

קַבַר אַבְרָהָם אֶת־שָּׂרָה אִשְׁתּוֹ אֶל־מְעָרַת שָׂרֵה הַמַּכְפֵּלָה

"Abraham buried Sarah his wife <u>in</u> the cave of the field at Machpelah" (Gen 23:19),

כּי אִם־אֶל־הַמָּקוֹם אֲשֶׁר־יִבְחַר יְהוֶה אֶֶלֹהֶיךְ לְשַׁכֵּן שְׁמוֹ שָׁם

"but <u>at</u> the place where YHWH your God chooses to establish his name" (Deut 16:6), הוא־בָא וַיַּעַמֹד אָל־אָדְנָיו, "he went in and stood <u>in the presence of</u> his master" (2 Kgs 5:25).

(g) Specification – clarifies or explains further what immediately precedes, which would otherwise be generally or ambiguously stated. This use of the preposition often requires "about" or "concerning": הַשָּׁמַע אֶת־הַשְׁמַעָה אֶל־הַלָּקַח אֲרוֹן הָאֱלֹהִים וּמֵת חָמִיהָ וְאַישָׁה "and she heard the news <u>concerning</u> the capture of the ark of God and the death of her father-in-law and husband" (1 Sam 4:19), לְכֵן כָּה־אָמָר יְהוָה אֶל־מֶלֶך אַשׁוּר (גוּפּר, thus says YHWH <u>concerning</u> the king of Assyria" (Isa 37:33), כי תוֹרִם אָל־הַדֶּרֶך הַמוֹרָה אֲשֶׁר יֵלְכוּ־בָה , "indeed, teach them <u>about</u> the good way in which they should walk" (2 Chr 6:27).

At times, the use of אָל־/אָל, for specification takes on a *causal* connotation: הָהְאָבֵל שְׁמוּאֵל אָל־שָׁאוּל, "Samuel grieved <u>concerning [because of</u>] Saul" (1 Sam 15:35), אָל־הַכַּעַס אֲשֶׁר הַכְעַסָתָ which you have provoked" (1 Kgs 21:22).

# אַצָל 4.13

*Spatial* – Expresses concepts of nearness or close proximity, usually requiring "beside," "with," or "near": ישׁיָל דָּגוֹן, "and he set it *beside* Dagon" (1 Sam 5:2), מוֹ מַעָּלִדְיָה אָצֶל הַפְּרוֹת עַל־שָׁפָת הַיָּאֹר, "and they stood <u>near</u> the other cows on the bank of the Nile" (Gen 41:3).

### אָת־ / אֶת 4.1.4

<sup>9</sup> This preposition is presumably based on the root אתת since suffixed forms double the -t (אתל), and note the Akkadian preposition *itti*). Thus, it should not be confused with the definite direct object marker, which may be based on the root אות וווא אות in light of its forms with suffixes (אוֹת).

(b) Possession – expressing ownership:

הְנֵה־הַכֶּסֶף אָתִי אֲנִי לְקַחְתִיו

"behold, the silver is <u>with</u> me, I took it" (Judg 17:2), קה אָקָנוּ, "What do we have [lit: What (is) <u>with us</u>]" (1 Sam 9:7), הַנְּבִיא אֲשֶׁר־אָתוֹ חֲלוֹם יְסַפּר חֲלוֹם, "The prophet who <u>has a dream</u> (literally: the prophet, to whom, <u>a dream is with him</u>), may tell the dream" (Jer 23:28).

(c) Complement – with verbs of speaking, dealing, or making: וְיָדְבֵּר אָתָם קָשׁוֹת, "and he spoke harshly <u>with</u> them" (Gen 42:7), וְיִדְשְׁתָם כִּי־אֲנִי יְהוָה בַּשֲשׁוֹתִי אָתְּכֶם, "Then you will know that I am Yhwh, when I have dealt <u>with you</u>" (Ezek 20:44).

(d) Spatial – denotes a sense of nearness or proximity: הָצְשְׁנוּים אֲשֶׁר אֶת־קָרָשׁ, "in Zaanannim, which is <u>near</u> Kedesh" (Judg 4:11 *Qere*), הָצָשְׁנוּים אָתִי יְהָוָה הְנֵה מָקוֹם אָתִי (Exod 33:21), "Yhwh said, 'See, there is a place <u>beside</u> me" (Exod 33:21), "Yhwh said, 'See, there is a place <u>beside</u> me" (Exod 33:21), "In the chariot, at the ascent of Gur, which is <u>by Ibleam</u>" (2 Kgs 9:27).

### 4.1.5 P

(*a*) *Spatial* – This preposition<sup>10</sup> indicates localization, especially "in," "at" or "on" a place:

לַעֲשׂות זְבָחִים בְּבֵית־יְהוֶה בִּירוּשָׁלַם

"to offer sacrifices <u>in</u> the house of YHWH <u>at</u> Jerusalem" (1 Kgs 12:27), יאָקֹב זָבַח בָּהָר, "Then Jacob offered a sacrifice <u>on</u> the mountain" (Gen 31:54), רְכַנְעֲנִי אָז בָּאָרֶץ, "the Canaanites were then <u>in</u> the land" (Gen 12:6).

<sup>10</sup> Approximately 58% of the uses of ∃ in the Bible are *spatial*, while only about 16% are temporal (Jenni 1992, 69). For a complete list of uses and numerous examples, see *HALOT* 1:104–05; *DCH* 2:82–86.

With certain verbs of motion or movement, I denotes the route "through": וַיָּעֵבֹר אָבָרָם בָּאָרץ, "and Abram passed through the land" (Gen 12:6), קום התהלך בארץ, "rise up, walk through the land" (Gen 13:17). At other times, refers to the movement "into" or rest "in" a place ("terminative"): הְבָאִים אָחֵרִיהֶם בַּיֵם, "those who went after them <u>into</u> the sea" (Exod 14:28), וְשָׁלֵה אָת־הַשָּׁעִיר בַמִּדְבַר, "and he shall send forth the goat *into* the wilderness" (Lev 16:22), רְבֵי אָדָם בְּצֵל כְנָפֵיך יְחֵסִיוּן, "all people may take refuge *in* the shadow of your wings" (Ps 36:8). Included in this category of localization are those times when ⊇ connotes "in the domain of," that is, inclusion "among" or "within" a group: והפיץ יהוה אתכם בעמים ונשארתם מתי מספר בגוים, "YHWH will scatter you *among* the peoples and you shall be left few in number among the nations" (Deut 4:27), אוֹדָךָ בָעַמִים אָדֹנָי, "I will give thanks to you, O Lord, among the peoples" (Ps 57:10). At times, I appears to connote movement or action away "from" a place, especially in poetry: אָרני מַעון אַתָּה הַיִית לַנוּ בִּרר וַדר, "Lord, you have been our dwelling place <u>from</u> all generations" (Ps 90:1), אין ישוּשָׁתָה לו באלהים, "he has no salvation <u>from</u> God" (Ps 3:3), דַיַן אַלְמָנוֹת אֵלהִים בַּמְעוֹן קָדְשׁוֹ, "a protector of widows is God from his sanctuary" (Ps 68:6), הַלִיכוֹת אָלִי בַקֹרָשׁ, "the processions of my God, my King, from the sanctuary" (Ps 68:25).<sup>11</sup>

(b) Temporal – expresses a moment or point of time when an action takes place: בְּיוֹם אֶרֶץ וְשֶׁמְיִם, אֶרֶץ וְשֶׁמְיִם, "<u>in</u> the day when YHWH God made earth and the heavens" (Gen 2:4), בִּיוֹם בְּרֹא אֶלֹהִים אָרֶם, "<u>in</u> the day when God created humankind" (Gen 5:1), בִּימִי קְצִיר־חָשִׁים, "<u>in</u> the days of wheat harvest" (Gen 30:14), (הָבָא יַעֵקֹב מִן־הַשָּׁדֶה בָּעֶרֶב, (for source), (הָבָא יַעֵקֹב מִן־הַשָּׁדֶה בָּעֶרֶב,

<sup>&</sup>lt;sup>11</sup> The so-called double meaning of ⊐ ("in" and "from") has been heavily debated in light of Ugaritic evidence, which seems to support "from" in these poetic contexts, especially in the Psalms. For bibliography, see *DCH* 2:602. See also on ? in 4.1.10.

"and Jacob came from the field <u>in</u> the evening" (Gen 30:16), בְּכָל־עֵת אֹהֵב הָרָעָ, "a friend loves <u>at</u> all times" (Prov 17:17). When used with the infinitive construct, דְ frequently connotes "when," or "whenever": בְּהַבְּרָאָם, "<u>when</u> they [the heavens and the earth] were created" (Gen 2:4), בְּהַצָּרְיִם מְמַצְרִיִם, "<u>when</u> they [the Israelites] came out of Egypt" (Josh 5:4); and see the famous examples in the verses following the Shema: קוּבְרָתָבְרָקָוּבְרָקָנְמָך וּבְלֶכְתָּדְ בְדֶרֶקוּמָרָ you shall speak of them [God's words] <u>when</u> you sit at home and <u>when</u> you are on the road, <u>when</u> you lie down and <u>when</u> you rise" (Deut 6:7).

(c) Instrumental – refers to an inanimate object used to perform a certain action: בְּשֶׁרֶם 'יֵכוּ, "<u>with</u> a rod they will smite [him]" (Mic 4:14), בְשָׁרֶם בֶּחֶרֶם, "and I will kill you <u>with</u> the sword" (Exod 22:23). Similar to this use is the בְ of *material*, which indicates the material with which something is made or formed: וְצַׁרֶך בַּזָּהֶב יְרַקְּעָנוּ, "and a goldsmith overspreads it <u>with</u> gold" (Isa 40:19).

(d) Adversative – indicates a relationship of disadvantage: יָדוֹ בַכּל וְיֵד כַּל בּוֹ , "his hand shall be <u>against</u> everyone and everyone's hand <u>against</u> him" (Gen 16:12). In some contexts, the <u>adversative</u> connotes "in spite of" in translation: texts, the <u>adversative</u> connotes "in spite of" in translation: לא־יַאֲמִינוּ בִי בְּכֹל הָאֹתוֹת אֲשֶׁר עָשִׁיתִי בְּקְרְבּוֹ , "they do not believe in me <u>in spite of</u> all the signs that I have done among them" (Num 14:11).

(e) *Specification* – clarifies or explains further an immediately preceding verbal action, which would otherwise be generally or ambiguously stated:

וְשָׂמַחְתָּ בְכָל־הַמּוֹב אֲשֶׁר נָתַן־לְדָ יְהוָה

"and you shall rejoice <u>in</u> all the good which YHWH has given to you" (Deut 26:11). The ה of *specification* may also particularize the parts of which the general whole consists: וַיָּגַע כָּל־בָּשֶׁר הָרֹמֵשׁ עַל הָאָרֵץ בָּעוֹף וּבַבְּהֵמָה וּבַחָיָה וּבְכַל־הַשֶּׁרֵץ "and all flesh died that moved on the earth – birds, cattle, wild animals, all swarming creatures" (Gen 7:21),

וַיְּמָרְרוּ אֶת־חַיֵּיהֶם בַּעֲבֹדָה קָשָׁה בִּחֹמֶר וּבִלְבַנִים וּבְכָל־עֲבֹדָה בַּשָּׁדֵה

"and they made their lives bitter with hard service  $-\underline{in}$  mortar and  $\underline{in}$  brick and  $\underline{in}$  all field service" (Exod 1:14).

(f) Causal – identifies a cause or reason: אָאישׁ בְּחֶשְׁאוֹ יוּמָתוּ "each person shall be put to death <u>because of</u> his own sin" (Deut 24:16),

הֵן הֵנָּה הֶיוּ לִבְנֵי יִשְׂרָאֵל בִּדְבַר בִּלְעָם לִמְסָר־מַעַל בַּיהוָה

"behold, these women caused the Israelites, <u>because</u> of the word of Balaam, to act treacherously against YHWH" (Num 31:16),

וּרְאִיתֵם אֶת־כְּבוֹד יְהוָה בְּשָׁמְעוֹ אֶת־תְּלָנֹתֵיכֶם עַל־יְהוָה

"and you shall see the glory of YHWH, <u>because</u> he has heard your complaining against YHWH" (Exod 16:7), בְּעָזְרְכֶם אֶת־מִצוֹת 'הוָה, <u>because</u> you have forsaken the commands of YHWH" (1 Kgs 18:18). The preposition may be combined with אָשֶׁר to form a causal conjunction: אֲשֶׁר אַת־אָשֶׁר (Gen 39:9), לַבָּאַשֶׁר '<u>because</u> YHWH was with him" (Gen 39:23).<sup>12</sup>

(g) Accompaniment – denotes circumstances occurring "together with": יַצָאוּ בִּרְכָשׁ נָּדוֹל, "they shall come out <u>with</u> great possessions" (Gen 15:14), יַצָאוּ בָּרְכָשׁ, "stretch out your hand <u>with</u> your staff" (Exod 8:1), בָּעָבִיך בְּנִים, "<u>with</u> pain you shall bring forth children" (Gen 3:16), "<u>with</u> pain you shall bring forth children" (Gen 3:16), נוֹיֵצָא אֶדוֹם לִקְרָאתוֹ בְּעָם כָּבִד them <u>with</u> a large force" (Num 20:20). (h) Essence<sup>13</sup> – marks identity of a noun in the context, occasionally with a predicate, and connoting "(having the same nature) as" or "(consisting) of": נְעֵשֶׁה אָדָם בְּצַלְמֵנוּ (Gen 1:26), in the same nature] as our image" (Gen 1:26), מוֹ ה אָלהים עָשָׁה אָדָם בָּצַלְמֵנוּ, "and I appeared to Abraham, Isaac, and Jacob as God Almighty" (Exod 6:3), הְנָר שָׁם בָּמָתִי מְשָׁם, "behold, Lord YHWH comes as a mighty one" (Isa 40:10), הַנֶּעָשָׁם הָאָר מָשָׁם בַּמָתִי מָשָׁם גָּרָי הַשָּרָי הַשָּרָם הַצָּרָי הַשָּרָם בָּצַרְמָנוּ אָרָם בָּמָת מָשָׁם אָרָם הַצַרְהָם אָל־יִצָּקָב בָּאַל שָׁדָי, "and I appeared to Abraham, Isaac, and Jacob as God Almighty" (Exod 6:3), הַנָּה אָרָיָי יְהוָה בְּחָזָכ יָבוּאַ מָרַי הַשָּרָי הַשָּרָם הַאָרָי הַשָּרָי הַשָּרָם מָשָּר אָרָרָהָם אָל־יִצָּקָב בָּאַל שָׁדִי (Isa 40:10), הַנָּה מָסַר מָשָׁם בּמָתֵי מָשָׁם הַאָּרָי הַשָּרָי יָבוּה בַּחָזָכ יָבוּאַר מָשָר הַשָּרָי יָבוּה הַבּחַזָר (Deut 26:5).

(*i*) *Manner* – describes the way in which an action or situation is performed or the attendant circumstances of the action. In such cases, הברים בקול נָרוֹל , "weeping <u>aloud</u> [literally: <u>with</u> a great voice]" (Ezra 3:12), "weeping <u>aloud</u> [literally: <u>with</u> a great yoice]" (Ezra 3:12), "go back now, and go <u>peaceably</u> [literally: <u>in</u> peace]" (1 Sam 29:7).

(j) Price – indicating the cost or price of something, and at times connoting "at the risk of" or "in exchange for": בְּכָסֶךְ מְלֵא יִהְעָנָה לִי "in exchange for": בְּכָסֶךְ מְלֵא יִהְעָנָה לָן ווּם silver [i.e., full price]" (Gen 23:9), at the cost of full silver [i.e., full price]" (Gen 23:9), "I will serve you seven years <u>as the price of</u> your younger daughter Rachel" (Gen 29:18), וַיָּמִירוּ אֶת־כְּבוֹדָם בְּתַבְנִית שׁוֹר אֹכֵל עֵשֶׁב (Gen 29:18), וַיָּמִירוּ אֶת־כְּבוֹדָם בְּתַבְנִית שׁוֹר אֹכֵל עֵשֶׂב (Gen 29:18), ביָמָירוּ אָרַכּנוֹדָם בְּתַבְנִית שׁוֹר אֹכֵל עֵשֶׁב (Sen 29:18), היא אָרָבוֹדָם בְּתַבְנִית שׁוֹר אֹכֵל מֵשֶׁב (Gen 29:18), ביָמָשָׁב (Sen 29:18), היא געשׁב היא געשׁב אָרָדָם בָּתַבְנִית שׁוֹר אַכָל מַשָּׁב (Ps 106:20).

### בין 4.1.6

Denotes the interval "between" two points, usually *spatial* and occasionally *temporal*. In 80 percent of its

<sup>&</sup>lt;sup>13</sup> Known by older grammarians as *beth essentiae* (Kautzsch 1910, 379). See Joüon and Muraoka 1993, 486–87, and *DCH* 2:84–85.

occurrences, בָּין is paired with itself ("...וֹבֵין...וֹבָין") in order to denote the interval between two points or two parties, in which case the second בֵין is not repeated in translation:<sup>14</sup> הַין בִית־אַל וּבֵין הָשָי, "*between* Bethel and Ai" (Gen 13:3), וִיְהָי שָׁלוֹם בֵּין יִשָּׁרָאֵל וּבֵין הָאָאָרי, "and there was peace between Israel and the Amorites" (1 Sam 7:14), בְרִית בֵּינִי וְבֵינֵך (let there be] a treaty *between* me and you" (1 Kgs 15:19), יַנָּר סִינַי הָהָר בְּיָנוֹ וּבֵין בְּנֵי יִשָּׂר אָשֶׁר בָּהַר סִינַי, "the laws, which YHWH established between himself and the Israelites on Mount Sinai" (Lev 26:46). The second בין of the idiom may be repeated, in which case the idea appears to be "between A (on the one hand) and B and C (on the other)": וַיָּכָרת יָהוֹיַדַע אָת־הַבָּרית בֵּין יְהוָה וּבֵין הַמֵּלֶך וּבֵין הַעָם, "Then Jehoiada made a covenant between YHWH on the one hand, and the king and people on the other" (2 Kgs 11:17).<sup>15</sup>

In other occurrences, בִין is used only once, and stands before the plural or dual of a noun or with plural suffixes to refer to a group, in which case the preposition may denote "among": אֶלְהֵי אָרְרָהֶם וָאלֹהֵי נְחוֹר יָשָׁפְּטוּ בִינֵינוּ, "the God of Abraham and the God of Nahor judge <u>between</u> us" (Gen 31:53), רִיב בֵּין אָרָיִים בֵין אָרָים men" (Deut 25:1), בִין אָרָיִים בֵין אָרָים can one who sows discord <u>among</u> brothers" (Prov 6:19).<sup>16</sup>

### **4.1.7** בַּעַר / בַּעַר

(a) Spatial – indicates localization, especially "behind" something: הָאָרֶץ בְעֵרִי לְעוֹלָם, "[I went down to] the earth whose bars [are closed] <u>behind</u> me forever" (Jonah 2:7 [Eng. 2:6]). With the verb "close, shut" (סגר), the concept is to close off or shut up inside: רָיָהָוָה בַּעֲרוֹ, "and YHWH

- <sup>15</sup> For more on this, and other variations of the idiom, see *DCH* 2:146.
- <sup>16</sup> On such uses as "inclusive," and on the distinction between *inclusive* and *exclusive* uses of בִין, see Waltke and O'Connor 1990 (199–201).

<sup>&</sup>lt;sup>14</sup> For the statistics, see Waltke and O'Connor 1990 (199–201) and *DCH* 2:146–49.

closed it <u>behind</u> him" (Gen 7:16), וּבְאַרַק הַדֶּלֶת בַּעֲרַף, "and go in and shut the door <u>behind</u> you" (2 Kgs 4:4), "קנָר יְהוָה בְּעַר רַחְמָה, "Yhwh <u>closed</u> her womb" (1 Sam 1:6).

With verbs of motion (בוא (בוא, בוא, etc.), the preposition often denotes a sense of movement "through" an object: בָּעַר הַחַלּוֹנִים יָבֹאוּ כַּגַּנָ , (<u>through</u> the windows they enter like a thief" (Joel 2:9), וָהוֹרְדֵם בַּחֶבֶל בְּעַר הַחַלּוֹן "then she let them down by a rope <u>through</u> the window" (Josh 2:15).

Used metaphorically, the spatial sense of "around" or "round about" can take on an *advantage* idea related to the next usage:

הַלא־אַהָּה שַׂכְתָּ בַעֲדוֹ וּבְעַד־בֵּיתוֹ וּבְעַד כָּל־אֲשֶׁר־לוֹ מִסָּבִיב

"Have you not put a fence <u>around</u> him, his house, and all that he has, on every side?" (Job 1:10 *Qere*), אָאָהָה יְהוֶה מָגון בַּעֵרי, "You, O YHWH, are a shield <u>around</u> me" (Ps 3:4).

(b) Advantage – indicates a relationship of interest or advantage, often requiring "for" or "on behalf of": הַפַּלֵל בַּעַרְדָ וָחְיֵה ' "he will pray <u>for</u> you and you will live" (Gen 20:7), יְהָפַּלֶר בְּעַר הָעָם (Exod 32:30), "perhaps I can make atonement <u>for</u> your sin" (Exod 32:30), and make atonement <u>for</u> yourself and <u>for</u> the people" (Lev 9:7),

לְכוּ דְרְשׁוּ אֶת־יְהוֶה בַּעֲדִי וּבְעַד־הָעָם וּבְעַד כָּל־יְהוּדָה

"Go, inquire of YHWH <u>on my behalf</u>, and <u>on behalf of</u> the people and all Judah" (2 Kgs 22:13).

### יַעַן 4.1.8 יַעַן

*Causal* – More often used as a conjunction, יַשַן denotes a causal sense: יַשַן כָּל־תּוֹשְׁבֹתִיך: "<u>because</u> of all your abominations" (Ezek 5:9), יַשַן מָאַסְדָ מְמֶּלֶך מַמֶּלֶך "*Because* you rejected the word of YHWH, he has rejected you as king" (1 Sam 15:23). For more specific constructions in which the particle appears, refer to "Causal Clauses" (section 5.2.5).

# **4.1.9** ⊃

⊃, unlike other prepositions, has no spatial use, and has a temporal use only with the infinitive construct. Instead, it shows comparison or correspondence relationships.

(a) Agreement – The preposition often denotes a sense of agreement in quantity or measure: יֹסֵף עֵלִיכֶם כָּכֶם אָלֶף פָּעָמִים, "[may YHWH] make you a thousand times <u>as</u> many as you are" (Deut 1:11). This type of quantitative agreement can be *approximate* as well: רָשָׁרָ שָׁנִים, "they lived there <u>about</u> ten years" (Ruth 1:4), "שָׁרָאָר חַלוּצִי הַצָּבָא , "<u>about</u> forty thousand armed for war" (Josh 4:13).

Along with quantitative agreement, ⊃ also expresses agreement in *kind* or *quality* as well: וְהְיֵיתֶם כָּאלֹהִים, "you will be *like* God" (Gen 3:5),

הַלוֹא יִדַעְהֶם כִּי־נַחֵשׁ יְנַחֵשׁ אִישׁ אֲשֶׁר כָּמֹנִי

"Do you not know that a man <u>like me</u> practices divination?" (Gen 44:15), וּבְעֶרֶב יְהָיֶה עַלֹ־הַמִּשְׁכָן כְּמֵרְאָה־אַשׁ עַר־בֹּקָר, "In the evening, [the cloud] was over the tabernacle, <u>like the appearance of fire</u>, until morning" (Num 9:15).

(b) Correspondence – a comparison that establishes an equivalence between the things that are compared, and often clarifying the identity of what is compared: הוא כָאָישׁ אָכָת, "he <u>is truly</u> a man of truth" (Neh 7:2). This is especially common with the כָּאָרָיָ כָּאָרָיָן כָּאָרָיָן בָּאָרָיָן גַאָרָאָשׁ, "And it shall be, <u>as with</u> the people, <u>so with</u> the priest; <u>as with</u> the slave, <u>so with</u> his master" (Isa 24:2).

(c) Temporal – with the infinitive construct, expresses "when" or "as soon as": וְהָיָה כָּרְאוֹתוֹ כִּי־אֵין הַנַּעַר וְמֵת, "<u>As soon as he sees</u> that the boy is not with us, he will die" (Gen 44:31),

וַיְצו אַבְשָׁלוֹם אֶת־נְעָרָיו לֵאמֹר רְאוּ נָא כְּמוֹב לֵב־אַמְנוֹן בַּיַיִן וְאָמַרְתִּי אֲלֵיכֶם הַפּוּ אֶת־אַמְנוֹן וַהַמָּשֵם אֹתוֹ

"Absalom commanded his servants, saying 'Watch; <u>when</u> <u>Amnon's heart is merry</u> with wine, and I say to you "Strike Amnon," then put him to death'" (2 Sam 13:28).

### **4.1.10** <sup>7</sup>

(a) *Spatial* – most often, indicates *direction* "to" or "toward" the object of the preposition:

הַבוֹא אָנִי וְאִמְד וְאַחֶיך לְהִשְׁתַוֹת לְד אָרְצָה

"Will we indeed come, I and your mother and your brothers, to bow down <u>to you</u>, to the ground?" (Gen 37:10), קריעוּ לֵאלהִים בְּקוֹל רְנָה, "Shout <u>to</u> God, with a voice of joy" (Ps 47:2 [Eng. 47:1]),

With verbs of motion, spatial ל can take on a *terminative* sense, expressing movement wherein the goal is reached: הַיּוֹם בָּא לְעִיר, "he came <u>to</u> the city today" (1 Sam 9:12), הַיָּוֹם בָא לְעִיר], "and God sent a messenger <u>to</u> Jerusalem" (1 Chr 21:15). Occasionally, the *spatial* has an ablative function, indicating movement "away from" the object: הַלְעָרֵם בְּתָמִים הָלְצָרִם בְּתָמִים, "He does not withhold goodness <u>from</u> those who walk in uprightness" (Ps 84:12), וְלִיהוָה אֲרֹנִי לְמֶוֶת הוֹצָאוֹת, "Escape <u>from</u> death belongs to YHWH, the lord" (Ps 68:21 [Eng. 68:20]).

(b) Locative – locates the object of the preposition "in" or "at" a certain point: וְהֵשֶׁב לִימִינוֹ, "she sat <u>at</u> his right" (1 Kgs 2:19), לְפָתַח חַשָּׁאת רֹבֵץ, "Sin is crouching <u>at</u> the door" (Gen 4:7), וְיַשֵּׁלוּ וַיָּתֵרוּ אֶת־הָאָרֶץ מִמִּדְבַּר־צָן עַד־רְחֹב לְבֹא חֵמָת, "They went up and spied out the land from the wilderness of Zin to Reho, <u>at</u> Lebo Hamath" (Num 13:21).

(c) Temporal – similar to the *locative* use, temporal ל locates the object of the preposition "in," "at," or "during" a certain period of time: וְתָּבֹא אֵלִיו הֵיּוְנָה לְעֵת עֶרֶב, "the dove came to him <u>at</u> the time of evening" (Gen 8:11), dove came to him <u>at</u> the time of evening" (Gen 8:11), "fraction (Gen 8:11), "Bring your sacrifices <u>in</u> the morning" (Amos 4:4). Temporal ל also functions to indicate movement toward a moment in time, or the duration of an action/situation "until" a moment in time: בָּלֹיָמָיָך לְעוֹלָם "all your days <u>to eternity</u>" (Deut 23:7 [Eng 23:6]), "all your days <u>to eternity</u>" (Deut 23:7 [Eng 23:6]), ", "None of the flesh which you slaugher in the evening of the first day shall remain overnight <u>until</u> the morning" (Deut 16:4).

 (d) Purpose – frequently with the infinitive construct, in order to show the aim or goal of another verb: הַפָּאוֹר הַנְּרֹל לְמֵמְשֶׁלֶת הַיּוֹם, "the greater light <u>to rule</u> the day" (Gen 1:16), הַשָּׁאוֹר הַנְּרֹל לְמֵמְשֶׁלֶת הַיָּוֹם, (Gen 1:16), ויוֹצֵא מֹשֶׁה אֶת־הָעָם לְקָרָאת הָאֱלֹהִים מֵן־הַפַּחֲנֶה, "then Moses brought the people out from the camp <u>to meet</u> God" (Exod 19:17),

```
אַתָּה יָדַעְתָּ אֶת־דָּוִד אָבִי כִּי לֹא יָכֹל לִבְנוֹת בַּיִת לְשֵׁם יְהוָה אֶלֹהָיו
```

"You know that David my father was unable *to build* a house for the name of Үнwн his god" (1 Kgs 5:17 [Eng 5:3]).

Related to this use is the *causal*, where ל shows the motive behind an action: אִישׁ הָרַגְהִי לְפָצְעִי, "I have killed a man *for wounding me*" (Gen 4:23),

לא תְתְגֹּדְדוּ וְלֹא־תָשִׁימוּ קָרְחָה בֵּין עֵינֵיכֶם לָמֵת

"You shall not cut yourselves or shave your foreheads *because* of the dead" (Deut 14:1).

(e) Quasi datival –  $\stackrel{!}{>}$  has several datival uses. One of the more common indicates the indirect object of verbs of

giving, speaking, listening, or sending: אַקן־לו, "you will give <u>to him</u>" (Deut 15:14),

אָאֶלֶף לַמַּשֶּׁה אֶלֶף לַמַּשֶּׁה לְכֹל מַמּוֹת יִשְּׁרָאֵל תִּשְׁלְחוּ לַצְּבָא

"A thousand from each tribe of all the tribes of Israel you shall send *to the war*" (Num 31:4),

חַלא הוּא אָמַר־לִי אֲחֹתִי הָוא

"Did he himself not say *to me*, 'She is my sister'?" (Gen 20:5).

(e.1) Interest/advantage-Sometimes called the *ethical da*tive, לי marks the person or object for whom an action is directed or intended: אַל־תְּבְכּוּ לְמָת, "Do not weep <u>for</u> the dead" (Jer 22:10), הַמוֹב לְךָ כִּי־תַעֲשׁׁק, "Is it right <u>for you</u> to oppress?" (Job 10:3),

שִּׁימָה בְּפִיהֶם לְמַעַן תּהוֶה־לִּי הַשִּׁירָה הַוֹּאת לְעֵד בִּבְנֵי יִשְׂרָאֵל

"Put it in their mouths, so that this song will be a witness *for me* against the people of Israel" (Deut 31:19).

ל sometimes expresses the exact opposite, indicating that an action or situation is directed "against" someone or something: גָם־אָנֹכִי אוֹתְך מַחַטוֹ-לי, "I also kept you from sinning <u>against</u> me" (Gen 20:6).

(e.2) Product – with verbs of making, indicates a thing that is made, or a person who is altered, either in status or in form: אָאֶעָשָׁךְ לְגוֹי בָּרוֹל "I will make you [*into*] a great nation" (Gen 12:2), "I will make you [*into*] a great nation" (Gen 12:2), "I will make you [*into*] a great nation" (Gen 12:2), "And YHWH God made the rib which he took from Adam *into a woman*" (Gen 2:22), יָצָא מַמְלְמוֹ לְשׁהַם אַרְצַךְ לְשָׁמָה "He has gone out from his place to make your land *into a desolation*" (Jer 4:7).

(f) Possession – shows that the object of the preposition is in possession of something: אָיָהוּא לִירהוּא, "your silver and your gold are <u>mine</u> [literally: <u>to/for me</u>]" (1 Kgs 20:3), difference (1 Kgs 14:11), <u>Jeroboam</u> who dies in the city, the dogs will eat" (1 Kgs 14:11).

The *possessive* use of 2 is quite frequent with the particle of existence, to show possession, or with the particle of nonexistence, to show lack of possession. See sections 4.4.1 and 4.4.2.

(g) Genitival – nouns in a genitival relationship formed by the construct chain necessarily agree in definiteness. BH forms genitive relationships between nouns that do not agree in definiteness by יְרָשֶׁי בֵּית הַלְחְמִי : לְשָׁשִי בֵּית "Behold, I have seen <u>a son of Jesse</u>, the Bethlehemite" (1 Sam 16:18), וְאָהְכֶם יְהֵיוֹ אֵישׁ אֵישׁ לְמֵשֶׁה (Num 1:4).

(*h*) Specification – calls attention to the object of the preposition: וּלְיִשָׁמְעָתִיךָ, "With <u>regard</u> to Ishmael, I have heard you ...." (Gen 17:20),

שָׁלֹף חַרְבָּד וּמוֹתְתֵנִי פָּן־יֹאמָרוּ לִי אָשָׁה הֲרָגָתְהוּ

"Draw your sword and kill me, so that it will not be said <u>of me</u>, 'A woman killed him'" (Judg 9:54), לאיַטָּג כָּלְמוֹת, "If they do not speak out <u>concerning these things</u>, reproaches will not be turned back" (Mic 2:6).

(i) Normative – often translated "according to," classifies the object of the preposition, sometimes dividing a larger whole into parts: הְתְיַצְבוּ לְפֵנֵי יְהָוֶה לְשֶׁבְטֵיכֶם וּלְאַלְפֵיכֶם וּלְאַלְפֵיכָם "present yourselves before YHWH <u>according</u> to your tribes and <u>according</u> to your thousands" (1 Sam 10:19), "fruit trees bearing fruit of its kind" (Gen 1:11), בְּנֵי מְרָרִי לְמִשְׁפְּהֹתָם לְבֵית־אֲבֹתָם תִפְּלְד אֹתָם, "As for the Merarites, you shall number them <u>by their families</u> and <u>by their fathers's households</u>" (Num 4:29).

(j) Manner – expresses action that is in accordance with a certain standard or principles, which is denoted by

the preposition:

יַצֵּב גְּבֶלֹת עַמִּים לְמִסְפַּר בְּנֵי יִשְׂרָאֵל

"He fixed the boundaries of the people <u>according to</u> the number of the people of Israel" (Deut 32:8), היא־לְמָרְאָה שֵׁינָיו יוֹכְיחַ, "He will not judge <u>by</u> what his eyes will see, and he will not make a decision <u>by</u> what his ears hear" (Isa 11:3).

(k) Estimative – expresses the opinion or perception held by the object of the preposition: (cf. DCH 4:484a): אָרָהֶכִי הַיְהוּדִי מְשָׁנָה לָמֶּלֶךְ אֲחַשְׁוּרוֹשׁ וְנָדוֹל לְיָהוּדִים, 22:24), כִי מְרְהֵכִי הַיְהוּדִי מְשָׁנָה לָמֶלֶךְ אֲחַשְׁוּרוֹשׁ וְנָדוֹל לְיָהוּדִים, "For Mordecai the Jew was second to the King Ahasuerus and <u>great among the Jews</u>" (Esth 10:3). Where a divine name is the object of an *estimative* א לווים, the *superlative* is intended, great city" or "Nineveh was a great city <u>in the eyes of</u> God" (Jonah 3:3).

(*l*) Agent – With passive verbs, לְ often indicates the one who performs the action: בָּרוּךְ אַבְרָם לְאֵל שֶׁלְיוֹן לְנֵה שָׁמַיִם וָאָרֶץ "Blessed be Abram, <u>by God</u> most high, maker of heaven and earth" (Gen 14:19), וְנָבְחַר מֶוֶת מֵחַיִים לְכֹל הַשְׁאַרִית "Death will be chosen, rather than life, <u>by all the remnant</u>" (Jer 8:3).

(*m*) *Reflexive* – The לְ preposition also functions *reflexively*, where the object of the preposition is always the same as the subject of the verb. This use is especially common with verbs of motion or with imperatives. לְבָרְיָלָךְ אֶלֹ-לְבָן אָחִי חָרָנָה (Gen 12:1), קוּם בְּרַח־לְךָ אֶלֹ-לְבָן אָחִי חָרָנָה, (*Arise*, and *flee* to Laban, my brother; to Haran" (Gen 27:43), ויֹאמֶר יְהוָה אֶלֹ-מֹשֶׁה כְּתָב־לְךָ אֶת־הַדְּבָרִים הָאֵלָ

לְמַעַן 4.1.11

(*a*) *Purpose* – Paired with an infinitive form, the preposition functions to show *purpose:* 

וְיֵהוּא עָשָׂה בְעָקְבָּה לְמַעַן הַאָּבִיד אֶת־עֹבְדֵי הַבָּעַל

"But Jehu did it in cunning, *in order to destroy* the servants of Baal" (2 Kgs 10:19),

אָמְרִי־נָא אֲחֹתִי אָתְ לְמַעַן יִיטַב־לִי בַעֲבוּרֵך

"Please say that you are my sister, *so that* it will go well for me because of you" (Gen 12:13).

(b) Causal – denotes the cause of an action or situation: וַיְּהְעַבֵּר יְהוָה בִּי לְמַעַוְכֶם, "and Yhwh was angry with me <u>because</u> of you" (Deut 3:26),

הַשֵּׁבֶט הָאֶחָד יִהְיֶה־לּוֹ לְמַעַן עַבְדִי דָוִד וּלְמַעַן יְרוּשָׁלַח

"He will have one tribe, *because* of my servant David and because of Jerusalem" (1 Kgs 11:32).

לְפַנֵי 4.1.12

This preposition is a compound of the simple preposition  $\varsigma$  prefixed to the substantive  $\mathfrak{P}$  in the construct state.

(a) Locative – The basic locative sense of the preposition points to what is "before," or "in front of": ואַרְרָהָם עוֹדֶנוּ עֹמֵר לְפְנֵי יְהָוָה, "and Abraham was still standing <u>before</u> YHWH" (Gen 18:22),

וְהַנַּחְתָם בְּאֹהֶל מוֹעֵר לִפְנֵי הָעֵרוּת אֲשֶׁר אָוְעֵר לָכֶם שֶׁמָּה

"Deposit them in the tent of meeting *in front of* the testimony, where I meet with you" (Num 17:19 [Eng 17:4]).

Though translated "before," the particle can place emphasis not on the location of its object but on the fact that the object has been placed at one's disposal, or possession: הְגַּה־רִבְקָה לְפָּגֶיךְ קָח וָלֵךְ, "Here is Rebekah <u>before</u> you; take her and go" (Gen 24:51), אָהָגָר תְשָׁבוּ וְהָאָרֶץ תִהְהֶה לִפְּנֵיכֶם, "You shall dwell with us, and the land shall be open <u>before</u> you" (Gen 34:10).

(b) Temporal – points to an action that occurred in the past relative to the concurrent situation in the context, that is "before": שֶׁנָתַיִם לִפְנֵי הָרָשָׁשׁ, "two years <u>before</u> the earthquake" (Amos 1:1), לְפְנֵי קְצִיר, "<u>before</u> the harvest" (Isa 18:5).

(c) Perceptual – expresses someone's personal view, or introduces events or situations that comes to one's attention: אֶה־חַשָּׁאָתִי לִפְנֵי אָבִיךְ כִּי מְבַקָּשׁ אֶת־נַפְשׁׁ, "What is my sin <u>against</u> your father that he seeks my life?" (1 Sam 20:1), וַהַשְׁחַת הָאָרֵץ לִפְנֵי הָאֵלְהִים, "The earth was corrupt <u>in the sight</u> of God" (Gen 6:11),

מִכּּל חַמּאתֵיכֶם לִפְנֵי יְהוָה תִּמְהָרוּ

"You will be clean from all your sins *before* Унwн" (Lev 16:30).

#### (12 4.1.13

(a) Source – One of the most common uses of בן designates where something or someone originated: אָרְצָן מְבֵית לְחֶם, "Ibzan <u>from</u> Bethelehem" (Judg 12:8),

וְעַתָּה קְחוּ לָכֶם שְׁנֵי עָשָׂר אִישׁ מִשִּׁבְמֵי יִשְׂרָאֵל

"Now, take for yourselves twelve men *from* the tribes of Israel" (Josh 3:12).

מן can place emphasis not just on the point of origin but also movement *away from* that point of origin, making it *ablative*: לְהוֹצִיאָם מֵאֶרִץ מִצְרִים, "to bring them out <u>from</u> the land of Egypt" (Exod 12:42), them out <u>from</u> the land of Egypt" (Exod 12:42), "I went after him and I struck him and I rescued [the lamb] <u>from his mouth</u>" (1 Sam 17:35).

(b) *Temporal* – similar to *source*, marks the beginning point of a given temporal period:

"I have walked before you <u>from my youth</u> until this day" (1 Sam 12:2), וְמָלֵךְ יְהוֶה עֲלֵיהֶם בְּהַר צִיוֹן מֵעַהֶּה וְעַר־עוֹלָם, "Yhwh will reign over them in Mount Zion <u>from now</u>, until eternity" (Mic 4:7),

מִן־הַיּוֹם אֲשֶׁר הוֹצֵאתִי אֶת־עַמִּי אֶת־יִשְׂרָאֵל מִמִּצְרַיִם

"<u>Since the day</u> when I brought my people Israel out of Egypt..." (1 Kgs 8:16).

(c) Material – identifies the material with which an action is performed, most often with verbs of formation: אָרָהָאָרָם אָפָר מִן־הָאָרָמָה ווייצֶר יְהוָה אָלהִים אָת־הָאָדָם עָפָר מִן־הָאָרָמָה. "And Yhwh God formed man <u>from</u> the dust of the ground" (Gen 2:7), מַסַרָה מַכַּסְפָם. "Idols made <u>from their silver</u>" (Hos 13:2).

(d) Causal-shows the reason or rationale behind an action: הָרִים רָשְׁשׁוּ מְמָנוּ (Nah 1:5), הָרִים רָשָׁשׁה אָת־הַדְּבָר הַזֶּה לָארֹנִי, "The mountains quake <u>because</u> of him" (Nah 1:5), חָלִילָה לִי מִיהוָה אִם־אָשֶׁשֶׂה אָת־הַדְּבָר הַזֶּה לַארֹנִי, "Far be it from me, <u>because</u> of YHWH, to do this thing against my lord" (1 Sam 24:7 [Eng 24:6]), נוֹיַנַּח שָׁלֹמֵה אֶת־כָּלִ־הַכֵּלִים מֵרֹב מָאָר מָאָר (1 Kgs 7:47).

In showing cause, כָּמ בָּמָ בָּמָ can at times designate the agent of a passive verb: לא־יָכָרַת כָּל־בָּשֶׁר עוֹד מָמֵי הַמֵּבּוּל, "All flesh shall never again be cut off <u>by the water</u> of a flood" (Gen 9:11).

(e) Estimative – expresses the opinion of one who is making an evaluation or judgment:

כּּי־כָבֵד מִמְּד הַדָּבָר לא־תוּכַל עֲשׂהוּ לְבַדֶּד

"For the task is too hard *for you*; you are not able to do it alone" (Exod 18:18),

קַמּנְתִּי מִכּּל הַחֲסָדִים וּמִכָּל־הָאֱמֶת אֲשֶׁר עָשִׂיתָ אֶת־עַבְדֶדָ

"I am *unworthy of* the steadfast love and faithfulness you have shown to your servant" (Gen 32:11 [Eng 32:10]).

(f) Partitive – designates a part of a larger whole: יָצָאוּ מִן־הָעַם, "<u>Some</u> of the people went out..." (Exod 16:27), יָצָאוּ מִן־הָעָם, "<u>Some</u> of the servants of the king are dead, as well as your servant, Uriah the Hittite" (2 Sam 11:24), "The generous man is blessed, because he gives <u>some of his food</u> to the poor" (Prov 22:9).

(g) Privative – marks something missing or lacking from the object of the preposition: בְּתִיהֶם שָׁלוֹם מָפָּחַר, "their houses stand <u>without</u> fear" (Job 21:9), הָי־אָז תִּשָּׁא פָנִיך מְמוּם, (Job 21:9), "Then, indeed, you could lift up your face <u>without</u> blemish" (Job 11:15), יְכָל־קְצִינֵיְך נְרְדּוּ־יַחַר מִקַשָּׁת אָסָרוּ , "All your rulers have fled together, they have been captured <u>without</u> the bow" (Isa 22:3).

(i) Compound – מן is commonly used to form compound particles with other prepositions, sometimes carrying the force of only one of the particles, but at other times carrying the combined meaning of both particles. Compare: לא־יָסוּר שֶׁבֶט מִיהוּדָה וּמְחֹכֵק מְבֵין רַגְלִיו יָהְנָרָ מְבָט מִיהוּדָה וּמְחֹכֵק מְבֵין רַגְלִיו יָהְנָרָ מָבָט מִיהוּדָה וּמָחֹכֵק מְבֵין רַגְלִיו יָהָלָק מְבֵין הַגָּלִים מִהוּדָה וּמְחֹכֵק מְבֵין הַגְלָיו יָהָלָיָ מְבֵין מָבָלי מִיהוּדָה וּמָחֹכֵק מְבֵין הַגְלָיו יָהָלָי מָבַט מִיהוּדָה וּמָחֹכֵק מְבֵין הַגְלָיו יָהָלָי מְבָטוּר שֵׁבָט מִיהוּדָה וּמָחֹכֵק מְבֵין הַגְלָיו יָהָלָים מִהוּדָה וּמָחֹכֵק מְבֵין הַגָּלִים יִהְנוּרַקוֹל מָבוּין מְבָלים מָבַין מַבָּשָמִים מִמַעַל וְעַל־הָאָרֶין מִתְכוּין מָמָבּין מַנָּמוּם מוּשׁנּים אוֹש voices among the branches" (Ps 104:12), "They lift up voices among the branches" (Ps 104:12), "There is no god like you in heaven above, or on the earth below" (1 Kgs 8:23); ווּבוּצָארִי אֶרְכָם מִתַּחַת סְבָלֹת מִצְרִים ווּ וּשׁוּש מוּשַעָּ ווּבַש מוּשַעָּ ווּ וּשַרָם, "I will bring you out from under the burdens of the Egyptians" (Exod 6:6).

# מפני 4.1.14

This preposition is a compound construction, consisting of the simple preposition מָן prefixed to the substantive פָּנִים in the construct form.

(a) Ablative – with verbs of motion, indicates motion away from a position "before" or "in front of" the object of the preposition: הַאָרָי גְּבְרְהֵי אָלָכִי בְּרְחֵת, "and she said 'I am fleeing <u>from</u> my mistress Sarah'" (Gen 16:8), said 'I am fleeing <u>from</u> my mistress Sarah'" (Gen 16:8), cloud moved <u>from in front</u> of them and stood behind them" (Exod 14:19), אָלָכִי אוֹרִישֵׁם מִפְּנֵי בְּנֵי יִשְׁרָאֵל them out <u>from before</u> the people of Israel" (Josh 13:6).

(b) Spatial – marks a location "before" or "in front of" the object of the preposition: הָס מְפְנֵי אֲדֹנִי יְהוֶה, "Be silent <u>before</u> the lord, YHWH" (Zeph 1:7), אַל־תַּחָת מְפְנֵיהֶם, "Do not be dismayed <u>before</u> them" (Jer 1:17).

(c) Causal – denotes the cause of an action or situation: וּיָבַא נֹחַ וּבָנָיו וְאָשֶׁתּוֹ וּנְשֵׁי־בָנָיו אָתּוֹ אָל־הַתַּבָה מִפְּנֵי מִי הַמָּבּוּל "Then Noah and his wife and the wives of his sons with him went into the ark <u>because</u> of the waters of the flood" (Gen 7:7), אָרָ מִפְּנֵי הָעָם מְאָר מִפְנֵי הָעָם מָאָר in great fear, <u>because</u> of the people" (Num 22:3), וַיָּנָס יוֹתָם וַיִּבְרַח וַיֵּלֶךְ בְּאָרָה וַיֵּשֶׁב שָׁם מִפְּנֵי אֲבִימֶלֶךְ אָחִיו Jotham escaped and fled, and went to Beer, and remained there <u>because</u> of Abimelech, his brother" (Judg 9:21).

#### لالا 4.1.15

(a) Locative – marks the the extent, limit, or goal of movement ("as far as," "up to," "to"): עַר־צַוָּאר יַוּיּשָ, "it will reach <u>up to</u> the neck" (Isa 8:8), רַיָּשְׁבוּ שָׁם וּיָשָׁר שָׁר עַר־קָרָן וַיִּשְׁבוּ שָׁם. "They went <u>as far as</u> Haran, and settled there" (Gen 11:31), "You have not returned <u>to me</u>" (Amos 4:6).

(b) Temporal – indicates the duration of an action: יַצָּא אָדָם לְפָעֶלוֹ וְלַעֲבֹדָתוֹ עֲדֵי־עָרֶב work and their labor <u>until</u> evening" (Ps 104:23), לא־בָאָתֶם עַר־עָתָה אָל־הַמְנוּחָה, "You have <u>not yet</u> (literally: <u>until now</u>) come upon the inheritance (Deut 12:9), "עַר־מָתִי מֵאַנְהָ לְעָנֹת מִפָּנָי, "<u>How long</u> (literally: <u>until when</u>) will you refuse to humble yourself before me?" (Exod 10:3).

(c) Degree – Quite frequently, עווי is conjoined to the אר מאר to express a large quantity of a substantive, or an extreme quality of an action or attribute: סיב אר מון הביער היאר געריקאר ייד מאר ווּהָהי הַמִּלְחְמָה קַשְׁה עַר־מָאר בַּיוֹם הַהוּא (1 Kgs 1:4), אל הַקַעָר יָבָה עַר־מָאר battle was <u>very</u> fierce on that day" (2 Sam 2:17), the <u>exceedingly</u> angry, O YHWH" (Isa 64:8 [Eng 64:9]).

### עַל 4.1.16

(*a*) *Spatial/locative* – על shows a variety of spatial/locative uses:

(a.1) Vertical relationship – indicates a location "over" or "upon" an object: רוּהַ אֶלהִים מְרַהֶפֶּת עַל־פְּנֵי הַמָּיָם, "the

spirit of God was hovering <u>over</u> the surface of the water" (Gen 1:2), הָשֶׁמֶשׁ יָצָא עַל־הָאָָרֶץ וְלוֹט בָּא צֹעֲרָה, "The sun had risen <u>over</u> the earth, when Lot came to Zoar" (Gen 19:23), ויֵט אַהֵרן אָת־יָרוֹ עַל מֵימֵי מִצְרִים, "Aaron held out his arms <u>over</u> the waters of Egypt" (Exod 8:2), out his arms <u>over</u> the waters of Egypt" (Exod 8:2), בָּאַשֶׁר הְיִיתִי בִּימֵי חָרְפִּי בְּסוֹר אֵלוֹהַ עֵלֵי אָהֵלִי the prime of my days, my God was <u>over</u> my tent" (Job 29:4).

With some verbs of motion, עַל can take on a *terminative* sense, expressing the end goal of vertical motion: ויֵרֶר הָעַיִט עַל־הַפָּנָרִים, "the bird of prey came down <u>upon</u> the carcasses" (Gen 15:11),

> וַיִּשָּׂא בִלְעָם אֶת־עֵינָיו וַיַּרְא אֶת־יִשְׂרָאָל שֹׁכֵן לִשְׁבָשָׁיו וַתְּהִי עָלָיו רוּחַ אֱלֹהִים

"Baalam lifted his eyes and he saw Israel camping tribe by tribe. And the spirit of God came <u>upon him</u>" (Num 24:2).

(a.2) *Horizontal relationship* – indicates a location "at" or "beside": וְיָבֶן עֵל־קִיר הַבַּיָת, "He built <u>against</u> the walls of the house ...." (1 Kgs 6:5),

וַיֹּאמֶר יְהוָה אֶל־מֹשֶׁה וְאֶל־אַהֲרֹן בְּהֹר הָהָר עַל־גְּבוּל אֶרֵץ־אֶדוֹם

"Then Yhwh spoke to Moses and to Aaron at Mount Hor, <u>at the border</u> of the land of Edom" (Num 20:23), הוֹמָה הְיוּ עָלִינוּ, "They were a wall <u>around</u> us" (1 Sam 25:16).

(b) Duty – The preposition can mark a burden or duty that impinges on a person: אָלָי לְהֶת לְךָ עֲשָׁרָה כֶסֶך "I would <u>have had [literally: it would have been upon me]</u> to give you ten [pieces] of silver" (2 Sam 18:11), to give you ten [pieces] of give you ten [pieces] of silver" (2 Sam 18:11), to give you ten [pieces] of give you ten (c) Rank – indicates a person's rank, or responsibility over another: יוֹאָר בֶּן־אָרוּיָה עַל־הַצָּרָא, "Joab, son of Zeruiah, was <u>over</u> the army" (2 Sam 8:16),

הַגַּר אֲשֶׁר בְּקְרְבְּד יַעֲלֶה עָלֶידְ מַעְלָה מָעְלָה

"The alien in your midst shall rise <u>above you</u>, higher and higher" (Deut 28:43), אַרֹנָי בָּל־עָּלֵיך, "You are my lord; I have no good greater than you" (Ps 16:2).

(d) Causal – expresses the *cause* of an action or situation: הַנְּךְ מֵת עַל־הָאָשֶׁה, "You will certainly die <u>because</u> of this woman" (Gen 20:3),

"He called the name of the place Massah and Meribah <u>because</u> of the quarrel of the people of Israel" (Exod 17:7), <u>הייעלי</u>ה הרענו כָל־הַיוֹם, "<u>Because of you</u>, we are being killed all day long" (Ps 44:23 [Eng 44:22]), הייקועל־זאת, "... for he was angry with him <u>because</u> of this" (2 Chr 16:10).

The *causal* use of  $\underline{a}$  is frequent when conjoined with the particle בן:

הֵילִילוּ כִּי קָרוֹב יוֹם יְהוֶה כְּשֹׁר מִשַּׁרֵי יָבוֹא עַל־כֵּן כָּל־יָדֵיִם תִּרְפֶּינָה

"Wail, for the day of Yнwн is near. It will come as destruction from the Almighty. <u>Therefore</u>, all hands will fall limp" (Isa 13:6–7 [Eng 13:6]), על־כֵּן יַעָזָב־אָישׁ אֶת־אָבִיו (אֶת־אָבִין וְאֶת־אָבִי), "<u>For this reason</u>, a man will leave his father and his mother" (Gen 2:24).

(e) Manner – behavior or action according to a standard: נָשְׁבַּע יְהָוָה וְלֹא יְנָחֵם אֲתָּה־כֹהֵן לְעוֹלָם עַל־הְבְרָתִי מַלְכִּי־צֶדֶק "Yhwh has sworn, and he will not change his mind. You are a priest forever, <u>according</u> to the order of Melchizedek'" (Ps 110:4), הוֹצִיאוּ אֵת־בְנֵי יִשְׂרָאֵל מֵאֶרֵץ מִצְרִים עַל־צְבָאֹתָם הוֹצִיאוּ אַת־בְנֵי יִשְׂרָאֵל מֵאֶרֵץ מִצְרִים עַל־צָבָאֹתָם, "Bring out the people of Israel from the land of Egypt according to their hosts" (Exod 6:26),

עַל־כֵּן קֶרְאוּ לַיָּמִים הָאֵלֶה פּוּרִים עַל־שֵׁם הַפּוּר

"Therefore, they called these days Purim, *according* to the name of Pur" (Esth 9:26).

(f) Adversative – shows action directed *against* another: אָם־לא תַשָּׁלִים עַמָּך וְעָשָׁתָה עַמְדְ מָלְחָמָה וְצַרְתָּ עָלֶיק But if it does not make peace with you, and makes war against you, then you shall make siege <u>against</u> it" (Deut 20:12), you, then you shall make siege <u>against</u> it" (Deut 20:12), "Then Baasha, the son of Ahijah of the house of Issachar, conspired <u>against</u> him" (1 Kgs 15:27),

> נִיּעֶּךְ אֶת־יְהוֹרָם בֶּן־אַחְאָב מֶעֶךְ יִשְׂרָאֵל לַמִּלְחָמָה עַל־חָזָאַל מֶעֶדְ־אָרָם

"He went with Jehoram, son of Ahab, king of Israel, to fight *against* Hazael, king of Aram" (2 Chr 22:5).

Adversative עַל also shows that an action can occur, despite circumstances that make it seem unlikely to the speakers: וְעַהָּה יַשְׁרָאָל עַל־וֹאָת, "Yet now, there is hope for Israel, <u>in spite</u> of this" (Ezra 10:2), there is hope for Israel, <u>in spite</u> of all these things, you said 'I am innocent." (Jer 2:34–35).

(g) Accompaniment – denotes circumstances that occur with another: לא שָׁתָם עַל־צָאן לָבָן, "he did not set them <u>with</u> the flock of Laban" (Gen 30:40), לא ־תִשְׁחֵט עַל־חָמֵץ דֵּם־זִבְחִי "You shall not offer the blood of my sacrifice <u>with</u> leavened bread" (Exod 34:25), ואָת־מַלְכֵי מִרְיָן הָרְגוּ עַל־חַלְלֵיהֶם, "They killed the kings of Midian, <u>along with</u> the rest of their slain" (Num 31:8).

Similarly, עַל can express a sense of *addition*: שֶׁבֶר נַל־שֶׁבֶר נִקְרָא, "Disaster <u>upon</u> disaster is proclaimed" (Jer 4:20). (*h*) Interest – עָל' with verbs of thinking, feeling, or emotions marks the object of interest: יְגִיל עָלַיִדְ בְּרְנָה , "he will rejoice <u>over</u> you with shouts of joy!" (Zeph 3:17), "I told them of how the hand of my God had been gracious <u>upon me</u>" (Neh 2:18).

(i) Emotive – functions to highlight or emphasize the subject of the emotion, as well as the expression of emotion: בְּהַתְּשָׁמִרְ שָׁלֵי נַפְּשִׁי אֶת־יְהָה זֶכְרְתּ: "When my life was <u>ebbing away</u> (literally: in feebleness <u>upon me</u>), I remembered YHWH" (Jonah 2:8 [Eng 2:7]), נֶהְפָּרְ שָׁלֵי לְבִי (Hy heart is turned <u>over</u> within me" (Hos 11:8).

#### עם 4.1.17

(a) Accompaniment – This preposition expresses additional participants in the discourse through a sense of *accompaniment*.<sup>17</sup> וַיָּאֶכְלוּ וַיָּשֶׁתוּ הוּא וְהָאָנָשִׁים אֲשֶׁר־עָבּוֹ ("Then he and the men who were <u>with</u> him ate and drank" (Gen 24:54), the men who were <u>with</u> him ate and drank" (Gen 24:54), "Now, listen to my voice, I will give you counsel, and may God be <u>with</u> you" (Exod 18:19).

Similarly, עם expresses a sense of *addition*:

עם־עָרִיהֵם הֵחֵרִימָם יְהוֹשָׁעַ

"Joshua exterminated them <u>along with</u> their cities" (Josh 11:21), הַאָּך תַּסְפָה צַדִּיק עִם־רְשָׁע, "Will you indeed sweep away the righteous <u>along with</u> the wicked?" (Gen 18:23).

(b) Personal complement – resembles a datival function, to mark the recipient of an action: וְשַׁה־חֶסֶר עִם אֲרֹנִי אַבְרָהָם,

<sup>17</sup> BDB notes that the comitative/accompaniment sense of אָר is much more intimate and expresses closer association than the comitative sense of עם (37). "And show kindness <u>to</u> my master Abraham" (Gen 24:12), אָלא אָם־עוֹדֶנִּי חָי וְלא הַעַּטֶּה עִמָּרִי חֶסֶר יְהוָה וְלא אָמוּת, "If I am still alive, will you show <u>to me</u> the faithfulness of YHwH, that I may not die?" (1 Sam 20:14),

הַנֵּה אָנֹכִי בָּא אֵלֶיךּ בְּעַב הֶעָנָן בַּעֲבוּר יִשְׁמַע הָעָם בְּדַבְּרִי עִמָּך

"Behold, I will come to you in a pillar of cloud, so that the people will hear when I speak *with* you" (Exod 19:9).

Similarly, this preposition may indicate dealings or relationships with another: אָלְהֶיָה עָם יְהוֶה אֶלְהֶיה "Be blameless <u>before</u> YHWH, your god" (Deut 18:13), "Let your hearts be wholly devoted <u>to</u> YHWH, our god" (1 Kgs 8:61).

(c) *Locative* – marks a locality or position:

וַיֵּשֶׁב יִצְחָק עִם־בְּאֵר לַחַי רֹאִי

"and Isaac lived <u>near</u> Beer-lahai-roi" (Gen 25:11), וַיִשְׁלֵח יְהוֹשֻׁעַ אֲנָשִׁים מִירִיחוֹ הָעַי אֲשֶׁר עִם־בֵּית אָוָן, "Joshua sent out men from Jericho to Ai, which is <u>near</u> Beth-Aven" (Josh 7:2), הַמָּה עִם־בֵּית מִיכָה, "[When] they were <u>near</u> the house of Micah" (Judg 18:3).

(d) Restrictive – presents an exception to a circumstance or action: רְאוּ עַהָּה כִּי אֲנִי אֲנִי הוּא וָאֵין אֱלֹהִים עָפָּרִי, "See now that I, I am he, and there is no god <u>besides</u> me" (Deut 32:39), מִי־לִי בַשְׁמָיִם וְעָפְּדָ לֹא־חָפָצְתִי בָאָרֵיץ , "Whom do I have in heaven <u>but you</u>, and I desire nothing else on earth" (Ps 73:25), יְהוָה אֵין־עָפְדָ לְעָוֹוֹר (Ps 73:25), "O Yhwh, there is no one besides you to help" (2 Chr 14:10 [Eng 14:11]).

<u>بوت</u>ر 4.1.18

(a) Vertical relationship – This preposition locates an object "under" another: יְהֵיוּ אֲפֶר תַחֵת כַפּוֹת רְגְלִיכֶם, "they will be

ashes <u>under</u> the soles of your feet" (Mal 3:21 [Eng 4:3]), יַקָּחְ־נָא מְעָטִ־מִים וְרְחֲצוּ רַגְלִיכֶם וְהָשֶׁעֲנוּ תַּחָת הָעֵץ, "Let a little water be brought and wash your feet, and rest yourselves <u>under</u> the tree" (Gen 18:4), וְהָיא יוֹשֶׁבֶת תַּחַת־תִּמֶר דְּבוֹרָה (Judg 4:5), "She used to sit <u>under</u> the oak of Deborah" (Judg 4:5), solution in all that is done <u>under</u> the sun" (Eccl 9:3).

(b) Static position – The locative תַחָת can also express a static position of "on the spot," or "in that place": and we will not go up to them" (1 Sam 14:9), (יֶשְׁבוּ אֲלֵיהֶם "Remain, every person, <u>in their place</u>" (Exod 16:29), "עמו אָישׁ אָת־אָחִיו וְלֹא־קָמוּ אִישׁ מְתַחְתִיו שָׁלשֶׁת יָמִים, "They could not see one another, and they did not rise <u>from their place</u> for three days" (Exod 10:23).

(c) Metaphorical – expresses subordination or obedience to the influence of another: וְהַכְּנַע מוֹאָב בִּיוֹם הָהוּא תַּחָת יֵד יְשֶׁרָאֵל, "On that day, Moab was subdued <u>under</u> the hand of Israel" (Judg 3:30), הַחַר אָרָרָי אָחֲלִי כוּשָׁן, "I saw the tents of Cushan <u>under</u> distress" (Hab 3:7), saw the tents of Cushan <u>under</u> distress" (Hab 3:7), "I saw the tents of Cushan <u>under</u> distress" (Hab 3:7), saw the tents of Cushan <u>under</u> distress" (Hab 3:7), "I no man has lain with you, and if you have turned to uncleanness <u>under the authority</u> [literally: <u>under</u>] your husband..." (Num 5:19).

(d) Substitution – A common secondary use of הַחַת וֹּבָסֿ show substitution, that is, "instead of" or "in the place of": שֶׁת־לִי אֶלֹהִים זֶרַע אַחֵר תַּהָל "God has appointed to me another offspring <u>in place of</u> Abel" (Gen 4:25), וַיַּמְלֵךְ פַּרְעֹה נְכֹה אֶת־אֶלְיָקִים בֶּןְ־יֹאשִׁיָהוּ תַּחַת יֹאשׁיָהוּ אָבִיו "Pharaoh Neco made Eliakim, son of Josiah, king <u>in place of</u> Josiah, his father" (2 Kgs 23:34), וַיֵּשֶׁב שְׁלֹמה עַל־כָּפַא יְהוָה לְמֶלֶךְ תַחַת־רְדָוִיד אָבִיו Solomon sat on the throne of YHWH, as king, <u>in place of</u> David his father" (1 Chr 29:23).

#### 4.2 Adverbs

The adverb in any language is typically the most complicated and least understood of the traditional parts of speech. It is elusive due to its wide range of functions, modifying a single word or an entire clause, as well as because an adverb can be classified as other parts of speech. In BH, both levels of complexity are present in the adverbs. Many, if not most, of the adverbs in BH can function as conjunctions or prepositions. To complicate matters, adverbial function is not limited to the adverbs in BH; it can be performed by accusatives (section 2.3.2), by infinitives absolute (section 3.4.2,b and c), and by certain verb plus verb constructions (see section 4.3.3,g on hendiadys).<sup>18</sup> BH adverbs have two major divisions: Those that modify clauses or words in a way that is related to the discourse are the *clausal/item* adverbs.<sup>19</sup> Those that specify the time, place, or manner of the predicated situation are called the constituent adverbs. There are some adverbs that are classified in both groups; thus, the distinction seems to lie in syntax, not morphology.

The following are the most common adverbs and their uses.<sup>20</sup>

#### 4.2.1 × / '<u>`</u>

(a) **Temporal** – Typically rendered "then," this adverb<sup>21</sup> functions to indicate a subsequent action within the

- <sup>19</sup> Item adverbs, which modify single words, are quite rare in BH. Because they serve a function similar to the clausal adverbs, they should be considered together. See Waltke and O'Connor 1990, 656; van der Merwe, Naudé, and Kroeze 1999, 58 and 305–20.
- <sup>20</sup> Kautzsch 1910, 483–84; Waltke and O'Connor 1990, 655–73; van der Merwe, Naudé, and Kroeze 1999, 305–20; Williams 1976, 63–66; Bauer and Leander 1991, 630–34; Joüon and Muraoka 1993, 329–36. See also BDB, *HALOT*, and *DCH*.
- <sup>21</sup> אוֹי is the poetic form of the adverb.

<sup>&</sup>lt;sup>18</sup> Waltke and O'Connor 1990, 656. The authors note that the verbs משוב and סף are paired with other verbs to indicate a repetition or continuation of action.

"Then they believed in Yhwh and in Moses, his servant. <u>Then</u> Moses and the people of Israel sang" (Exod 14:31– 15:1),

```
וּלְשֵׁת גַּם־הוּא יֻלַּד־בֵּן וַיִּקְרָא אֶת־שְׁמוֹ אֶגוֹשׁ אָז הוּחַל לִקְרֹא
בְּשֵׁם יְהוָה
```

"To Seth, to him also, was born a son, and he called his name Enosh. <u>Then</u> people began to call on the name of YHWH" (Gen 4:26),

```
אָבִיך חֲלוֹא אָכַל וְשֶׁתָה וְעָשָׂה מִשְׁפָּט וּצְדָקָה אָז טוֹב לוֹ
```

"Did your father not eat and drink, and do justice and righteousness? <u>*Then*</u> it was well with him" (Jer 22:15).

(b) Logical – often marks a logical turn in the flow of the discourse: יְשָׁנְהָי אָז יְנוּהַ לִי, "I would have slept, <u>and [as a result]</u> I would have been at rest" (Job 3:13),

"This book of the law shall not depart from your mouth, but you shall meditate upon it day and night, so that you may be careful to do all that is written in it. <u>For then</u> you shall make your way prosperous, and <u>then</u> you shall succeed" (Josh 1:8),

> וִלֹא־יָסַף עוֹד מַלְאַךְ יְהוָה לְהֵרָאֹה אֶל־מָנוֹחַ וְאֶל־אָשְׁתּוֹ אָז יָדַע מְנוֹחַ כִּי־מַלְאַךְ יְהוֶה הוּא

"The messenger of YHWH did not appear again to Manoah and his wife; <u>then</u> Manoah knew that it was the messenger of YHWH" (Judg 13:21).

(c) Condition – can also introduce the apodosis of a conditional statement, and often implies the fulfillment of

that condition:

```
אִם־תְּבַקְשֶׁנָּה כַכָּסֶף וְכַמַּמְמוֹנִים תַּחְפְּשֶׁנָּה אָז תָּבִין יִרְאַת יְהוָה
וְדַעַת אֶלהִים תִּמְצָא
```

"If you seek her as silver, and search for her as hidden treasure, <u>then</u> you will understand the fear of YHWH and you will find the knowledge of God" (Prov 2:4–5), אַחֲלֵי אֲרֵנִי לְפְנֵי הַנָּבִיא אֲשֶׁר בְּשׁמְרוֹן אָז יָאֲסֹך אֹתוֹ מַאָּרַשְׁהוֹ , "Ah that my lord were before the prophet that is in Samaria, <u>then</u> he would cure him of his leprosy" (2 Kgs 5:3), לּוּלֵי תוֹרָתִך שַׁעַשָׁשָׁי אָז אָבַדְהִי בְעָנִי , "If your law had not been my delight, <u>then</u> I would have perished in my affliction" (Ps 119:92).

# 4.2.2 JX

(a) **Restrictive** – often functions to establish a clarification of or mild contrast with ideas preceding in the discourse. It is a restriction or limitation, however, and not a strong disjunction because there is not a complete reversal of what has preceded. Rather, there is a limitation placed on the previous clause(s):

"Every first issue of the womb of all flesh, which they offer to YHWH, whether man or animal, shall be yours. <u>Nevertheless</u> the firstborn of man you shall redeem." (Num 18:15), the firstborn of man you shall redeem." (Num 18:15), (Num 18:15), "<u>Only</u> Noah, and those who were with him in the ark, [were] left" (Gen 7:23),

"Do not fear. You have committed all this evil; <u>yet</u> do not turn from serving Үнwн" (1 Sam 12:20).

(b) Asseverative – Often rendered "surely," <sup>3</sup> introduces a statement or expression of truth, or highlights an

unexpected truth, quite often in speech or colloquial language, by expressing a conviction as to its correctness:<sup>22</sup> אָדָ הָנָה מְשָׁרוּ (1 Sam 16:6), אָךְ מֶלֶך־יִשְׁרָאֵל הוּא (<u>Surely</u> it is the king of Israel" (1 Kgs 22:32), אָךְ מֶלֶרִישָׁרִי אָהָה, (<u>Surely</u> it is the king of Israel" (1 Kgs 22:32), אַךְ מֶלֶרִי אָהָה שַׁצְמִי וּרְשָׁרִי אָהָה wy bone and my flesh" (Gen 29:14).

# אַל 4.2.3

While translated much like אָל ("not"), אָל may be found in specific constructions.

(a) Prohibition – typically negates imperatival forms to denote a specific or immediate prohibition: אַל־תִּשְׁמִח יִשְׁרָאָל "<u>Do not rejoice</u>, O Israel" (Hos 9:1), גָּלִיתַּקָרָאָל־תַּחָרֹג, (<u>Do not kill</u> the innocent and the righteous" (Exod 23:7), "<u>Do not kill</u> the innocent and the righteous" (Exod 23:7), "<u>The not commanded you?</u> Be strong and courageous! <u>Do not tremble</u>, and <u>do not be dismayed</u>" (Josh 1:9).

(b) Negative volition – As the negation of the modals jussive and cohortative, אל expresses their same nuances negatively, usually in the form of a negative wish or negative prayer: אל־יָמָשׁלוּ־בִי, "<u>Do not let them</u> rule over me" (Ps 19:14), אל־יָבָא הָרִיָה בִּינִי וְבִינֶיךָ "<u>Let there be no strife</u> between me and you" (Gen 13:8), "<u>Let there be no strife</u> between me and you" (Gen 13:8), their place on the seventh day" (Exod 16:29).

### 4.2.4 J×

The first of two major coordinating adverbs ( $\Box$ ) being the second),  $\neg$ <sup>23</sup> is closer to the simple  $\rceil$  conjunction, and

<sup>&</sup>lt;sup>22</sup> Waltke and O'Connor 1990, 670.

<sup>&</sup>lt;sup>23</sup> Generally more common in poetry and later prose, while  $\Box$  is more common to prose.

may at times be translated simply "and." In general, it also associates its clause (i.e., one that follows it) with a preceding clause (Waltke and O'Connor 1991, 663).

(a) Addition – indicates that an entity is added to another: אַרְ־אָנִי בַחֲלוֹמָ, "I <u>also</u> had a dream [literally: <u>also</u> I in my dream]" (Gen 40:16),

```
וְזָכַרְתִּי אֶת־בְּרִיתִי יַשְׁקוֹב וְאַף אֶת־בְּרִיתִי יִצְחָק וְאַף אֶת־בְּרִיתִי
אַרְרָהֶם אֶזְכֹּר
```

"Then I will remember my covenant with Jacob; I will remember <u>also</u> my covenant with Isaac and <u>also</u> my covenant with Abraham" (Lev 26:42), אר־לְךָ לְיָלָה, "Yours is the day, yours <u>also</u> the night (Ps 74:16).

When the entity added is unexpected, אָר may be translated "even" and is sometimes said to be an *emphatic* use of the particle, although the concept of emphasis is too broad to be of much use:<sup>24</sup> יאָרֶכֶם אַרְ־אָנָי, "I, <u>even</u> I, will punish you myself" (Lev 26:28). Such a use may occur in questions: הַאַרְ אַרְנָם אַרֵּד וַאַני זָכְנָת: bear a child, now that I am old?" (Gen 18:13).

(b) Asseverative<sup>25</sup> – Especially in poetry, אָר introduces a statement or expression of truth and expresses a conviction as to its correctness: אָר־נְחָלָת שָׁפְרָה שָׁלָ, "<u>Surely</u> my heritage is beautiful to me" (Ps 16:6), אַר מִן־קָמִי הָרוֹמְמֵני, "<u>Indeed</u> you exalted me above my adversaries" (Ps 18:49 [Eng 18:48]).

(c) Rhetorical – A variation of the *asseverative*, this particle is often combined with יכָ to express a comparative assertion in which two clauses are related, the second one bearing persuasive force: גָּבָרְבְוָמָה יְבִיאָנוּ , "the

<sup>&</sup>lt;sup>24</sup> Joüon and Muraoka 1993, 618.

<sup>&</sup>lt;sup>25</sup> The asseverative/emphatic use of ¬X, as well as □ (4.2.5), may not be original to either particle, but a nuance of their primary use as particles denoting addition (Muraoka 1985, 142-43).

sacrifice of the wicked is an abomination; <u>how much more</u> when he brings it with evil intent" (Prov 21:27). This may also be expressed negatively (i.e., with a negated first clause):

הַנֵּה הַשְּׁמַיִם וּשְׁמֵי הַשָּׁמַיִם לֹא יְכַלְכָּלוּך אַף כִּי־הַבַּיִת הַזֶּה אֲשֶׁר בְּנִיתִי

"Even heaven and the highest heavens cannot contain you; *how much less* this house that I have built!" (1 Kgs 8:27).

This *rhetorical* use of אָר כִי may actually introduce a *rhetorical question*, in which the assertion of the clause is confirmed in light of the preceding clause:

הַנֵּה אֲנַחְנוּ פֹה בִּיהוּדָה יְרֵאִים וְאַף כִּי־נֵלֵךְ קְעָלָה אֶל־מַעַרְכוֹת פְּלְשְׁתִים

"Look, we are afraid here in Judah; <u>how much more</u> then if we go to Keilah against the armies of the Philistines?" (1 Sam 23:3), אָך כִּירָהָעָב וְנָאֲלָח אִישׁ־שֹׁהָה כַמַּיִם עַוְלָה one loathsome and foul, a person who drinks wrongdoing like water?" (Job 15:16).

## 4.2.5

The second of two major coordinating adverbs  $(\neg \aleph$  being the first),  $\Box \clubsuit$  has quite similar uses, although it is much more common in prose, while  $\neg \aleph$  is more common in poetry. In general,  $\Box \clubsuit$  may modify either a word or clause, in which case it also associates its clause (i.e., one that follows it) with a preceding clause.

(a) Addition – often used to point out an additional participant or party within the discourse: וְּתְּהֶן נַם־לְאִישָׁה עָבָּה "and she gave <u>also</u> to her husband with her "(Gen 3:6). When the additional event or statement is unexpected or illogical, "even" may be the best translation: ויָבָן פַּרְעָה וְיָבֹא אֶל־בֵּיתוֹ וְלֹא־שֶׁת לְבֵוֹ נַם־לְוֹאָת "Then Pharaoh turned and went into his house, and he did not take <u>even</u> this to heart" (Exod 7:23), אָין עַשָּה־טוֹב אָין נָם־אָחָד, "there is no one who does good, not *even* one" (Ps 14:3).

The use of אַם as a "double conjunction" is a variation on the *addition* use, and typically denotes the inclusion of *both* entities preceded by the particle:<sup>26</sup> הירָכֶב גַם־פָּרָשִׁים, "*both* chariots <u>as well as</u> charioteers went up with him" (Gen 50:9), גַם־תָּבָן גַם־תָּסָפּוֹא רַב עָמָנו fodder" (Gen 24:25).

(b) Asseverative – adds emphasis or certainty to an idea: וְהָיָה אָם־לֹא יָאֲמִינוּ גַּם לְשְׁנֵי הָאֹתוֹת הָאֵבֶה, "But, if they will not believe <u>even</u> these two signs" (Exod 4:9), will not believe <u>even</u> these two signs" (Exod 4:9), גָם בֵּין הָעֲמָרִים הְּלַקֵּט וְלֹא תַכְלִימוּהָ, "Let her glean <u>even</u> among the sheaves and do not insult her" (Ruth 2:15).

When found at the end of a speech or exposition, or at the end of lists, אַם־בְּרוֹשִׁים שָׁמְחוּ לְך <u>ג</u>ם־בְּרוֹשִׁים שָׁמְחוּ לְך <u>ג</u>ם־בְרוֹשִׁים שָׁמְחוּ לְך <u>ג</u>ם־בְרוֹשִׁים שָׁמְחוּ לְך <u>ג</u>מי לָרוּ שָׁבְרוּ אֶת־יְהוָה רַק צֹאנְכֶם וּבְקַרְכֶם יָצָּג נַם־טַפְּכֶם יֵלֵךְ עָמְכָם , לְכוּ שָׁבְרוּ אֶת־יְהוָה רַק צֹאנְכֶם וּבְקַרְכֶם יָצָג נַם־טַפְּכֶם יֵלֵך Go, serve Yhwh, only let your flocks and herds be detained; <u>even</u>your little ones may go with you" (Exod 10:24).

(c) Concessive – indicates that an action that is or was expected to lead to another action will not, or leads to an unexpected action: בְּחָנוּנִי נַם־רָאוּ פְּעֵלִי, "they tested me, <u>even though</u> they saw my works" (Ps 95:9), הָבְשׁוּ נַם־בּוֹשׁ לֹא־יִבְשׁוּ (They acted

<sup>&</sup>lt;sup>26</sup> See van der Merwe, Naudé, and Kroeze 1999, 239 and 314–15, and especially point 5.2(ii) on page 316.

shamefully. They committed abomination. <u>Yet</u> they were not at all shamed" (Jer 8:12).

#### הַרְבֵה 4.2.6

Degree – As an adverb, בְּרְבֵה<sup>27</sup> expresses a great, abundant, or extreme degree of action, and is often used with מָאָר<sup>28</sup> הַנָּה הַסְכַּלְתִי וָאָשְׁגֶה הַרְבֵה מְאֹר, "Yes, I have been foolish, and have erred <u>very greatly</u>" (1 Sam 26:21), גָאָיָרָא הַרְבֵה מְאֹר "Then I was <u>very much</u> afraid" (Neh 2:2).

The particle also functions *adjectivally*, to indicate a large amount of a substantive: וְנֵם־הַרְבָּה נָפַל מִן־הָעָם וַיָּמָתוּ "Also, *many* of the people fell and they died" (2 Sam 1:4), אַל־תִּיָרָא אַבְרָם אָנֹכִי מְגַן לָךְ שְׁכָרְף הַרְבָּה מְאֹר Abram, I am a shield to you. Your reward shall be *very great*" (Gen 15:1).

#### יוֹמֶם 4.2.7

*Temporal locative* – This adverb<sup>29</sup> functions to designate a general point in time, "by day" or "daytime," when an action takes place: וְשָׁבֵי יְהָרָה שָׁלֵיהֶם יוֹמָם, "the cloud of YHWH was over them <u>by day</u>" (Num 10:34),

```
ַ
וַיְהִי כַּאֲשֶׁר יָרֵא אֶת־בֵּית אָבִיו וְאֶת־אַנְשֵׁי הָעִיר מֵעֲשׂוֹת יוֹמָם וַיַּעַשׂ
לְיִלֶה
```

"Because he was too afraid of his father's house and of the men of the city to do it *by day*, he did it at night" (Judg 6:27),

<sup>&</sup>lt;sup>27</sup> This adverb illustrates the complexity of Hebrew adverbs generally, since it is in form a Hiphil infinitive absolute. Along with the adverbial function, the infinitive also functions as an adjective, as well as a substantive.

<sup>&</sup>lt;sup>28</sup> HALOT 1:255.

<sup>&</sup>lt;sup>29</sup> The frequentative or repetitive use of the מיֹמָם 'i' is noted in BDB (401) and Seow (44). BDB notes that the repetitive use of this adverb is most often found in poetry, and within a merism, where it is paired with ליְלָילָה It may be, however, that the repetitive force comes not from the adverb alone, but is expressed by the merism instead. For more on merisms in BH, see Krašovec (1977 and 1983).

ןהוֹצֵאָת כֵלֶיךְ כַּכְלֵי גוֹלָה יוֹמָם לְעֵינֵיהֶם, "Bring your baggage out <u>during the day</u> in their sight, like baggage for exile" (Ezek 12:4).

לה 4.2.8

(a) Manner – indicates that an action takes place or took place in a certain manner, and often takes "thus" in translation. With verbs of speech, this use often introduces the content of speech: לָבְרָיִיָּרָאָל, "thus you shall say to the children of Israel" (Exod 3:14), you shall say to the children of Israel" (Exod 3:14), לְבָּרֵיךָ

וִיָּסְבּוּ אֶת־הָעִיר בַּיּוֹם הַשֵּׁנִי פַּעַם אַחַת וִיָּשָׁבוּ הַמַחֲנָה כֹּה עָשׂוּ שֵׁשֶׁת יָמִים

"On the second day, they surrounded the city once and returned to camp. They did *thusly* for six days" (Josh 6:14).

(b) Demonstrative/locative – points demonstratively to the place of action or discourse: שָׁים כָּה בָנֶך אַהַי, "Set it <u>here</u> before my brothers" (Gen 31:37), קּרָיַצֵּב כָּה עַל־עַלְתָדָ, "Stand <u>here</u>, by your burnt offering" (Num 23:15), "Stand <u>here</u>, by your burnt offering" (Num 23:15), "אָמֶר הַמָּלֶך סָב הַתִיצֵב כָּה וַיָּסָב וַיַּעֲמֹד around and stand <u>here</u>.' So he turned around and stood still" (2 Sam 18:30).

Rarely, this locative sense refers to a temporal period: דְהַנָּה ל'א־שָׁמַעְהָ עַר־כֹּה, "Indeed, you have not listened until <u>now</u>" (Exod 7:16).

## ₹' 4.2.9

The particle 'יֹ has a tremendous variety of uses, several of which have adverbial force, which are detailed under "Conjunctions" (see section 4.3.4).<sup>30</sup>

30 DCH 4:383-91

2[ 4.2.10

(a) Comparative – typically used in the apodosis of a comparison: בְּעֵינֵי שֶׁבְּחָה אֶל־יָד גְּבְרְתָּה כֵּן עֵינֵינוּ אֶל־יְהוָה. "As the eyes of a maid to the hand of her mistress, <u>so</u> our eyes look to YHWH" (Ps 123:2), וְיָהִי כַּאֲשֶׁר פְּתַר־לְנוּ כֵן הָיָה, "As he interpreted to us, <u>so</u> it turned out" (Gen 41:13).

In variation of this use, כָ appears in a clause that is not a comparative apodosis, but nevetheless has a comparative force (hence, *manner*): כָּל־אוֹיְבֶיךּ יְהָוָה *"Thus* perish all your enemies, O YHWH" (Judg 5:31), *"Thus* perish all your enemies, O YHWH" (Judg 5:31), ''and let them be for lights in the expanse of heaven to give light onto the earth.' And it was <u>so</u>" (Gen 1:15).

(b) Compound forms of כִן – The compound form כִּכִן typically indicates a response to a statement of conditions ("the foregoing being the case, therefore..."):<sup>31</sup> ("But they do not consider the deeds of YHWH nor do they consider the work of his hands; <u>therefore</u>, my people go into exile" (Isa 5:12-13),

"You have forsaken me and served other gods; *therefore*, I will no longer deliver you" (Judg 10:13),

ַיַעַן אֲשֶׁר־הִכְרַתִּי מִמֵּך צַּדִּיק וְרָשֶׁע לְכֵן תֵּצֵא חַרְבִּי מִתַּעְרָה אֶל־כָּל־בָּשָׁר מִנֶגָב צָפּוֹן

"Because I will cut off from you the righteous and the wicked; <u>therefore</u>, my sword will go out from its sheath against all flesh from south to north" (Ezek 21:9 [Eng 21:4]).

The compound form שָׁל-כֵּן often introduces a statement of effect, linking two clauses with a causal link:

"Because of the tithe of the sons of Israel, which they offer as an offering to YHWH, I have given to the Levites for an inheritance; <u>therefore</u>, I have said concerning them, 'They shall have no inheritance among the children of Israel' " (Num 18:24), עַל־כֵּן קָרָא שֶׁמָה בָּכֵל כִּי־שָׁם בָּלַל יְהוָה שִׁפַת כָּל־הָאָרֵץ, "<u>Therefore</u>, its name is called Babel, because there, YHWH confused the languages of the whole earth" (Gen 11:9).

# לא 4.2.11

Negation – The primary use of this adverb is to negate verbal clauses.<sup>32</sup> The particle functions to negate a verbal idea in an independent verbal clause: רְשָׁה לֹא רָאָינוּ, "we did <u>not</u> see evil" (Jer 44:17), דְשָׁה כָּאַחָת שָׁפּחֹתֶיך, "we "I am <u>not</u> like one of your maidservants" (Ruth 2:13), "I am <u>not</u> like one of your maidservants" (Ruth 2:13), disgrace, and his reproach will <u>not</u> be blotted out" (Prov 6:33).

The negative force of לא כמח also be used with commands to denote a *general* or *permanent prohibition*: לא תאכל מְמָנוֹ , "you <u>shall not</u> eat from it" (Gen 2:17), fut anything that does not have fins and scales, *you shall not eat*" (Deut 14:10), וּבִיּוּם הַשְּׁבִיעִי מִקְרָא־קְדֶשׁ יִהְיֶה לְכֶם כָּל־מְלָא כָת עֲבֹדָה לֹא תַעֲשׁוּ And on the seventh day you shall have have a holy convocation; <u>you shall not do</u> any laborious work" (Num 28:25).

<sup>&</sup>lt;sup>32</sup> Occasionally, and in contrast with the negative particle לא אל will be used to negate only one word, typically a nominal form: וֹקאָרָם לְבָרוֹ , "it is not good for the man to be alone" (Gen 2:18).

138

At times, לא negates verbless clauses or prepositional phrases: לא בִי־הִיא, "it is <u>not</u> in me" (Job 28:14), לא־טוֹב הַדְּבְר אֲשֶׁר אַתָּה עשֶׁה, "The thing you are doing is <u>not good</u>" (Exod 18:17),

וַתַּעֲשֶׂה אָדָם כִּדְגֵי הַיָּם כְּרֶמֶשׂ לֹא־מֹשֵׁל בּוֹ

"You have made humanity like fish in the sea, like crawling things with *no ruler* over them" (Hab 1:14).

לאר 4.2.12

**Intensive** – The intensive adverbial use of מאר points to a high degree or magnitude in the force of the verb: אַבְרָם כָּבֵד מָאר, "Abram was <u>very</u> rich" (Gen 13:2), "Abram was <u>very</u> rich" (Gen 13:2), "They will be afraid of you; so be <u>very</u> careful" (Deut 2:4), "They will be afraid of you; so be <u>very</u> careful" (Deut 2:4), "They <u>were</u> <u>overcome with fear</u>" (2 Kgs 10:4),

וּיִקְצֹף הַמֶּלֶך מְאֹד וַחֲמָתוֹ בָּעֲרָה בוֹ

"Then the king became *very angry*, and his anger burned within him" (Esth 1:12).

The particle also intensifies an adjective:

הָאָרֶץ אֲשֶׁר עָבַרְנוּ בָה לָתוּר אֹתָה טוֹבָה הָאָרֶץ מְאֹד מְאֹד

"The land which we went through to spy out is a *very good* land" (Num 14:7),

וִיְהִי כְּכַלּוֹת יְהוֹשֶׁעַ וּבְנֵי יִשְׂרָאֵל לְהַכּוֹתָם מַכָּה גְּדוֹלָה־מְאֹד

"When Joshua and the people of Israel had finished slaying them with a <u>great slaughter</u> ... " (Josh 10:20), קי־נָרוֹל יוֹם־יְהוָה וְנוֹרָא מְאֹר וּמִי יְכִילֶנוּ, is great and <u>very terrible</u>; who can endure it?" (Joel 2:11).

In addition to the adverbial use, מָאֹר can also function adjectivally to denote a "large amount," frequently when paired with either הרבה סירה רב סירה הרבה, וּמְבֶּטָח וּמְבֵּרֹתֵי עָרִי חֲדַדְעָזֶר לָקַח הַמֶּלֶךְ דָּוִד נְחֹשֶׁת הַרְבֵּה מְאֹר , וּמְבֶּטָח וּמְבֵּרֹתֵי עָרִי חֲדַדְעָזֶר לָקַח הַמֶּלֶךְ דָוֹד David took a <u>large amount</u> of copper" (2 Sam 8:8), ואָהַן לָמָּלֶךְ מַאָּהוְעָשָׂרִים כָּכָּר זָהָבוּבְשָׁמִים הַרְבָה מָאָר וּאָהַן לָמָּלֶךְ מַאָּהוְעָשָׂרִים כָּכָּר זָהָבוּבְשָׁמִים הַרְבָה מָאָר, "She gave to the king 120 talents of gold and a <u>large amount</u> of spices" (1 Kgs 10:10).

## עוד 4.2.13 עוד

**Manner** – expresses a *continual* or *persistent* nature of the verbal clause: אָבְרָהָם עוֹדֶנּוּ עֹמֵר לִפְנֵי יְהָוָה, "and Abraham was <u>still</u> standing before YHWH" (Gen 18:22), as the day when Moses sent me" (Josh 14:11), עוֹדֶנִי הַיּוֹם חָזָק כַּאֲשֶׁר בִּיוֹם שָׁלֹחַ אוֹתִי מֹשֶׁה , "So, circumcise the foreskins of your hearts, and stiffen your neck <u>no longer</u>" (Deut 10:16),

עוֹדֶנּוּ הָאֶָרֶץ לְפָנֵינוּ כִּי דְרַשְׁנוּ אֶת־יְהוָה אֱלֹהֵינוּ

"The land is *still* ours, because we have sought YHWH, our God" (2 Chr 14:6).

This particle also indicates a *repeated* nature of a verbal clause: וְיֵרָע אָרָם עוֹך אָת־אָשָׁתוֹ, "Adam had relations with his wife <u>again</u>" (Gen 4:25),

בִּי אֲדוֹנָי אִישׁ הָאֶלהִים אֲשֶׁר שָׁלַחְתָ יָבוֹא־נָא עוֹד אֵלֵינוּ

"O lord, please let the man of God, whom you sent, come to us <u>again</u>" (Judg 13:8), וַתַּלֶר בַת, "She conceived <u>again</u> and gave birth to a daughter" (Hos 1:6).

## עַקָּה 4.2.14

(a) Temporal – places focus on time that is concurrent with the perspective of the discourse, and is most often rendered "now": אַשָּה מָשִׁיחוֹ אָשָׁיחוֹ (Ps 20:7), געָהָר וַעָד־עַהָּה זָה (So she came and she stayed from this morning until <u>now</u>" (Ruth 2:7), אָהָה עַקָה מָלוּכָה עַל־יִשְׂרָאַל "Do you <u>now</u> reign over Israel" (1 Kgs 21:7), וְנָם־עַתָּה נְאָם־יְהוָה שֶׁבוּ עָדֵי הְּכָל־לְבַרְכָם וּרְצוֹם וּרְבְרִי וּבְמָסְפֵּר , (נָם־עַתָּה נָאָם־יְהוָה שֶׁבוּ עָדֵי הְכָל־לְבַרְכָם וּרְצוֹם וּרְבָרִי , Yet even <u>now</u>,' declares YHWH, 'Return to me with all your hearts, and in fasting, weeping, and mourning'" (Joel 2:12).

(b) Logical – typically occurs through the compound form and usually indicates a shift in the argument or flow of the discourse without a break in the theme. Frequently, this is also accompanied by a temporal shift as well, when one reflects on past events and commits to present or future action:

"You yourselves have committed a great sin; and <u>now</u> I will go up to YHWH, perhaps I can make atonement for your sin" (Exod 32:30),

```
הֵן הָאָדָם הָיָה כְּאַחַד מִמֶּנוּ לָדַעַת מוֹב וָרָע וְעַתָּה פֶּן־יִשְׁלַח יָדוֹ
וְלָקַח גַּם מֵעֵץ הַחיִים
```

"Indeed, humanity has become like one of us, knowing good and evil; *and now*, they might stretch out their hand and take also from the tree of life" (Gen 3:22),

"Why do you not repair the damages to the house? <u>Now, therefore</u>, do not take any money from your acquaintances" (2 Kgs 12:8 [Eng 12:7]).

## רק 4.2.15

(a) Restrictive – functions to place a limit on an idea: רָק אָהָכָם יָדַעָהִי מִכּּל מִשְׁפָּחוֹת הָאֲדָמָה, "You, <u>only</u>, have I known among all the nations of the earth" (Amos 3:2), געש הָרַע בְּעֵינֵי יְהוָה רַק לֹא כְּמַלְכֵי יִשְׂרָאֵל אֲשֶׁר הִיוּ לְפָנָיו, "He did evil in the eyes of YHWH, <u>though</u> not like the

kings of Israel who were before him" (2 Kgs 17:2), רק הָבָּמוֹת לֹא־סָרוּ עוֹד הָעָם מְזָבָחִים וּמְקַמְרִים בַּבָּמוֹת (<u>Only</u> the high places were not removed. The people still sacrificed and burned incense on the high places" (2 Kgs 14:4).

Occasionally, this restrictive idea can be expanded into a complete contrast between two ideas: אָבָרוָד אָבִין הָיָשָׁר בְּעֵינֵי יְהוָה רַק לא בְּדָוִד אָבִיו ויַשַּשׁ הַיָּשָׁר בְּעֵינֵי יְהוָה רַק לא בְּדָוִד אָבִיו, "He did right in the eyes of YHWH, <u>but</u> not like his father David" (2 Kgs 14:3).

With discourse involving instruction, the restrictive force of רָק can also indication a *clarification*: רְגָּה כָּל־אֲשֶׁר־לוֹ בְּיָדֶךְ רַק אֲלָיו אַל־הַשְׁלֵח יְדֶךְ הְנָה כָּל־אֲשֶׁר־לוֹ בְּיָדֶךְ רַק אֲלָיו אַל־הַשְׁלֵח יְדָךָ, "See, all that he has is in your hands; *only*, do not send forth your hand upon him" (Job 1:12),

אָנֹכִי אֲשַׁלַּח אֶתְכֶם . . . רַק הַרְחֵקוּ לָאֶ־תַרְחִיקוּ לָלֶכֶת

"I will let you go...<u>only</u>, do not go very far away" (Exod 8:24),

"If the woman is not willing to go with you, you will be free from this my oath. <u>*Only*</u>, you must not take my son back there" (Gen 24:8).

(b) Asseverative – expresses a conviction as to the correctness of an observation:

אָמַרְתִּי רַק אֵין־יִרְאַת אֱלֹהִים בַּמָּקוֹם הַזֶּה

"I thought, there is <u>certainly</u> [literally: I <u>only</u> thought there is] no fear of God at all in this place" (Gen 20:11), רק עַם־חָכָם וְנָבוֹן הַגוֹי הַגָּרוֹל הַזֶּה, "<u>Surely</u>, this great nation is a wise and discerning people" (Deut 4:6).

#### **4.2.16** □₩

(a) Locative – points demonstratively to a place, "there," which is the realm of action: וּבָאַת עַד־בָּבֶל שָׁם תַּנְצֵלי, "Go

to Babylon, *there* you will be rescued" (Mic 4:10),

וַיָּשֶׂם שֶׁם אֶת־הָאָדָם אֲשֶׁר יָצָר

"<u>There</u> he placed the man whom he had formed" (Gen 2:8), אָשָׁם תּאַכְלְךָ אָשׁ תַּכְרִיתֵךְ חֶרֶב consume you, the sword will cut you down" (Nah 3:15), ווַיַּעַשׁ לוֹ שָׁם סָכָה ווַיָּשָׁב תַּחְתֶּיהָ הַּצֵּל עַד אֲשֶׁר יִרְאָה מַה־יִהְיֶה הָּעִיר "He made for himself <u>there</u> a booth, and he sat under it until he could see what would happen to the city" (Jonah 4:5).

With the relative pronoun אָשֶׁר, בשָׁ functions to introduce a relative clause that points to location:

הַעֲרָכֶּל אֲשֶׁר־שָׁם הָאֱלהִים

"the thick cloud <u>where</u> God was" (Exod 20:21), ויֵלֶך לְמַסָּעָיו מְנֶגֶב וְעַד־בֵּית־אָל עַד־הַמָּקוֹם אֲשֶׁר־הָיָה שָׁם אָהֵלוֹ, "He went on his journeys, from the Negeb to Bethel, to the place <u>where</u> his tent had been" (Gen 13:3 *Qere*).

(b) Terminative – Typically with verbs of motion, and suffixed with the *he locale*, שָׁשָׁ has a *terminative* sense, placing emphasis on the end goal of movement, rather than a static location: אָמְלְשָׁה נָא שָׁמָה, "Please let me escape <u>to there</u>" (Gen 19:20), אָמְלְשָׁה נָא שָׁמָה (Gen 19:20), יְהוֹשֶׁע בִּן־נוּן הָעֹמִד לְפָנֵיך הוּא יְבֹא שָׁמָה of Nun, who stands before you; he will enter <u>there</u>" (Deut 1:38), וְיָנָסוּ שָׁמָה כָּל־הָאַנָשׁים וְהַנָּשׁים וְכֹל בַעֵלֵי הָעִיר, "All the men and women, and all the leaders of the city fled to <u>there</u>" (Judg 9:51).

קַמִיד 4.2.17

(a) Manner – most often expresses a *continual* or *persistent* manner of the action: אָשָׁוִיתִי יְהוֶה לְעֶגְדִי תְמִיד "I have set YHWH *continually* before me" (Ps 16:8), "איני תְמִיד אָל־יְהוָה, "my eyes are *ever* toward YHWH" (Ps 25:15), עַל־מָי לֹא־עָבְרָה רָשָׁתְך תְמִיד, "on whom has your evil not passed *continually*" (Nah 3:19).

## 4.3 Conjunctions

Conjunctions join words, phrases, or clauses and express relations between them. They may be of two types: coordinate conjunctions and subordinate conjunctions. The first of these conjoin grammatically equivalent nouns or clauses; in BH, these are l and אָאָמ The second type, subordinate conjunctions, conjoin a subordinate clause to the main clause, and BH has a number of examples ('בָ, (בָּר, אָם, 'בָּ, etc.). However, in BH, whether a clause is coordinate or subordinate is often interpretive because of the high frequency of the *waw consecutive* in verbal coordination (see section 3.5). Thus, subordinate clauses are frequently unmarked grammatically because the conjunctions are often imprecise.<sup>34</sup>

#### 4.3.1 الأ

*Alternative* – typically functions to provide an option between two or more substantives:

"Ask for a sign from Yhwh your God, as deep as Sheol <u>or</u> as high as heaven" (Isa 7:11), הורק היא יחידה אין־לו ממנו בן אוֹ־בָת, child; except for her, he [Jephthah] had no son <u>or</u> daughter" (Judg 11:34), האָם־עֶּבֶר יַנָּח הַשׁוֹר אוֹ אָמָה אוֹ מס gore a male slave <u>or</u> a female slave" (Exod 21:32).

#### 4.3.2 DX

(a) Conditional/contingency – introduces the protasis (the "if" statement) of a conditional statement: אָם־אָמְצָא בְסָרֹם חֲמָשִׁים צַדִּיקם, "<u>if</u> I find

<sup>&</sup>lt;sup>33</sup> At times אָם and י⊃ may serve as coordinates.

<sup>&</sup>lt;sup>34</sup> Meyer 1992, 181–82; van der Merwe, Naudé, and Kroeze 1999, 57–58 and 294–305; and for comparison with epigraphic Hebrew, see Gogel 1998, 223–30.

in Sodom fifty righteous people" (Gen 18:26), אם־בְּדְרָכִי תַּלְדְ וְאָם אֶת־מַשְׁמְרְתִּי תַשְׁמִר וְנֵם־אַתָּה תְּדִין אֶת־בֵּיתִי "If you walk in my ways, and *if* you keep my requirements, then you will govern my house" (Zech 3:7), היאָם־אָנָ יוֹם אָנָי אָם־חָכָם לְבֶּדְ יִשְׁמָת לְבֵי נֵם־אָנִי wise, my heart also will be glad" (Prov 23:15). This use is negated by the particle לא יחוילא חָפַצָּחָ בָּה , *"if* you are <u>not</u> pleased with her" (Deut 21:14), אַם־לֹא יִמְצֵא הַנַּבְּ וְנָקְרָב בַּעַל־הַבֵּיִת אֶל־הָאֶלהִים, "If the thief is <u>not</u> caught, the master of the house shall appear before the judges" (Exod 22:7 [Eng 22:8]).

(b) Concessive – אם כמח function to indicate that an action that is expected to lead to another action does not: לווח אם אם יעלוּ השָׁמִים משָׁם אוֹרִידֵם, "<u>though</u> they go up to the heavens, I will bring them down from there" (Amos 9:2), אם יעַמֹר משָׁה וּשָׁמוּאַל לְפָנֵי אֵין נַבְּשִׁי אָל־הָעָם הָוֶה (<u>Even if</u> Moses and Samuel were before me, my heart would not be with this people" (Jer 15:1).

(c) Alternative – In an interrogative clause, □X when repeated functions as an alternative, marking off substantives as options/alternates:

הַתָּבוֹא לְד שֶׁבַע שָׁנִים רָעָב בְּאַרְצֶד אִם־שְׁלשָׁה חֲדָשִׁים נָסְד לִפְנֵי־צָרֶיד וְהוּא רֹדְפֶד וְאִם־הֵיוֹת שְׁלשֶׁת יָמִים דֶּבֶר בְּאַרְצֶד

"Shall seven years of famine come to you in your land? <u>Or</u> will you flee three months before your foes while they pursue you? <u>Or</u> shall there be three days' pestilence in your land?" (2 Sam 24:13), הַלְצָרִינוּ אָם־לְצָרִינוּ "Are you for us <u>or</u> for our enemies?" (Josh 5:13), "Are you for us <u>or</u> for our enemies?" (Josh 5:13), c. הַלְבָן מֵאָה־שָׁנָה יָנָלֵר וָאָם־שָׁנָה הַלָר , "Can a child be born to a man 100 years old, <u>or</u> will Sarah, who is 90 years old, give birth?" (Gen 17:17).

(d) Exceptive – Frequently with כי סי בּלְתָּי, שָּׁם גָבְלְתָּי marks an action that will not take place unless another action

is taken: לא אַשַׁלֵחָדְ כִּי אָם־בֵּרְכְתָּנִי אוֹם הַרַכְתָּנָי go <u>unless</u> you bless me" (Gen 32:27 [Eng. 32:26]), הוואס היין היינקן כְּפִיר קוֹלוֹ מִמְעֹנְתוֹ בְּלְתִי אָם־לְכָר (Amos 3:4). Similarly, the exceptive use of אָם בערייי (Amos 3:4). Similarly, the exceptive use of שאָ expresses a situation in which there are no other alternatives or possibilities, frequently to emphasize the situation at hand: אָין זאת בְּלְתִי אָם־הֶרֶב גִּרְעוֹן בֶּן־יוֹאָשׁ אִישׁ יִשְׂרָאֵל , "This is <u>nothing less</u> than the sword of Gideon, son of Joash, a man of Israel" (Judg 7:14), האין זֶה כִּי אָם־בֵּית אֵלֹהִים is <u>nothing less</u> than the house of God" (Gen 28:17).

(e) Maledictory – Specifically in oath statements, אַם אָאַם אָז פֿאַריָקָשָרי אַבּעָקי לאָבֹקָם an oath emphatically, indicating that an action certainly will not take place: אָם־יִרְאוֹ אָת־הָאָרִין אָשֶׁר נְשָׁבַעְקִי לְאָבֹקָם, "[as I live]... they will not see the land which I swore to their ancestors" (Num 14:23), הַיָּרָר הַזֶּה אָת־הָדָרָר הַזֶּה (Num 14:23), הַיָּרָר הַזֶּה אָת־הָדָרָר הַזֶּה (Num 14:23), הַיָּרָר הַזֶּה אָת־הָדָרָר, "By your life, and by the life of your soul, *I will not do* this thing!" (2 Sam 11:11), הַרָּירְהָנָה אָם־יִהָּנָה אַם־יִהָּנָה אַם־יִהָּנָה אַם־יִהָּנָה אַם־יִהָּנָה אַם־יִהָרָה (1 Sam 19:6).

(f) Oath – Contrasted with the maledictory use of אָם אָ, it may be paired with י⊃ or לא to introduce a positive oath, an action that one has committed oneself to take: מָשְׁבָע יְהוֶה צְּבָאוֹת בְּנֵפְשׁוֹ כִּי אָם־מֵלֵאתִיך אָדָם, "Yhwh of hosts swears by himself, 'I will <u>surely</u> fill you with people'" (Jer 51:14),

אִם־לא הָאָרֶץ אֲשֶׁר דָּרְכָה רַגְלְדְ בָּה לְדְ תִהְיֶה לְנַחֵלָה וּלְבָנֶידְ עַר־עוֹלָם

"<u>Surely</u> the land that your foot has walked upon will be an inheritance to you and your children to eternity" (Josh 14:9), אָם־לֹא יָסָחָבוּם צְעִירִי הַצֹּאן אָם־לֹא יִסָחָבוּם צְעִירִי הַצֹּאן אָם־לֹא יִסָחָבוּם צָעִירִי הַצֹּאן אַם־לֹא יִסָחָבוּם צַעִירִי הַצָּאן אַם־לֹא יִסָּקַבוּם צַעִירִי הַצָּאן אַם־לֹא יִסָּקַבוּם בַּנַהָּם בּוּהַעָּשִים עַלִיהָם נוּהָשּׁים אַבּיקַים בּוּהַם צַעִירִי הַצָּאן אַם־לֹא יִסָּקַבוּם צַעִירִי הַצָּאן אַם־לַא וּשׁים עַלִיהָם בּוּנָקּם אַנוּקּש

(g) Interrogatory – אָם will often introduce a question: אָם־תַּתַּן עֵרְבוֹן עֵר שֶׁלְחָדְ אָם־תַּתַּן עֵרְבוֹן עֵר שֶׁלְחָדָ (Gen 38:17), אָם באָת אָרֹנִי הַמָּלֶדְ נְהְיָה הַדְּבָר הַזֶּה (Gen 38:17), אָם באָת אָרֹנִי הַמָּלֶדְ נְהְיָה הַדְּבָר הַזֶּה (Has this thing been done by my lord, the king?" (1 Kgs 1:27), אַם־יָתָּכָע שוֹפָר בְּעִיר וְעָם לֹא יֶחֲרָדוּ, "If a trumpet blows in the city, <u>will</u> not the people tremble?" (Amos 3:6).

## 4·3·3 ]

This conjunction functions on two levels, to conjoin nouns and to conjoin clauses. The *waw* conjoined to a finite verbal form can have unique functions (see section 3.5).

(a) Adversative – often introduces contrasting or antithetical ideas: נְחַמְתִי כֵּי עֲשִׁיתִם וְנֹחַ מְצָא חֵן בְּעֵינֵי יְהוָה. "I am sorry I have made them.' <u>But</u> Noah found favor in the eyes of YHWH" (Gen 6:7–8),

וּנְתַתִּיו לְגוֹי גָּדוֹל וְאֶת־בְּרִיתִי אָקִים אֶת־יִצְחָק

"and I will make him a great nation; <u>but</u> my covenant I will establish with Isaac" (Gen 17:20–21), רְאָה נָא אָנֹכִי יוֹשֵׁב בְּבִית אָרָזִים וַאָּרוֹן הָאֶלֹהִים ישֵׁב בְּתוֹך הַזְרִיעָה "See now, I live in a house of cedar, <u>but</u> the ark of God dwells within a tent" (2 Sam 7:2).

(b) Conjunctive – describes two or more interrelated concepts or situations that may otherwise not be related: (קנה שָׁמִים וָאָרֶץ, "Maker of heaven <u>and</u> earth" (Gen 14:19), "הְנְיֵי גְרָשׁ מְפָּיֶדְ אֶת־הָאָמְרִי וְהַכְּנְאָרֵי וְהַחְתִי וְהַקּרְיִי וְהַחְנִי וְהַיְבוּאָרֵי (See, I am going to drive out from before you the Amorite, <u>and</u> the Canaanite, <u>and</u> the Perizzite, <u>and</u> the Hittite, <u>and</u> the Jebusite" (Exod 34:11), "Sean son of Kareah <u>and</u> all the commanders of the armies took the entire remnant of Judah" (Jer 43:5).

(c) Alternative – marks off alternatives or options: וְלָא אוֹשִׁיעֵם בְּקָשֶׁת וּבְחָרֶב וּבְמָלְחָמָה בְּסוּסִים וּבְפָרָשִׁים, "I will not save them by bow <u>or</u> by sword <u>or</u> by war <u>or</u> by horses <u>or</u> by horsemen" (Hos 1:7), וְלֹא יְדָע בְּשָׁכְבָה וּבְקוּמָה וּבְקוּמָה, "He did not know when she lay down <u>or</u> when she arose" (Gen 19:33), הִכָּה דְוִד אֶת־הָאָרֵץ וְלֹא יְחֵיֶה אִישׁ וְאָשָׁה (I Sam and he did not leave man <u>or</u> woman alive" (1 Sam 27:9).

When the conjunction is repeated with each alternative, it is typically rendered "whether...or": אָנְכִי אֲחַשֶּׁנָה מִיָּדִי תְּבַקְשֶׁנָה נְגָבְתִי יוֹם וּגְנֵבְתִי לְיְלָה , "I bore the loss myself; from my hand you required it, <u>whether</u> stolen by day <u>or</u> stolen by night" (Gen 31:39).

(d) Epexegetical – introduces a clause or phrase that clarifies, expands, or paraphrases the clause that precedes it: אָבָל אָשָׁה־אַלְמָנָה אָנִי וַיָּמָת אִישׁ, "Alas, I am a widow, <u>that is</u>, my husband died" (2 Sam 14:5), a widow, <u>that is</u>, my husband died" (2 Sam 14:5), ועַתָּה שָׁמַע יַעַקְב עַבְהִי וְיִשָּׁר אַל בָּחַרְתִּי בוֹ Jacob my servant, [<u>that is</u>] Israel, whom I have chosen" (Isa 44:1), כל אַשֶׁר־דְּבֶר יְהוָה נַעֲשָׁה וְנִשְׁמָע יַעַקְב עַבְהִי וְיִשָּׁר אַל בָּחַרְתִי בוֹ (Isa 44:1), כל אַשֶׁר־דְּבֶר יְהוָה נַעֲשָׁה וְנִשְׁמָע יָנָקַר בַּרָהִי וָשָׁרָאַל בָּחַרְתִי בוֹ has spoken, we will do, <u>that is</u>, we will obey" (Exod 24:7), נַיַיִּלְקוּט shepherd's bag, <u>that is</u>, in the pouch" (1 Sam 17:40).

(e) Circumstantial – details the circumstances under which a certain action takes place (see section 5.2.11): אָלָאָךְ הָאֶלֹהִים עוֹד אָל־הָאָשָׁה וְהִיא יוֹשֶׁבֶת בַּשֶׂרֶה ", "so the messenger of God came again to the woman <u>while</u> she was sitting in the field" (Judg 13:9), <u>שׁׁחוֹפ</u> אָר וּמַגְדָל וְרֹאשׁוֹ בַשְׁמֵיִם , "Come, let us build ourselves a city and a tower, <u>with</u> its top in the heavens" (Gen 11:4).

(f) Conditional – introduces the apodosis of a conditional sentence: בְּבֹאָה רַגְלֵיְךָ הָעִירָה וּמֵת הַיָּלֶר, "When your feet enter the city, <u>then</u> the child will die" (1 Kgs 14:12),

אָם־כּה 'אַמַר נְקָדִים יִהְיֶה שְׁכָרֶךְ וְיָלְדוּ כָּל־הַצֹּאַן נְקָדִים, "If he said 'The speckled shall be your wages,' <u>then</u> all the flock brought forth speckled" (Gen 31:8).

(g) Hendiadys – The waw conjunctive can function to conjoin two or more words into a construction that refers to a single idea, or points to a single referent. Thus, the conjoined words take the place of a single word with modifiers. This expression, called hendiadys,<sup>35</sup> can be constructed with two or more nouns, or two or more verbs. *Nominal* hendiadys is a common label for the former. The latter construction, however, which is labeled here as *verbal* hendiadys, is sometimes categorized as an adverbial use of the finite verbal forms of certain roots.<sup>36</sup> Verbal hendiadys, however, may also be considered a nuance of the verbal complement function of the infinitive construct (see section 3.4.1,a3).

(g.1) Nominal hendiadys – אָאָמָת הַאָּמָכָם הָסָר וָאָמָת יַעַשּׂ־יְהוָה עַפְּכָם הָסָר וָאָמָת "Now, may Yhwh show you <u>true faithfulness</u>" (2 Sam 2:6), האָאַל הַגָּאָמָן שׁמֵר הַבְּרִית וְהַחָסָר, "The faithful God, who keeps his <u>covenant loyalty</u>" (Deut 7:9), הָאָר קוֹבְהוּ הָהוּ נָבֹהוּ earth was a <u>formless void</u>" (Gen 1:2).

(g.2) Verbal hendiadys – one verb functioning adverbially to modify the idea of the other verb. The aspect of the verbs can be identical: וְּמָהֵר וַתְּרֶד מֵעֵל הַחְמָהָר (אַרֶד מֵעָל הַחְמָהָר (Gen 24:18), "She <u>quickly lowered</u> her jar" (Gen 24:18), וְּמָהֵר וַתַּרֶד מֵעֵל הַחְמוֹר, "She <u>quickly dismounted</u> from the donkey" (1 Sam 25:23), "אָרָבָּר וְהַרָד מֵעֵל הַחָמוֹר, "Also, the people of Israel <u>wept again</u>" (Num 11:4); or the two verbal forms can also be of distinct aspect: וְשָׁמַעָהָ בְּקוֹל יְהוָה אָמָה הָשׁוּב וְשָׁמַעָהָ בְּקוֹל יְהוָה ), "And you shall <u>again obey</u> the voice of YHWH" (Deut 30:8). The root "כּרָ

<sup>36</sup> See Kautzsch 1910, 386; Joüon and Muraoka 1993, 650.

<sup>&</sup>lt;sup>35</sup> The term "hendiadys" derives from a Latin modification of the Greek words *hen* ("one") *dia* ("through") and *dyoin* ("two").

infinitive construct: אָלא־הּסִיף עוֹד מֶלֶך מִצְרִים לְצֵאת מֵאַרְצוֹ. "The king of Egypt <u>did not come out of his land again</u>" (2 Kgs 24:7), לְכֵן לא־אוֹסִיך לְהוֹשִׁיעַ אֶתְכֶם, "Therefore, I will not <u>deliver you again</u>" (Judg 10:13).

## **4·3·4** '⊃

(a) Causal – This conjunction<sup>37</sup> forms a causal link between two clauses, introducing the reason an situation takes place, or providing action or the motivation for why something should be done: יקט מָפּנֵיהֶם הָאָרֵץ הַאָלָאָה הָאָרֵץ הַמָּס מָפּנֵיהֶם, "the end of come before because all flesh has me the earth is filled with violence because of them" (Gen 6:13), אָהָה לַיוֹם כִּי קָרוֹב יוֹם יְהוַה, "Alas for the day! For the day of YHWH is near" (Joel 1:15), הָנֵעַת אַפֵּיך תֹאכַל לֵחֵם עַר שׁוּבִך אֵל־הָאָדָמָה כִּי מִמֵנָּה לָקָחָתָ, "By the sweat of your brow, you shall eat food until you return to the earth; because you were taken from it" (Gen (בָּקָרָבְּכָם, 2:19), לא תַעֵלוּ וְלא־תְלָחֲמוּ כִּי אֶינֵנִי בָּקָרָבְּכָם, "Do not go up, or fight, because I am not among you" (Deut 1:42).

(b) Evidential <sup>38</sup> – Although translated similarly to the *causal*, the evidential use of っ presents the evidence or motivation that lies behind a statement, rather than presenting the cause of an action or situation. Thus, the causal link is with the action of speech, not the contents of speech; the focus is not on what is spoken but on the reason the speaker is saying something. As such a link, っ provides the

<sup>37</sup> This particle '⊃ is thought to have originated as a demonstrative particle (Kautzsch 1910, 305; Meyer 1992, 436; Brockleman 1956, 151). The various uses of the particle may be classified into two major groups: either some type of emphatic use (Muilenberg 1961, 135–60, with the most extensive treatment), or as a connective (Aejmelaeus 1986, 205). However, Muraoka states that the particle is basically demonstrative, with the emphatic nuance derived secondarily (1969, 132).

<sup>38</sup> Claasen 1983, 37–44; Aejmelaeus 1986, 203.

evidence for the whole or a specific part of a statement, a particular word in the statement, or even the emotional tone of the statement:

"David said to Abner, 'Are you not a man? Who is like you in Israel? Why did you not guard your lord, the king? *For* one of the people came to kill the king, your lord.' " (1 Sam 26:15),

"Nathan said, 'My lord, the king. Did you say "Adonijah will rule after me, and he shall sit on my throne?" <u>For</u> he went down today, and he sacrificed oxen, fatted cattle and sheep in abundance'" (1 Kgs 1:24-25).

(c) *Clarification* – introduces a subordinate clause that clarifies or explains the main clause:

זֶה־לְּדָ הָאוֹת כִּי אָנֹכִי שְׁלַחְתִּידָ

"this will be the sign <u>that</u> I have sent you" (Exod 3:12), הַמְעַם כִּי הָאֶרִץ וְבַת חָלָב וּרְבַשׁ you brought us out from a land flowing with milk and honey?" (Num 16:13).

(*d*) **Result** – introduces clauses that express the outcome or consequence of the action or situation of the main clause: וְנָם־פּה לֹא־עָשׁיתִי מְאוּמָה כִּי־שָׁמוּ אֹתִי בַּבּוֹר, "Even here, I have not done anything, <u>that</u> they should put me into the dungeon" (Gen 40:15). This use of the particle is prevalent after questions: מָה רָאִיתָ כִּי שָׁשִׁיתָ אֶת־הַדְּכָר הַזֶּה. "What have you seen, <u>that</u> you have done this thing?"

(Gen 20:10), מִי אָל־פָּרְעֹה, "Who am I <u>that</u> I should go to Pharaoh?" (Exod 3:11).

(e) **Temporal** – indicates that the time reference of a subordinate clause is contemporary with the situation of the main clause (see section 5.2.4):

וִיְהֵי כִּי־הֶחֵל הָאָדָם לָרֹב עַל־פְּגֵי הָאָדָמָה וּבְנוֹת יֵלְדוּ לָהֶם וַיִּרְאוּ בְגִי־הָאֵלהִים אֶת־בִּנוֹת הָאָדָם כִּי טֹבת הֵנָּה

"<u>When</u> people began to multiply on the face of the earth and daughters were born to them, the sons of God saw that the daughters of humanity were beautiful" (Gen 6:1-2), the daughters of humanity were beautiful" (Gen 6:1-2), center of the daughters of humanity were beautiful", (Gen 6:1-2), center of the daughters of the daughter of the daughters of the daughter of the daughters of the daughters of the daughter of the daughters of the daughter of the da

When dealing with future situations, the temporal use is quite similar to the conditional use of כָּי See (f), which follows.

(f) Conditional – introduces the protasis of conditional statements – the "if" statement: פִּי־תְמְצָא אִישׁ לֹא תְבָרְכֶנּוּ (2 Kgs 4:29), בִּי־תְמְצָא שָׁנִים יַעֲבֹר (2 Kgs 4:29), פִי תַקְנָה עֶבֶר עִבְרִי שֵׁשׁ שָׁנִים יַעֲבֹר, "If you buy a Hebrew slave, he shall serve six years" (Exod 21:2), Hebrew slave, he shall serve six years" (Exod 21:2), "If a person opens a pit, or if a person digs a pit and does not cover it" (Exod 21:33), מָרִיקְנָה יָשׁ־לִי תְקָנָה יָשׁ־לִי חָקָנָה יָשׁרִי יָשׁרָי (Ruth 1:12).

The distinction between temporal and conditional is somewhat vague, especially when dealing with future temporal statements.<sup>39</sup> Consider: נאמרים לבנף could be translated as "*If* your son asks you in time saying...then you shall answer your son" or "*When* your sons asks you in time saying... then you shall answer your son" (Deut 6:20–21).

With the particles עָּהָת סַ עָּהָג, יאָ introduces the apodosis – the "then" statement – of a conditional sentence: הי שָׁבְנוּ זֶה פַּעֲמָים, "For if we had not delayed, [<u>then</u>] we could have returned twice" (Gen 43:10), delayed, [<u>then</u>] we could have returned twice" (Gen 43:10), חִי הָאֶלהִים כִּי לוּלֵא דְּבַרְתָ כִי אָו מֵהַבֹּקָר נַעֲלָה הָעָם lives, if you had not spoken, <u>then</u> the people would have gone away in the morning" (2 Sam 2:27).

(g) Adversative – introduces an antithetical statement after a negative clause. It often gives an alternative for the negative statement: אָלא־תַקָרָא אֶת־שָׁמָה שָׁרִי כִּי שָׁרָה שָׁמָה "Do not call her name Sarai, <u>but</u> her name will be Sarah" (Gen 17:15), לא־תַגַּע בּוֹ יָד כִּי־סָקוֹל יִסְקַל אוֹ־יָרָה יִיָּרָה "No hand shall touch him, <u>but</u> he shall surely be stoned or shot through" (Exod 19:13), לא־אֶכָּה אֶת־כָּל־הַמַּמְלָכָה מִיָּדוֹ כִּי נָשִׁיא אֲשׁתֵנוּ כֹּל יְמִי חַיָּיו "I will not take the kingdom from his hand, <u>but</u> I will make him ruler all the days of his life" (1 Kgs 11:34).

The *adversative* sense is also denoted by בָּי אָם again, noting an alternative to a negative statement:<sup>40</sup> לא יִיָרָשֶׁך וֶה כִּי־אָם אֲשֶׁר יְצֵא מְמֵעֶ'ך withis man will not be your heir, <u>but</u> one who will come forth from your body..." (Gen 15:4), לא יִעַקֹב יָאָמֵר עוֹד שָׁמָך כֵּי אָם־יִשְׂרָאֵל Your name shall no longer be Jacob, <u>but rather</u> Israel" (Gen 32:29 [Eng. 32:28]).

(h) Concessive – introduces a clause that should, or is expected to, lead to the action of the main clause but, in fact, does not. Thus, the main clause occurs in spite of the clause introduced by כָּייִפּל לֹא־יוּטָל: כָּי

<sup>&</sup>lt;sup>40</sup> The longer construction Cギ メコ is more common in denoting an adversative sense. The adversative force of this construction likely comes from the adversative use of で, which is further enhanced by a pleonastic use of Choors 1981, 251–52).

"<u>though</u> he falls, he will not stumble" (Ps 37:24), הי אָנָפְק דִי יָשׁׁב אָפָּך וּתְנָחֲמֵנִי (Ps 37:24), ה, you were angry with me, your anger turned away and you comforted me" (Isa 12:1), מַאַתוֹת הַשְׁמֵים אַל־תַחָתוּ כִּי־יַחַתוּ הַגּוֹיִם מַהַמָּה, be terrified by the signs of heaven, <u>though</u> the nations are terrified by them" (Jer 10:2).

(*i*) Asseverative – emphasizes the clause it modifies. This use of כָּי may have originated in oath statements, to indicate an action that one will take:<sup>41</sup> הייהוָה כִּי בֶן־מָוֶת הָאִישׁ הָעשֶׁה וֹאַת, "As Yhwh lives, <u>surely</u> the man who did this thing deserves to die" (2 Sam 12:5), "May God do this and more; <u>You shall surely die</u>, Jonathan" (1 Sam 14:44), בּוֹי בַּרְעָה כָּי בְרָעָה כָּי מָרָגָלִים אָהָם, "As Pharaoh lives, you are <u>surely</u> spies" (Gen 42:16).

The construction  $\Box \aleph$   $\Im$  highlights a negative oath, an action that one will not take, or an action/situation that one pledges to not allow:

ואוּלַם חַי־יִהוָה אֵלֹהֵי יִשָּׂרָאֵל ... כִּי אָם־נוֹתַר לְנָבָל

"Nevertheless, as YHWH, the God of Israel lives . . . <u>there would not have been</u> left to Nabal . . ." (1 Sam 25:34). On occasion, however, the construction כָּיאָם־רַצָּתִי אָםרִין וְלָלְקָחְתִי מָאָתּוֹ מְאוֹמָה tive י⊃ to indicate an action that one will take: דייהָוָה כִּי־אָם־רַצָּתִי אָחַרִין וְלָלְקָחְתִי מָאָתּוֹ מְאוֹמָה lives, <u>I will run</u> after him and take something from him" (2 Kgs 5:20), אָם־הַעֲשׁוֹן כָּזֹאָת כִּי אָם־נִקַמְתִי בְּכֶם, "Since you act like this, <u>I will surely take revenge</u> on you" (Judg 15:7).

The asseverative כָּי can function in other types of discourse as well, to place emphasis on the surety of a fact or situation: כָּי יִבֹשׁוּ מָאַילִים אָשֶׁר חַמִדְהָם, "<u>surely</u> they will be ashamed of oaks that they have desired" (Isa 1:29), אם לא תַאָמִינוּ כִּי לא הָאָמַנוּ כִי לא הָאָמַנוּ

will not last" (Isa 7:9), כָּי אָם־יָשׁ אָחַרית "*Surely* there is a future" (Prov 23:18).

(j) Perceptual – With verbs of perception (seeing, hearing, believing, feeling, etc) 'כ can function to introduce a subordinate clause that indicates the object of perception: רְיָהָא אֶלהִים כִּי־מוֹב, "God saw <u>that</u> [this] was good" (Gen 1:10), ויַרָא אָיָבָל כִּי־סָכֵּל נְבוֹת וַיָּמֹת, "When Jezebel heard <u>that</u> Naboth was stoned and died" (1 Kgs 21:15), "You shall know <u>that</u> I am YHWH, your God" (Exod 6:7).

(k) Subject – introduces clauses that function as a subject in a sentence: שוֹב כִּי־תִהְיָה־לְנוּ מֵעִיר לֵעְזוֹר, "It is better <u>that you send us</u> help from the city" (2 Sam 18:3 Qere), אַל־יָחָר בְּעֵינֵי אֲרֹנִי כִּי לוֹא אוּכַל לָקוּם מִפְּנֵיךָ, tet not my lord be angry <u>that I am not able to rise</u> before you" (Gen 31:35).

(*l*) *Recitative* – introduces direct speech, and is therefore usually left untranslated:

וַיּאֶמֶר חֲזָהאֵל כִּי מָה עַבְדְּדָ הַכֶּלֶב כִּי יַאֲשֶׂה הַדָּבָר הַגָּדוֹל הַזֵּה

"Hazael said <<u>that</u>>, 'What is your servant, a dog, that he should do this great thing?'" (2 Kgs 8:13),

"Surely, as I swore to you by Yhwh, God of Israel, saying <<u>that</u>>, 'Solomon, your son, will be king after me'" (1 Kgs 1:30), כן אַמְרָהֶם לֵאמֹר כִּי־פְּשָׁעֵינוּ וְחַמֹאתֵינוּ עָלֵינוּ, "Thus, you have said <<u>that</u>>, 'Our trangressions and our sins are upon us'" (Ezek 33:10).

With the interrogative particle כָּי, יָס can introduce a question posed by a speaker in context: אַמָר דָּוִד הֵכִי יָשׁ־עוֹד אֲשָׁר נוֹתַר לְבֵית שָׁאוּל, "David said <that>, 'Is there anyone still left of the house of Saul?'"(2 Sam 9:1), ויָאָקר הָרִי קָרָא שְׁמוֹ יַשְׁקֹב , "Then [Esau] said<that>, 'Is he not rightly named Jacob?'" (Gen 27:36).

(m) Exceptive – indicates an action or situation that will not take place unless accompanied or followed by another action/situation. The exceptive " is often translated "except" or "unless" in English: וִלֹא יֹאכַל מִן־הַקֵּדַשִׁים כִּי אָם־רַחַץ בִּשָׂרוֹ בַמַיָם, "And will not eat from the holy [gifts] he unless he washes his body with water" (Lev 22:6), אָין־חֵפָץ לַמֶּלֶך בְּמֹהַר כִּי בְּמֵאָה עָרְלוֹת פָּלִשְׁתִּים, "The king does not desire any dowry, except for 100 foreskins of the Philistines" (1 Sam 18:25).

After a general negative statement, אָם יָסָ often introduces a specific situation that is opposed to, or reverses, the negative statement:<sup>42</sup> לא־תְרָאָה אֶת־לְכָנִי הֲבִיאָךְ אָת מִיכָל, "You shall not see my face, <u>unless</u> you bring Michal" (2 Sam 3:13), not see my face, <u>unless</u> you bring Michal" (2 Sam 3:13), נְהַג וְלָךָ אַל־תַּעֲצָר־לִי לְרְכָב כִּי אָם־אָמָרְתִי לְךָ forward; do not slow down the pace for me <u>unless</u> I tell you" (2 Kgs 4:24).

(*n*) Interrogative – Occasionally, יכָּ will function to introduce an interrogative clause: כָּי הָאָרָם עֵץ הַשָּׁרָם עֵץ, "Are the trees of the field as human beings?" (Deut 20:19), כּי־הָצִילוּ אָת־שׁמְרוֹן מִיָדִ, "Have they delivered Samaria from my hand?" (2 Kgs 18:34).

## 4.3.5 ]₽

*Consequential* – This conjunction most often indicates an undesirable action or situation that arises from another action as a consquence: לא תאכלו מַמָּנוּ וְלא תְגְעוּ בּוֹ פֶּן־תְּמָתוּן, "You shall not eat from it or touch it, <u>lest</u> you die"

(Gen 3:3), אַל־תַּשְׁמַע קוֹלְך עִמְנוּ פָּן־יִפְּגְעוּ בָכֶם אֲנָשִׁים מְרִי נָפָשׁ, "Do not let your voice be heard among us, <u>or else</u> fierce men will attack you" (Judg 18:25), אָל־תּוֹכָח לֵץ פּן־יִשְׁנָאָ "Do not reprove a scoffer, <u>lest</u> he hate you" (Prov 9:8).

#### 4.4 Particles of Existence/Nonexistence

Hebrew uses the two words אין and מאין to denote existence or nonexistence.  $^{43}$ 

#### 4.4.1 <u>}</u>

(a) Nonexistence – denies the existence of a substantive, in which case it usually occurs in apposition with a preceding noun: אָרֶם אָין לַעֲבֹר אָת־הָאָרָמָה, "<u>there was no</u> human to cultivate the ground" (Gen 2:5). However, it may also occur in the construct state with a following noun or pronoun: אָרֶם בָּמַקוֹם הָאָה "<u>There is no</u> fear of God in this place" (Gen 20:11).

(*b*) *Nonpossession* – The particle is also used with the ל preposition to show that the object of the preposition does not possess a substantive: אָין לָה וָלָד (Gen 11:30), אָין לָאִישׁ גָּאָין לָה וָלָד וָאָם־אָין לָאִישׁ גָּאַל (Mum 5:8).

(c) Negative – The particle can also negate verbal clauses when bound with a pronominal suffix. In this construction, the bound pronominal suffix is the subject of the verbal action that is being negated, usually a participle: אינוי נכן לָכָם הָבָן, "<u>I am not giving</u> you straw" (Exod 5:10), אינוי נכתן, "<u>I will not listen</u>" (Isa 1:15).

<sup>&</sup>lt;sup>43</sup> Joüon and Muraoka 1993, 576 and 604–05; Williams 1976, 67–68 and 79; van der Merwe, Naudé, and Kroeze 1999, 320–21; Seow 1995, 107–08.

#### 4.4.2 2

(*a*) Existence – Antithetical to אַין, the particle שׁיֵ affirms the existence of a substantive. The particle can function in the same way as אָכן ישׁ in modifying a nominal clause: אָכן ישׁ יְהוָה בָּמָקוֹם הָוָה, "Surely, YHWH is in this place" (Gen 28:16). With the pronominal suffixes, it indicates the subject of a verbless clause: אָר יָשׁנוֹ פֿה, "with <u>the one who is</u> here" (Deut 29:14 [Eng 29:15]).

(b) Possession – With the preposition ל bound to a pronominal suffix, שָׁ expresses possession, with the pronominal suffix indicating the owner: מָה־יָּשׁ־לֶך, "What <u>do you have</u>?" (2 Kgs 4:2 *Qere*), גָשׁ־לִי רָב, "<u>I have</u> enough" (Gen 33:9).

(c) Predicate – When bound to a pronominal suffix, שָׁיַ can function to express the subject of a participle. This usage is normally preceded by אָם אָם to expresses intention or desire: אָם־יָשָׁרָ מְשַׁלֵח אָת־אָם", "If you will send our brother with us" (Gen 43:4), אָם־יָשָׁכָם עֹשִׁים הָסָר , (עַקּה אָם־יָשָׁכָם עֹשִׁים הָסָר , Now then, if you will deal loyally" (Gen 24:49).

#### 4.5 The Particles הַנָּה and הַנָּה

Though traditionally rendered in English as "Behold" or "Lo," neither הבה חס הבה is limited to a demonstrative function. In fact, the demonstrative nature of either particle is not even the primary aspect. Their primary function, rather, is as a "particle of interest,"<sup>44</sup> which calls special attention to an element of the context, either a single word or an entire statement. As the particle calls attention to an object, it also indicates a shift in perspective within a narrative. In essence, the particle introduces movement away from the narratival perspective to present vividly the perspective of a particular speaker or figure within the narrative. Consequently, הבה with introductory or

<sup>&</sup>lt;sup>44</sup> Waltke and O'Connor 1990, 300; Pratico and Van Pelt 2001, 148-49.

transitional signals within a dialogue, as the unique view of each speaker is laid out (see "*perception*" in section 4.5.2).<sup>45</sup> Furthermore, the particle has a strong "overtone of feeling"<sup>46</sup> based on its nature as an exclamation. Thus, there is no single rendering in English that adequately captures the force of הַבָּה or הַבָּה, since the specific use of either particle is dependent not only on textual context but also on the emotional mindset of the speaker and the other figures of the narrative as well.

#### **4.5.1** آ<u>أ</u><u>أ</u>

The particle without the *waw* conjunction is unique in that it is usually used with the verb אמר <sup>47</sup>. Hence, it serves to introduce direct speech, much like the recitative function of the particle בי (see section 4.3.4,1). Within this use, there are a variety of nuances that the particle המר can express:

(a) Exclamatory – functions quite frequently as a "presentative exclamation,"<sup>48</sup> emphasizing "immediacy" or "hereand-now-ness,"<sup>49</sup> often as a response to a summons. This exclamatory emphasis is often related to the presence of people or certain objects, calling attention to and focusing upon an object:<sup>50</sup> הְנָנִי שֶׁלְחֵנִי, "<u>Here I am</u>, send me!" (Isa 6:8), הְנָה הָאָשׁ וְהָעֵצִים וְאַיֵּה הָשָׁה לְעֹלָה (Gen 22:7), יָאָמֶר הַנֵּנִי, "אַכֶּר הָנֵיַי, "God] said to him 'Abraham,' and [Abraham] said '<u>Here I am</u>." (Gen 22:1).

(b) Immediacy – Related to the exclamatory, הנה used with verbs or participles can point to the immediacy of the action of the verb or participle: הנה אָרִיך חֹלָה, "Your father is

- <sup>45</sup> Schneider 1989, 261–68.
- <sup>46</sup> McCarthy 1980, 331.
- 47 Zewi 1996, 21-37.
- <sup>48</sup> Waltke and O'Connor 1990, 674.
- 49 Lambdin 1971a, 168.
- <sup>50</sup> Berlin 1983, 91. See also Andersen 1974, 94.

<u>now</u>sick" (Gen 48:1), הַנָּה מִבְנֵי יִשְׂרָאֵל, מִבְנֵי הַנָּיְלָה מִבְנֵי יִשְׂרָאֵל, "Some men from the people of Israel have come here <u>just</u> tonight" (Josh 2:2).

4.5.2

The form of the particle with the conjunction occurs quite often with verbs of sight, or related contexts, typically introducing the object of perception.<sup>51</sup>

(a) Immediate perception – In some narrative situations, וְהָבָה points not to an immediate action but to the immediate perception of an action or the results of an action. Thus, it functions to indicate "suddenness in the presentation of perception," not a sense of immediacy or suddenness in the events of the narrative: גער הַבָּרָלָה וְיָהָרִד הָאִישׁ וּיָלָפָת וְהְבָּה אָשָׁה שֹׁכֶבָת מַרְגָלֹתְיו, "It happened in the middle of the night that the man was startled and bent forward; and <u>behold</u> there was a woman lying at his feet" (Ruth 3:8), וּבְעָז הָלָיָלָה הָשַׁעָר וְיָשֶׁב שֶׁם וְהָבָה הַגֹּאָל עָבָר var now Boaz went to the gate and sat there, <u>and the kinsmen-redeemer was passing by</u>" (Ruth 4:1).<sup>52</sup>

(b) **Perception**–With verbs of perception or within a perceptual situation, און יהנה will introduce a perception from a perspective that is distinct from the narrator or other figures in the narrative<sup>53</sup> and unique to the speaker: what may be termed the "participant perspective."<sup>54</sup> Unlike "immediate

<sup>51</sup> Zewi 1996, 37.

<sup>52</sup> Berlin 1983, 91. Berlin notes that in the two verses, the emphasis of suddenness is not placed on the action of Ruth lying down or the redeemer passing by. Rather, the emphasis is on the fact that, at a certain moment, Boaz noticed these actions occurring. Thus, the particle indicates "Boaz's perception as he becomes aware of Ruth's presence" and points out that "Boaz suddenly saw the *goel*, and not that the *goel* arrived immediately."

<sup>53</sup> Berlin 1983, 60.

<sup>54</sup> Andersen 1974, 94.

perception," this use of הְבָּה refers to the unique perspective of the speaker as he or she perceives an action, and not necessarily the immediate nature of the action. Based on the emotionality of the particle, the object of perception is often colored with a sense of excitement: perception is often colored with a sense of excitement: perception is often colored with a sense of excitement: perception is often colored with a sense of excitement: perception is often colored with a sense of excitement: perception is often colored with a sense of excitement: perception is often colored with a sense of excitement: perception is often colored with a sense of excitement: perception is often colored with a sense of excitement: perception is often colored with a sense of excitement: perception is often colored with a sense of excitement: perception is often colored with a sense of excitement: perception is often colored with a sense of excitement: perception is often colored with a sense of excitement: perception is often colored with a sense of excitement: perception is often colored with a sense of excitement: perception is often colored with a sense of excitement: perception is often colored with a sense of excitement: perception is often colored with a sense of excitement. perception is often colored with a sense of excitement. perception is often colored with a sense of excitement. perception is often colored with a sense of excitement. perception is often colored with a sense of excitement. perception is often colored with a sense of excitement. perception is often colored with a sense of excitement. perception is often colored with a sense of excitement. perception is often colored with a sense of excitement. perception is often colored with a sense of excitement. perception is often colored with a sense of excitement. perception is often colored with a sense of excitement. perception is often colored with a sense of excitement. perception is often colored with a sense of excitement. perception is often colo

```
וְלֹא־הֶאֶמְנְתִּי לַהְּבָרִים עַד אֲשֶׁר־בָּאתִי וַתִּרְאֶינָה עֵינַי וְהַנֵּה
לא־הַנַּר־לִי הַחֵצִי
```

"I did not believe the reports until I came and my eyes had seen *that the half was not told to me*" (1 Kgs 10:7).

(c) Logical – As it introduces perceptions, the particle also has various *logical functions*, connecting two ideas together:

(c.1) Causal – וְיָקְרָע אֶת־בְּנָדִיוֹסֵף בַּבּוֹר וַיִּקְרָע אֶת־בְּנָדִיוֹסֵף (<u>Because</u>), <u>Because</u> Joseph was not in the pit, he tore his clothes" (Gen 37:29), "Then [Aaron] ran into the midst of the assembly <u>because</u> the plague had begun among the people" (Num 17:12 [Eng. 16:47]).

(c.2) Occasion/circumstantial-

וְהַנֵּה אֲנַחְנוּ מְאַלְמִים אָּלָמִים בְּתוֹך הַשָּׂדֶה

"<u>While</u> we were binding sheaves in the field (Gen 37:7), וָיָבֹא אֶל־הָאִישׁ וְהְנֵה עֹמֵר עַל־הַגְּמֵלִים עַל־הָעָיָן, "He went to the man <u>while</u> he was standing by the camels at the spring" (Gen 24:30).

(c.3) Conditional-

וְרָאָהוּ הַכֹּהֵן בַּיוֹם הַשְּׁבִיעִי וְהַנֵּה הַנֶּגַע עָמַד בְּעֵינָיו לֹא־פָּשָׂה

"The priest will look at him on the seventh day, and *if*, in his eyes, the infection has not changed" (Lev 13:5), ווהגה גלך ומה־נָרִיא לָאִיש, f we go, what shall we take for the man?" (1 Sam 9:7).

(c.4) Temporal -

וְהִנֵּה אָנֹכִי בָא בִּקְצֵה הַמַּחֲנֶה וְהָיָה כַאֲשֶׁר־אֶעֱשֶׂה כֵּן תַּעֲשׂוּן

"When I arrive to the outskirts of town, do as I do" (Judg 7:17), אָיָרְהָאָל וְהַנֵּה אָנֹכִי יֹרֵד אָלֶיך (Go down before me to Gilgal, and [<u>then</u>] I will go down to you" (1 Sam 10:8).

(c.5) Result – הַן לִי לֹא נָתַתָּה זָרַע וְהְנֵה בֶן־בֵּיתִי יוֹרֵשׁ אֹתִי, "You have given me no offspring, <u>so</u> one born in my house is my heir" (Gen 15:3), קרְתָה בְרִיתְדָ אָתִי וְהַנֵּה יְדִי עַבָּזְדְ לְהָסֵב אֵלֶיךָ אֶת־כָּל־יִשְׂרָאֵל "Make your covenant with me, <u>so that</u> my hand will be with you to bring to you all of Israel" (2 Sam 3:12).

(c.6) Adversative -

```
ּכִּי לֹא אֶחְמוֹל עוֹד עַל־יֹשְׁבֵי הָאָרֶץ נְאָם־יְהוֶה וְהִנֵּה אָנֹכִי מַמְצִיא
אֶת־הָאָדָם אִישׁ בְּיַד־רֵעֵהוּ
```

"'For I will no longer have pity on the inhabitants of the land,' declares Үнwн, '*but rather* I will cause the men to fall each into the hand of a neighbor'" (Zech 11:6),

> וַיּקֶרָא אֲדֹנָי יְהוָה צְּבָאוֹת בַּיוֹם הַהוּא לִבְכִי וּלְמִסְבֵּד וּלְקָרְחָה וְלַחֲגֹר שָׂק וְהַנֵּה שָׁשׁוֹן וְשִׁמְחָה

"The lord, YHWH of hosts, called on that day to weeping and lamenting, to shaving the head and wearing sackcloth. *Instead* there is rejoicing and merriment" (Isa 22:12–13).

# 5 Clauses and Sentences

Thus far in this book we have treated the syntax of individual words and phrases. By "phrases" we mean the function of individual words and the way they relate to each other in larger units to create noun phrases, verb phrases, adverbial phrases, and prepositional phrases. In most cases, a phrase is a series of words capable of serving the syntactical function of a single word. In this final section, we move beyond phrases to clauses and sentences.<sup>1</sup>

The very definitions of "clause" and "sentence" may be problematic, and so we begin by clarifying a few more terms. Put simply, a clause is a group of words containing a subject and only one predicate.<sup>2</sup> This distinguishes a clause from

- <sup>1</sup> As elsewhere in this grammar, we have retained traditional terminology and explanations for most grammatical functions. However, the research of a growing number of scholars employing discourse analysis (or "discourse linguistics," also known in Europe as "text linguistics"; see Lowery 1995, 107) have highlighted the inadequacy of the traditional word-oriented and sentence-oriented study of BH grammar, emphasizing instead the need to include longer units of text in linguistic study (van der Merwe 1994, 14–15; Andersen 1974, 17–20). Selected studies have been fruitful (see especially Andersen 1970 and 1974, Miller 1996 and 1999, and to some degree the grammar of van der Merwe, Naudé, and Kroeze 1999), but after thirty years of research, this burgeoning discipline lacks consensus of methodology and terminology. See further the various articles in Bodine 1995 and Bergen 1994.
- <sup>2</sup> Behind our brief and functional definitions remain unsettled debates among linguists. For rigorous definitions and explanations of how these

a phrase, which makes no specific grammatical predication. Having clarified clause and phrase, we need to qualify the statement with which we began this unit, "Thus far in this book we have treated the syntax of individual words and phrases." Exceptions to this statement exist in our treatment of the finite verbal forms (sections 3.2 and 3.3) and verbal sequences (section 3.5). Since Hebrew finite verbs contain implied subjects (embedded by means of pronominal morphemes), the subject and predicate of a clause are both contained in the verb, making it impossible to study verbal syntax without considering clause-level relationships.<sup>3</sup> Nonetheless, our study of BH syntax prior to this point in the book has been *primarily* focused on the phrase-level relationships of Hebrew nouns, verbs, and particles.<sup>4</sup> In section 5, we now turn to clausal and interclausal relationships.

A sentence is comprised of one or more clauses and is the largest grammatical structural unit, except for the discourse or the text itself. Thus, sentences combine to create texts, which may be further analyzed in terms of text-linguistic conventions of cohesion and rhetorical organization.<sup>5</sup> So the grammatical hierarchy moves from word to phrase to clause to sentence to discourse.<sup>6</sup>

issues relate specifically to the grammar of BH, see Andersen 1974, 21–28; Joüon and Muraoka 1993, 561–64; van der Merwe, Naudé, and Kroeze 1999, 59–65; Waltke and O'Connor 1990, 77–80.

- <sup>3</sup> So, for example, כָּחַב in Joshua 8:32 is used in reference to the law of Moses, which <u>he wrote</u> in the presence of the Israelites. Since the subject "he" is implied in the verb, there is in the Hebrew syntax no noun phrase separate from the verb phrase "wrote in the presence of the Israelites." Thus, our treatment of the verbs necessarily went beyond the phrase level and considered the subjects with their verbs in clause-level relationships.
- <sup>4</sup> Thus, until this point in the book, we have limited ourselves to a consideration of morphosyntactical relationships, as much as possible. "Morphosyntax" considers the way two words relate to each other, or in some cases, complex combinations of words in phrases (Richter 1978–80, 2:3–83, esp. 4 for Richter's definition of *Wortgruppe*, "phrase").
- <sup>5</sup> Van der Merwe, Naudé, and Kroeze 1999, 21, 51–52, and 65–66.
- <sup>6</sup> Or more fully, from morpheme (the smallest unit of meaningful speech) to word to phrase to clause to sentence to discourse.

Further definition may be needed to clarify the types of sentences possible in BH. A sentence with only one clause may be termed a "simple" sentence, which may be either nominal or verbal (see section 5.1). So, for example, the following nominal clause in Psalm 121:5 could be taken separately as a simple sentence: יְהוָה שׁמְרָדְ, "Yhwh (is) your keeper." Likewise, the verbal clause in Genesis 1:1 could be isolated as a simple sentence: בָרָא אָלהים, "God created." A sentence with two or more clauses of equal syntactical status (or function) is a "compound" sentence.7 Consider the following compound sentence from 1 Samuel 3:9: וַיָּלֶך שָׁמוּאָל וַיִשְׁכָב בַמְקוֹמוֹ, "Then Samuel went and lay down in his place." The two clauses have the same subject, and are of equal syntactical status, albeit in the second clause, the subject is implied and the action is modified by a prepositional phrase. BH narrative uses the ubiquitous conjunction waw in a flexible way to link verbs in chains and sequences in such compound constructions (see section 3.5). A sentence with two or more clauses of unequal status is a "complex" sentence, in which one clause is subordinated to the other, the independent or main clause.

#### 5.1 Nominal and Verbal Clauses

Clauses may be further classified as coordinate or subordinate.<sup>8</sup> The two categories included in this section are clauses that may be coordinated or juxtaposed in apposition with other clauses to create compound sentences. Thus, the *nominal* and *verbal* clauses are the most basic types of clauses. The

<sup>7</sup> Andersen is correct in stating that this traditional definition is inadequate for encompassing the realities of BH (1974, 24–28). But the traditional definition of a compound sentence is still useful for our purposes. The reader should also be aware that grammarians have used the designation "compound sentence" in very different ways, as reference to Kautzsch and Meyer will attest (Kautzsch 1910, 457–58; Meyer 1992, 355–57).

<sup>8</sup> Joüon and Muraoka 1993, 561; Andersen 1974, 24–27

various subordinate clauses possible in BH will be covered in section 5.2.9

#### 5.1.1 Nominal Clause

BH has the ability to form a clause without the use of a finite verb.<sup>10</sup> The subject of such nominal clauses will be a noun or pronoun. Its predicate will be another noun, pronoun, prepositional phrase, adverb, or infinitive construct.<sup>11</sup>

Nominal clauses can be classified in two general categories. *Identification* clauses signify the nature or identity of the subject, while *description* clauses speak of the quality or attributes of the subject.<sup>12</sup> The former typically answers "who" or "what," while the latter describes what the subject is like. The expected word order of a nominal clause may be generally classified according to the type of nominal clause. Identification clauses, in general, are ordered subject-predicate and take a definite predicate, while description clauses are generally ordered predicate-subject and have an indefinite predicate. However, there are many exceptions to the ordering, depending on the presence of additional syntactical elements, an intended emphasis

- <sup>9</sup> Joüon and Muraoka 1993, 561–86; Meyer 1992, 348–57; Kautzsch 1910, 450–57; Williams 1976, 96–99; Chisholm 1998, 113–14; van der Merwe, Naudé, and Kroeze 1999, 59–65 and 336–50; and for epigraphic Hebrew, see Gogel 1998, 273–92.
- <sup>10</sup> Such verbless clauses were also present in the Canaanite dialect of the Amarna letters (Rainey 1996, 1:180).
- <sup>11</sup> And so more accurately, this could be called a "verbless" clause, since the predicate may be nearly anything except a verb (as per Andersen 1970, 17–30). Recently it has been doubted whether nominal clauses should be considered a distinct clausal type in BH. Rather, it is argued, the copula קָיָה was assumed in such clauses, but was optional and so frequently omitted as unnecessary. In this view, the so-called nominal or verbless clause is simply a variety of the verbal clauses without the copula included (Sinclair 1999, 75).
- <sup>12</sup> Andersen preferred to speak of "identification" and "classification" clauses (1970, 31–34).

on one of the elements, or, in some cases, the rhetorical function of the nominal clause.<sup>13</sup>

(a) Noun as predicate (see predicate nominative, section 2.1.2) – אָאָדָה הָאִישׁ, "You <u>are the man</u>" (2 Sam 12:7), "יְהוָה מָיָלִי, "Yhwh <u>is king</u>" (Ps 10:16). The predicate may be a proper noun: יְהוָה מָיָלִי, "I <u>am Yhwh</u>" (Gen 15:7), "I <u>am Joseph</u>" (Gen 45:3). Note the predicate-subject word order in *description* nominal clauses: נְבִיא הוּא <u>נְבִי</u> הָרָאָרָיָ (Gen 20:7).

Sometimes, a *pleonastic* pronoun is situated between the subject and the predicate. The pronoun is redundant since it is not necessary grammatically to complete the predication (compare יְהוֶה הָאֱלֹהִים, "Yhwh <u>is God</u>" [Josh 22:34] with יְהוָא הָאֱלֹהִים, "Yhwh <u>is God</u>" [1 Kgs 18:39]): יְהוָא הָאָלוֹהים, "Esau <u>is Edom</u>" (Gen 36:8).<sup>14</sup>

(b) Adjective as predicate (see section 2.5.2) – הַנָּה אָנַחְנוּ פֹּה בִּיהּנְדָה יְרָאִים, "Look, we <u>are afraid</u> here in Judah" (1 Sam 23:3), הַנָּה אָנַחְנוּ פֹּה בִיהּנְדָה יְרָאִים (1 Kgs 2:38), מוֹב חַדָּבָר, "The water <u>is bad</u>" (2 Kgs 2:19), "The girl <u>was</u> very <u>beautiful</u>" (1 Kgs 1:4). The adjective regularly serves as predicate in clauses of description: הָרָה אָנָרָה יָפָה מַרַיּמָאַר, "I <u>am pregnant</u>" (2 Sam 11:5), הָרָה אָנָרָא הוֹאַ, "he <u>is unclean</u>" (Lev 13:11).

(c) Participle as predicate (see section 3.4.3, b) – עֹבְרִים אָנַחְנוּ מִבִּית־לֶחֶם יְהוּדָה עָד־יַרְכְּתֵי הַר־אָפְרַיִם, "We <u>are</u> <u>passing</u> from Bethlehem of Judah to the remote part of the hill country of Ephraim" (Judg 19:18), דְּהָוֶה שׁׁמְרָד , "Yhwh <u>is your keeper</u>" (Ps 121:5), מָרִים רְעָה (Zech 11:16), הָכִי הְנֵה־אָנֹכִי מֵקִים רֹעֶה , "For I am about to <u>raise up</u> a shepherd" (Zech 11:16), מָרוּעַ פְּנֵיך רָעִים וְאַתָּה אֵינָד חוֹלֶה , though you <u>are not sick</u>?" (Neh 2:2).

<sup>&</sup>lt;sup>13</sup> See Waltke and O'Connor 1990, 130–35; Joüon and Muraoka 1993, 568– 76; Kautzsch 1910, 454; Williams 1976, 98–99.

<sup>&</sup>lt;sup>14</sup> Such pleonastic uses of the pronouns are apparently not intended to denote emphasis. See Muraoka 1985, 15.

With the passive participle, there is quite often a desiderative sense in the predication: בְּרוּך יְהוָה "<u>Blessed is</u> YHWH" or "May YHWH <u>be blessed</u>" (Ruth 4:14), אָרְרֶריָק אָרוּר וּמְבְרַכֶיך בְּרוּך, "<u>Cursed be</u> those who curse you, and <u>blessed be</u> those who bless you" (Gen 27:29).<sup>15</sup>

(d) Prepositional phrase as predicate – הָאֶלהִים בַּשְׁמִים, "God is <u>in heaven</u>" (Eccl 5:1 [Eng 5:2]), יקוֹל מִלְחָמָה בָּאָרְץ, "The noise of battle <u>is in the land</u>" (Jer 50:22), ג'יהוָה הוּא "It <u>is YHWH's</u>" (Lev 27:26), לא בְרַעַש יְהוָה, "YHWH <u>was</u> <u>not in the earthquake</u>" (1 Kgs 19:11), יְהוָה עִמְּך (Judg 6:12).

5.1.2 Verbal Clause

The subject of a verbal clause will be a noun or pronoun, and its predicate will be a finite verb (i.e., perfect or imperfect aspect), or a nonfinite verb functioning as one.

Because the verbal clause is so prevalent in BH, we offer here a survey of its forms.

#### (a) Subject of the verbal clause

(a.1) Noun as subject – The subject of a verbal clause is most often expressed by a noun: וויצו יהושע אָת־הַכֹּהְנִים, "<u>Joshua</u> commanded the priests" (Josh 4:17), וויְכָהֹב משֶׁה אָת כָּלֹדְרְבְרִי יְהוָה "<u>Moses</u> wrote down all the words of YHWH" (Exod 24:4).

(a.2) *Pronoun as subject*—When an independent personal pronoun functions as the subject of a verb, there is generally a sense of prominence, since the subject is already inherent in the verbal form.

Specifically, the pronoun as subject can be *disjunctive*, expressing a strong contrasting or disjunctive sense, so that

the subject of the verb is highlighted against another:

וּיָשֶׂם אֶת־הַשְׁפָחוֹת וְאֶת־יַלְדֵיהֶן רִאשׁנָה וְאֶת־לֵאָה וִילָדֶיהָ אַחֲרֹנִים וְאֶת־רָחֵל וְאֶת־יוֹסֵף אַחֲרֹנִים וְהוּא עָבַר לִפְּנֵיהֶם

"He put the maids and their children in front, and Leah and her children next, and Rachel and Joseph next. <u>But he</u> passed on before them..." (Gen 33:2–3).

The pronoun as subject can also be *emphatic*, highlighting the subject: הוא ישופון אַהָה הְשׁוּפְנוּ שָׁקָה, "<u>He</u> shall bruise your head, and <u>you shall bruise</u> his heel" (Gen 3:15), הוא יַשֵׁבֹר לִפְנֵי הָעָם הַוָּה, "<u>he shall go</u> across before this people" (Deut 3:28).<sup>16</sup>

With the first-person pronoun, the emphatic reference to the self can be *assertive*, connoting a sense of self-consciousness or assertiveness:<sup>17</sup> הָרָשָה (<u>But I</u> have dealt wickedly with you" (1 Sam 24:18), (<u>But I</u> have dealt wickedly with you" (1 Sam 24:18), אָנֹכִי אָשֶׁב עַר שׁוּבֶך (Judg 6:18).

(a.3) Indefinite subject – There are situations in which the subject of a verb points to no person in particular, but to a general "someone," "one," or "anyone." There are several constructions that can express this indefinite<sup>18</sup> or vague<sup>19</sup> subject. First, the *third person masculine singular active verb*, either with a substantival participle as subject, or without a stated subject, often expresses an indefinite subject: יעל־בֵּן קָרָא שֶׁמָה בָּבֶל , "Therefore <u>one calls</u> its name Babel" (Gen 11:9), יעל־בֵּן קָרָא שֶׁמָה דָּבָל it <u>anyone falls</u> from it [your roof]" (Deut 22:8). Second, the *third person masculine singular passive verb* also expresses an indefinite subject: *ו*בָּכָל הָבָל הָבָל וְלָא יוּכָל לְרָאֹת אֶת־הָאָרֶץ , "And they will cover the surface of the land so that <u>no one will be able to see</u> the land" (Exod 10:5), הַדָּהָה אָרָהָאָרָץ, "Then

- <sup>17</sup> Joüon and Muraoka 1993, 539–40.
- <sup>18</sup> Waltke and O'Connor 1990, 70.
- <sup>19</sup> Joüon and Muraoka 1993, 578.

<sup>&</sup>lt;sup>16</sup> Muraoka 1985, 48.

people began to call on the name of YHWH" (Gen 4:26). Third, the third person masculine plural verb, again either with a substantival participle as the subject, or an unstated subject, expresses an indefinite subject: מִן־הַבְּאֵר הַהָוֹא יִשְׁקוּ הָעֲדָרִים, "<u>They watered</u> the flock from that well" (Gen 29:2), שָׁמָה קַברוּ אֶת־אַבְרָהָם ואֶת שֶׂרָה, "There they buried Abraham and Sarah" (Gen 49:31). Fourth, the singular and plural participles may also express an indefinite subject: לְבֵנִים אֹמְרִים לְנוּ עֲשׁוּ, "<u>They</u> keep saying to us 'Make bricks!'" (Exod 5:16), אַלִי קרא משעיר, "Someone is calling me from Seir" (Isa 21:11). Finally, the anarthrous word איש may also serve as an indefinite subiect of a finite verb: לקטו ממנו איש לפי אָכלו, "Gather from it, everyone, as much as they should eat" (Exod 16:16), אָם־נַשָּך הַנַּחָשׁ אָת־אִישׁ וְהָבִּיט אָל־נָחַשׁ הַנָּחשׁת וַחַי, "If the serpent bites *anyone*, and he looks to the bronze serpent, he will live" (Num 21:9).

# (b) Predication of the verbal clause

As noted, predication in the verbal clause is achieved by the finite verbs. The full treatment of the syntax of verbs is located in Chapter 3. This discussion will be limited to considering the construction of the verbal clause.

(b.1) Word order – The typical word order for a verbal clause is verb – subject – object (VSO): יְיָאֶהֶב יְצְחֶק אֶת־עֵשָׁו "Isaac loved Esau" (Gen 25:28). Frequently, temporal emphasis will be marked by a temporal particle or adverbial phrase placed before the verb, describing the circumstances under which the predication of the verb takes place: בְּרָא אֶלֹהִים אֶת הַשְׁמִים (אַת הָאָרִץ), "In the beginning, God created heaven and earth" (Gen 1:1).

(b.2) The typical order of VSO will often be altered for several reasons. First, the subject can precede the verb to place *emphasis on the subject*: וְנִח מָצָא חֵן בְּעֵינֵי יְהוָת. "<u>But Noah</u> found favor in the eyes of YHWH" (Gen 6:8); וויהוָה הַמִיל רוּחַ־גְּרוֹלָה אֶל־הַיָּם. upon the sea" (Jonah 1:4).<sup>20</sup> Second, the subject will often precede the verb when there is a *change of subject* in narratival flow:

וַיִּשְׁבַע לוֹ וַיִּמְפֹר אֶת־בְּכֹרָתוֹ לְיַעֲקֹב וְיַעֲקֹב נָתַן לְעֵשִׂו לֶחֶם וּנְזִיד עֲדָשִׁים

"So he [Esau] swore to him, and sold his birthright to Jacob. <u>Then Jacob</u> gave Esau bread and lentil stew" (Gen 25:33-4).<sup>21</sup> Third, the object may also precede the verb to place *emphasis on the object* of the verb: the verb to place *emphasis on the object* of the verb: (Gen 3:10), אֶת־לְלְךָ שֶׁמַעָּהִי בַּגָן (Gen 9:5). Fourth, care must be taken with *poetic texts* because word order in such texts is extremely variable: (Ps 24:2), כִי־הוּא עַל־יַמִים יָסָרָה (Ps 24:2), כִי־הוּא עַל־יַמִים יָסָרָה (Ps 122:2). Fifth, in answers to a question, the essential part of the *reply* usually comes first: ויאָקרוּ אַלֶר הְנָה בָאֹהֶר וֹא צָלִיוֹ אַיָּה (*Ps said* to him 'Where is Sarah, your wife?' He said, '<u>She is in the tent</u>'" (Gen 18:9),

> וַיֹאמֶר חֲזָאֵל מַדּוּעַ אֲדֹנִי בֹכֶה וַיֹאמֶר כִּי־יָדַעְתִּי אֵת אֲשֶׁר־תַעֲשֶׂה

"Hazael said 'Why is my lord weeping?' And he said, "<u>Because I know</u> what you will do'" (2 Kgs 8:12). Finally, *interrogative particles*, introducing both direct and indirect questions, will often come first in the verbal clause: הבירם אָקָה (<u>Who</u> told you that you were naked?" (Gen 3:11), מָי הַגָּיר אָרָהָהָכִיהָ אָרָ־אָהָרָיָ struck your donkey?" (Num 22:32).

<sup>20</sup> However, it is likely that only the redundant use of the pronoun when a noun is subject marks real emphasis (5.1.2,a2; and see Andersen 1970, 24).

<sup>21</sup> Gesenius, however, notes that oftentimes when the subject precedes the verb, it is not because of a shift in the narrative but because it is a description of the state of the subject within the narrative. (Kautszch 1910, 455).

#### **5.2 Subordinate Clauses**

Subordinate clauses are dependent upon and modify an independent clause (or main clause) in a combination of two or more predications (see "complex" sentence at the beginning of this chapter). The following categories are helpful in considering the different types of subordination. Because BH makes no grammatical distinction between subordination and coordination, the reader should note carefully the variety of particles used to introduce them. However, it should also be kept in mind that most of these clauses can be introduced by a simple juxtaposition of clauses (asyndetically, or without conjunctions), or by the *waw* conjunctive.<sup>22</sup>

In general, there are two types of subordinate clauses: *complement* clauses and *supplement* clauses. Complement clauses are subordinate, but are closely linked to the main clause, and in fact cannot be omitted without changing the meaning of the main clause. Supplement clauses, on the other hand, are subordinate clauses that serve an adverbial function and could be omitted without changing the meaning of the main clause.<sup>23</sup> The first of these categories, the *substantival clause*, is a complement clause, and the rest are supplement clauses.<sup>24</sup>

#### 5.2.1 Substantival Clause

BH can use either a nominal or verbal clause to serve as a noun substitute in nominative, genitive, or accusative

- <sup>22</sup> The ubiquitous *waw* in BH narrative joins clauses paratactically (in coordination) or hypotactically (in subordination). Niccacci contends that many BH clauses are identified as subordinate due to contemporary conventions in European languages, even though the Hebrew syntax does not require subordination (Niccacci 1990, 128).
- <sup>23</sup> Van der Merwe, Naudé, and Kroeze 1999, 64–65 and 367. A further distinction is that a speaker cannot perform a speech act in a supplement clause, but can in a complement clause.
- <sup>24</sup> Joüon and Muraoka 1993, 589–645; Meyer 1992, 432–60; Waltke and O'Connor 1990, 632–46; Kautzsch 1910, 467–506; Williams 1976, 83– 96; Chisholm 1998, 119–35; and for epigraphic Hebrew, see Gogel 1998, 277–91.

functions in relation to the main clause to which it is subordinate. As in our discussion of nouns (Chapter 2), the case names should not imply that BH marks these clauses morphologically as nominative, genitive, or accusative, only that the clauses function syntactically in these distinct roles.

(b) Genitive – substitutes for a noun in a genitive function (see section 2.2), following and modifying a noun or clause that precedes it. The relationship between the genitive clause and the element it modifies can be marked by a noun in construct followed by the genitive clause, or a simple juxtaposition of clauses: בָּלֹיְמֵי הְתַהַלְכָנוּ אָתָם, "all of the days <u>we went about with them</u>" (1 Sam 25:15). The genitive clause may follow the relative particle בָּלֹיְמֵי אָשֶׁר הַנָּנַע בּוֹ יִטְמָא טָמֵא הוּא following: אָשֶׁר הַנָּנַע בּוֹ יִטְמָא טָמַא הוּא following: בָּלֹיְמֵי אָשֶׁר הַנָּנַע בּוֹ

(c) Accusative – substitutes for a noun in an accusative function, indicating the object of a verb (section 2.3.1): (יַרָא יְהוָה כִּי רַבָּה רְעַת הָאָדְם, "Yhwh saw that <u>the evil of humanity was great</u>" (Gen 6:5). An accusative acting as the object of a verb is frequently introduced by cor

יוָשָּׁם דְּוַיָּאָל אַלָּרָבּוֹ אָשֶׁר לֹא־יִחְנָאָל <sup>25</sup> אָשֶׁר (literally: "placed upon his heart") that <u>he would</u> <u>not defile himself</u>" (Dan 1:8). The particle י is especially common with verbs of perception, or in marking direct narration: יוֹב הָעֵץ מוֹב הָעֵץ, "The woman saw that <u>the</u> <u>tree was good</u>" (Gen 3:6), וְתֵּרֶא הָאָשָׁה כִּי מוֹב הָעֵץ , "He said '<u>Take these seven ewe lambs</u>" (Gen 21:30). Such clauses may also serve adverbially to modify the action of a verb (section 2.3.2): וְיָאמֶר הַוּא מַשְׁכִיל מָאָד (1 Sam 18:15). The accusative function is the most frequent of the substantival uses.

### 5.2.2 Conditional Clause

Conditional sentences (expressing the "if ... then ..." relationship) are introduced by subordinate *conditional clauses* detailing the condition(s), that is, the "protasis." Several particles are used to introduce conditional clauses, including occasionally the simple *waw* conjunctive. At times the apodosis will be introduced by the simple *conditional waw* (section 4.3.3,f): אָרָאָיָתִי מָה וְהַגּּרְהַיִּ יְלָן I see anything, then I will tell you" (1 Sam 19:3), I see anything, then I will tell you" (1 Sam 19:3), then has all this befallen us?" (Judg 6:13).

The two primary types of conditional clauses are the real and unreal. The former points to an action or situation that either has been fulfilled in the past or has the potential of being fulfilled, while the latter is an action or situation that cannot be fulfilled, or stands contrary to what has occurred in the past.

(a) Real conditional clauses are most often introduced by the particles 'כָּ, אָם תְּהְיוּ כָמֹנוּ :הֵן אָם אָם אָם אָם <u>become</u> like us..." (Gen 34:15), כִּי־תְמְצָא אִישׁ לֹא תְבְרְכָנּוּ "<u>If you encounter a man</u>, do not greet him" (2 Kgs 4:29),

<sup>&</sup>lt;sup>25</sup> At times even with the DDO, 1 Kgs 18:13.

מָה־נֹאכָל בַּשָׁנָה הַשְׁבִיעָת הֵן לא נְזְרָע, "What shall we eat in the seventh year *if we cannot sow?*" (Lev 25:20).

On occasion, אָשֶׁר can also function to indicate real conditional clauses:

אֶת־הַבְּרָכָה אֲשֶׁר תִּשְׁמְעוּ אֶל־מִצְוֹת יְהוֶה אֱל הֵיכֶם

"the blessing [will obtain], *if you listen to the commands of* <u>*YHWH*, your God</u>" (Deut 11:27).

A negative real conditional clause can be introduced by אָם־לא חָפַצְתָ בָה אָם־לא חָפַצְתָ בָה (*If you are not pleased* with her..." (Deut 21:14).

(b) Unreal conditional clauses are primarily marked by the particle לּוֹלֵא הַתְמַהְמָהַנוּ :לוֹלֵא הַתְמַהְמָהַנוּ :לוֹלֵא הַתְמַהְמָהַנוּ : לוֹלֵא הַתְמַהְמָהַנוּ : לוֹלֵא הַתַּמַהְמָהַנוּ : '<u>if we had not delayed</u>" (Gen 43:10,)

לוּ הַחֵיִתֵם אוֹתָם לא הָרַגְתִּי אֶתְכֶם

"*if you had let them live*, I would not kill you" (Judg 8:19). The particle 'can also function in this manner.

# 5.2.3 Final Clause

BH does not always make a sharp distinction between final and result clauses, which taken together may also be called "telic clauses."<sup>26</sup> A primary means of introducing final clauses is with the infinitive construct prefixed with the 'preposition: וְיַשֶׁלָה אֶהְאָב לָאֶכל וְלָשָׁתוֹת, "so Ahab <u>went up to eat and to drink</u>" (1 Kgs 18:42, a final clause), לְמָה לֹא־מָצָתִי חֵן בְּעֵינֵיךְ לְשׁׁתוֹם אֶת־מַשָּׁא כָּל־הָעָם תָזֶה עָלָי have I not found favor in your eyes, <u>that you laid the burden</u> of all this people on me?" (Num 11:11, a result clause).

(a) Final clauses – express the purpose or motivation for the action or situation of the main clause. They are often introduced by the particle אַשֶׁר The particles בַּעֲבוּר or לְמַעָן also function in this manner,

<sup>&</sup>lt;sup>26</sup> Joüon and Muraoka 1993, 633–35; Williams 1967, 86.

often with the particle אַשֶׁר, but not necessarily so: אַשָּׁר יִלְמְדוּן לְיִרְאָה אֹתִי וּזַאַשְׁמִעֵם אֶת־דְּבָררָי אֲשֶׁר יִלְמְדוּן לְיִרְאָה אֹתִי, "I will make them hear my words, <u>so that they may learn to fear me</u>" (Deut 4:10), לְמַעַן יַאָמִינוּ כִּי־נִרְאָה אֵלֶיךְ יְהוָה, "<u>so that they might believe</u> that Yhwh appeared to you" (Exod 4:5).

Purpose can also be expressed by an imperative with a simple prefixed *waw* conjunctive:

מָה אֶעֱשֶׁה לָכֶם וּבַמָּה אֲכַפֵּר וּבָרְכוּ אֶת־נַחֲלַת יְהוָה

"What can I do for you, and how can I make atonement *that you will bless the inheritance of YHWH*?" (2 Sam 21:3).

(b) Result clauses – express the end result or consequence of the action or situation of the main clause. Like final clauses, result clauses are introduced by אָשֶׁר אָשָׁר אָשָׁר אָשָׁר הַיּוֹם, governing a finite or infinite verb, also functions to introduce result clauses:<sup>27</sup> nite verb, also functions to introduce result clauses:<sup>27</sup>, "Abraham called the name of that place 'YHWH will provide,' <u>so that it is said to this day</u> ..." (Gen 22:14).

Often, if the main clause is a question, the result clause is marked by כָּדְרָזֶכְרְנוּ :כָּי תַזְכְרְנוּ :What is man <u>that you remember him</u>?" (Ps 8:5),

> הַאֶּלֹהִים אָנִי לְהָמִית וּלְהַחֲיוֹת כִּי־זֶה שֹׁלֵחַ אָלַי לֶאֶסף אִישׁ מִצַרַעִּתּוֹ

"Am I God, to kill and to make alive, *that this man is sending* word to me to cure a man of his leprosy?" (2 Kgs 5:7).

(c) Negative final/result clauses – serve as the negative of either the final or result clause, and are introduced with ther the final or result clause, and are introduced with <u>understand each other</u>" (Gen 11:7, a negative final clause), <u>understand each other</u>" (Gen 11:7, a negative final clause), <u>understand each other</u>" (Gen 11:7, a negative final clause), <u>so that no one will say</u> 'This is Jezebel'" (2 Kgs 9:37, a negative result clause).

<sup>&</sup>lt;sup>27</sup> Waltke and O'Connor 1990, 638.

The particles לְבַלְתִּי , בְּלְתִי , בָּלְתִי , function to mark a negative final clause: בְּלְבִי אָמֶרְתֶךּ לְמַעַן לֹא אָחֶטָא־לָךָ. "I have hidden your word in my heart, <u>so that I will not sin</u> <u>against you</u>" (Ps 119:11, a negative final clause).

#### 5.2.4 Temporal Clause

The subordinate *temporal clause* expresses the time frame of an action or situation, and how it relates to the idea of the main clause. A primary means of expressing temporal clauses is the infinitive with the prepositions ק. ק. אָחָר׳, אָחָר׳, אָחָר׳, אָחָר׳, אָחָר׳, אָחָר׳, ק. אָחָר׳, אָחָר׳, אָחָר׳, אָחָר׳, well. As demonstrated by the variety of temporal clauses that follow, so the infinitive can express a simultaneous, preceding, or succeeding action (see section 3.4.1,b): preceding, or succeeding action (see section 3.4.1,b): ק. הָבָעֶרֶב כְּבוֹא הַשֶׁמָשׁ שָׁנָה דָוֹר בְּמָלְכוּ diameter he began to reign (2 Sam 5:4), שָׁנָה דָוֹר בְּמָלְכוּ הוn the evening, <u>when the sun sets</u> (Deut 16:6), הָיָמָרָב הָבוֹת אָת סִיחֹן, "after he had defeated Sihon" (Deut 1:4), ווִימָלָא שֶׁבְעַת יָמִים אָחַרִי הַכּוֹת־יְהָוָה אָת־הַיָּאָר, "seven days passed <u>after</u> YHWH <u>had struck</u> the Nile" (Exod 7:25).

In addition to the infinitive, there are also other means by which a temporal clause is expressed.

(a) Contemporary action or situation – Introduced by כָּאָשֶׁר or יכָ, a subordinate clause can describe an action or situation that is contemporary with the main clause: ויְהָי כַּאֲשֶׁר כָּלָה לְהַקְרִיב אֶת־הַמִּנְחָה וִיָשָׁלֵח אֶת־הָשָׁ [*Ehud*] presented the tribute, he sent the people away" (Judg 3:18), וִיְהִי כַּאֲשֶׁר הַקְרִיב לְבוֹא מִצְרִימָה, "<u>When he</u> "[*Abraham*] came near to Egypt" (Gen 12:11).

(b) Later or succeeding situation – Introduced by עד, which is also used in the compound forms ער אָם, ער אָם the *temporal* clause may express action that is *later* than the action of the main clause. The particle שַרָּטָ, also found prefixed with the preposition ], also expresses a later situation. The idea of succeeding action is also expressed by the *wayyiqtol* pattern, as well as the conjunction לפני and the adverb בראשנה. It should be noted that the translation of the various particles ranges from "until" and "before" to "after." What is critical to note, however, is that the subordinate clause refers to a situation that comes temporally after the main clause, regardless of how the particles are translated in English: תָּמִים אַתָּה בִּדָרֶכֶיךָ מִיוֹם הִבָּרְאָךְ עַד־נִמְצָא עַוְלָתָה בָּד "You were blameless in your ways from the day you were created, *until iniquity was found in you*" (Ezek 28:15), ולא־הָאָמַנּהִי לַדְבָרִים עַד אֲשֶׁר־בָאתִי וַהְרָאֵינָה שֵׁינִי, "I did not believe things <u>until I came and my eyes saw</u>" (1Kgs 10:7), ויהי כאשר כלו הגמלים לשתות ויקח האיש נזם זהב "When the camels had finished drinking, the man took a gold ring" (Gen 24:22), וִיָּלְנוֹ שֶׁם טֶרֶם יַשְׁבֹרוֹ, "They camped there before they crossed over" (Josh 3:1).

The construction אַחָר הַדְּבָרִים is often used in narrative to indicate a later situation:

אַחַר הַדְּבָרִים הָאֵלֶה הָיָה דְבַר־יְהוָה אֶל־אַבְרָם

<u>After these things</u>, the word of YHWH came to Abram" (Gen 15:1).

(c) Preceding situation – shows a situation that occurred before the action of the main clause and is introduced most often by אָחֶרי/אָחָר אָחָרי, אוֹחר can be used with אָשֶׁרי/אַחָרי, אָשֶׁר can be used in this manner. Again, it is important to note that the subordinate clause refers to a situation coming before the main clause: הַאָר הָכָּתָה הָעִיר הָכָּתָה הָעִיר הַיָּרָבע עָשָׁרָה שָׁנָה אָחָר הָבָּעָר הָבָעָר הָאָדָר הָבָעָה הָעִיר הַבָּעָר הָבָעָה הָעָיר הַבָּעָר הָבָעָה הָעִיר הַבָּעָר הָבָעָר הַבָּעָה הָעִיר הַבָּעָר הָבָעָה הָעִיר הַבָּעָה הָעִיר הַבָּעָה הָעִיר הַבָּעָה הָעִיר הַבָּעָה הָעָיר הַבָּעָה הָעָיר הַבָּעָה הָעָיר הַבָּעָה הַעָּיר הַבָּעָה הָעָיר הַבָּעָה הַבָּעָר הָבָעָם מִדַרְכוֹ הָרָעָה הַרָעָה הַעָּרָה הַבּעָר הָדָבָע בַעָהַר הַבָּעָה הָעָיר הַבָּעָה הַעָּיר הַבָּעָה הַבָּעָה הַבָּעָר הַבָּעָה הַבָּעָה הַבָּעָר הַדָּבָע בַעָה הַבָּעָה הַבָּעָה הַבָּעָה הַבָּעָה הַבָּעָה הַבָּעָה הַבָעָה הַבָּעָה הַרָּעָה הַבָּעָה הַבּעָה הַבָּעָה הַבּעַר הַדָּבעָה הַבּעַר הַדָּבעָה הַבּעַר הַדָּבעָה הַבּעַר הַדָּבעָה הַבּעַר הַדָּבעָה הַבּעַר הַדָּבעָה הַבּעַר הַבָּע בַעָר הַבָּעָה הַבּעַר הַדָּעָה הַבּעַר הַבּעָר הַדָּעָר הַבּעָר הַדָּעָר הַבָּעָר הַיָּג הַרָר הַבָּעָה הַעָּג הַרָי הַיָ הַרַעָ הַיַיָר הַבָּעָר הַרָיָ הַרָ הַבָּג הַרָר הַבָּג הַיַר הַבָּעָר הַיָּעָר הַיַר הַיַר הַיָ הַיבּע הַיב הַיָר הַיַ הַרָי הַיַי הַבּג הַרָר הַבָּעָר הַיַ הַיַי הַיב הַיָר הַי הַיַר הַיָר הַיַי הַרַר הַיָ

וַיְהִי בָרָד וְאֵשׁ מִתְלַקַחַת בְּתוֹךְ הַבָּרָד כְּבֵד מְאֹד אֲשֶׁר לא־הָיָה כָמהוּ בְּכָל־אֶרֶץ מִצְרַיִם מֵאָז הָיְתָה לְגוי "So there was hail and fire flashing continually in the midst of the hail, very severe, such as had not been in all the land of Egypt <u>since it became a nation</u>" (Exod 9:24).

Often, the *waw* conjunctive plus subject (noun or pronoun) plus perfect denotes present perfect or pluperfect in English (only context dictates which is intended):<sup>28</sup> וֹיָשֶׁב אָהֵרֹן אָל־משָׁה אָל־מָשָׁר וְהַמָּגַפָּה נָשֵׁצְרָה, "Aaron returned to Moses at the entrance of the tent of meeting; now the plague <u>had been checked</u>" (Num 17:15 [Eng. 16:50]), הַבֹּקַרִים נָשָׂא אָת־הָאַרְבָּה, "When it was morning, the east wind <u>had (abready) brought</u> the locusts" (Exod 10:13).

#### 5.2.5 Causal Clause

The subordinate *causal* clause shows the reason or basis for the action or situation of the main clause. BH has a variety of causal clauses that are introduced by a simple *waw* conjunctive,<sup>29</sup> an infinitive clause (typically an infinitive construct with the  $\uparrow$  preposition), a verbal clause (with perfect plus *waw* consecutive), or the particles  $\bigcirc$  and  $\bigcirc$  are construct in the preposition  $\square$  and  $\bigcirc$  and

 $<sup>^{28}</sup>$  So waw + S(ubject) + perfect = anterior action or situation. Zevit has demonstrated the clarity with which authors of BH could thus mark anterior action, that is, action that began prior to the action of the main clause (Zevit 1998, 15–37, and see Joüon and Muraoka 1993, 624).

<sup>&</sup>lt;sup>29</sup> Kautzsch 1910, 492.

<sup>3</sup>º Joüon and Muraoka 1993, 638.

<sup>&</sup>lt;sup>31</sup> Waltke and O'Connor 1990, 640.

Examples: כָּי עָשִׁיהָ זֹּאָת אָרוּר אַהָּה (*Because* you did this, you are cursed" (Gen 3:14), יַעַן אָמְרְכֶם אֶת־הַדְּבָר הַזֶּה, <u>י</u>עַן אַמְרְכֶם אָת־הַדְּבָר הַזֶּה, <u>because</u> you speak this word" (Jer 23:38),

וְגֵר לֹא תִלְחָץ וְאַתֶּם יְדַעְתֶּם אֶת־נֶפֶשׁ הַגֵּר

"You shall not oppress a stranger, *since* you know the life of a stranger" (Exod 23:9).

(a) Joüon notes that the particle עֵקֶר which is sometimes with אָשֶׁר, has a special causal nuance, expressing "in recompense for the fact" of some action, or in a negative manner, "in punishment for the fact":<sup>32</sup> וְהַתְבָרֵכוּ בְוֹרְעֵך כֹּל גּוֹיֵי הָאָרֶץ עֵקָב אֲשֶׁר שֶׁמַעָהָ בְּקֹל. "By your seed all the nations of the earth will be blessed, <u>because</u> you obeyed my voice" (Gen 22:18), "In So you shall perish <u>because</u> you have not listened to the voice of YHWH your God" (Deut 8:20).

(b) The preposition מִן can also function to indicate cause: נְּשָׁלֵינוּ בְּלוּ מֵרֹב הַדֶּרֶךְ מְאֹד, "Our sandals are worn out <u>because</u> of our long journey" (Josh 9:13), לא־יָכֹל עוֹד לְהָשִׁיב אֶת־אַבְנֵר דְּבָר מִיָּרְאָתוֹ אֹתוֹ, "He could no longer speak a word to Abner, <u>because</u> he was afraid of him" (2 Sam 3:11).

# 5.2.6 Comparative Clause

(a) A subordinate clause will often function to make a comparison with the action or situation of the main clause. The situation that is the standard of comparison is usually stated in the main clause, or apodosis. The compared situation is typically denoted by the subordinate clause, the protasis. The most common construction for the comparative clauses is ראש באשר פתר קנו בן הָיָה, "And *just as* he interpreted for us, <u>so</u> it happened" (Gen 41:13). The reverse order, where the apodosis comes first, also occurs: בָן תַּעֲשֶׁה כַאֲשֶׁר דְּבַרְתָּ, "<u>So</u> do, <u>just as</u> you have spoken" (Gen 18:5).

(*b*) The comparative relationship between clauses is also denoted by the particles בָ and בָ, where the former governs the main clause, typically a verbal or infinitival clause, <sup>33</sup> and the latter governs the subordinate clause:

ַכְעֵינֵי עֲבָדִים אֶל־יַד אֲדוֹנֵיהֶם כְּעֵינֵי שִׁפְחָה אֶל־יַד גְּבִרְתָּה כֵּן עֵינֵינוּ אֶל־יְהוֶה

"*Just as* the eyes of the servants [look] to the hand of their masters; *just as* the eyes of the maidservant [look] to the hand of her mistress; *so* our eyes [look] to YHWH" (Ps 123:2).

The subordinate clause of the comparative relationship can also be introduced by the preposition ק: כָּבָרְעָה; for <u>you are like Pharaoh</u>" (Gen 44:18), יְהָיָה לָפְנֵי יְהָיָה YhwH" (Num 15:15).

# 5.2.7 Exceptive Clause

This subordinate clause functions to present an exception to the idea, action, or situation that is presented in the main clause. The particle אָ, used either alone or with the particles בָּלְתוֹ (which also functions alone) or כָּ and כָּ אָפָס כָּ mat אָפָס כָּ אָם־בָּרָכָתִנ, as well as אָפָס כָּ and רָק ntroduce these clauses (see sections 4.2.2 on אָפָס י כָ, 4.2.15 on כָּ, 4.2.15 on (רַק וּרָק וּרָ אַם־בַּרְכָתִני, "I will not release you, <u>unless you</u> <u>bless me!</u>" (Gen 32:27 [Eng 32:26]),

"YHWH has taken away your sin; you will not die. <u>*However*</u>, because by this deed you have given occasion to the enemies of YHWH to blaspheme ...." (2 Sam 12:13–14).

#### 5.2.8 Restrictive Clause

This clause places a limitation on the idea or action of the main clause. These clauses are most often introduced by the particles אָרָ אָבָס בָי (see section 4.2.2,a on אַרָ 4.2.15,a on בק סוב):

```
וְהָשְׁמִדְתִּי אֹתָה מֵעַל פְּנֵי הָאֲדָמָה אֶפֶס כִּי לֹא הַשְׁמֵיד אַשְׁמִיד
אֶת־בֵּית יַעֲלְב
```

"I will destroy it from the face of the earth, <u>but</u> I will not completely destroy the house of Jacob" (Amos 9:8), אָקְרָא שָׁמְךָ עָלֵינוּ (We will eat our own bread, and wear our own clothes; <u>only</u> let us be called by your name (Isa 4:1), clothes; <u>only</u> let us be called by your name (Isa 4:1), clothes; <u>only</u> let us be called by your name (Isa 4:1), clothes; <u>only</u> let us be called by your name (Isa 4:1), clothes; <u>only</u> let us be called by your name (Isa 4:1), clothes; <u>nevertheless</u>, the honor shall not be yours" (Judg 4:9).

#### 5.2.9 Intensive Clause

This subordinate clause expands and adds to the idea of the main clause. The primary markers of these clauses are אר אר אר (see sections 4.2.4,b on אר אר גם) אר אר געפה ויאכל (גם), "She took from its fruit and ate, and she <u>gave also</u> to her husband with her, and he ate" (Gen 3:6); אר געפה (<u>I also saw</u> in my dream" (Gen 40:16).

#### 5.2.10 Adversative Clause

This subordinate clause presents an antithesis to the idea of the main clause. The relationship between those clauses can be marked by  $\vec{x}'$ ,  $\vec{\gamma}$  (which is

also used with אָרָם אוּרָם), or the particles אַרָּלָם אוּרָם, גָּם־הוּא יִהְיֶה־לְּעָם וְגַם־הוּא יִגְדָּל וְאוּלָם אָחִיו הַקְּמֹן יְגָדֵּל מִמְנוּ, "He also will become a people, and he, too, will be great. *But, his younger brother shall be greater than he*" (Gen 48:19), (Gen 48:19), "He said 'No. *Rather, I come now* as the captain of the army of YHWH" (Josh 5:14).

The waw conjunction can also perform this function: וְלֹא־יִקְרָא עוֹד אֶת־שָׁמְדָ אַבְרָם וְהָיָה שִׁמְדָ אַבְרָהָם will your name be Abram, <u>but</u> it will be Abraham" (Gen 17:5), אַל־תָּלֶן הַלַּיְלָה בְּעַרְבוֹת הַמִּדְבָּר וְגַם עָבוֹר תַּעֲבוֹר (Gen not spend the night at the fords of the wilderness, <u>but</u> by all means, cross over" (2 Sam 17:16).

### 5.2.11 Circumstantial Clause

This subordinate clause describes the circumstances under which the action or situation of the main clause takes place. Frequently, the simple conjunction ן functions to introduce a circumstantial clause (see section 4.3.3,e): סָּרָאָרָי וְהָאָדֹיָי נָצָּב עַל־חוֹמָת אָנָדְ וְרָאָשׁוֹ בָשָׁמִים, "a tower *with its top in the sky*" (Gen 11:4), כָּה הְרָאָנִי וְהָנָה אָדֹיָי נָצָב עַל־חוֹמָת אָנָדְ וֹבְיָדוֹ אָנָדָ וֹ

Quite often, the circumstantial clause is juxtaposed to the main clause with no connecting particle: אָהָלה בִית־אָל מִיָם ווים אָהֲלה בִית־אָל מִיָם, "He pitched his tent, <u>with Bethel on</u> <u>the west</u>" (Gen 12:8), ווים אָהֲלוֹם עוֹדֶנּוּ חַי, "and he thrust them through Absalom's heart <u>while he was still alive</u>" (2 Sam 18:14).

As illustrated, the predicate of a circumstantial clause that is concomitant with the situation of the main clause is expressed by nouns, adjectives, or prepositional phrases. The participle serves this same function: וְיָבֹא אֵלִישָׁע דַמָּשֶׂק וּבֶן־הַרֵד מֶלֶךְ־אָרֶם חֹלֶה, "Elisha

came to Damascus *while Ben-Hadad, King of Aram was ill*" (2 Kgs 8:7),

וַיִּרָא אֵלָיו יְהוָה בְּאֵלנֵי מַמְרֵא וְהוּא יֹשֵׁב פֶּתַח־הָאֹהֶל כְּחֹם הַיּוֹם

"Now YHWH appeared to him by the oaks of Mamre, *while he was sitting* at the tent door in the heat of the day" (Gen 18:1).

Finite verbal forms are used to express circumstances that are either in the past or the future.<sup>34</sup> A perfect verb as the predicate of the circumstantial clause expresses anterior action (see section 3.5.4,c): אל האָאָרָרוֹ אֹתִי וִיהוָה הְצָלִיח דְרְכִי since <u>Yhwh has prospered</u> my way" (Gen 24:56), since <u>Yhwh has prospered</u> my way" (Gen 24:56), took captive the women who were in [the city], both small and great, <u>without killing anyone</u>" (1 Sam 30:2).

An imperfect verb as the predicate expresses future circumstances:

# הַמְכַפֶּה אֲנִי מֵאַבְרָהָם אֲשֶׁר אֲנִי עֹשֶׁה וְאַבְרָהָם הָיוֹ יִהְיֶה לְגוֹי נֶּדוֹל וְעָצוּם

"Shall I hide from Abraham what I am about to do, <u>since</u> <u>he will become</u> a great and mighty nation?" (Gen 18:17–18), אָאָשְהְרָה הָאַרְצֶך בַּדֶרֶך בַּדֶרֶך אַלֵך לא אָסוּר יָמִין וּשְׂמאול, "Let me pass through your land; I will travel only on the highway; <u>I will not turn</u> to the right or to the left" (Deut 2:27).

#### 5.2.12 Concessive Clause

This subordinate clause denotes "causal contrast";<sup>35</sup> that is, it presents an action or situation that would

<sup>&</sup>lt;sup>34</sup> Williams 1976, 83. Gesenius further notes that circumstantial clauses employing a verb particularize the action of the main clause, or provide a reason for the action of the main clause (Kautzsch 1910, 490).

<sup>&</sup>lt;sup>35</sup> Joüon and Muraoka 1993, 641.

seemingly lead to, or be expected to cause, the action or situation of the main clause, but, in fact, does not. The clause can be asyndetic, that is, without any markers, by juxtaposition, or it can take a simple waw conjunctive: הַנֵּה־נָא הוֹאַלְתִּי לְדַבֵּר אֶל־אֲדֹנֵי וְאָנֹכִי עָפָר וָאֵפֶר, "Certainly, now, I have ventured to speak to the lord, although I am dust and ashes" (Gen 18:27). The particles 'ج, or גם כי also express the idea of "although," while the נכי, or על particles אם and ים express the notion of "even though" (see sections 4.3.2,b and 4.3.4,h): גם כִּי־תַרְבּוּ תִכְּלֶה אֵינֵנִי שׁמֵע "Even though you multiply prayers, I will not listen" (Isa 1:15), וְלֹא־נָחָם אֶלהִים הַרֵך אֶרֵץ פִּלְשָׁתִים כִּי קָרוֹב הוּא (Isa 1:15), "God did not lead them by the way of the Philistines, even though it was near" (Exod 13:17), כִּי אָם־תְּכַבִּסִי בַּנֵּתֶר וְתֵרְבִּי־לֵךְ בֹּרִית נְכָתֵם עֲוֹנֵךְ לְפַנֵי, "<u>Although</u> you wash yourselves with lye and use much soap, the stain of your iniquity is before me" (Jer 2:22).

# 5.2.13 Relative Clause

This clause stands subordinate to a main clause and attributes a quality, state, or verbal idea to the main clause. This group can be further divided into *limiting* and *nonlimiting* relative clauses.<sup>36</sup> A *limiting* relative clause functions to make a distinction between more than one member of a group or class: הַמִים אֲשֶׁר מִעָל לְרָקִיעָ הַמָּיִם אֲשֶׁר מַעָל לְרָקִיעָ *which were above the firmament*" (Gen 1:7). A *nonlimiting* relative clause simply marks a general attribute of the antecedent without setting it off against other members of its "class":<sup>37</sup>

 $<sup>^{36}</sup>$  Gesenius prefers the categories "complete" and "incomplete" relative clauses, the former used for "the nearer definition of a noun," and the latter, relative clauses that are not dependent on a noun (Kautzsch 1910,  $_{485}$ ).

<sup>&</sup>lt;sup>37</sup> Joüon and Muraoka 1993, 592.

"And now, you are cursed from the ground <u>which has</u> <u>opened its mouth</u>" (Gen 4:11). (a) The

most frequent indicator of the relative clause is the particle אָשֶׁר In poetry, or in later texts, the prefix -ש identifies the relative clause. Within more archaic poetry, the particles אָר היאָר אָלָר אָשָׁר אָלָר בָּאָר אָלָר בָּאָר tive use of the participle can also mark a relative clause. Examples: אָשָר אָלָר בָּאָר אָשֶׁר אָלָר בָּאָר אָלָר בָּאָר הַרָּהַשָּם הַאָּה אָל־הָאָר אָשָׁר אָלָר בָּאָר (Josh 1:2), אָשָר אָלָר בָּאָר אָשֶׁר אָלָר בָּאָר (Josh 1:2), הַבְּרָשֶׁר בָּלַר בָּעָר בָּלַר בָּאָר (I Kgs 2:11), הַרָּשְׁתוּזוּ הוּא שָׁרָה הוּא שָׁרָה בָּאָר לוו the net <u>in which they hid</u>" (Ps 9:16), הַבָּשֶׁר הָאָר בָּאָר לָאָר הָדָאָר לָא לְהָם (Eccl 1:9). The relative clause can also be asyndetic: בָּאֶר לָא לְהָם הָרָאָר לָא לְהָם

(c) **Paronomasia** – The repetition of a word both in the main clause and the relative clause can be used to express

<sup>&</sup>lt;sup>38</sup> Waltke and O'Connor 1990, 333–35; Joüon and Muraoka 1993, 595– 600; Kautzsch 1910, 444–46.

a sense of indeterminateness:<sup>39</sup> הּוֹלֵךְ עַל אֲשֶׁר־אֲנִי הּוֹלֵך ". . . while I go <u>wherever I will go</u>" (2 Sam 15:20), "I some sense of indetermination (2 Sam 15:20), "I some sense of the sense of the sense of the sense of the sense (1 Sam 23:13), "I they went <u>wherever</u> they could go" (1 Sam 23:13), "I mey went <u>wherever</u> they could go" (1 Sam 23:13), "I mey went <u>wherever</u> they could go" (1 Sam 23:13), "I will be gracious to <u>whomever I will</u> [literally: to whom I will be gracious] and I will show compassion on <u>whomever I will</u> [literally: to whom I will show compassion]" (Exod 33:19).

# 5.2.14 Disjunctive Clause

This subordinate clause represents an idea that is alternative to, or stands apart from, the idea of the main clause. Typically, the conjunction אל denotes the idea of "or": דיקו לי ואָפָנָה עַל־יָמִין או עַל־שָׁמאל, "Let me know, that I may turn to the right <u>or</u> to the left" (Gen 24:49). The idea of "or" can also be expressed by the waw conjunctive: נימי במוסים וּבְפָרָשִׁים בּקַשָׁת וּבְחָרֶב וּבְמָלְחָמָה בְּסוּסִים וּבְפָרָשִׁים , "I will not save them by bow <u>or</u> by sword <u>or</u> by war <u>or</u> by horses <u>or</u> by horsemen" (Hos 1:7). The notion of "either ... or" or "whether ... or" can be expressed by consecutive waw conjunctions: מות יוּמָת וּמָצָא בְיָדוֹ מוֹת יוּמָת וֹנְגַב אָישׁ וּמְכָרוֹ וְנְמָצָא בְיָדוֹ מוֹת יוּמָת kidnaps a man, <u>whether he sells him or</u> he is found in his possession ..." (Exod 21:16). The notion of "whether ... or" is also expressed by consecutive Exp

"Wherever my lord the king may be, *whether for death or for life*, there your servant will be" (2 Sam 15:21).

#### 5.3 Additional Sentence Types

At the beginning of this chapter, we provided working definitions for "simple," "compound," and "complex" sentences, as

<sup>39</sup> Joüon and Muraoka 1993, 599.

well as the slippery term "sentence" itself. In what follows, we offer an overview of other important sentence types in BH.

# 5.3.1 Interrogative Sentences

An interrogative clause, in its most basic sense, poses a direct question, which can be pointed to the reader or confined within the narrative. This question can be genuine, or it can be rhetorical, in which case it is not expected to be answered. Interrogative sentences are often distinguished only by intonation, as they are not marked with any morphological tags: "אָאָר שָׁלֹם בּוֹאָן". "And [Samuel] said, '<u>Do you come peacefully?</u>" (1 Sam 16:4). However, the following are particles for marking interrogatives.

(a) Interrogative clauses may be marked by the particle הַ to indicate direct questions: אַל־הִיָרָאוּ כִי הַתַחַת אָלהִים אָנִי
 "Do not fear, <u>for am I in the place of God?</u>" (Gen 50:19),
 "Do not fear, <u>fightian לַנַּעַר לְאַרְשָׁלוֹם הַנַּעַר לְאַרְשָׁלוֹם הַוּשַלוֹם הַוּשַ</u>

(b) In the case of a disjunctive question, "X or Y?" the particle אום אום introduce the second option: הַלָּצָרִינָן, "Are you with us <u>or with our enemies?</u>" (Josh 5:13).

(c) In addition to the particle אָן, there is a group of interrogative pronouns that introduce specific types of questions, including אָן, אָדָן, אָדָן, אָלָן, אָנָר.:

מֶה חֲרִי הָאַף הַנְּרוֹל הַזֶּה

"<u>Why</u> this great anger?" (Deut 29:23 [Eng. 29:24]), קי הַגִּיד לְךָ כִּי עֵירֹם אָתָה, "<u>Who</u> told you that you were naked?" (Gen 3:11), אָיך האַמַר אָהַרָתִיך, "<u>How</u> can you say 'I love you?'" (Judg 16:15), אָי הֶבֶל אָחִיך, "<u>Where</u> is Abel, your brother?" (Gen 4:9). (d) The particle אָפוֹא often follows an interrogative to give an additional emphasis to the question: מִי־אָפוֹא הָאָר־צַיָר, "<u>Who was it, then</u>, who hunted game?" (Gen 27:33).

These interrogative markers can also mark *exclamatory* clauses: הְנָגְלֹה נְנְלֵחִי אֶלֹ־בֵּית אָבִיךָ, "<u>I certainly revealed myself</u> to the house of your fathers" (1 Sam 2:27), to the house of your fathers" (1 Sam 2:27), <u>המלקיה המלקיקה (How the mighty have fallen</u> in the midst of the battle!" (2 Sam 1:25), המלקיקה <u>How beautiful</u> are your tents, O Jacob!" (Num 24:5). Additionally, a *substantive* can function as an exclamation. This use is not marked by morphology, and so is context sensitive: ווֹאמֶר אֶלֹי אָבִיוֹ רֹאשִׁי רֹאשִׁי רֹאשִׁי (2 Kgs 4:19).

Interrogative clauses can also function to denote *indirect questions*, which report or refer to the question of another speaker, although that first question need not be explicitly stated. Because indirect questions logically refer to another's speech, they are not meant to be answered. These clauses are introduced in the same manner as direct questions: לא יָדְשָׁת מָאַין הָמָה (I do not know <u>where</u> <u>they are from</u>" (Josh 2:4), לא יָדְשָׁת כְּסְפַּנוּ בְּאַמְתְּחֹתִינוּ מָאַין הָמָה (Gen 43:22), יָשָׁל מָה אָשֶׁשֶׁה לָה אָשֶׁשֶׁה לָק ה

# 5.3.2 Oath Sentences

A maledictory oath, or curse, is often introduced by the formulaic statement – "May God add to this and more...." The content of the oath itself, the apodosis, is marked by  $\Box \aleph$  if the oath statement is negative, expressing an action that one will not take, whereas  $\aleph \Box \aleph$  and  $"\supseteq$  function to introduce positive statements, actions that one will take.

**Examples:** 

כּה יַשְשָׁה יְהוָה לִי וְכֹה יֹסִיף כִּי הַמָּוֶת יַפְּרִיד בֵּינִי וּבֵינֵך

"Thus may YHWH do to me and more <u>if anything but death</u> <u>part you and me</u> [that is, Nothing but death will part you and me.]" (Ruth 1:17),

"Thus may God do to me and more also *if I taste bread or anything else before the sun goes down* [that is, I will not taste bread or anything else...]" (2 Sam 3:35).

#### 5.3.3 Wish Sentences

Desiderative clauses express a strong wish or desire. Often, the expression of a wish can take on the character of a question or exclamation.

(a) The most common means of expressing a desire in BH is through the use of the volitional verbs (see section 3.3): הְפָּרֶר נָא מֵעָלִ', "<u>Let there be</u> light" (Gen 1:3), יָהִי אוֹר "<u>Please separate</u> from me" (Gen 13:9). Occasionally, an optative clause introduced by a finite verb is marked with or וויאמֶר אַרְרָהָם אָל־הָאֵלֹהִים לוּ יִשְׁמָעָאל יִחְזֶה לְפָנֶיך אַם זי "Abraham said to God, '<u>O that Ishmael might live</u> before you!" (Gen 17:18), מִרְרַיִם מָצַרִים, "<u>Would that we died</u> in the land of Egypt!" (Num 14:2). (b) The expression of a wish can often resemble a question. Thus, the interrogative particles, in particular יב, can be used to introduce a strong wish or desire: בָּאָרֵץ, can be used to introduce a strong wish or desire: מָי־יָשָׁמָנִי שׁפַט בָּאָרֵץ, "<u>O that someone would appoint</u> me judge in the land" (2 Sam 15:4). The construction is a notable peculiarity in BH expressing an optative sense, sometimes with the sense of "God grant!":<sup>40</sup> is a notable peculiarity in BH expressing an optative sense, sometimes with the sense of "God grant!":<sup>40</sup>, "<u>Would that</u> this people were under my authority!" (Judg 9:29), "<u>Would that</u> all YHWH's people were prophets!" (Num 11:29).

(c) The expression of a wish can also be constructed with a participle or a prepositional phrase acting as the predicate in the clause: אֹרְרֶיךָ אָרוּר וּמְרָרֵכֶיךָ אָרוּר וּמְרָרֵכֶיךָ, <u>"Cursed be</u> those who curse you and <u>blessed be</u> those who bless you" (Gen 27:29), שָׁלוֹם לֶכֶם, "<u>Peace to you</u>" or "<u>May you have peace</u>" (Gen 43:23).

(d) Similar to the statements of oath are *asseverative clauses*, which also express a strong assertion. This statement may be an oath, in which case the asseverative clause functions as an elliptical oath statement. However, the clause can simply be a reinforcement or affirmation of a fact. In addition to the particles already noted, אָרָל אָמָנָה אָמָנָם הָאָר, אָמָנָה אָמָנָם הָאָר, and הַבָּה are all used to mark affirmation: הַנָּה אָמָנָה בָּמָקום הָאָר מין אַרָל, "<u>Surely</u> there is no fear of God in this place" (Gen 20:11), אָרָל אָשָׁמִים אַנְחְנוּ עַל־אָחִינוּ הָנָה בַּמָקוֹם הָאָר (Gen 20:11), אָרָל אָשָׁמִים אַנְחְנוּ עַל־אָחִינוּ (Gen 20:11), אָרָל אָשָׁמִים אַנְחְנוּ עַל־אָחִינוּ הָנָה בַמָּקוֹם הָאָר (Gen 20:11), אָרָל אָשָׁמִים אַנִחְנוּ עַל־אָחִינוּ (Gen 42:21), הַנָּה מוּטַר brother" (Gen 42:21), הַנָּה מוֹשָׁרָן יִהְיָה מוֹשָׁרָן הַנָּמָוֹם הָאָר (Gen 28:16), הַנָּה מוֹשָׁרָן יִהְיָה מוֹשָרָן אַר מוֹשָרָן אָרָזים בָמָמָנוּ מוּטַר אָרָן אָרָזים הַנָּמָוֹם הַאָּר מוּטַר מוּטָר מוּטַר מוּטָשָרָן (Gen 23:16), אָרָן יִהָיָה מוּטַעָר מוּטַר הַנָּה מוּטַעָר מוּטַר מוּשַר מוּטַר מוּטַר מוּטַר מוּטַר מוּטַר מוּטַר מוּטַר מוּטַר מוּטַר מוּשַר מוּשָר מוּטַר מוּטַר מוּטַר מוּטַר מוּטַר מוּטַר מוּשָר מוּשָר מוּטַר מוּטַר מוּטַר מוּטַר מוּטַר מוּטַר מוּטַר מוּטַר מוּט מוּט מוּט מוּטַר מוּט מוּט מוּטַר מוּטַר מוּטַר מוּטַר מוּטַר מוּטַר מוּטַר מוּטַר מוּ

<sup>4</sup>º Joüon and Muraoka 1993, 616.

# 5.3.4 Existential Sentences

There are two main ways in which the existence of a thing or person is expressed in BH.

(a) For *past or future time*, a form of the verb הָיָה expresses existence: הַנָּחָשׁ הָיָה עָרוּם מִכֹל חַיַּת הַשָּׁרָה, "The serpent <u>was</u> more crafty than any other animal of the field" (Gen 3:1), more crafty than any other animal of the field" (Gen 3:1), אָיָשׁ הָיָה הְאָרֶץ־עוּץ (Job 1:1), כָה יִהְיֶה וַרְעֶךָ, "So <u>shall</u> your descendants <u>be</u>" (Gen 15:5).

(b) For *present time*, the verb הָּיָה is not used. Rather, the particle of existence שָׁיֵ is used (see section 4.4.2). This particle is not negated, but rather, the absence of something in present time is expressed by the particle אָין (section 4.4.1): אָין הָוָה בַּמְקוֹם הַאָּר אָר הָאָרָטָה, "Surely, YHWH *is* in this place" (Gen 28:16), אָרָה אָרָטָה אָרָטָר אָין לַעַבר אָר הַאָרָטָה (Gen 28:16), הווא אָרָטָר אָין לַעַבר אָר

# 5.3.5 Negative Sentences

(a) Typically, verbal clauses are negated by אָל, although אַל is used with the imperfect as a functional equivalent to a negated imperative or the other volitional verbs (see sections 4.2.3, 4.2.11): לְאָרֶם לֹא־מָצָא עֵזֶר כְּנֶגְדוֹ Adam, <u>there was not found</u> a suitable helper" (Gen 2:20), Adam, <u>there was not found</u> of them" (Josh 11:6).

(b) Nominal clauses are typically negated by אין, although לא לא although אין לה וָלָד, "She <u>had no</u> child" (Gen 11:30), לא־מוֹב הֵיוֹת הָאָדָם לְבַדּוֹ for the man to be alone" (Gen 2:18).

(c) Specifically for negation of infinitives construct, לְבְלְתִּי סוּר־מִמֶּנוּ יָמִין וּשָׂמֹאול so that you <u>might not turn</u> from it to the right or to the left" (Josh 23:6), הַשְׁמֶר לְךָ פֶּן־תִּשְׁכֵח אֶת־יְהוֶה אֱלֹהֶיךָ לְבָלְתִי שְׁמֹר מִצְוֹתִיו care that you do not forget YHWH your God <u>by not keeping</u> his commandments" (Deut 8:11).

# 5.3.6 Elliptical Clauses and Sentences

At times, BH omits certain parts of speech from clauses. Close attention to context generally reveals the omitted element(s), and ambiguity is seldom a result.

(a) Substantives are often omitted in *comparisons*, particularly in the clause presenting the compared situation: עוֹשָׁוֶה רַגְלֵי כָּאַיָלוֹת, "He makes my feet like <u>hinds [feet]</u>"
 (2 Sam 22:34 Qere), וְהָרֶם כָּרְאָים כְּרָגִים קרָנִי, "You have exalted my horn like <u>[that of] the wild ox</u>" (Ps 92:11).

(b) Pronouns are often omitted when their antecedents are clear from the context, primarily in two situations: as the subject of nonfinite verbs: בַּהַלִישַׁב בָּהַן לוֹט . <u>When he overthrew</u> the cities in which Lot lived" (Gen 19:29), עַמֵר עַל־הַגְּמַלִים עַל־הָעָיָן, "<u>He was standing</u> by the camels at the well" (Gen 24:30); and when functioning as the object of the verb: רָיָבא אֶל־הָאָרָם, "He brought <u>them</u> to the man" (Gen 2:19).

(c) Certain expressions of quantitative measure, such as "day," "month," or "year" are omitted after numerals: אָך בָּשָׁבִישָׂר לַחֹרָשׁ הַשְּׁבִישָׂר, "On exactly the <u>tenth [day]</u> of the seventh month" (Lev 23:27),

וְלֹא אָנֹכִי שֹׁמֵל עַל־כַּפַּי אֶלֶף כָּסֶף

"Even if I should receive a *thousand* [*pieces*] of silver" (2 Sam 18:12 *Qere*).

(*d*) Ellipsis also occurs frequently with negatives: אָל בְּנֹתִי "<u>No</u>, my daughters (Ruth 1:13), הֵיָשׁ יְהוֶה בְּקְרְבֵנוּ אֲם־אָיִן, "Is Yhwh in our midst or <u>not [in our midst]</u>?" (Exod 17:7).

# APPENDIX I: Stem Chart

		<u>Horizontal Axis</u> – "Type"			
		Simple	Factitive- Causative	Causative	
Vertical Axis - "Voice"	Active	קל <b>QAL</b> (G)	פּעֵל PIEL (D)	הָפְּעִיל HIPHIL (H)	
	Passive	נפְעַל NIPHAL (N)	פעל PUAL (Dp)	הָפְעַל <b>HOPHAL</b> (Hp)	
	Reflexive	נפְעַל NIPHAL (N)	הְתְפַיֵּעַל <b>HITHPAEL</b> (HtD)	Ø	

*Sources:* Joüon and Muraoka 1993, 124; Greenberg 1965, 42; Goshen-Gottstein 1969, 70–91, esp. 74–75; Harper 1888, 71; and Waltke and O'Connor 1990, 353–54.

# APPENDIX II: Expanded Stem Chart

	Axis of Causation – Voice of the Undersubject				
		No Causation	Brings about a <i>State</i>	Causes an <i>Action</i>	
		No Generated Predication	Generates a <b>Nominal</b> Predication	Generates <b>Verbal</b> Predication	
		No Undersubject	Patiency Nuance	Agency Nuance	
		QAL Stative	PIEL	HIPHIL 2 Place Ingressive	
<u>Axis</u> of <u>Voice</u> - Voice of the Primary Subject	Active	Stative         Fientive         > Intransitive         > Transitive	<ul> <li>Factitive         <ul> <li>makes 2 be x                 Outcome                describable by                <i>adjectival</i> clause</li> </ul> </li> <li>Resultative         <ul> <li>makes 2 be x-ed</li> <li>Outcome describable                by participle clause</li> </ul> </li> </ul>	<ul> <li>2 Place Ingressive 1 makes 2 become x 1 Place Internal 1 makes self do/become x 2 Place Causative 1 makes 2 do x</li> <li>3 Place Causative 1 makes 2 do x to 3</li> </ul>	
	Passive	(Qal Passive) NIPHAL	<b>PUAL</b> Passive of Piel	HOPHAL Passive of Hiphil	
	Middle	<b>NIPHAL</b> Middle Adjectival Reflexive-Reciprocal	HITHPAEL Reflexive-Reciprocal of Piel Also later Passive	<b>HISHTAPHEL</b> only in השְׁתַחֵוָה	

 $\mathit{Source:}$  Based on Waltke and O'Connor 1990 (343–61), and with the help of Lawson G. Stone.

- **Ablative** Expresses a relationship of separation, or movement away from a substantive. 4.1.10; 4.1.13; 4.1.14
- **Absolute Noun** The basic, lexical form of a noun in Hebrew; can function as a nominative, dative, or accusative.
- Accompaniment Expresses addition to a noun, or to a verbal action or situation. 4.1.4; 4.1.5; 4.1.16; 4.1.17
- Accusative The noun case that marks the direct object, the recipient of action, of a verb. Example: "I hit *the ball.*" 2.3
- Active Voice Construction of a verbal clause in which the subject performs an action. Example: "He cut the grass." See Passive Voice. 3.1.1.
- Addition See Accompaniment. 4.1.2; 4.1.16; 4.2.4; 4.2.5
- Advantage Specifically here, an action or situation that is meant for the benefit of another. 4.1.2; 4.1.7; 4.1.10
- **Adverb** A word that modifies a verb or adjective, but also other adverbs (item adverbs) and, in many cases, entire clauses (clausal adverbs); typically expresses how an action or situation takes place. 4.2
- **Adversative** Expresses opposition or antithesis. 4.1.5; 4.1.16; 4.3.3; 4.3.3; 4.5.2
- Agency Nuance With Hiphil or Hophal causatives, indicates that the secondary subject takes action toward a third object; the secondary subject is effected to take action toward a third object. See Patiency Nuance. 3.1; 3.1.3; 3.1.6
- **Agent** The one who performs the action of a verb. 3.1.2; 3.5.4; 4.1.10; 4.1.13
- Agreement Expresses similarity. 4.1.9
- Alternative Expresses a choice between two or more options. 4.3.1; 4.3.2; 4.3.3
- **Anaphoric** A syntactical element, typically a pronoun, used to refer to something or someone that has been mentioned earlier in the text.

Anarthous See Indefinite.

- **Aorist** Greek verbal tense, generally referring to past time, that focuses on the completed nature of a verb, rather than its duration. 3.2.1
- **Apodosis** The main clause of a conditional statement; the "then" statement. 3.5.2; 4.2.1; 4.2.10; 4.3.3; 4.3.4; 5.2.6; 5.3.2
- Apposition Construction in which two nouns are juxtaposed, refer to the same element, and have the same grammatical function; the second noun (the apposition) typically modifies the first noun (the leadword).2.4
- **Aspect** From the German *Aspekt;* the manner in which the action or situation of a verb relates to time, in terms of completeness or lack of, in Hebrew and other Semitic languages. 3.2
- Assertive Expresses a notion of confidence or boldness. 5.1.2
- Asseverative Expresses emphasis or surety. 4.2.2; 4.2.4; 4.2.15; 4.3.4; 5.3.3
- **Asyndetic** A string of words, phrases, or clauses that are not connected by a conjunction that is expected. 5.2.1,a; 5.2.12; 5.2.13, a
- **Attributive** Construction that functions to describe a quality or attribute. 2.2; 2.2.5; 2.3.1, C 2.4.2; 2.5.1; 3.4.3, a; 5.2.13, a
- Benediction A blessing spoken by one, meant to bless another. 3.3.1,c
- **Biblical Hebrew** The language of the Hebrew Bible/Old Testament; contrast with Epigraphic Hebrew, which focuses on extra-biblical inscriptions; or Rabbinic Hebrew, which focuses on Hebrew texts written after the Bible.
- **Cardinal Number** A number used in counting, to indicate the quantity of objects. 2.7.1
- **Case** The function of a noun within a sentence; see Accusative, Dative, Genitive, Nominative. 2.0
- **Causal** Expressing the cause of an action or situation. 2.2.7; 4.1.2,g; 4.1.5,f; 4.1.8; 4.1.10,d; 4.1.11,b; 4.1.13,e; 4.1.14,c; 4.1.16,d; 4.2.10,b; 4.3.4,a, b; 4.5.3,c1; 5.2.5
- **Causation** Refers either to the act or process of causing an action/idea to take place. 2.3.1,e; 3.1; 3.1.3; 3.1.6
- **Causative** A verbal notion that expresses the subject of the verb causing another (the secondary subject) to take some action. 2.3.1,d; 3.1.3; 3.1.4; 3.1.5; 3.1.6; 3.1.7
- **Circumstantial** Refers to the conditions under which, or the manner in which, an action is performed. 4.3.3,e; 4.5.2,c2; 5.2.11
- **Clause** A group of words with a subject and only one predicate, forming the basic unit of predication; can be verbal or verbless, dependent or independent. See Nominal Clause, Verbal Clause, Subordinate Clause, Independent Clause. 5.1

- **Cognate Accusative** Construction in which the verb and its object are derived from the same root. Example: "I *dreamt* a *dream*." 2.3, 1,c
- **Cohortative** An expression of strong wish or desire directed to the first person, the self. 3.3.3
- Comitative See Accompaniment.
- **Comparative** A relationship in which the quality or quantity of two or more nouns is compared. 2.5.4; 4.1.13,g; 4.2.4,c; 4.2.10,a; 5.2.6
- **Complement Accusative** An object that is used with certain intransitive verbs, which are modified to function as transitive verbs. 2.3.1,d
- **Complete Passive** A passive voice construction in which the agent of the verb is specified. Most English passive constructions are complete passives, whereas most Hebrew passive constructions are incomplete passives. Example: "The book was written *by the author*." See Incomplete Passive, Passive voice. 3.1.2,a
- **Complex Sentence** A sentence in which two or more clauses are joined together, and in which one clause is subordinate to another. 5.0
- **Concessive** Expresses an action or situation that is expected, or may lead to another action but, in fact, does not. 4.2.5,c; 4.3.2,b; 4.3.4,h; 5.2.12
- **Conditional** An action or situation that may or may not occur, dependent on the completion, or incompletion, of another action. Real conditions are those that can potentially be fulfilled, whereas unreal conditions are those that cannot or will not be fulfilled. 5.2.2
- **Conjugation** Different forms of the verb that make it specific for person, gender, and number, as well as aspect and type of action. 3.0
- **Conjunction** A particle that joins two or more words, or clauses, together. 4:3
- **Consequential** Expresses the potential or actual consequences of an action/situation. 3.5.1,b; 3.5.2,b; 3.5.3,b; 4.3.5
- **Construct Chain** A phrase with a construct noun followed by an absolute noun, expressing a genitive relationship between the two nouns. 2.3
- **Construct Noun** A form of the noun in Hebrew that functions to form genitive relationships. 2.3
- **Contingent** An action or idea that is dependent on another action/idea. 3.2.2,d
- **Coordinate** Expresses some notion of equality, either in stature, rank, or quality. 5.1
- **Coordinate Conjunction** A conjunction that joins nouns or clauses that have identical grammatical functions. 4.3
- **Coordinate Relationship** A linkage of two or more verbs, in which the leading, or governing, verb influences the aspect and mode of the linked verb. 3.5

- **Coordinating Adverb** An adverb that functions to show coordination between two clauses. 4.2
- **Correspondence** Expresses a complete equality in identity and quality between two substantives. 4.1.9,b
- **Customary** Describes action occurring in past or present time with regular repetition. 3.2.2,b
- **Dative** Case that marks the indirect object of a verb often the recipient in verbs of speaking or giving. Example: "I threw the ball *to my dog*." 4.1.10,e
- **Declarative** An expression of assessment, judgment, or evaluation. 3.1.3,d; 3.1.6,c; 4.1.2,c
- **Declensions** Different forms placed on nouns or adjectives that give them specific function and reference according to gender and number.
- **Definiteness (or Determination)** A noun having the definite article; refers to a specific or known person, place, or thing. 2.6
- **Degree** With the preposition ער expresses a large quantity or an extreme quality. 4.1.15,d
- **Demonstrative** A substantive that points to, or highlights, other substantives. Also labeled "deictic." Example: "He wrote *this* book." 2.5.1; 2.6.6; 4.2.8,b
- **Denominative** A verbal form that is derived from a noun. Recall that in Hebrew, the verbal form is primary, while nominal forms are derived from the verbal form. 3.1.3,b; 3.1.5,d; 3.1.6,d
- Desiderative Expresses a wish or desire. 5.3.2
- Determination See Definiteness. 2.6
- **Direct Object** The recipient of verbal action. Example: "He kicked *the ball.*" 2.3.1
- **Disjunctive** Establishes a contrast between words or clauses. 5.2.14; 5.3.1,b
- Distributive Reference to each and every member of a group. 2.7.1,d
- **Double Accusative** The use of two accusatives, or direct objects, with factitive or causative verbs; one receives action from the main subject, while the other receives action from the secondary subject. See Agency Nuance, Patiency Nuance, Secondary Subject. 2.3.1,e1
- **Duration/Durative** Expresses progressive, continuous action. 3.4.2,b; 3.4.3,b
- Duty An action that one is compelled to take. 4.1.16,b
- **Elliptical** Describes a clause from which certain parts of speech have been omitted. 5.3.5
- **Emotive** With the preposition על, indicates an expression of emotion. 4.1.16,i
- **Epexegetical** Providing an additional explanation or clarification. 2.2.6; 3.5.1,d; 4.3.3,d

- **Essence** Specifically with the  $\beth$  preposition, marks the predicate of a clause. 4.1.5,h
- **Estimative** Expresses judgment or assessment of the value, worth, significance, or status of a substantive. 4.1.10,k; 4.1.13,d

**Ethical Dative** Action that is on behalf of another; action that is meant to benefit another. 4.1.10,e1

- **Euphonic** A particle or part of speech that has no syntactical function, but is added for an audible effect. 3.3.2.
- **Evidential** Expresses the evidence or motivation for speech; does not focus on what is said, but the reason for speaking. 4.3.4,b
- **Exceptive** Taking out or omitting something from a situation. 4.3.2,d; 4.3.4,m; 5.2.7
- **Exclamatory/Exclamation** A sharp or sudden utterance; a sudden expression of emotion. 4.5; 5.3.1; 5.3.3
- Exhortation Speech or language intended to encourage. 3.3.3,c
- **Existential** Expresses the existence or presence of a substantive. 4.4; 5·3·4
- **Experience** Expressed by a verb, a state of mind, such as "love," "hate," "fear." 3.2.1,c
- **Explicative** A substantive that typically precedes the subject, and explains what is contained in the subject. 2.2.12; 2.4.5
- Factitive A verbal form that expresses the causing of a state. 3.1.3,a
- Fientive A verb that describes an action or change of state. 3.1.1,a
- **Final Clauses** Subordinate clauses that express the purpose or motivation behind the idea of the main clause. 5.2.3
- **Finite** A verb that is inflected for aspect, person, gender, and number; can stand as a predicate in independent clauses.
- **Frequentative** Describes action that is repeated, either over time or space. See Iterative, Pluralic. 3.1.3,c
- **Generic** Refers to a class or group of similar or identical substantives, rather than a specific individual or type. 2.6.5
- **Genitive** A relationship between nouns, often denoting attribute or possession, but other relationships as well; usually noted in English by the particle "of." 2.2
- **Gentilic** A noun that refers to a single member of a collective group, typically an ethnic or national group. 2.6.3.
- **Governing Verb** The first verb in a coordinate relationship within a narrative text; typically influences the aspect and usage of linked verbs that follow. 3.5
- *Hapax Legomenon* From Greek for "read once," an expression, word, or form that is used only once in a text.
- **Hendiadys** Expression of a singular idea by two independent words. 4.3.3,g

- **Immediacy** Expresses action that takes place with little or no loss of time; emphasis on the "present," relative to the narrative. 4.5.1,a
- **Imminence** An action or situation that is impending (taking place soon, relative to the context). 3.4.1,f
- **Imperative** An expression of will directed to another; expressing a command, entreaty, or wish to another party. 3.3.2
- **Incomplete Passive** A passive voice construction in which the agent of the verb is not expressed. Example: "*The book was written*." See Complete Passive. 3.1.2,a
- **Indefinite** Describes a noun without the definite article; typically refers to a general or generic, rather than specific, individual. 2.6
- **Indefinite Subject** An unspecified agent of a verb, usually expressed in English by "someone" or "anyone" 5.1.2,a3
- **Independent Clause** A clause that can stand on its own, with complete meaning. 5.2
- Indeterminate see Indefinite. 2.6
- **Indicative** A verbal form that expresses actions or states objective facts. 3.2.2
- **Indirect Object** A noun representing the secondary goal of the action of the verb. See Dative.
- **Indirect Questions** Phrases that report or refer to a question from another; the first question need not be explicitly stated. Example: "She knows *where the cat is hiding.*" 5.1.2,b2; 5.3.1
- **Infinitive** A verbal form, functioning as a verbal noun, that is, not limited to person, gender, or number. 3.4.1; 3.4.2
- **Inflexion** A morphological change in words that marks distinctions such as case, gender, number, tense, person, mood, or voice. 2.0
- Ingressive Expresses entry into a state. 3.1.2; 3.1.6,d
- **Instrumental** Identifies a person, place, or thing used as a means, agent, or tool with which an action is performed. 4.1.5,c
- **Intensive** Adding force or emphasis; specifically with Piel verb forms, expressing a repeated or enhanced action. 3.1.3,c; 4.2.12; 5.2.9
- Interest Action that is meant for the benefit of another; with the preposition ניל, indicates the one for which an emotion is intended. See Ethical Dative, Advantage. 4.1.16,i
- **Internal Accusative** An abstract noun that is identical to the action expressed by the verb. Example: "He will live *life*." See Cognate Accusative. 2.3.1,c
- **Interrogative** Expresses a question; often introduced by interrogative particles, which introduce questions. 4.3.4,n; 5.3.1
- Intransitive Verb A verb that does not have a direct object.
- Iterative Expresses repetition of an action over time. 3.1.3,c
- **Jussive** Expresses a command or wish predominantly to the third but to the second person as well. 3.3.1

- **Juxtaposition** Position of two parts of speech directly next to each other; the position of being directly next to another.
- **Leadword** The first noun in an appositional construction; typically modified by the second noun, termed the apposition. 2.4
- **Linked Verb** A verb in a coordinate relationship that is typically influenced by, and successive to, the governing verb. 3.5
- Locative Expresses the location of a noun or clause at a point in time or point in space. 4.1.10,b; 4.1.12,a; 4.1.15,a; 4.1.16,a; 4.1.17,d; 4.1.18,b; 4.2.7; 4.2.8,b; 4.2.16,a
- **Logical** Expresses the expected consequence or result of an action or situation. 3.3.2; 3.5; 4.2.1,b; 4.2.14,b; 4.5.3,c
- Maledictory An oath statement that express a curse or ill will. 5.3.2
- **Manner** The fashion, custom, or procedure of an action. 2.3.2,c; 3.4.2,c; 4.1.1,b; 4.1.5,i; 4.1.10,j; 4.1.16,e; 4.2.8,a; 4.2.10,a; 4.2.13,a; 4.2.17
- **Material** The matter with which some other thing is created or formed. 2.2.10; 2.3.2,f; 2.4.3; 4.1.5,c; 4.1.13,c
- **Means** Points to a substantive by or with which an action is performed. 2.2.9
- Measure Expresses the quantity of a substantive. 2.2.11; 2.4.4; 2.7.1,a
- **Merism** A construction involving two polar nouns to express the totality between those two poles. Example: "night and day," referring to totality of time, not just the two periods of "night" and "day." 4.2.7, n.29
- **Middle Voice** Verbal construct in which the subject and the object are identical, but without the emphasis on the subject acting upon itself, as with the reflexive. 3.1.2,b
- Modal A verbal form that expresses a strong wish or desire. 3.3
- **Mood** A verbal inflexion that expresses the factuality of the verb, either real, as with indicative verbs, or unreal, as with volitional or modal verbs.
- **Morpheme** The smallest meaningful unit of language; typically refers to a single "word" or particle, which is formed by a group of phonemes.
- **Morphology** The study of the formation of parts of speech.
- **Naming** With the definite article, the indication that a common noun is to be viewed as a proper noun. 2.6.3
- Narrative Presentation of events or theme as a story.
- **Nominal Clause** A unit of predication that does not include a finite verb; a predication formed by two substantives. 5.1.1
- Nominative A noun case that typically marks the subject of the verb. 2.1
- **Normative** Expresses classification of a substantive according to a standard; oftentimes, will express division of a larger group according to smaller divisions. 4.1.10,i
- **Northwest Semitic** Group of languages, including Aramaic, Ugaritic, and Phoenician, to which Hebrew belongs.
- **Oath** A solemn commitment to the truth of one's words; a commitment to take, or not to take, a certain action. 5.3.2

- **Object** see Direct Object; Indirect Object.
- **Obligation** An action that should or should not be done, according to societal, cultural, or historical traditions. 3.2.2,d3
- Optative Expresses a strong wish or desire. 5.3.3
- **Ordinal Number** A number designating the place of order occupied by an item in a sequence. 2.7.2
- **Paronomasia** A word play; figure of speech that involves a play on words, often with words that sound similar; within relative clauses, paranomasia often functions to express indefiniteness. 5.2.13,c
- **Participle** A part of speech having the characteristic of both verb and adjective. 3.4.3
- **Particle** A unit of speech with a connective or limiting function articles, prepositions, conjunctions, adverbs; or as utterances interjections and exclamations.
- Partitive A portion or part of a larger group. 4.1.13,e
- **Passive Voice** Construction in which the subject of the verb receives the action of the verb. Example: "The building *was built* by the contractor." 3,1.2,a
- **Patiency Nuance** With factitive verbs, indication that the secondary subject does not have a role in the action of the verb; the secondary subject is affected, only, and does not take other action. See Agency Nuance. 3.1.3
- **Perceptual** Marking the object of perception; identifying the thing that is perceived. 4.1.2,d; 4.3.4,j; 4.5.2,b
- **Performative** Describes an action that is completed by speech or utterance. Example: "Your name will no longer be Abram, but Abraham." 3.2.1,f
- **Permissive** Describes an action that is agreeable to the object, and allowed to occur by the subject. 3.1.6,e
- Personal Complement Marker of the indirect object of a verb. 4.1.17,c
- **Perspective** The specific view or perception of a speaker within the narrative.
- **Phoneme** A unit of letters in a language that functions to produce a single, distinctive sound in that language.
- **Phonology** The study of the sounds of a language.
- Phrase A group of words that can syntactically function as one word. 5.0
- **Pleonasm/Pleonastic** An expression that employs unnecessary words; a redundant expression. 5.1.1,a
- Pluralic Describes action that is intensified over space. 3.1.3,c
- **Possession** Expresses that a person, place, or thing belongs to another. 4.1.4,b; 4.1.10,f; 4.1.12,a; 4.4.1,b; 4.4.2,b
- **Postpositive** The location of a part of speech after a specified position; a postpositive infinitive, for example, comes after the finite verb it

modifies. An infinitive absolute that comes before the finite verb would be *prepositive*. 3.4.2,b

- **Predicate** A substantive or verb that makes an assertion about the subject of a sentence. 2.1.2; 2.5.2; 3.4.3,b; 4.4.2,c; 5
- **Predicate Adjective** An adjective that makes an assertion about the substantive it modifies. 2.5.2
- **Preposition** A part of speech that expresses various spatial, temporal, logical, or comparative relationships between nouns or clauses. 4.1
- **Preterite** Verbal form referring specifically to indicative action in the past. 3.2.2
- Privative Expresses removal or extraction. 4.1.13,f
- **Prohibition** An act of restraining; a command meant to restrain or stop an action. 4.2.3; 4.2.11
- **Promise** With imperatives, an assurance that the addressed party will take some action in the future. 3.3.2,c
- **Protasis** The introductory clause of a conditional statement the "if" clause. 5.3.2; 5.2.2; 5.2.6
- **Purpose** An expression of the reason or rationale behind an action. 2.2.8; 3.4.1,c; 3.5.5,b; 4.1.10,d; 4.1.11,a
- **Quasi-Active** Describes verbs that are normally passive, but used with identical subject and object. See Middle Voice. 3.1.2,b
- **Quasi-Datival** Describes constructions that indicate the indirect object of verbs. 4.1.10,e
- Rank With the preposition , על, indicates one's position or responsibility. 4.1.16,c
- **Reciprocal** Describes action in which two or more parties both affect each other; action that is mutual. 3.1.2,c; 3.1.5,b.
- **Recitative** With the preposition '>, introduces the contents of direct speech; functionally, marks a direct quotation. 4.3.4,1
- **Referential** With the definite article, refers to a person, place, or thing that has been previously mentioned in the narrative. 2.6.1
- **Reflexive** Describes action that is directed to the self; hence, the subject and the object of the verb are identical. 3.1.2,c; 3.1.5,a

Relative Clause A dependent clause that modifies the main clause. 5.2.13

Resolve Expression of self-determination. 3.3.3,a

- **Restrictive** Expressing a limitation, exception, or restriction of an action or idea. 4.1.17,e; 4.2.2,a; 4.2.15,a; 5.2.8.
- **Result** The end/outcome of an action. 2.2.9; 3.4.1,d; 4.3.4,d; 4.5.2,c5
- **Resultative** With Piel verbs, describes the bringing about of the outcome of an action designated by the verbal root. 3.1.3
- **Result Clause** A subordinate clause that expresses the end result of the idea of the main clause. 5.2.3,b

- **Rhetorical** Specifically with the particle '>, expresses a comparison established between two clauses, with the second clause typically indicating some persuasion. 4.3.4,c
- Rhetorical Question A question with no expected answer. 4.3.4,c; 5.3.1
- **Secondary Subject** With causative or factitive verbs, the substantive in which an action, state, or change of state is brought about by the main subject of the verb; also referred to as "under subject." 3.1
- **Sentence** A grammatical unit comprised of one or more clauses. See Simple Sentence and Complex Sentence. <u>5</u>; <u>5.3</u>
- **Simple Sentence** A sentence consisting of only one clause. See Complex Sentence. 5; 5.3
- **Simultaneous** Describes action that occurs, or a state that exists, at the same time as another. 3.4.1,b1; 3.4.1,b2; 3.4.2,c; 5.2.4
- **Solitary** With the definite article, identifies common nouns that refer to a singular entity, and hence, are on their way to becoming proper nouns. 2.6.4
- Source The point of origin, or the initiator of action. 4.1.13,a
- **Spatial** Expresses how a noun is related to space. 4.1.1,a; 4.1.2,f; 4.1.3; 4.1.4,d; 4.1.5,a; 4.1.6; 4.1.7,a; 4.1.10,a; 4.1.14,b; 4.1.16,a
- **Species** Classification of a substantive with others according to their type, group, or quality. 2.4.1
- **Specification** A clarification or explanation of verbal action or situation; an expression that specifically points out a substantive. 2.2.6; 2.3.2,e; 3.4.1,g; 3.5.1,d; 4.1.2,g; 4.1.5,e; 4.1.10,h
- **Stative Verb** A verb that expresses a state, rather than an action or change of state; usually, in English, this function is completed through predicate adjectives. 3.1.1,b
- **Stem (or Derived Stem)** A vowel pattern applied to a Hebrew verbal root, which functions to specify type of action. 3.1
- Subject The agent who or that performs the action of a verb. 2.1.1
- **Subjunctive** A verbal form that expresses nonfactual actions or situations, such as wishes, expectations, possible actions, and conditional actions. 3.2.2
- **Subordinate Clause** A clause that is associated with, and modifies the meaning of, an independent clause. 5.2
- **Subordinate Conjunction** A conjunction that joins a subordinate clause to a main, independent clause. 4.3
- Substantive A word or a group of words functioning as a noun.

Substitution The replacement of one substantive for another. 4.1.18,d

**Succession** In coordinate relationships, the primary relationship formed between governed and linked verbs denoting an action or situation that follows, either temporally or logically, another action or situation. 3.5

- Superlative The highest degree of an adjective. Example: "*tallest*." 2.2.13; 2.5.4,b; 4.1.10,k
- **Syntax** The organization of grammatical parts of speech to form meaningful units, such as phrases or clauses.
- **Telic** Expresses a goal, or movement toward a goal; with subordinate clauses, typically expresses the result of an action, or the motive behind an action.
- **Temporal** Expresses a substantive's relationship to time; with clauses, can express how the time of a subordinate clause relates to the time of the main clause. 3.4.1,b; 3.5.1,a; 3.5.2,a; 3.5.5,a; 4.1.1,b; 4.1.5,b; 4.1.6; 4.1.9,c; 4.1.10,c; 4.1.12,b; 4.1.13,b; 4.1.15,b; 4.2.1,a; 4.2.7; 4.2.14,a; 4.3.4,e; 4.5.2,c4; 5.2.4.
- **Tense** Distinction in the form of a verb to express the time or duration of action. Note that in BH, verbs are not strictly marked for tense; rather, their relationship to time is noted by aspect.
- **Terminative** Expressing movement, with special emphasis on the completed goal of that movement. 4.1.2,a; 4.1.5,a; 4.1.10,a; 4.1.16,a; 4.2.16,b
- **Transitive** A fientive verb with an identified direct object. 3.1.1,a
- **Verbal Clause** A clause in which the predicate is formed by a finite verb. 5.1.2
- **Verbal Complement** An infinitive that functions to complete the meaning of certain verbs. Example: "I know how *to play* the piano." 3.4.1,a3
- Vocative Grammatical case that marks direct address. 2.1.3; 2.6.2
- **Voice** Verbal inflection that indicates the relationship of the subject of the verb to the action of the verb. In English, voice is established, typically, by word order, whereas in BH, it is established by the derived stems. See Active Voice, Passive Voice. 3
- **Volitional** Describes a verbal form that expresses a choice or decision to which one has committed or will commit. 3.5.2,c

# Sources Consulted

- All abbreviations may be found in Patrick H. Alexander et al., eds., *The SBL Handbook of Style for Ancient Near Eastern, Biblical, and Early Christian Studies.* Peabody, Mass.: Hendrickson, 1999, 121–152.
- Aejmelaeus, Anneli. 1986. "Function and Interpretation of ∵ in Biblical Hebrew." *Journal of Biblical Literature* 105:193–209.
- Andersen, Francis I. 1970. *The Hebrew Verbless Clause in the Pentateuch*. Journal of Biblical Literature Monograph Series 14. Nashville, Tenn.: Published for the Society of Biblical Literature by Abingdon Press.
  - 1974. *The Sentence in Biblical Hebrew*. Janua Linguarum, Series Practica 231. The Hague: Mouton.
- Andersen, Francis I., and A. Dean Forbes. 1983. "'Prose Particle' Counts of the Hebrew Bible." Pages 165–83 in *The Word of the Lord Shall Go Forth.* Edited by Carol L. Meyers and M. O'Connor. Winona Lake, Ind.: Eisenbrauns.
- Bauer, Hans, and Pontus Leander. 1991. Historische Grammatik der hebräischen Sprache des Alten Testamentes. Halle: Niemeyer, 1918–22. Reprint, Hildesheim: Olms.
- Ben Zvi, Ehud, Maxine Hancock, and Richard Beinert. 1993. Readings in Biblical Hebrew: An Intermediate Textbook. New Haven, Conn.: Yale University Press.
- Bergen, Robert D., ed. 1994. Biblical Hebrew and Discourse Linguistics. Winona Lake, Ind.: Eisenbrauns for the Summer Institute of Linguistics.
- Bergsträsser, Gotthelf. 1962. Hebräische Grammatik mit Benutzung der von E. Kautzsch bearbeiteten 28. Auflage von Wilhelm Gesenius' hebräischer Grammatik. 2 vols. Leipzig: Hinrichs, 1918/1929. Reprint, Hildesheim: Olms.
  - 1983. *Introduction to the Semitic Languages*. Translated and supplemented by Peter T. Daniels. Winona Lake, Ind.: Eisenbrauns.

- Berlin, Adele. 1983. *Poetics and Interpretation of Biblical Narrative*. Sheffield, Eng.: Almond Press.
- Beyer, Klaus. 1969. Althebräische Grammatik: Laut- und Formenlehre. Göttingen: Vandenhoeck und Ruprecht.
- Blau, Joshua. 1976. A Grammar of Biblical Hebrew. Ponta Linguarum Orientalium NS XII. Wiesbaden: Otto Harrassowitz.
- Bodine, Walter R., ed. 1995. *Discourse Analysis of Biblical Literature: What It Is and What It Offers*. SemeiaSt. Atlanta: Scholars Press.
- Brockelmann, Carl. 1956. *Hebräische Syntax*. Neukirchen: Neukirchener Verlag.
- Chisholm, Robert B., Jr. 1998. From Exegesis to Exposition: A Practical Guide to Using Biblical Hebrew. Grand Rapids, Mich.: Baker.
- Claassen, W. T. 1983. "Speaker-Oriented Functions of KI in Biblical Hebrew." *Journal of Northwest Semitic Languages* 11:29–46.
- Comrie, Bernard. 1976. Aspect: An Introduction to the Study of Verbal Aspect and Related Problems. Cambridge Textbooks in Linguistics. Cambridge: Cambridge University Press.
- Crawford, Timothy G. 1992. *Blessing and Curse in Syro-Palestinian Inscriptions* of the Iron Age. New York: Peter Lang.
- Dobson, John H. 1999. *Learn Biblical Hebrew*. Dallas, Tex.: SIL International.
- Emerton, J. A. 1994. "New Evidence for Use of Waw Consecutive in Aramaic." Vetus Testamentum 44:255–58.
  - 2000a. "Two Issues in the Interpretation of the Tel Dan Inscription." *Vetus Testamentum* 50:29–37.
  - 2000b. "The Hebrew Language." Pages 171–99 in *Text in Context: Essays by Members of the Society for Old Testament Studies.* Edited by A. D. H. Mayes. New York: Oxford University Press.
- Endo, Yoshinobu. 1996. The Verbal System of Classical Hebrew in the Joseph Story: An Approach from Discourse Analysis. SSN 32; Assen: van Gorcum.
- Fassberg, Steven E. 1999. "The Lengthened Imperative קַטְּלָה in Biblical Hebrew." *Hebrew Studies* 40:7-13.
- 2001. "The Movement from *Qal* to *Pi''el* in Hebrew and the Disappearance of the *Qal* Internal Passive." *Hebrew Studies* 42:243–55.
- Garr, W. Randall. 1985. *Dialect Geography of Syria-Palestine, 1000–586 B.C.E.* Philadelphia: University of Pennsylvania.
- Goetze, Albrecht. 1942. "The So-Called Intensive of the Semitic Languages." Journal of the American Oriental Society 62:1–8.
- Gogel, Sandra Landis. 1998. A Grammar of Epigraphic Hebrew. SBLRBS 23. Atlanta: Scholars.
- Goshen-Gottstein, M. H. 1969. "The System of Verbal Stems in the Classical Semitic Langauges." Pages 70–91 in *Proceedings of the*

International Conference on Semitic Studies Held in Jerusalem, 19–23 July 1965. Jerusalem: Israel Academy of Sciences and Humanities.

- Greenberg, Moshe. 1965. Introduction to Hebrew. Englewood Cliffs, N.J.: Prentice-Hall.
- Harper, William R. 1888. *Elements of Hebrew by an Inductive Method.* 8th ed. New York: Charles Scribner's Sons.
- Harris, Zellig S. 1939. *Development of the Canaanite Dialects*. American Oriental Series 16. New Haven, Conn.: American Oriental Society.
- Hetzron, Robert. 1969. "The Evidence for Perfect \**Y'aqtul* and Jussive \**Yaqt'ul* in Proto-Semitic." *Journal of Semitic Studies* 14:1–21.
- Horsnell, Malcolm J. A. 1999. A *Review and Reference Grammar for Biblical Hebrew.* Hamilton, Ont.: McMaster University.
- Hostetter, Edwin C. 2000. An Elementary Grammar of Biblical Hebrew. Biblical Languages: Hebrew, 1. Sheffield, Eng.: Sheffield Academic Press.
- Huehnergard, John. 1987. "'Stative,' Predicative Form, Pseudo-Verb." Journal of Near Eastern Studies 46:215–32.

1988. "The Early Hebrew Prefix-Conjugations." Hebrew Studies 29:19-23.

- Jenni, Ernst. 1968. Das hebräische Pi'el: Syntaktisch-semasiologische Untersuchung einer Verbalform im Alten Testament. Zurich: EVZ.
  - 1981. Lehrbuch der hebräischen Sprache des Alten Testaments. Basel: Helbing und Lichtenhahn.
  - 1992. Die hebräischen Präpositionen, I: Die Präposition Beth. Stuttgart: Kohlhammer.
- Johnson, Bo. 1979. Hebräisches Perfeckt und Imperfekt mit vorangehendem w<sup>e</sup>. Lund: Gleerup.
- Joosten, J. 1998. "The Functions of the Semitic D Stem: Biblical Hebrew Materials for a Comparative-Historical Approach. *Or* 67:202–30.
- Joüon, Paul, and Takamitsu Muraoka. 1993. A Grammar of Biblical Hebrew.
  2 vols. Subsidia biblica 14/1–2. Rome: Editrice Pontificio Istituto Biblico.
- Kaufman, Stephen A. 1996. "Semitics: Directions and Re-Directions." Pages 273–82 in *The Study of the Ancient Near East in the Twenty-First Century: The William Foxwell Albright Centennial Conference*. Edited by Jerrold S. Cooper and Glenn M. Schwarz. Winona Lake, Ind.: Eisenbrauns.
- Kautzsch, Emil, ed. 1910. *Gesenius' Hebrew Grammar*. Translated and revised by A. E. Cowley. 2d English ed. Oxford: Clarendon.
- Kelley, Page H. 1992. *Biblical Hebrew: An Introductory Grammar*. Grand Rapids, Mich.: Eerdmans.
- Kelley, Page H., Terry L. Burden, and Timothy G. Crawford. 1994. A Handbook to Biblical Hebrew: An Introductory Grammar. Grand Rapids, Mich.: Eerdmans.

- Kittel, Bonnie Pedrotte, Vicki Hoffer, and Rebecca Abts Wright. 1989. *Biblical Hebrew: A Text and Workbook*. New Haven/London: Yale University Press.
- Kouwenberg, N. J. C. 1997. Gemination in the Akkadian Verb. Studia Semitica Neerlandia. Assen: Netherlands: Van Gorcum.
- Krašovec, Jože. 1977. Der Merismus im Biblisch-Hebräischen und Nordwestsemitischen. BibOr 33. Rome: Editrice Pontificio Istituto Biblico.

1983. "Merism – Polar Expression in Biblical Hebrew." Biblica 64:231–39.

- Kroeze, Jan H. 2001. "Alternatives for the Nominative in Biblical Hebrew." Journal of Semitic Studies 46:33–50.
- Lambdin, Thomas O. 1969. "Review of Ernst Jenni, Das hebräische Pi'el: Syntaktisch-semasiologische Untersuchung einer Verbalform im Alten Testament." Catholic Biblical Quarterly 42: 388–89.
  - 1971a. Introduction to Biblical Hebrew. New York: Charles Scribner's.

1971b. "The Junctural Origin of the West Semitic Definite Article." Pages 315–33 in *Near Eastern Studies in Honor of William Foxwell Albright.* Edited by Hans Goedicke. Baltimore: The Johns Hopkins Press.

- Landes, George M. 2001. Building Your Biblical Hebrew Vocabulary: Learning Words by Frequency. Atlanta: Scholars Press.
- Lowery, Kirk E. 1995. "The Theoretical Foundations of Hebrew Discourse Grammar." Pages 103–30 in *Discourse Analysis of Biblical Literature: What It Is and What It Offers*. Edited by Walter R. Bodine. SemeiaSt. Atlanta: Scholars Press.
- Martin, James D. 1993. *Davidson's Introductory Hebrew Grammar.* 27th ed. Edinburgh: T & T Clark.
- McCarthy, Walter. 1980. "The Uses of *w<sup>e</sup> hinneh* in Biblical Hebrew." *Biblica* 61:330–42.
- McFall, Leslie. 1982. *The Enigma of the Hebrew Verbal System*. Sheffield, Eng.: Almond Press.
- Meier, Samuel A. 1992. Speaking of Speaking: Marking Direct Discourse in the Hebrew Bible. VTSup 46. Leiden: E. J. Brill.
- Meyer, Rudolf. 1992. *Hebräische Grammatik*. Berlin: Walter de Gruyter. [English trans: *Hebrew Grammar*. 3d ed. Translated by P. T. Daniels. Winona Lake, Ind.: Eisenbrauns.]
- Miller, Cynthia L. 1994. "Introducing Direct Discourse in Biblical Hebrew Narrative." Pages 199–241 in *Biblical Hebrew and Discourse Linguistics*. Edited by R. D. Bergen. Winona Lake, Ind.: Eisenbrauns for the Summer Institute of Linguistics.
  - 1996. The Representation of Speech in Biblical Hebrew Narrative: A Linguistic Analysis. HSM 5.5. Atlanta: Scholars Press.
  - 1999. The Verbless Clause in Biblical Hebrew: Linguistic Approaches. Linguistic Studies in Ancient West Semitic 1. Winona Lake, Ind.: Eisenbrauns.

- Moscati, Sabatino, ed. 1980. An Introduction to the Comparative Grammar of the Semitic Languages: Phonology and Morphology. 3d ed. Porta Linguarum Orientalium 6. Wiesbaden: Harrassowitz.
- Muilenberg, James. 1961. "The Linguistic and Rhetorical Usages of the Particle ℃ in the Old Testament." *Hebrew Union College Annual* 32:135– 60.
- Muraoka, T. 1978. "On the So-Called Dativus Ethicus in Hebrew." Journal of Theological Studies 29:495–98.
  - 1985. *Emphatic Words and Structures in Biblical Hebrew*. Jerusalem and Leiden: Magnes Press, The Hebrew University, E. J. Brill.
- Niccacci, A. 1990. *The Syntax of the Verb in Classical Hebrew Prose*. JSOTSup 86. Sheffield, Eng.: JSOT Press.
- Polzin, Robert. 1976. Late Biblical Hebrew: Toward an Historical Typology of Biblical Hebrew Prose. HSM 12. Missoula, Mont.: Scholars.
- Pratico, Gary D., and Miles V. Van Pelt. 2001. *Basics of Biblical Hebrew Grammar*. Grand Rapids, Mich.: Zondervan.
- Rainey, Anson F. 1986. "The Ancient Hebrew Prefix Conjugation in Light of Amarnah Canaanite." *Hebrew Studies* 27:4–19.
  - 1988. "Further Remarks on the Hebrew Verbal System." *Hebrew Studies* 29:35-42.
  - 1996. Canaanite in the Amarna Tablets: A Linguistic Analysis of the Mixed Dialect Used by the Scribes from Canaan. 4 vols. Handbuch der Orientalistik 25/1-4. Leiden: Brill.
- Revell, E. J. 1989. "The System of the Verb in Standard Biblical Prose." *Hebrew Union College Annual* 60:1–37.
- Richter, Wolfgang. 1978–80. *Grundlagen einer althebräischen Grammatik.* 3 vols. St. Ottilien: EOS Verlag.
- Rocine, B. M. 2000. *Learning Biblical Hebrew: A New Approach Using Discourse Analysis.* Macon, Ga.: Smyth and Helwys.
- Rooker, Mark F. 1990. *Biblical Hebrew in Transition: The Language of the Book* of *Ezekiel*. JSOTSup 90. Sheffield, Eng.: JSOT Press.
- Ross, Allen P. 2001. *Introducing Biblical Hebrew*. Grand Rapids, Mich.: Baker.
- Rubinstein, Eliezer. 1952. "A Finite Verb Continued by an Infinitive Absolute in Hebrew." *Vetus Testamentum* 2:262–67.
  - 1998. *Syntax and Semantics: Studies in Biblical and Modern Hebrew.* Texts and Studies in the Hebrew Language and Related Subjects, 9. Tel Aviv: The Chaim Rosenberg School of Jewish Studies.
- Sáenz-Badillos, Angel. 1993. *A History of the Hebrew Language.* Translated by John Elwolde. Cambridge: Cambridge University Press.
- Schneider, Wolfgang. 1989. Grammatik des Biblischen Hebräisch. Munich: Claudius Verlag.

- Schoors. A. 1981. "The Particle 'C." Pages 240–76 in *Remembering All the Way*. Oudtestamentische Studiën 21. Edited by B. Albrektson. Leiden: Brill.
- Segal, M. H. 1927. A Grammar of Mishnaic Hebrew. Oxford: Clarendon.
- Segert, Stanislav. 1984. A Basic Grammar of the Ugaritic Language. Berkeley: University of California Press.
- Seow, C. L. 1995. A Grammar for Biblical Hebrew. 2d ed. Nashville, Tenn.: Abingdon.
- Shulman, Ahouva. 2001. "Imperative and Second Person Indicative Forms in Biblical Hebrew Prose." *Hebrew Studies* 42:271–87.
- Sinclair, Cameron. 1999. "Are Nominal Clauses a Distinct Clausal Type?" Pages 51–75 in *The Verbless Clause in Biblical Hebrew: Linguistic Approaches.* Edited by Cynthia L. Miller. Linguistic Studies in Ancient West Semitic 1. Winona Lake, Ind.: Eisenbrauns.
- Smith, Mark S. 1991. The Origins and Development of the Waw-Consecutive: Northwest Semitic Evidence from Ugarit to Qumran. HSS 39. Atlanta, Ga.: Scholars Press.
- Thomas, D. Winton. 1954. "A Consideration of Some Unusual Ways of Expressing the Superlative in Hebrew." *Vetus Testamentum* 3:209–24.
- Van der Merwe, Christo H. J. 1994. "Discourse Linguistics and Biblical Hebrew Linguistics." Pages 13–49 in *Biblical Hebrew and Discourse Linguistics*. Edited by R. D. Bergen. Winona Lake, Ind.: Eisenbrauns for the Summer Institute of Linguistics.
- Van der Merwe, Christo H. J., Jackie A. Naudé, and Jan H. Kroeze. 1999. A Biblical Hebrew Reference Grammar. Biblical Languages: Hebrew 3. Sheffield, Eng.: Sheffield Academic Press.
- Waltke, Bruce K., and M. O'Connor. 1990. *An Introduction to Biblical Hebrew Syntax.* Winona Lake, Ind.: Eisenbrauns.
- Weingreen, J. 1959. A Practical Grammar for Classical Hebrew. 2d ed. Oxford: Clarendon Press.
- Westermann, Claus. 1991. *Basic Forms of Prophetic Speech*. Translated by H. C. White. Louisville, Ky.: Westminster/John Knox.
- Williams, Ronald J. 1976. *Hebrew Syntax: An Outline*. 2d ed. Toronto: University of Toronto.
- Zevit, Ziony. 1988. "Talking Funny in Biblical Henglish and Solving a Problem of the *yaqtul* Past Tense." *Hebrew Studies* 29: 25–33.
  - 1998. *The Anterior Construction in Classical Hebrew*. SBLMS 50. Atlanta: Scholars Press.
- Zewi, Tamar. 1996. "The Particles און הגָה in Biblical Hebrew." *Hebrew Studies* 37:21–37.

# Subject Index

Page numbers in **bold italics** refer to discussions that define the subject.

accusative case, 4-5, 13-14, 13-21, 26, 171-73, 195 adverbial accusative, 18, 18-21 cognate accusative, 15, 197 objective accusative, 14-15, 14-18 adjectives, 8, 24, 24-28 attributive adjective, 8, 10, 16, 25-26, 78-79, 196 morphology of, 24 predicate adjective, 26, 79-82, 203 substantive adjective, 26, 82-83 adverbs, 14, 127, 127-42, 195 adversative clauses, see clauses and sentences agency, 12, 37, 48-49, 195 Akkadian, 4 D stem in, 42 n18 Amarna letters, 165 n10 Amorite, 4 apposition, 7, 21, 21-24, 26, 196 Arabic, 4 Aramaic, 13 asseveratives, see intensives casus pendens, see nominative absolute causal clauses, see clauses and sentences causation, 11, 105, 108-9, 115, 117, 119-20, 122, 149, 160, 196 circumstantial clauses, see clauses and sentences Classical Biblical Hebrew, 3

clauses and sentences, 162-64, 196 adversative, 181-82 causal, 178-79 circumstantial, 182-83, 196 comparative, 179–80 concessive, 183-84, 197 conditional, 173-74 coordinate, 164-65 disjunctive, 186 elliptical, 192 exceptive, 180-81 existential, 191 final, 174–76, 199 intensive, 181 interrogative, 187–88 negative, 191–92 nominal, 165–67, 201 oath, 188–89 relative, 184–86, 203 restrictive, 181, 203 subordinate, 143, 164-65, 204 substantival, 171-73 temporal, 176-78 verbal, 167–70, 205 predication of, 167-169 subject of, 169-170 wish, 189–90 cohortative, 65-66, 197 comparative, 27, 118, 136, 197 comparative clauses, see clauses and sentences complement clauses, see subordinate clauses concessive clauses, see clauses and sentences conditional clauses. see clauses and sentences

conditionality, 128, 143-44, 147, 151-52, 160, 173-74 conjunctions, 143, 143-56, 197 construct state, 5, 8, 197 contingency, 59–60, 197, 203 contrast, 92–93, 104, 123, 146, 152, 161 definite article, 7 definite direct object marker, 5, 13, 24, 198 demonstrative, 32, 135, 157, 198 description, see nominal clause; predicate nominative determination, 28–29, 28–32, 198 directive he, see he locale disjunctive clauses, see clauses and sentences elliptical clauses and sentences, see clauses and sentences emphasis, 34, 74-76, 169-70 epexegesis, 86-87, 147, 198 exceptive clauses, see clauses and sentences exclamation, 158, 189 existential sentences, see clauses and sentences explicative, 12, 24, 199 final clauses, see clauses and sentences focus marker, see nominative absolute genitive case, 4-5, 8, 8-13, 26, 113, 171-72, 199 he locale, 4-5, 18 hendiadys, 148-49, 199 hinneh, 157–61 with participles, 80 ngg Hiphil, 48–52 Hithpael, 47-48 Hophal, 52–53 identification, see nominal clause; predicate nominative imperative, 7, 63–65, 200 imperfect, 56-60 plus waw consecutive, 84-87 indefinite article, 28 indirect object, 98, 111-112 infinitive asbolute, 73-77 infinitive construct, 67–73 instrument, see means intensive clauses, see clauses and sentences intensives, 129–30, 131, 133, 138, 141, 153, 190

interrogative sentences, see clauses and sentences interrogatives, 146, 155, 170, 187-88 Jussive, 61-63, 200 Late Biblical Hebrew, 3 material, 12, 20, 23, 201 means, 11, 104, 201 measure, 12, 23, 33, 201 merism, 134, 201 multiplication, 34 morphology, 1, 201 negation, 137, 156, 175, 191-92 negative sentences, see clauses and sentences Niphal, 38-41 nominal clause, 7, 26, 164-67, 201; see also clauses and sentences nominative absolute, 5, 7 nominative case, 4–5, 6, 26, 171–72, 201 Northwest Semitic languages, 67 n69 noun cases, 4-6, 14, 196 numerals, 23, 32–35, 196, 202 oath sentences, 188-89 oaths, 145, 201 paronomasia, 185–86, 202 participle, 77-83, 202 particles, 95, 95-161, 202 of existence/nonexistence 156 - 57of interest (hinneh), 157-61 patiency, 37, 41-43, 202 perfect, 54-56 plus waw consecutive, 87-91 phrases, 162, 202 Piel, 41-45 poetry, 170 possession, 8-9, 32, 102, 112, 157, 202 postconstructus, see genitive case predicate nominative, 5, 6 for description, 6 for identification, 6 prepositions, 95-96, 203 product, 21 prohibition, 63, 130, 203 Pual, 46–47 purpose, 11, 71, 91-92, 111, 115, 203 Qal, 37-38 Qal passive, 38–39

#### Subject Index

relative clauses, see clauses and sentences restrictive clauses, see clauses and sentences result, 11, 71, 85, 88, 150-51, 161, 175, 203 resumptive pronoun, 185 sentences, 162-64, 204; see also clauses and sentences sequence, 84-85, 88, 91 species, 22 specification, 10-11, 20, 23, 72, 100-101, 104, 113, 204 stative verbs, 6 subject nominative, 6 subordinate clauses, 7, 171, 171-86, 204 substantival clause, see clauses and sentences superlative, 13, 28, 205 supplement clauses, see subordinate clauses syntax, 1, 205 temporal clauses, see clauses and sentences temporals, 69-70, 97, 103, 110, 111, 116, 117, 120, 127, 134, 139-40, 151, 160-61, 169, 176-78, 205 Ugaritic, 4, 62

verbal clause, 167–70; *see also* clauses and sentences

verbal sequences, 83-94 verbs aspect, 36, 53–60, 196 causative, 49-50, 53 declarative, 45, 46, 51 denominative, 44, 46, 48, 51, 53, 198 factitive, 44, 46, 199 fientive, 38, 199 frequentative, 45, 46, 199 future, 58 imperative, 63-64 iterative, 48 middle voice, 40, 201 modals, 60–61, 60–66, 201 nonfinites, 66-67, 66-83 passive voice, 39, 200, 202 permissive, 51 preterite, 60, 203 reciprocal, 48 reflexive voice, 40, 47 resultative, 46 stative, 38, 41, 51, 53, 55, 204 stem, 37, 204 tense, 36, 205 voice, 205 vocative, 5, 7, 30, 205 waw consecutive, 83 wish sentences, see clauses and sentences word order, 5-6, 169-70 in poetry, 170

# Scripture Index

Brackets indicate English references where English versification differs from the MT.

Genesis		3:19	70, 149
		3:22	140
1:1	6, 14, 31, 164, 169	3:24	$4^{8}$
1:2	120–121, 148	4:1	84, 101
1:3	6, 61, 189	4:2	24
1:3-4	29-30	4:7	110
1:4	15	4:8	24, 87
1:5	35, 55, 93	4:9	55, 187
1:7	184	4:11	184-185
1:16	31, 111	4:13	26, 27
1:22	61, 73	4:15	34
1:26	66n68, 106	4:23	111
1:29	15	4:25	126, 129
1:31	26	4:26	128, 1689–9
2:1	85	5:1	103, 106
2:4	30, 103	5:4	97
2:5	60, 156, 191	5:22	31, 101
2:5	18, 21, 38, 117	5:24	38
2:8	142	5:32	10
2:10	80	6:1-2	151
2:15	12	6:5	172
2:15		6:7-8	146
2:10	59 67, 137	6:8	169
2:17	66n68, 137n32, 191	6:9	48
		6:10	33
2:19	58, 71, 192	6:11	116
2:20	191	6:13	6, 149
2:22	112	6:14	59
2:24	122	6:17	20, 23, 81
3:1	99, 191	6:22	45
3:3	155-156	7:1	45 63
3:5	80, 109	7:4	19, 81
3:6	26, 132, 173, 181	7:7	119–120
3:8	47	7:9	34-5
3:10	170	7:16	34-5 107-8
3:11	170, 187	7:21	107-05
3:12	7	7:23	129
3:14	118, 179	7:23 8:7	76
3:15	168	8:10	12
3:16	105	0.10	14

8:11	111	17:00 01	1.46
8:15		17:20–21 18:1	146 86, 183
	73	18:2	
9:5 9:6	170 39n10	18:4	33, 80 126
e e	82	18:5	120
9:10			
9:11	39110, 117	18:7-8	30
10:11	18	18:9	170
10:12	25	18:13	131
10:18	31	18:17-18	183
10:25	12	18:18	75 6 -
10:32	31	18:21	65
11:3	15, 66	18:22	115, 139
11:4	147, 182	18:23	124
11:7	175	18:26	59, 143-144
11:9	137, 168	18:27	184
11:30	156, 191	18:30	63, 91
11:31	120	19:1	80
12:1	63, 114	19:9	75
12:2	44, 112	19:13	81
12:6	102, 103	19:19	50
12:7	99	19:20	62, 142
12:8	182	19:22	70
12:11	72, 176	19:23	121
12:12	93	19:29	192
12:13	115	19:33	147
12:14	69–70	20:3	122
12:15	18-19	20:5	112
13:2	138	20:6	112
13:3	107	20:7	108, 166
13:3 Qere	142	20:9	59
13:8	22, 63, 130	20:10	150-151
13:9	189	20:11	141, 156, 190
13:17	103	20:13	9
14:19	79, 82, 114, 146	21:20	101
15:1	9, 134, 177	21:30	173
15:2	19	22:1	158
15:3	161	22:7	158
15:4	152	22:12	63
15:5	191	22:14	175
15:7	50, 71, 166	22:16	40
15:11	121	22:18	179
15:12	31, 72	23:6	13
15:13	185	23:9	106
15:14	105	23:10	82
15:15	58	23:19	100
15:18	12, 56	24:8	141
16:8	õ	-	104 105
16:11	59, 80, 119	24:12	124 - 125
	59, 80, 119 99	24:12 24:18	124-125 148
16:12	59, 80, 119 99 104		
	99	24:18	148
16:12 17:5 17:10	99 104	24:18 24:25	148 133
17:5	99 104 182	24:18 24:25 24:30	148 133 160, 192
17:5 17:10	99 104 182 77	24:18 24:25 24:30 24:31	148 133 160, 192 59
17:5 17:10 17:11 17:15	99 104 182 77 20	24:18 24:25 24:30 24:31 24:42	148 133 160, 192 59 19
17:5 17:10 17:11	99 104 182 77 20 152 144	24:18 24:25 24:30 24:31 24:42 24:49 24:51	148 133 160, 192 59 19 157, 186
17:5 17:10 17:11 17:15 17:1 <u>7</u>	99 104 182 77 20 152	24:18 24:25 24:30 24:31 24:42 24:49	148 133 160, 192 59 19 157, 186 62, 116

<b>a</b> • /			
Genesis (con		33:9	157
<sup>2</sup> 4:55	97	33:11	53
24:56	183	33:18	19
25:11	125	34:10	116
25:28	169	34:15	173
25:33-34	170	35:4	30
26:11	53	36:7	27
26:28	62	36:8	166
27:3	18	37:7	80–81, 160
27:4	92	37:10	99, 110
27:28	12	37:21	20
27:29	167, 190	37:25	80
27:33	16, 188	37:29	160
27:34	70	37:34	48
27:36	155	37:35	20
27:39	190	38:11	20
27:42	53	38:17	146
27:43	114	39:2	85
28:6	931115	39:6	10
28:16	157, 190, 191	39:9	105
28:17	145	39:23	80, 105
28:20-21	90	40:10	70
29:2	169	40:15	31, 150
29:7	67	40:16	131, 181
29:14	130	40:23	86-87
29:14	106	41:1	30
29:19	67	41:3	101
30:6	18	41:13	136, 179–180
30:7		41:19	130, 179-100
	35	41:28	81
30:14	103-104	-	62
30:16	104	41:33	20
30:21	97 62	41:40	20 56
30:24		41:41	
30:40	123	41:43	77
31:3	63	42:7	102
31:8	148	42:13	26, 34
31:23	19 86	42:16	153, 189
31:26		42:21	190
31:32	55	$4^{2:35}$	81
31:33-34	93-94	43:3	75
31:35	154	43:4	157
31:37	135	43:10	152, 174
31:39	147	43:12	79
31:53	107	43:14	26
31:54	102	43:22	188
32:4	98	43:23	190
32:11	118	44:15	109
[32:10]		44:18	180
32:15	23	44:28	75
32:25	70	44:31	110
32:27	145, 180	44:33	62
[32:26]		47:31	32
32:29	152	48:1	158-159
[32:28]		48:9	98
33:2-3	168	48:10	69
33:3	34	48:11	52

	. 9 .		
48:19	182	10:24	133
49:10	119	11:5	7 <sup>8</sup> –79
49:31	169	12:15	35
50:5-6	64	12:16	39n10
50:9	133	12:41	8–9
50:19	187	12:42	116
		12:46	39
		13:17	184
Exodus		14:5	172
Exouus		14:10	51
1:7	88	14:19	119
1:10	151	14:28	103
1:14	105	14:31-15:1	127-128
2:10	30, 86	15:1	60
2:11	22, 97	15:19	50
2:18	67	16:7	105
3:1	81	16:10	98
3:3	92	16:16	169
3:5	185	16:27	118
3:11	151	16:29	126, 130
3:12	150	17:7	122, 192
		18:11	27
3:14	135	18:16	90
4:5	175	18:17	138
4:6	46	18:17	0
4:9	133		27, 117–118
4:10	10	18:19	62, 124
4:14	$\frac{71}{6}$	19:5	90
4:18	64, 66	19:9	125
5:1	97	19:13	152
5:10	156	19:17	111
5:15	135	20:8	77
5:16	169	20:15	60
6:2	6	20:21	142
6:3	41, 106	21:2	151
6:6	88, 119	21:16	186
6:7	154	21:32	143
6:11	63	21:33	151
6:26	122-123	22:7	9, 144
7:3	50	[22:8]	
7:10	85	22:23	104
7:16	135	23:7	130
7:23	132	23:9	179
7:25	70, 176	23:17	34
8:1	105	23:24	45
8:2	121	24:3	19
8:4	92	24:4	167
8:24	141	24:5	22
9:7	38	24:7	147
9:23	92-93	25:18	21
9:24	177-178	28:3	11
9:27	32	29:33	46
9:31	3- 46	29:37	13
10:3	120	30:36	76
10:5	168	31:13	44
10:13	178	31:16	44 73
10:13	126	31:18	73 79
10.23	140	30	19

Exodus (cont	.)
32:3	16
32:8	77
32:9	10
32:26	41
32:30	108, 140
32:34	88
33:7	76 186
33:19	
33:21 34:6	102 10 <b>-</b> 11
34:0 34:11	146
34:12	40
34:25	123
34:27	114
35:35	17
36:4	26
39:4	$4^{6}$
39:10	35
39:17	23
40:32	14
Leviticus	
1:1	86
9:7	108
10:16	$4^{6}$
11:4	31
13:5	160
13:11	166
13:44 13:46	45 172
16:22	103
16:30	116
18:18	100
21:9	22
22:4	12
22:6	155
23:27	192
24:5	21
25:20	174
25:39	40
26:18 26:28	34
20:28 26:42	131
20:42 26:46	131 107
20:40	167
	- 1
Numbers	

#### Numbers

1:1	86	54	-1
1:4 3:23	113 96	Deuteror	nomy
3:23 4:19	89 89	1:4	176
4:29	113	1:11	109

5:8	156
5:19	126
6:24	15, 62
6:25	98
7:17	23
9:15	109
10:34	134
11:4	148
11:10	20
11:11	174
11:29	190
11:33	16
12:5	19
13:21	110-111
13:30	89
13:30	189
	138
14:7 14:8	
	89-90
14:11	104
14:23	145
15:15	180
15:35	77
16:13	150
17:12	160
[16:47]	0
17:15	178
[17:16]	
17:19	115
[17:4]	
18:2	101
18:15	129
18:24	137
20:20	105
20:23	121
21:9	169
22:3	120
22:8	32
22:32	170
23:15	135
24:2	121
24:5	188
24:9	31
24:17	56
25:13	87
28:19	23
28:25	137
	112
31:4 31:8	123
31:16	105
32:14	98
32:51 33:51	151
33.51 34:2	<sup>1</sup> 51 24
34.4	-4
Develo	

1:18	18
1:28	118
1:37	133
1:38	142
1:42	149
1:43	51
1:45	52
2:4	138
2:27	66, 183
	. 0
3:2	80
3:25	25
3:26	115
3:28	50, 168
4:5	17, 44
4:6	25, 141
4:10	175
4:19	31
4:26	75
4:27	88, 103
4:34	78
6:7	104
6:8	88
6:20-21	151-152
7:6	
	71
7:9	148
8:11	191–192
8:20	179
9:24	81
10:1	12
10:16	139
10:19	88
11:4	44
11:10	15n25
11:27	174
12:9	120
13:2	98
13:4 [13:3]	60
14:1	111
14:10	137
14:12	172
15:14	112
16:4	111
16:6	100, 176
16:21	23
17:14	97
18:13	125
20:12	123
20:19	155
21:14	144, 174
22:8	168
22:9	15125
23:7	111
[23:6]	
24:16	105
-4.10	
25:1	51, 107
26:5	106
26:9	18

26:11	104
27:14	16
28:12	9
28:43	9 122
28:49	185
28:54	28
29:14	32, 157
[29:15]	
29:19	69
29:23	187
[29:24]	•
30:8	148
31:16	81
31:19	112
32:6	
	87
32:8	114
32:24	11
32:39	125
33:28	19
33:29	39n10
Lochuo	
Joshua	
1:2	185
1:8	89, 128
1:9	130
2:2	159
	188
2:4	
2:5	72, 76
2:10	50
2:15	108
3:1	177
3:12	116
3:17	76
4:4	23
4:13	109
4:17	167
5:4	104
5:13	144, 187
5:14	98, 182
6:10	50
6:13 Qere	76
6:14	135
0.14	
	133
6:27	86
6:27 7:2	86 125
6:27 7:2 7:25	86 125 21
6:27 7:2 7:25 8:32	86 125 21 163n3
6:27 7:2 7:25 8:32 9:1	86 125 21 163n3 30
6:27 7:2 7:25 8:32 9:1 9:13	86 125 21 163n3 30 179
6:27 7:2 7:25 8:32 9:1	86 125 21 163n3 30
6:27 7:2 7:25 8:32 9:1 9:13	86 125 21 163n3 30 179
6:27 7:2 7:25 8:32 9:1 9:13 10:20 11:6	86 125 21 163n3 30 179 138
6:27 7:2 7:25 8:32 9:1 9:13 10:20	86 125 21 163n3 30 179 138 191 31
6:27 7:2 7:25 8:32 9:1 9:13 10:20 11:6 11:9 11:21	86 125 21 163n3 30 179 138 191 31 124
6:27 7:2 7:25 8:32 9:1 9:13 10:20 11:6 11:9 11:21 13:6	86 125 21 163n3 30 179 138 191 31 124 119
6:27 7:2 7:25 8:32 9:1 9:13 10:20 11:6 11:9 11:21 13:6 14:9	86 125 21 163n3 30 179 138 191 31 124 119 145
6:27 7:2 7:25 8:32 9:1 9:13 10:20 11:6 11:9 11:21 13:6	86 125 21 163n3 30 179 138 191 31 124 119

Joshua (cont.	)	1:11	9
20:6	31	1:23	62
22:34	166	2:6	44
23:2	38	2:16	56
23:6	191	2:27	188
24:20	90	2:34	34
	5	3:9	164
Judges		4:19 5:2	72, 101 101
3:7	87	5:6	98
3:18	176		90 70
3:19	7	5:9 7:9	12
3:20	32, 126	7:3 7:14	107
4:5	126	7:14	85-6
4:9	181	8:1	21
4:11 Qere	102	9:2	27
$5:21^{-2}$	20		-7 102, 160
5:31	136	9:7 9:12	1102, 100
6:12	167	10:8	61, 161
6:13	173	10:19	113
6:18	168	10:19	62
6:26	58	11:3	90
6:27	134	12:2	90 117
7:5	59	12:12	58
7:14	145	12:12	90 97
7:17	160-161	12:14	97 129
8:19	174	13:14	45
8:30	23	13:17	45 20
9:21	120	13:20	18
9:29	190	14:9	126
9:51	142	14:11	47
9:53	34	14:44	153
9:54	113	15:23	108-109
10:13	136, 149	15:28	88
11:1	10	15:30	89
11:12	99	15:34	93
11:34	143	15:35	101
12:8	116	16:4	187
13:8	139	16:6	130
13:9	147	16:11	28
13:21	128	16:18	113
14:18	27, 118	17:16	5 <sup>2</sup>
15:7	153	17:28	55
16:15	187	17:35	116
17:2	102	17:40	147
18:3	125	17:50	85
18:12	96	17:55	7
18:25	156	17:58	30
19:13	89	18:4	48
19:14	30	18:6	69
19:18	166	18:10	32
19:20	121	18:15	173
20:37	99	18:22	16
1 Samuel		18:25	155
		19:3	173
1:1	29, 33-34, 86	19:6	145
1:6	108	19:10	87

20:1	116
20:14	125
20:31	10
22:5	92
23:1	80
23:3	132, 166
23:13	186
24:7	117
[24:6]	
24:18	168
25:7	39n10
25:8	91
25:16	121
25:23	148
25:34	153
26:12	13n19
26:13	85
26:15	99, 150
26:16	189
26:21	134
26:22 Qere	7
27:9	147
28:3	94
29:2	35
29:7	106
30:2	183

#### 2 Samuel

1:4	38, 134
1:25	188
2:6	62-63, 148
2:17	120
2:27	152
3:11	179
3:12	161
3:13	155
3:16	76
3:21	98
3:35	189
5:3	31
5:4	176
5:13	97
7:2	146
7:12	38
7:19	38
7:25	29
8:8	138-139
8:16	122
9:1	154-155
11:5	166
11:10	97
11:11	145
11:14	100
11:24	118
12:5	153
0	00

12:6	34 6, 166
12:7	6, 166
12:13-14	180-181
13:12	59
13:23	9
13:28	110
14:4	30
14:5	86, 147
14:21	56
15:4	190
15:20	186
15:21	186
16:7	10
16:9	66
17:16	182
18:3 Qere	154
18:11	72, 121
18:12 Qere	192
18:14	182
18:30	135
18:32	187
19:38	62
[19:37]	
21:3	175
21:17	15
21:20	35
22:24	114
22:34 Qere	192
24:3	34
24:13	144

#### 1 Kings

	0
1:2	89
1:4	120, 166
1:5	80
1:24-25	150
1:27	146
1:30	154
1:34	31
1:40	40
2:3	73
2:11	185
2:19	110
2:27	71
2:31	89
2:36	89
2:38	166
3:7	67
3:9	$7\dot{8}$
3:17	34
3:21	160
5:10	9
5:11 [4:31]	118
5:17 [5:3]	111
6:5	121
0.9	

1 Kings (cont	.)	4:24	155
		4:29	151, 173
6:37	8	5:1	25
7:14	32 21	5:2	20
7:27		5:3	129
7:47 8:10	117 38	5:7	175
8:16	50 117	5:10	89
8:18	51	5:11	75
8:23	119	5:20	153
8:27	132	5:24-25	93
8:32	19	5:25	100
8:55	16	7:1	23 70
8:61	125	7:6 7:0	50 80
8:64	27	7:9 8:7	182-183
13:12	85	8:12	170
13:33	177	8:13	20, 154
14:9	67	9:4	20, 194
14:11	112	9:4 9:8	38
14:12	147	9:29	35
14:18	72-73	9:37	175
15:19	107	10:4	138
15:23	20	10:18	77
15:27	123	10:19	115
16:24	23	11:17	107
18:13	173n25	12:8	140
18:18	105 82	[12:7]	
18:19	02 18, 21	13:2	97
18:32 18:39	7, 166	14:3	141
18:42	174	14:4	141
19:11	167	14:8	$4^{8}$
20:3	112	16:17	23
20:13	34, 81	17:2	140-141
21:1	97	17:6	35
21:7	139-140	17:17	71
21:15	154	17:18 18:4	48
21:22	101	18:34	44
22:7	91	19:21	155 12–13
22:12	64	22:13	108
22:13	19	23:34	126
22:22	88, 101	24:7	149
22:30-31	94	25:9	26
22:32	130	25:19	23
22:34	53	0 0	0
24:5	82	Isaiah	
2 Kings		1:7 1:15	11–12 55, 156, 184
2:9	188	1:15	16 150, 104
2:19	166	1:24	40
4:2 Qere	157	1:29	153
4:4	108	4:1	181
4:9	25	4:4	74
4:16 Qere	81	5:1	65
4:19	188	5:3	$6_{4}$

5:12-13	136	4:7	112
5:13	56	4:20	123
6:1	87	6:10	66
6:2	35	6:29	75
6:5	10	8:3	114
6:8		8:7	
	158	0.7	56
7:6	98	8:12	133-134
7:9	153-154	10:2	153
7:11	143	10:5	74
7:13	27	10:10	22
7:14	88	15:1	144
7:15	74	16:16	25
8:8	120	18:18	66
10:6	10	18:20	67
11:3	114	22:10	112
12:1	153	22:13	48, 82
13:6-7	122	22:15	128
		0	
14:8	133	23:17	75
14:10	133	23:18	91
18:5	116	23:28	102
19:21	41	23:38	179
19:22	40	28:9	185
21:11	169	28:16	32
22:2	11	43:5	146
22:3	118	44:4	55
22:12-13	161	44:17	137
24:2	109	47:4	79 79
29:8	31	49:20	145
30:16	26		
30.10	20	50:22	167
	0		
35:9	82	51:14	145
37:30	$6_4$	51:14	145
37:30 37:33	64 100–101		145
37:30 37:33 38:18	64 100–101 82	51:14 <b>Ezekiel</b>	145
37:30 37:33	64 100–101		145 40
37:30 37:33 38:18	64 100–101 82	Ezekiel	
37:30 37:33 38:18 38:20	64 100-101 82 72	Ezekiel 1:1 2:9	40 98
37:30 37:33 38:18 38:20 40:2	64 100-101 82 72 99	Ezekiel 1:1 2:9 3:2	40
37:30 37:33 38:18 38:20 40:2 40:7 40:10	64 100-101 82 72 99 56 106	Ezekiel 1:1 2:9 3:2 5:9	40 98 17 108
37:30 37:33 38:18 38:20 40:2 40:7 40:7 40:10 40:19	$ \begin{array}{c} 6_4 \\ 100-101 \\ 8_2 \\ 7_2 \\ 99 \\ 56 \\ 106 \\ 104 \\ \end{array} $	Ezekiel 1:1 2:9 3:2 5:9 12:4	40 98 17 108 135
37:30 37:33 38:18 38:20 40:2 40:7 40:10 40:19 44:1	$ \begin{array}{c} 6_4 \\ 100-101 \\ 8_2 \\ 7^2 \\ 99 \\ 56 \\ 106 \\ 104 \\ 147 \\ \end{array} $	Ezekiel 1:1 2:9 3:2 5:9 12:4 16:4	$40 \\ 98 \\ 17 \\ 108 \\ 135 \\ 53$
37:30 37:33 38:18 38:20 40:2 40:7 40:10 40:19 44:1 51:3	64 100-101 82 72 99 56 106 104 147 13019	Ezekiel 1:1 2:9 3:2 5:9 12:4 16:4 18:9	$40 \\ 98 \\ 17 \\ 108 \\ 135 \\ 53 \\ 45$
37:30 37:33 38:18 38:20 40:2 40:7 40:10 40:19 44:1 51:3 51:17	64 100-101 82 72 99 56 106 104 147 13n19 11	Ezekiel 1:1 2:9 3:2 5:9 12:4 16:4 18:9 20:9	40 98 17 108 135 53 45 99
37:30 37:33 38:18 38:20 40:2 40:7 40:10 40:10 40:19 44:1 51:3 51:17 53:4	64 100-101 82 72 99 56 106 104 147 13n19 11 12	Ezekiel 1:1 2:9 3:2 5:9 12:4 16:4 18:9 20:9 20:44	40 98 17 108 135 53 45 99 102
$\begin{array}{c} 37:30\\ 37:33\\ 38:18\\ 38:20\\ 40:2\\ 40:7\\ 40:10\\ 40:19\\ 44:1\\ 51:3\\ 51:17\\ 53:4\\ 53:5\end{array}$	64100-1018272995610610414713019111211	Ezekiel 1:1 2:9 3:2 5:9 12:4 16:4 18:9 20:9 20:44 21:9	40 98 17 108 135 53 45 99
$\begin{array}{c} 37:30\\ 37:33\\ 38:18\\ 38:20\\ 40:2\\ 40:7\\ 40:10\\ 40:19\\ 44:1\\ 51:3\\ 51:17\\ 53:4\\ 53:5\\ 53:11 \end{array}$	$\begin{array}{c} 6_4\\ 100-101\\ 8_2\\ 7_2\\ 99\\ 56\\ 106\\ 104\\ 147\\ 13n19\\ 11\\ 12\\ 11\\ 26\\ \end{array}$	Ezekiel 1:1 2:9 3:2 5:9 12:4 16:4 18:9 20:9 20:44 21:9 [21:4]	40 98 17 108 135 53 45 99 102 136
$\begin{array}{c} 37:30\\ 37:33\\ 38:18\\ 38:20\\ 40:2\\ 40:7\\ 40:10\\ 40:19\\ 44:1\\ 51:3\\ 51:17\\ 53:4\\ 53:5\\ 53:11\\ 55:5 \end{array}$	64100-1018272995610610414713n191112112697	Ezekiel 1:1 2:9 3:2 5:9 12:4 16:4 18:9 20:9 20:44 21:9 [21:4] 23:5	40 98 17 108 135 53 45 99 102 136
$\begin{array}{c} 37:30\\ 37:33\\ 38:18\\ 38:20\\ 40:2\\ 40:7\\ 40:10\\ 40:19\\ 44:1\\ 51:3\\ 51:17\\ 53:4\\ 53:5\\ 53:11\\ 55:5\\ 61:7\end{array}$	$\begin{array}{c} 6_4\\ 100-101\\ 8_2\\ 7_2\\ 99\\ 56\\ 106\\ 104\\ 147\\ 13n19\\ 11\\ 12\\ 11\\ 26\\ \end{array}$	Ezekiel 1:1 2:9 3:2 5:9 12:4 16:4 18:9 20:9 20:44 21:9 [21:4] 23:5 26:7	40 98 17 108 135 53 45 99 102 136
$\begin{array}{c} 37:30\\ 37:33\\ 38:18\\ 38:20\\ 40:2\\ 40:7\\ 40:10\\ 40:19\\ 44:1\\ 51:3\\ 51:17\\ 53:4\\ 53:5\\ 53:11\\ 55:5\\ 53:11\\ 55:5\\ 61:7\\ 64:8 \end{array}$	64100-1018272995610610414713n191112112697	Ezekiel 1:1 2:9 3:2 5:9 12:4 16:4 18:9 20:9 20:44 21:9 [21:4] 23:5	40 98 17 108 135 53 45 99 102 136
$\begin{array}{c} 37:30\\ 37:33\\ 38:18\\ 38:20\\ 40:2\\ 40:7\\ 40:10\\ 40:19\\ 44:1\\ 51:3\\ 51:17\\ 53:4\\ 53:5\\ 53:11\\ 55:5\\ 61:7\end{array}$	64 100-101 82 72 99 56 106 104 147 13n19 11 12 26 97 10	Ezekiel 1:1 2:9 3:2 5:9 12:4 16:4 18:9 20:9 20:44 21:9 [21:4] 23:5 26:7	40 98 17 108 135 53 45 99 102 136 100 13
$\begin{array}{c} 37:30\\ 37:33\\ 38:18\\ 38:20\\ 40:2\\ 40:7\\ 40:10\\ 40:19\\ 44:1\\ 51:3\\ 51:17\\ 53:4\\ 53:5\\ 53:11\\ 55:5\\ 53:11\\ 55:5\\ 61:7\\ 64:8 \end{array}$	64 100-101 82 72 99 56 106 104 147 13n19 11 12 26 97 10	Ezekiel 1:1 2:9 3:2 5:9 12:4 16:4 18:9 20:9 20:44 21:9 [21:4] 23:5 26:7 28:3 Qere	40 98 17 108 135 53 45 99 102 136 100 13 27 177
$\begin{array}{c} 37:30\\ 37:33\\ 38:18\\ 38:20\\ 40:2\\ 40:7\\ 40:10\\ 40:19\\ 44:1\\ 51:3\\ 51:17\\ 53:4\\ 53:5\\ 53:11\\ 55:5\\ 53:11\\ 55:5\\ 61:7\\ 64:8\\ [64:9] \end{array}$	64 100-101 82 72 99 56 106 104 147 13019 11 12 11 26 97 10 120	Ezekiel 1:1 2:9 3:2 5:9 12:4 16:4 18:9 20:9 20:44 21:9 [21:4] 23:5 26:7 28:3 Qere 28:15 33:10	40 98 17 108 135 53 45 99 102 136 100 13 27 177 154
$\begin{array}{c} 37:30\\ 37:33\\ 38:18\\ 38:20\\ 40:2\\ 40:7\\ 40:10\\ 40:19\\ 44:1\\ 51:3\\ 51:17\\ 53:4\\ 53:5\\ 53:11\\ 55:5\\ 53:11\\ 55:5\\ 61:7\\ 64:8\\ [64:9]\\ 65:1\end{array}$	64 100-101 82 72 99 56 106 104 147 13019 11 12 11 26 97 10 120	Ezekiel 1:1 2:9 3:2 5:9 12:4 16:4 18:9 20:9 20:44 21:9 [21:4] 23:5 26:7 28:3 Qere 28:15 33:10 36:9	$\begin{array}{c} 40\\ 98\\ 17\\ 108\\ 135\\ 53\\ 45\\ 99\\ 102\\ 136\\ 100\\ 13\\ 27\\ 177\\ 154\\ 98 \end{array}$
$\begin{array}{c} 37:30\\ 37:33\\ 38:18\\ 38:20\\ 40:2\\ 40:7\\ 40:10\\ 40:19\\ 44:1\\ 51:3\\ 51:17\\ 53:4\\ 53:5\\ 53:11\\ 55:5\\ 53:11\\ 55:5\\ 61:7\\ 64:8\\ [64:9] \end{array}$	64 100-101 82 72 99 56 106 104 147 13019 11 12 11 26 97 10 120	Ezekiel 1:1 2:9 3:2 5:9 12:4 16:4 18:9 20:9 20:44 21:9 [21:4] 23:5 26:7 28:3 Qere 28:15 33:10 36:9 36:37	$\begin{array}{c} 40\\ 98\\ 17\\ 108\\ 135\\ 53\\ 45\\ 99\\ 102\\ 136\\ 100\\ 13\\ 27\\ 177\\ 154\\ 98\\ 40\\ \end{array}$
37:30 37:33 38:18 38:20 40:2 40:7 40:10 40:19 44:1 51:3 51:17 53:4 53:5 53:11 55:5 61:7 64:8 [64:9] 65:1 Jeremiah	64100-1018272995610610414713n1911121126971012040-41	Ezekiel 1:1 2:9 3:2 5:9 12:4 16:4 18:9 20:9 20:44 21:9 [21:4] 23:5 26:7 28:3 Qere 28:15 33:10 36:9 36:37 38:14	$\begin{array}{c} 40\\ 98\\ 17\\ 108\\ 135\\ 53\\ 45\\ 99\\ 102\\ 136\\ 100\\ 13\\ 27\\ 177\\ 154\\ 98\\ 40\\ 69 \end{array}$
37:30 37:33 38:18 38:20 40:2 40:7 40:10 40:19 44:1 51:3 51:17 53:4 53:5 53:11 55:5 61:7 64:8 [64:9] 65:1 <b>Jeremiah</b> 1:17	64 100-101 82 72 99 56 106 104 147 13019 11 12 11 26 97 10 120 40-41 119	Ezekiel 1:1 2:9 3:2 5:9 12:4 16:4 18:9 20:9 20:44 21:9 [21:4] 23:5 26:7 28:3 Qere 28:15 33:10 36:9 36:37 38:14 39:27	$\begin{array}{c} 40\\ 98\\ 17\\ 108\\ 135\\ 53\\ 45\\ 99\\ 102\\ 136\\ 100\\ 13\\ 27\\ 177\\ 154\\ 98\\ 40\\ 69\\ 26\\ \end{array}$
37:30 37:33 38:18 38:20 40:2 40:7 40:10 40:19 44:1 51:3 51:17 53:4 53:5 53:11 55:5 61:7 64:8 [64:9] 65:1 <b>Jeremiah</b> 1:17 1:19	64100-101827299561061041471301911121126971012040-41	Ezekiel 1:1 2:9 3:2 5:9 12:4 16:4 18:9 20:9 20:44 21:9 [21:4] 23:5 26:7 28:3 Qere 28:15 33:10 36:9 36:37 38:14 39:27 40:1	$\begin{array}{c} 40\\ 98\\ 17\\ 108\\ 135\\ 53\\ 45\\ 99\\ 102\\ 136\\ 100\\ 13\\ 27\\ 177\\ 154\\ 98\\ 40\\ 69\\ 26\\ 177\\ \end{array}$
37:30 37:33 38:18 38:20 40:2 40:7 40:10 40:19 44:1 51:3 51:17 53:4 53:5 53:11 55:5 61:7 64:8 [64:9] 65:1 <b>Jeremiah</b> 1:17 1:19 2:22	$ \begin{array}{c} 6_4\\ 100-101\\ 8_2\\ 7^2\\ 99\\ 56\\ 106\\ 104\\ 147\\ 1_{3}n19\\ 11\\ 12\\ 11\\ 26\\ 97\\ 10\\ 120\\ 40-41\\ \end{array} $ $ \begin{array}{c} 119\\ 98-99\\ 184\\ \end{array} $	Ezekiel 1:1 2:9 3:2 5:9 12:4 16:4 18:9 20:9 20:44 21:9 [21:4] 23:5 26:7 28:3 Qere 28:15 33:10 36:9 36:37 38:14 39:27 40:1 43:6	$\begin{array}{c} 40\\ 98\\ 17\\ 108\\ 135\\ 53\\ 45\\ 99\\ 102\\ 136\\ 100\\ 13\\ 27\\ 177\\ 154\\ 98\\ 40\\ 69\\ 26\\ 177\\ 48 \end{array}$
37:30 37:33 38:18 38:20 40:2 40:7 40:10 40:19 44:1 51:3 51:17 53:4 53:5 53:11 55:5 61:7 64:8 [64:9] 65:1 <b>Jeremiah</b> 1:17 1:19	64100-101827299561061041471301911121126971012040-41	Ezekiel 1:1 2:9 3:2 5:9 12:4 16:4 18:9 20:9 20:44 21:9 [21:4] 23:5 26:7 28:3 Qere 28:15 33:10 36:9 36:37 38:14 39:27 40:1	$\begin{array}{c} 40\\ 98\\ 17\\ 108\\ 135\\ 53\\ 45\\ 99\\ 102\\ 136\\ 100\\ 13\\ 27\\ 177\\ 154\\ 98\\ 40\\ 69\\ 26\\ 177\\ \end{array}$

#### Hosea

1:6	
1:7	
9:1	
11:8	
13:2	

139

124

117

147, 186 130

#### Joel

1:15	149
2:9	108
2:11	138
2:12	140

#### Amos

1:1	116
3:2	140
3:4	145
3:6	146
4:4	111
4:6	120
5:2	12
5:5	75
5:21	55
7:7	182
9:2	144
9:8	181

## Jonah

1:1	86
1:4	169-170
2:7 [2:6]	107
2:8 [2:7]	124
3:2	99
3:3	114
3:5	28
4:5	142

#### Micah

1:4	53
2:6	113
3:1	72
4:7	117
4:10	141-142
4:14	104

#### Nahum

1:3	75
1:5	117
3:15	142
3:19	142

#### Habakkuk

1:4	46-47
1:14	138
3:7	126

#### Zephaniah

1:7	119
3:17	124

### Zechariah

1:13	22
3:7	144
4:10	23
9:15	11
11:6	161
11:16	166

#### Malachi

3:12 [4:3] 125-126

#### Psalms

1:4	31
2:9	12
3:3	103
3:4	108
5:4	19
5:8	98
5:8 8:5	175
9:16	185
10:11	55
10:16	6, 166
12:2	7
[12:1]	
12:6	9
13:5	16
14:1	51
14:3	133
14:5	15
15:1	59
16:2	122
16:6	131
16:8	142
18:49	131
[18:48]	
19:2	79
[19:1]	
19:8	9
[19:9]	
19:9	6
[19:10]	

19:13	45
19:14	130
20:7	139
23:4	59
24:2	170
24:7	10
25:14	$7^{2}$
25:15	142
27:6	66
28:5	
20.5	99 67
31:8	65
33:4	26
36:8	103
37:24	152-153
38:7	45
44:23	11, 122
	11, 122
[44:22]	
47:2	110
[47:1]	
48:2	10
[48:1]	
50:23	9
50.25	
57:10	103
61:5	19
66:19	51
68:6	103
68:21	110
[68:20]	
68:25	100
0	103
73:25	125
74:16	131
76:9	50
76:9 84:3	50
84:3	50 99–100
84:3 84:12	50 99–100 110
84:3 84:12 90:1	50 99–100 110 103
84:3 84:12 90:1 92:11	50 99–100 110 103 192
84:3 84:12 90:1 92:11 95:8	50 99–100 110 103 192 50
84:3 84:12 90:1 92:11 95:8 95:9	50 99–100 110 103 192
84:3 84:12 90:1 92:11 95:8	50 99–100 110 103 192 50
84:3 84:12 90:1 92:11 95:8 95:9	50 99–100 110 103 192 50 133
84:3 84:12 90:1 92:11 95:8 95:9 98:5 103:18	50 99-100 110 103 192 50 133 44-45 82
84:3 84:12 90:1 92:11 95:8 95:9 98:5 103:18 104:12	50 99-100 110 103 192 50 133 44-45 82 119
84:3 84:12 90:1 92:11 95:8 95:9 98:5 103:18 104:12 104:23	50 99-100 110 103 192 50 133 44-45 82 119 120
84:3 84:12 90:1 92:11 95:8 95:9 98:5 103:18 104:12 104:23 105:8	50 99-100 110 103 192 50 133 44-45 82 119 120 29
84:3 84:12 90:1 92:11 95:8 95:9 98:5 103:18 104:12 104:23 105:8 106:3	50 99-100 110 103 192 50 133 44-45 82 119 120 29 82
84:3 84:12 90:1 95:8 95:9 98:5 103:18 104:12 104:23 105:8 106:3 106:20	50 99-100 110 103 192 50 133 44-45 82 119 120 29
84:3 84:12 90:1 92:11 95:8 95:9 98:5 103:18 104:12 104:23 105:8 106:3	50 99-100 110 103 192 50 133 44-45 82 119 120 29 82
84:3 84:12 90:1 95:8 95:9 98:5 103:18 104:12 104:23 105:8 106:3 106:20	50 99-100 110 103 192 50 133 44-45 82 119 120 29 82 106
84:3 84:12 90:1 92:11 95:8 95:9 98:5 103:18 104:12 104:23 105:8 106:3 106:3 106:20 109:10	50 99-100 110 103 192 50 133 44-45 82 119 120 29 82 106 45
84:3 84:12 90:1 92:11 95:8 95:9 98:5 103:18 104:12 104:23 105:8 106:3 106:20 109:10 109:14 110:4	50 99-100 110 103 192 50 133 44-45 82 119 120 29 82 106 45 99 122
$\begin{array}{c} 84:3\\ 84:12\\ 90:1\\ 92:11\\ 95:8\\ 95:9\\ 98:5\\ 103:18\\ 104:12\\ 104:23\\ 105:8\\ 106:3\\ 105:8\\ 106:3\\ 105:8\\ 106:3\\ 105:10\\ 109:14\\ 110:1\\ 119:11 \end{array}$	50 99-100 110 103 192 50 133 44-45 82 119 120 29 82 106 45 99 122 176
$\begin{array}{c} 84:3\\ 84:12\\ 90:1\\ 92:11\\ 95:8\\ 95:9\\ 98:5\\ 103:18\\ 104:12\\ 104:23\\ 105:8\\ 106:3\\ 106:3\\ 106:20\\ 109:14\\ 110:4\\ 119:11\\ 119:92 \end{array}$	50 99-100 110 103 192 50 133 44-45 82 119 120 29 82 106 45 99 122 176 129
$\begin{array}{c} 84:3\\ 84:12\\ 90:1\\ 92:11\\ 95:8\\ 95:9\\ 98:5\\ 103:18\\ 104:12\\ 104:23\\ 105:8\\ 105:8\\ 106:3\\ 106:20\\ 109:10\\ 109:10\\ 109:14\\ 110:4\\ 119:11\\ 119:92\\ 120:2\\ \end{array}$	50 99-100 110 103 192 50 133 44-45 82 119 120 29 82 106 45 99 122 176 129 22
84:3 84:12 90:1 92:11 95:8 95:9 98:5 103:18 104:12 104:12 104:23 105:8 106:3 106:3 106:20 109:10 109:14 110:4 119:11 119:92 120:2 121:5	50 99-100 110 103 192 50 133 44-45 82 119 120 29 82 106 45 99 122 176 129 22 82, 164, 166
$\begin{array}{c} 84:3\\ 84:12\\ 90:1\\ 92:11\\ 95:8\\ 95:9\\ 98:5\\ 103:18\\ 104:12\\ 104:23\\ 105:8\\ 105:8\\ 106:3\\ 106:20\\ 109:10\\ 109:10\\ 109:14\\ 110:4\\ 119:11\\ 119:92\\ 120:2\\ \end{array}$	50 99-100 110 103 192 50 133 44-45 82 119 120 29 82 106 45 99 122 176 129 82 176 129 82 176 129 82 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 129 122 176 129 127 176 129 127 176 129 177 129 176 129 177
84:3 84:12 90:1 92:11 95:8 95:9 98:5 103:18 104:12 104:12 104:23 105:8 106:3 106:3 106:20 109:10 109:14 110:4 119:11 119:92 120:2 121:5	50 99-100 110 103 192 50 133 44-45 82 119 120 29 82 106 45 99 122 176 129 22 82, 164, 166
84:3 84:12 90:1 92:11 95:8 95:9 98:5 103:18 104:12 104:12 104:23 105:8 106:3 106:20 109:10 109:14 110:4 119:11 119:92 120:2 121:5 122:2	50 99-100 110 103 192 50 133 44-45 82 119 120 29 82 106 45 99 122 176 129 82 176 129 82 176 129 82 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 122 176 129 129 122 176 129 127 176 129 127 176 129 177 129 176 129 177

# Job

Job	
1:1	191
1:2	23
1:5	58
3:13	128
1:10 Qere	108
1:12	141
1:14	81
5:13	26
5:23	53
10:1	65–66
10:3	112
11:15	118
15:16	132
19:19	55
21:9	118
28:14	138
29:4	121
31:7	97
33:32	45
40:5	34
Proverbs	
1:17	10
2:4-5	129
6:19	107
6:33	137
7:14	121
9:8	156
10:1	58
14:28	9
15:12	74
15:31	78
16:10	9
17:17	104
17:26	67
21:16	74
21:19	67
21:27	131-132
22:6	51
22:9	118
22:21	22
23:15	144
23:18 28:21	154
20.21	74
Ruth	
1:1	82
1:4	109
1:12	151
1:13	192
1:17	39, 188–189

#### Ruth (cont.)

20
66
139
71
137
133
159
10
159
167

#### Song of Songs

1:1	13
1:8	28
2:5	11
2:8	19
2:9	96

#### Ecclesiastes

1:4	80	1 Chronic	les
1:9	185	5:0	9
2:5	90-91	5:9	24
3:4	67	15:5	31
3:17	26	15:19	23
4:9	118	16:35	7
4:17 [5:1]	74	21:15	11
5:1[5:2]	167	29:12	44
5:4	172	29:22	9
9:3	126	29:23	12
5.5		20:24	2/

#### Lamentations

1:8	15	2 Chronicles	;
Esther		6:27 14:6	101 139
1:12	138	14:10	125
2:3	62	[14:11]	
4:2	72	14:12	41
5:2	70	[14:13]	
9:26	123	16:10	122
10:3	114	20:9	91–92
10.5	114	21:17 22:5	28 123
Daniel		24:25	48
1:8	$^{173}_{64}$	30:3	47
1:13		33:3-4	91

#### Ezra

1:3	61-2
2:62	46
3:12	106
10:2	123

#### Nehemiah

1:4	81
2:2	134, 166
2:8	100
2:18	124
4:5 [4:11]	92
6:16	101
7:2	109
8:11	41
13:3	87
13:18	72
13:20	34

5:9	24
15:5	31
15:19	23
16:35	7
21:15	110
29:12	44
29:22	9
29:23	126
29:24	24